A Shoe-Integrated Sensor System for Wireless Gait Analysis and Real-Time Therapeutic Feedback

by

Stacy J. Morris

M.S., Mechanical Engineering, Massachusetts Institute of Technology, 1999

B.S., Mechanical Engineering, Massachusetts Institute of Technology, 1996

SUBMITTED TO THE HARVARD-MIT DIVISION OF HEALTH SCIENCES AND TECHNOLOGY IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF

DOCTOR OF SCIENCE IN MEDICAL ENGINEERING AT THE MASSACHUSETTS INSTITUTE OF TECHNOLOGY

JUNE 2004

© 2004 Massachusetts Institute of Technology All rights reserved.

Signature of Author
Harvard-MIT Division of Health Sciences and Technology
May 10, 2004
Certified by
Joseph A. Paradiso, PhD
Associate Professor
Sony Career Development Professor of Media Arts and Sciences
MIT Media Laboratory
Thesis Supervisor
Accepted by
Martha L. Gray, Ph.D.
Edward Hood Taplin Professor of Medical and Electrical Engineering
Co-Director Harvard-MIT Division of Health Sciences and Technology

A Shoe-Integrated Sensor System for Wireless Gait Analysis and Real-Time Therapeutic Feedback

by

Stacy J. Morris

Submitted to the Harvard-MIT Division of Health Sciences and Technology on May 10, 2004 in Partial Fulfillment of the Requirements for the Degree of Doctor of Science at the Massachusetts Institute of Technology

ABSTRACT

Clinical gait analysis currently involves either an expensive analysis in a motion laboratory, using highly accurate, if cumbersome, kinematic systems, or a qualitative analysis with a physician or physical therapist making visual observations. There is a need for a low cost device that falls in between these two methods, and can provide quantitative and repeatable results. In addition, continuous monitoring of gait would be useful for real-time physical rehabilitation.

To free patients from the confines of a motion laboratory, this thesis has resulted in a wireless wearable system capable of measuring many parameters relevant to gait analysis. The extensive sensor suite includes three orthogonal accelerometers, and three orthogonal gyroscopes, four force sensors, two bi-directional bend sensors, two dynamic pressure sensors, as well as electric field height sensors. The "GaitShoe" was built to be worn on any shoes, without interfering with gait, and was designed to collect data unobtrusively, in any environment, and over long periods of time.

Subject testing of the GaitShoe was carried out on ten healthy subjects with normal gait and five subjects with Parkinson's disease. The calibrated sensor outputs were analyzed, and compared to results obtained simultaneously from The Massachusetts General Hospital Biomotion Lab; the GaitShoe proved highly capable of detecting heel strike and toe off, as well as estimating orientation and position of the subject. A wide variety of features were developed from the calibrated sensor outputs, for use with standard pattern recognition techniques to classify the gait of the subject. The results of the classification demonstrated the ability of the GaitShoe to identify the subjects with Parkinson's disease, as well as individual subjects. Real-time feedback methods were developed to investigate the feasibility of using the continuous monitoring of gait for physical therapy and rehabilitation.

Thesis Supervisor: Prof. Joseph A. Paradiso Associate Professor of Media Arts and Sciences

A Shoe-Integrated Sensor System for Wireless Gait Analysis and Real-Time Therapeutic Feedback

by

Stacy J. Morris

The following people served as committee members for this thesis:

• • • • • • • • • • • • • • • • • •

Neville Hogan, Ph.D. Professor of Mechanical Engineering Professor of Brain and Cognitive Sciences Massachusetts Institute of Technology

David E. Krebs, D.P.T., Ph.D.

Professor, MGH Institute of Health Professionals Director, The Massachusetts General Hospital Biomotion Laboratory

Rosalind W. Picard, Sc.D. Associate Professor of Media Arts and Sciences MIT Media Laboratory

ACKNOWLEDGMENTS

To the Whitaker Foundation for supporting my education and research, to CIMIT for supporting this research, and to the Media Lab for additional support.

To Mike Harty and Thomas Papakosta at Tekscan, Inc., for sharing data about pressures in the insole.

To Jack Memishian at Analog Devices, for generously providing the demo yaw gyroscopes.

To Donna Scarborough for her enthusiasm and guidance of the clinical testing, to Dov Goldvasser for his help in the subject testing, and to the rest of the Biomotion Lab for their assistance.

To Dr. Leslie Shinobu for her enthusiasm and ideas for the use of the GaitShoe for persons with Parkinson's disease, and to her and her colleagues in the MGH Neurology Department, for helping to recruit volunteers with Parkinson's disease.

To the many volunteers for subject testing at the gait lab, for making this thesis possible.

To Dan Lovell for his tireless work on the ultrasound sensor, and to Erik Asmussen for his creativity in implementing the real-time feedback.

To Egon Pasztor for generously sharing his C code for the stripchart program.

To Ari Benbasat, for teaching me most of everything I know about electronics, microcontrollers, and debugging, for the initial work on the "stack", for all the great times in our office, and for his friendship.

To Hong Ma for his help and smiles, and to the rest of the Responsive Environments Group, for all the fun times. I look forward to seeing what you all do next!

To Asha Balakrishnan for her help in continuously finding yet another piece of equipment useful for calibration, and to her, Heather Gunter, and Amy Kerdok, for their friendship and support.

To the HST/MEMP program for providing the framework to make this research possible, and for a most interesting education.

To my wonderful committee: thank you, Neville Hogan, Dave Krebs, and Roz Picard for all your advice, brainstorming, suggestions, and ideas.

To Joe Paradiso, for bringing me into the group and the Media Lab, for opening my eyes everyday with his energy, curiosity, and never ending creativity for new technology and applications. I am honored to have done my doctorate research with you.

To MIT, for being my home for the last twelve years, and for being everything I could have dreamed of wanting for my education and more.

To my parents for their eternal support and encouragement.

And, to Ebbe for being my rock.

8

TABLE OF CONTENTS

Abstract	t	
Acknow	ledgme	nts
List of I	llustrati	ons
List of T	ables	
Chapter	1.	Introduction
1.1	Thesis	Statement
1.2	Motiva	tion
1.3	Project	Description
1.4	Importa	ant Gait Parameters
1.5	Summa	ry of Contributions
Chapter	2.	Background
2.1	Prior W	
2.1	2.1.1	On-Shoe Research Systems
	2.1.2	Off-Shoe Approaches
	2.1.3	Gait Recognition Systems
	2.1.4	Commercial Systems
2.2	2.1.5 Cumoni	Expressive rootware. Instrumented insole and wurtiple bloe-based behaois 50
2.2	2 2 1	Observational Gait Analysis 38
	2.2.1	Overview of Clinical Gait Analysis
	2.2.3	Optoelectronic Systems
	2.2.4	Videographic Systems
	2.2.5	Electromagnetic Systems
2.3	Clinica	l Need
	2.3.1	Utility of On-Shoe Device in Subjects with Parkinson's Disease 44
	2.3.2	Studying Gait Outside of the Motion Lab 46
Chapter	3.	Hardware Design
3.1	Sensor	Selection
3.2	Physica	ll Implementation
	3.2.1	The Stack
	3.2.2	Insole Design
	3.2.3	Shoe Attachment
	3.2.4	The Basestation

2.2	C	See if estimation (2)
5.5	Sensor	Accelerometer 63
	332	Acceleronieter
	333	Force Sensitive Resistors 71
	334	Bend Sensors 75
	3.3.5	Polyvinvlidene Fluoride Strips
	3.3.6	Electric Field Sensor
	3.3.7	Ultrasound Sensor
34	Additi	onal Components 86
5.4	3.4.1	Microcontroller 86
	3.4.2	Radio Frequency Transceiver 88
	3.4.3	Power
	3.4.4	Other Components
35	The G	aitShoa System
5.5		anonoe System
Chapter	4.	Sensor Analysis
4.1	Data P	Processing
	4.1.1	Truncation
	4.1.2	Time Adjustment
	4.1.3	Data Adjustment
	4.1.4	Calibration and Analysis
4.2	Analys	sis Model
	4.2.1	Coordinate Systems
	4.2.2	Inertial Measurement Evaluation
4.3	Gyroso	copes
	4.3.1	Calibration
	4.3.2	Analysis of the Pitch
4.4	Accel	arometers 122
4.4		Calibration 122
	442	Orientation 125
	4.4.3	Velocity and Stride Length Analysis
15	Eeree	Sensitive Designers 124
4.5	Force	Calibration 134
	4.3.1	
4.6	Bend S	Sensors
	4.6.1	Calibration
	4.6.2	Placement of the Bend Sensors
	4.6.3	Plantar Flexion and Dorsiflexion
4.7	PVDF	Strips
4.8	Electri	c Field Sensor
4.9	Heel S	trike and Toe Off Timing
	-	
Chapter	э.	Gait Parameter validation

5.1

5.3	Heel S	trike and Toe Off	169
5.4	Discus	sion	170
			1
Chapter	r 6.	Pattern Recognition Analysis	175
6.1	Patterr	Recognition Techniques	175
	0.1.1 6 1 2	Classification and Regression frees (CAR1)	173
	6.1.3	Support Vector Machines (SVM)	179
	6.1.4	Neural Networks	181
6.2	Data S	ets and Classifications	184
	6.2.1	Gait Variation	185
	6.2.2	Subjects and Data Sets	186
6.3	Featur	e Set	187
6.4	Hypot	heses	192
6.5	Trainii	ng and Testing Groups	193
6.6	Result	s	195
	6.6.1	Hypothesis 1	196
	6.6.2	Hypothesis 2	203
	6.6.3	Hypothesis 3	205
	6.6.4	CART Feature Information	206
	6.6.5	Additional Neural Net Studies	209
6.7	Discus	sion	212
Chapter	r 7.	Real-Time Therapeutic Feedback	215
7.1	Overv	iew	215
7.2	Rhyth	mic Auditory Stimulator	216
	7.2.1	The RAS system	217
	7.2.2	RAS Feedback	218
7.3	Conclu	usions	220
	. 0		221
Chapter	rð.		221
Append	lix A.	Medical Information	227
A.1	Termir	nology	227
A.2	Parkin	son's Disease	228
Append	ix B.	Subject Testing	231
rr	Study	Design	231
D.1	Cultin	t Information	201
В.2	Subjec	a mormation	230
Append	lix C.	Pattern Recognition Information	239
C.1	Termir	10logy	239
C.2	Compl	lete Results	241

Append	lix D. Hardware Information	255
D.1	Schematics and Board Layouts	255
D.2	Component Information	269
D.3	Other Hardware Information	271
Append	lix E. Ultrasound Sensor	277
E.1	Circuit Boards	277
E.2	Microcontroller Code	281
E.3	Initial Results	283
Append	lix F. Code	285
F.1	Microcontroller Code	285
F.2	Matlab Code	289
Referen	nces	307

LIST OF ILLUSTRATIONS

Figure 2.1	Expressive Footware and its sensor outputs	37
Figure 2.2	A subject at the MGH Biomotion Lab	41
Figure 3.1	Schematic of the GaitShoe system	54
Figure 3.2	The stack	55
Figure 3.3	Photo of the front and back of the Main board	55
Figure 3.4	Photo of the front and back of the IMU board	56
Figure 3.5	Photo of the front and back of the Tactile board	56
Figure 3.6	Photos of the front and back of the transmit Ultrasound board for the right foot (right photos and the front and back of the receive Ultrasound board for the left foot (left photos) .	s), 57
Figure 3.7	Photo of the front and back of the Power board	58
Figure 3.8	Photograph of an insole sensor	58
Figure 3.9	Bend sensor in ankle strap	59
Figure 3.10	Cross section showing the electric field sensor components	59
Figure 3.11	Photos of three versions of the shoe attachment	60
Figure 3.12	Final prototype of the GaitShoe attachment	61
Figure 3.13	Photo of the GaitShoe system on two shoes	61
Figure 3.14	Accelerometer response to impact at heel	62
Figure 3.15	Photo of the basestation	62
Figure 3.16	Schematic of the ADXL202E	65
Figure 3.17	Photo of the IMU board, with the ADXL202E components highlighted	65
Figure 3.18	Schematic of the ENC-03J	68
Figure 3.19	Photo of the IMU board, with the Murata ENC-03J components highlighted	68
Figure 3.20	Schematic of the ADXRS150	70
Figure 3.21	Photo of the IMU board, with the ADXRS150 components highlighted	70
Figure 3.22	Schematic of the FSR	74
Figure 3.23	Photo of the Tactile board, with the FSR components highlighted	74
Figure 3.24	Bend sensors shown back to back	75
Figure 3.25	Schematic of the bi-directional bend sensor	76
Figure 3.26	Photo of the Tactile board, with the bi-directional bend sensor components highlighted .	77
Figure 3.27	Schematic of the PVDF strips	78
Figure 3.28	Photo of the Tactile board, with the PVDF strip components highlighted	78
Figure 3.29	Schematic showing electrode coupling to both the ground and the floor	80
Figure 3.30	Schematic showing the second electrode configuration coupling to the floor	81

Figure 3.31	Schematic of the electric field sensor
Figure 3.32	Photo of the Tactile board, with the electric field components highlighted
Figure 3.33	Example of an electrode set up on the bottom of a shoe to measure heel height $\ldots \ldots 83$
Figure 3.34	Example of two electrode set up on the bottom of a shoe to measure heel and toe height. $.83$
Figure 3.35	Ultrasound sensor for distance and angle between two shoes
Figure 3.36	Schematic of the Cygnal C8051F206
Figure 3.37	Photo of the main board, with the Cygnal C8051F206 components highlighted 87
Figure 3.38	Schematic of the RF Monolithics DR3000-1
Figure 3.39	Photo of the Main board, with the components of the RF Monolithics DR3000-1 highlighted
Figure 3.40	Schematic of the Power Board
Figure 3.41	Header interconnection net names
Figure 3.42	GaitShoe hardware
Figure 3.43	High level block diagram of the GaitShoe system
Figure 3.44	Sensor outputs across all sensors for both feet
Figure 4.1	Complete raw output
Figure 4.2	Truncated section of the raw output
Figure 4.3	Outlier identified in bend sensor output 103
Figure 4.4	Outlier identified in accelerometer output
Figure 4.5	Bend sensor data with a number of adjustments
Figure 4.6	Frames of reference used in evaluation 107
Figure 4.7	The gait cycle
Figure 4.8	Comparison of IMU outputs during walking gait 111
Figure 4.9	Illustration for linear integration
Figure 4.10	Photo of the turntable used for the gyroscope calibration
Figure 4.11	Line fits for determining the sensitivity of the gyroscopes (shown for IMU-1) 115
Figure 4.12	Sample of direct linear integration of the z-gyroscope
Figure 4.13	Z-gyroscope output where angular velocity is non-zero during stance 117
Figure 4.14	Sample of linear integration of the z-gyroscope, with iteration 118
Figure 4.15	Sample of linear integration of the z-gyroscope, with iteration and spline-fit 119
Figure 4.16	Pitch of the foot during the gait cycle 120
Figure 4.17	Sample of linear integration of the z-gyroscope, with iteration and post-integration spline-fit
Figure 4.18	Final results, compared to BML data 121
Figure 4.19	Comparison of BML pitch to GaitShoe pitch 122
Figure 4.20	Demonstration of accelerometer calibration
Figure 4.21	Sample accelerometer calibration for determining sensitivity and zero-offset
Figure 4.22	Determination of the orientation of the accelerometers

Figure 4.23	Dynamic acceleration along Xroom only	127
Figure 4.24	Dynamic acceleration with both Xroom and Yroom components	128
Figure 4.25	Integration of the acceleration in Xroom and Xshoe	131
Figure 4.26	Comparison of BML Xroom-displacement to GaitShoe Xroom- and	
	Xshoe- displacement	132
Figure 4.27	Integration of the acceleration in Yroom	133
Figure 4.28	Comparison of BML Yroom-displacement to GaitShoe Yroom-displacement	133
Figure 4.29	Force applicators (above) and FSRs (below)	136
Figure 4.30	Test set-up for calibration of FSRs	137
Figure 4.31	Sample FSR-402 calibration data	137
Figure 4.32	Various line-fits to the FSR-402 calibration data	138
Figure 4.33	FSR-402 calibration curve	139
Figure 4.34	FSR-400 calibration curve	140
Figure 4.35	95% confidence intervals for FSR sensitivity curves	141
Figure 4.36	Method of calibration of bend sensors	142
Figure 4.37	Data from calibration of sensitivity of the bend sensor	143
Figure 4.38	Line fit for determining the sensitivity of bend sensors	144
Figure 4.39	Bend calibration routine, on a new sensor with and without tape	145
Figure 4.40	Bend sensor, calibrated with different pivot points	147
Figure 4.41	Plantar Flexion and Dorsiflexion from the GaitShoe (left) and the BML (right)	148
Figure 4.42	Calibrated insole bend sensor output	149
Figure 4.43	Various PVDF output plots	150
Figure 4.44	Reebok running shoe, Ecco walking shoe, and Dansko clog.	151
Figure 4.45	Comparison of PVDF output in three different types of shoes	151
Figure 4.46	Comparison of PVDF response to FSRsum response	152
Figure 4.47	Various electric field sensor output plots	154
Figure 4.48	Comparison of the electric field sensor output to foot height.	155
Figure 4.49	Sample output from the second electrode design	156
Figure 4.50	Comparison of the sum of the four calibrated [N] FSR outputs to the force plate output	157
Figure 4.51	Comparison of the sum of the four calibrated FSR outputs in pressure units of psi	158
Figure 4.52	Determination of heel strike and toe off from FSRsum	159
Figure 4.53	First difference of spline-fit, zoomed-in	160
Figure 5.1	Comparison of GaitShoe pitch to BML foot array pitch	164
Figure 5.2	Histograms of the pitch validation results	166
Figure 5.3	Histograms of the stride length validation results	168
Figure 5.4	Histograms of the vertical displacement validation results	168
Figure 5.5	Histograms of the heel strike time and toe off time validation results	170

Figure 6.1	Examples of a single input neuron and a multiple input neural layer	182
Figure 6.2	Plots of maximum and minimum pitch, by age	191
Figure 6.3	Subject data of two most informative features in CART analyses	207
Figure 6.4	Subject data of two most informative features in CART analyses, individual PD subjects	208
Figure 7.1	Erik Asmussen and the Rhythmic Auditory Stimulator	216
Figure 7.2	Screenshots of the RAS system, showing the main menu (upper left), therapy configuration menu (upper right), and feedback control menu (lower)	217
Figure A.1	The Gait Cycle	228
Figure B.1	MGH consent form for subjects with healthy gait	233
Figure B.2	MGH consent form for subjects with difficulty walking	234
Figure B.3	MIT COUHES Consent Form	235
Figure B.4	Subject Testing Protocol, page 1 (left) and page 2 (right)	235
Figure C.1	Sample CART tree	239
Figure D.1	Schematic of the IMU board	256
Figure D.2	Layout of the top side of the IMU board	257
Figure D.3	Layout of the bottom side of the IMU board	258
Figure D.4	Pin mappings for the commercially available ADXRS150	259
Figure D.5	Schematic of the Tactile board	260
Figure D.6	Layout of the top side of the Tactile board	261
Figure D.7	Layout of the bottom side of the Tactile board	262
Figure D.8	Schematic of the Main board	263
Figure D.9	Layout of the top side of the Main board	264
Figure D.10	Layout of the bottom side of the Main board	265
Figure D.11	Schematic of the Power board	266
Figure D.12	Layout of the top side of the Power board	267
Figure D.13	Layout of the bottom side of the Power board	267
Figure D.14	Schematic of the Programming board	268
Figure D.15	Layout for the Programming board	268
Figure D.16	Component information, part one.	269
Figure D.17	Component information, part two	270
Figure D.18	Bend sensors soldered to wire	271
Figure D.19	Bend sensors with a "gob" of hot glue	271
Figure D.20	Hot glue flattened by fingers	272
Figure D.21	Trimmed hot glue on bend sensor solder connection	272
Figure D.22	Sketch of the pattern for both the left and right GaitShoe attachments	273
Figure D.23	Pattern traced onto PTG for forming the GaitShoe Attachment	273
Figure D.24	PTG cut to shape with heavy duty shears	274

Figure D.25	Heat gun and wooden block used to shape the PTG	274
Figure D.26	Sketch of the pattern for the battery enclosure	275
Figure D.27	GaitShoe attachment and battery enclosure; viewed separately from the back (left), and viewed together from the front (right)	275
Figure E.1	Photo of ultrasound hardware mounted on the GaitShoe hardware	278
Figure E.2	Schematic of the ultrasound transmit board.	278
Figure E.3	Schematic of the ultrasound transmit daughter board	279
Figure E.4	Schematic of the ultrasound receive board	279
Figure E.5	Schematic of the ultrasound receive daughter board	279
Figure E.6	Microcontroller code for the ultrasound transmit board.	281
Figure E.7	Microcontroller code for the ultrasound receiver board.	282
Figure E.8	Sample data from the two ultrasound sensors	283
Figure E.9	Data collected from all sensors, during slow gait	284
Figure F.1	206.h Code	285
Figure F.2	Basestation microcontroller code	286
Figure F.3	Stack microcontroller code, part one	287
Figure F.4	Stack microcontroller code, part 2	288
Figure F.5	Timing issues, as controlled by the basestation	289

LIST OF TABLES

TABLE 3.1	Sensor selection	52
TABLE 3.2	Relevant parameters of the Analog Devices ADXL202E accelerometer [71]	66
TABLE 3.3	Relevant parameters of the Murata ENC-03J gyroscope [73]	69
TABLE 3.4	Relevant parameters of the Analog Devices ADXRS150 gyroscope [75]	72
TABLE 3.5	Relevant parameters of the Interlink FSR-400 and FSR-402 [77]	74
TABLE 3.6	Relevant parameters of The Images Co. FLX-01 [78]	77
TABLE 3.7	Relevant parameters of the Measurement Specialties LDT0 PVDF strip [80]	79
TABLE 3.8	Relevant parameters of the Motorola MC33794DH [82]	84
TABLE 4.1	Final sensor order	101
TABLE 4.2	Order of calibrated and analyzed data	106
TABLE 4.3	Gyroscope sensitivities and zero offsets	115
TABLE 4.4	Accelerometer sensitivities and zero offsets	125
TABLE 4.5	Bend sensor sensitivities and zero offsets	145
TABLE 5.1	Pitch Validation Results	165
TABLE 5.2	Displacement Validation Results	167
TABLE 5.3	Heel Strike and Toe Off Validation Results	169
TABLE 6.1	Gender differences in gait [105]	185
TABLE 6.2	Summary of subject groupings	186
TABLE 6.3	Classifications of the ten female subjects	187
TABLE 6.4	Feature set	189
TABLE 6.5	Measurement exclusion due to dropped data	192
TABLE 6.6	Training and testing groups	193
TABLE 6.7	CART Hypothesis 1 results, using modified leave one out	196
TABLE 6.8	Naïve Bayes Hypothesis 1 results, using modified leave one out	197
TABLE 6.9	Neural Net Hypothesis 1 results, using modified leave one out	197
TABLE 6.10	CART Hypothesis 1 results, by class, using leave one subject out	198
TABLE 6.11	CART Hypothesis 1 results, by subject, using leave one subject out	198
TABLE 6.12	CART Hypothesis 1 results, using "free gait" as the test set	199
TABLE 6.13	CART Hypothesis 1 results, using "distracted gait" as the test set	199
TABLE 6.14	CART Hypothesis 1 results, using "paced gait" as the test set	199
TABLE 6.15	SVM Hypothesis 1.1 results, using modified leave one out	200
TABLE 6.16	SVM Hypothesis 1.2 results, using modified leave one out	200
TABLE 6.17	SVM Hypothesis 1.3 results, using modified leave one out	200

TABLE 6.18	SVM Hypothesis 1.1 results, using leave one subject out	201
TABLE 6.19	SVM Hypothesis 1.2 results, using leave one subject out	201
TABLE 6.20	SVM Hypothesis 1.3 results, using leave one subject out	201
TABLE 6.21	SVM Hypothesis 1.2 results, using "paced gait" as the test set	202
TABLE 6.22	SVM Hypothesis 1.3 results, using "paced gait" as the test set	202
TABLE 6.23	SVM Hypothesis 1.3 results, using "free gait" as the test set	202
TABLE 6.24	CART Hypothesis 2 results, using modified leave one out	203
TABLE 6.25	SVM Hypothesis 2 results, using modified leave one out	203
TABLE 6.26	Naïve Bayes Hypothesis 2 results, using modified leave one out	203
TABLE 6.27	Neural Net Hypothesis 2 results, using modified leave one out	203
TABLE 6.28	CART Hypothesis 2 results, using leave one subject out	204
TABLE 6.29	SVM Hypothesis 2 results, using leave one subject out	204
TABLE 6.30	Neural Net Hypothesis 2 results, using leave one subject out	204
TABLE 6.31	Neural Network Hypothesis 3 results, by subject, using cross-validation	205
TABLE 6.32	Informative features, as identified by CART	206
TABLE 6.33	Neural Net Hypothesis 1 results, using 10-fold cross-validation, and the top two features	209
TABLE 6.34	Neural Net Hypothesis 2 results, using 10-fold cross-validation, and the top two features	209
TABLE 6.35	Neural Net Hypothesis 3 results, by subject, using 10-fold cross-validation, and the top six features	210
TABLE 6.36	Contrived Groupings	211
TABLE 6.37	Neural Net results, using 10-fold cross-validation, and the top two features, with "Contrived Groups A"	211
TABLE 6.38	Neural Net results, using 10-fold cross-validation, and the top two features, with "Contrived Groups B"	212
TABLE 6.39	Neural Net results, using 10-fold cross-validation, and the top two features, with "Contrived Groups C"	212
TABLE 8.1	Comparison between the GaitShoe and the MGH Biomotion Lab	225
TABLE B.1	Information about volunteers for the subject testing	237
TABLE C.1	Sample results presented in a confusion matrix	240
TABLE C.5	CART Hypothesis 1 results, by class, using leave one subject out	242
TABLE C.2	CART Hypothesis 1 results, using modified leave one out	242
TABLE C.3	Naïve Bayes Hypothesis 1 results, using modified leave one out	242
TABLE C.4	Neural Net Hypothesis 1 results, using modified leave one out	242
TABLE C.7	CART Hypothesis 1 results, using "free gait" as the test set	243
TABLE C.8	CART Hypothesis 1 results, using "distracted gait" as the test set	243
TABLE C.6	CART Hypothesis 1 results, by subject, leave one subject out	243
TABLE C.10	SVM Hypothesis 1.1 results, using modified leave one out	244

TABLE C.11	SVM Hypothesis 1.2 results, using modified leave one out	244
TABLE C.12	SVM Hypothesis 1.3 results, using modified leave one out	244
TABLE C.9	CART Hypothesis 1 results, using "paced gait" as the test set	244
TABLE C.13	SVM Hypothesis 1.1 results, using leave one subject out	245
TABLE C.14	SVM Hypothesis 1.2 results, using leave one subject out	245
TABLE C.15	SVM Hypothesis 1.3 results, using leave one subject out	245
TABLE C.16	SVM Hypothesis 1.1 results, using "free gait" as the test set	245
TABLE C.19	SVM Hypothesis 1.1 results, using "distracted gait" as the test set	246
TABLE C.20	SVM Hypothesis 1.2 results, using "distracted gait" as the test set	246
TABLE C.21	SVM Hypothesis 1.3 results, using "distracted gait" as the test set	246
TABLE C.17	SVM Hypothesis 1.2 results, using "free gait" as the test set	246
TABLE C.18	SVM Hypothesis 1.3 results, using "free gait" as the test set	246
TABLE C.22	SVM Hypothesis 1.1 results, using "paced gait" as the test set	247
TABLE C.23	SVM Hypothesis 1.2 results, using "paced gait" as the test set	247
TABLE C.24	SVM Hypothesis 1.3 results, using "paced gait" as the test set	247
TABLE C.25	CART Hypothesis 2 results, using modified leave one out	248
TABLE C.26	SVM Hypothesis 2 results, using modified leave one out	248
TABLE C.27	Naïve Bayes Hypothesis 2 results, using modified leave one out	248
TABLE C.28	Neural Net Hypothesis 2 results, using modified leave one out	248
TABLE C.29	CART Hypothesis 2 results, using leave one subject out	249
TABLE C.30	CART Hypothesis 2 results, by subject, using leave one subject out	249
TABLE C.31	SVM Hypothesis 2 results, using leave one subject out	249
TABLE C.32	SVM Hypothesis 2 results, by subject, using leave one subject out	250
TABLE C.33	Neural Net Hypothesis 2 results, using leave one subject out	250
TABLE C.34	Neural Net Hypothesis 2 results, by subject, using leave one subject out	250
TABLE C.35	CART Hypothesis 2 results, using "free gait" as the test set	251
TABLE C.36	SVM Hypothesis 2 results, using "free gait" as the test set	251
TABLE C.37	CART Hypothesis 2 results, using "distracted gait" as the test set	251
TABLE C.38	SVM Hypothesis 2 results, using "distracted gait" as the test set	251
TABLE C.39	CART Hypothesis 2 results, using "paced gait" as the test set	251
TABLE C.41	Neural Network Hypothesis 3 results, by subject, using cross-validation	252
TABLE C.40	SVM Hypothesis 2 results, using "paced gait" as the test set	252
TABLE C.42	CART Hypothesis 3 results, by subject, using cross-validation	253
TABLE C.43	Naïve Bayes Hypothesis 3 results, by subject, using cross-validation	253
TABLE D.1	Insole sensor mapping	259
TABLE D.2	Vendor information	271

Chapter 1

INTRODUCTION

Clinical gait analysis is the investigation of the pattern of walking. At present, gait analysis is primarily carried out in one of two ways: in a motion laboratory, with full analysis of the motion of all body segments using highly accurate computer-based force sensors and optical tracking systems, or in an office with the clinician making visual observations. The first method is expensive, requires the maintenance of a dedicated motion lab, and uses cumbersome equipment attached to the patient, but produces well-quantified and accurate results for short distances. The second method is inexpensive and does not require any equipment, but the results are qualitative, unreliable, and difficult to compare across multiple visits.

There is a need for a low cost device that falls in between these two methods, and is capable of providing quantitative and repeatable results. In addition, there is a need for long term monitoring of gait, as well as quick diagnosis of chronic walking problems. Also, there is a need to be able to quantitatively analyze gait for patients who do not have access to motion analysis labs, such as is the case in economically disadvantaged locations.

This thesis discusses the development of an on-shoe system for continuous monitoring of gait. This system includes an instrumented insole and a removable instrumented shoe attachment. The data are sent wirelessly, providing information about the three-dimensional motion, position, and pressure distribution of the foot. The system was independently calibrated and analyzed, and was tested on fifteen subjects. The results from these

subjects were compared to the results from the gait analysis system at the Massachusetts General Hospital (MGH) Biomotion Lab.

1.1 Thesis Statement

The goal of this thesis was to design, build, calibrate, analyze, and use a wireless wearable system capable of measuring an unprecedented number of parameters relevant to gait. The system was designed to collect data unobtrusively, and in any walking environment, over long periods of time. It was built to be worn on the shoes, without interfering with gait. The sensors were calibrated, and the calibrated data were analyzed for information about the gait of the user, and the results of the gait analysis were validated against results from the optical tracking system in use at the MGH Biomotion Lab. The calibrated data were also used to generate features, which were used to classify the gait of the subject, using standard pattern recognition techniques. The system was also used to investigate real-time therapeutic feedback.

1.2 Motivation

Quantitative evaluation of gait is currently limited by the availability and the size of motion analysis labs. Motion analysis labs are expensive to maintain, and are typically only found in hospitals in large urban areas. Typically, patients can only walk about 7-10 meters per trial, and have one chance per trial to step on a disguised force plate. Alternatively, many physicians and physical therapists rely instead on observational gait analysis to evaluate patients. While well-trained medical specialists are undoubtedly capable of discerning a great deal of information about their patients' gait, small changes may be hard to detect, and a qualitative observation is difficult to compare between office visits or different specialists. Evaluation of common podiatric problems would be enhanced by an inexpensive method of quantitative evaluation. For instance, people with diabetes are often fitted with orthotics to improve their gait and reduce their chances of developing ulcerations on their feet; a straightforward and repeatable method of evaluating gait before

and after use of an orthotic would be desirable to optimize its shape and placement. Historically, orthotics have not been designed following systematic design procedures, but in an *ad hoc* manner, relying on the individual expertise of the orthotist [1].

In addition, gait and changes in gait are surrogate markers for a variety of other medically important phenomena: developmental maturation, likelihood of falling, and recovery from a stroke. Change in gait over extended time is used in neurological exams to diagnose dementias, and can be used to assess the adequacy of pharmacologic therapy in a number of neurologic/psychiatric disorders.

Finally, the development of a wearable wireless system has been greatly enabled by the many recent and on-going advances in sensor technology that have resulted in sensors which are small and inexpensive.

1.3 Project Description

The research sought to create a system that will provide instrumented gait analysis outside of traditional, expensive motion labs. Such a system has the potential to be highly informative by allowing data collection throughout the day in a variety of environments, thus providing a vast quantity of long-term data not obtainable with current gait analysis systems.

The top-level functional requirements for this system are:

- 1. Effect no change in gait.
- 2. Characterize the motion of both feet.
- 3. Be untethered.
- 4. Allow the subject to use his or her own shoes.

To meet these requirements, an on-shoe system has been designed and developed. The onshoe components were configured in such a way that gait was minimally affected, and such that they could be readily fixed to a variety of typical walking shoes. The system was replete with sensors, with the goal of measuring more parameters than would otherwise be necessary for any one application, essentially providing a wearable podiatric laboratory. A power source was contained on-shoe, and the system used wireless protocols to communicate between shoes and to transmit the data to a base-station; no cables of any sort were attached to either shoe.

This research evaluated the system both in persons with normal gait, and in elders with Parkinson's disease (PD). Subjects with PD were included for the purpose of evaluating the data in a population with altered gait. As indicated in recent research, the PD population would benefit from having a system which would allow evaluation of gait at home, by providing better information about gait abnormalities present in everyday life that have not traditionally been captured in analyses carried out in motion laboratories [2]. For example, this could provide the ability to titrate medication doses to the patient's current, rather than the average, needs.

1.4 Important Gait Parameters

As mentioned above, this thesis sought to create a system capable of providing clinically relevant information about gait. "Clinically relevant" is, of course, a subjective term which is certainly defined in many different ways. Therefore, to direct the design of the system, certain parameters of gait were identified (through a review of the literature about gait analysis and meetings with the physical therapists in the MGH Biomotion Lab) as important for the system to measure. The following gait measurements were identified:

- 1. Heel strike timing.
- 2. Toe off timing.
- 3. Dorsi-/plantar- flexion.
- 4. Stride length.
- 5. Stride velocity.

The system described within this thesis is designed so that, at a minimum, it is capable of characterizing these specific parameters of gait. The results for these and other parameters and were compared to those obtained by the system in use at the MGH Biomotion Lab;

this comparison was used to validate the system and to make a statement about the clinical relevance of the information resulting from the on-shoe system.

1.5 Summary of Contributions

The work completed for this thesis has resulted in the following:

- 1. A robust wireless two-shoe system, capable of measuring many gait-relevant parameters, and including shoe attachments, insoles, and base-station.
- 2. Data collection from ten subjects with normal gait, and from five subjects with Parkinson's disease (PD).
- 3. Techniques for calibration of the system.
- 4. Methods for analysis of gait features, including "important gait parameters" (heel strike timing, toe off timing, dorsi-/plantar- flexion, stride length, and stride velocity).
- 5. Determination of features from the sensor outputs, classification of gait as gait of healthy subjects or gait of subjects with Parkinson's disease, and classification of the gait of ten individual subjects.
- 6. Identification of two features which distinguish normal gait from the gait of subjects with Parkinson's disease.
- 7. Investigation into the use of interactive real-time therapeutic feedback.

Chapter 2

BACKGROUND

To provide an understanding for the need of a wireless gait system, this chapter discusses the prior work in the field, as well as the current state of the art. In addition, the clinical need for this system is discussed.

2.1 Prior Work

There is extensive prior research investigating alternatives to the traditional motion lab for gait analysis. The obvious advantage of directly measuring the pressure distribution beneath the foot has driven many of the early shoe-based systems. The shrinking size of data storage has further encouraged the development of non-tethered systems.

2.1.1 On-Shoe Research Systems

Efforts to take measurements more directly at the foot interface go back to at least the 1960's, with most early work focusing on various pressure sensors on an insole to gauge the pressure distribution beneath the foot.

While there are obvious advantages in taking measurements directly, there are some potential disadvantages with instrumented shoes. For instance, if accurate measurements of pressure underneath anatomical landmarks are required, sensor placement must either be guessed at, or an initial test must be done to determine correct placement. The placement of the sensors must be durable enough to prevent movement within the shoe during walking. The sensors themselves must be robust enough to withstand the normal and shear forces of walking, as well as the warm, humid climate inside the shoe. In addition, consideration must be made so that the instrumentation itself does not affect the gait. These limitations need to be taken into consideration during the design of any instrumented shoes [3].

Instrumented Insole for Pressure Distribution

In 1990, Wertsch *et al* [4] developed an exceptional system for measuring the pressure distribution beneath the foot. They first had each subject walk on inked paper to determine the locations of seven high pressure points corresponding to the five metatarsal heads, the big toe, and the heel center. They then placed seven force sensitive resistors (FSRs) at these locations, creating a specific insole for each foot of each subject. A seven channel amplification circuit was attached to each lower leg, and one shielded cable ran up each leg to a belt anchor. Both of the shielded cables were 10 m in length, and extended to connect to the analog-to-digital converter in a PC and to a power supply for the amplification circuit. Computer software was developed to collect and store the data, as well as to display the readings of all fourteen sensors in real-time, in two formats: bar graphs showing the pressure amplitude, and strip charts showing pressure vs. time. Although limited by requiring the subject to be tethered, this system gave detailed information about the pressure distribution beneath the foot, and provided those results in real-time.

Data collected with their device has led to a number of papers, including one quantifying the differences between shuffling and walking [5], and between sensate and insensate (no or little sensation in the foot) subjects [6]. In the latter study, the results led to a caution against drawing conclusions from a short segment of gait analysis in patients with sensory impairment, as a large step-to-step variation was found in these patients. This further emphasizes the need for a device capable of collecting data over a long time period.

Instrumented Insole for Gait Timing

In 1994, Hausdorff *et al* [7] developed a simple standalone "footswitch" system capable of detecting several of the temporal gait parameters. Their system consisted of two FSRs on an insole. The insole was cut from tracings of the subjects' feet on a manila folder, and the two FSRs (each square, 1.5 inches per side) were positioned under the heel and in the general area under the toes and metatarsals. The initial work used a circuit with a battery and data storage that was placed in the pants pocket of the subject; following work resulted in a single pack worn on the ankle [8] [9]. After collecting data, it was analyzed and compared to data taken simultaneously on commercial force plates. Calculations by their device found stance duration to be within 3% and swing and stride duration within 5% as compared to the results from the force plate.

Because the outputs of the FSRs were connected in parallel for hardware simplicity, they act as a single combined sensor. This does not affect the case where both sensors are active, or where both sensors are not active. However, this results in a loss of information if only one of the sensors is active, because it cannot distinguish between the two. For the calculations of gait timing, they did not find this to be a drawback; however the outputs of the FSRs on our insole were not combined, so that all the information can be utilized.

They have used the data from their insole to find patterns in gait [10], which they have been able to use to predict the maturation of gait in children [8], and the likelihood of falling in the elderly [9]. This simple device demonstrates that with only two FSRs, some types of abnormalities in gait can be distinguished from normal gait. This device is currently limited by the lack of real-time feedback.

Instrumented Insole for Conditions at the Foot Interface

More recent work resulting in shoe-based sensor systems with increasingly sophisticated measurement capabilities have been driven by sub-specialty interests in gait analysis. For diabetics, Morley *et al* [11] have developed an insole-based system to quantify the condi-

tions inside the shoe, with the goal of being able to predict progression of skin breakdown and ulceration in diabetic patients with peripheral neuropathy.

The laminated insole developed by Morley *et al* had pressure, temperature and humidity sensors designed to investigate the conditions at the foot interface. Combined pressure and temperature sensors were located beneath the heel and the region of the medial metatarsal head, pressure sensors were additionally located in the region of the central and lateral medial metatarsal heads, and a single humidity sensor was located centrally at the toes. Flexible wiring connected the insole to an electronics module and two AA batteries. These were located in a plastic enclosure, which was strapped to the calf of the subject. The data were stored on-board and uploaded to a computer via the serial port. It can currently store 4.5 hours worth of data, but with the implementation of data compression schemes, the data storage is expected to increase to 12-16 hours, to be able to cover a full day. They foresee a potential use of the device as an activity monitor for patients with diabetes, coronary heart disease, and/or obesity, to see if the subjects meet prescribed activity levels.

In initial work with their device [12], they were able to detect quantitatively distinct variations in pressure patterns that corresponded to different activities, and were able to correlate their results with previous studies. They have not yet published work investigating the tracking of the temperature and humidity sensors. Limitations of this device include restricted data storage capacity, a reported breakdown of connections, and the lack of realtime feedback.

Instrumented Insole and Shoe-based Gyroscope Device for Detection of Gait Timing

Another area of research driving devices capable of capturing information about gait is the development of neuroprosthetics used for walking assistance. Neuroprosthetics require inputs to trigger the functional electrical stimulation (FES) used to assist the patient in making the walking motions. Pappas *et al* [13] have developed a shoe and insole device capable of detecting four events during walking: stance, heel-off, swing, and heel-strike, as well as detecting whether the subject is walking or standing. Three FSRs are located on

an insole, one under the heel, and two at the inner and first and fourth metatarsal heads. The two FSRs at the metatarsal heads provided information about asymmetrical loading of the foot. The FSRs were taped onto a 3mm insole, and their positions were adjusted for each subject. Their system also included a gyroscope, which was attached to the back of the shoe, placed such that the sensing axis was perpendicular to the sagittal plane, providing measurements of rotation in the sagittal plane.

They implemented a pattern recognition algorithm with their system. In this algorithm, they divided the gait cycle into two distinct phases (stance, swing) and two distinct events (heel-off, heel-strike). There were seven possible transitions between these (stance to heel-off or directly to swing, heel off to swing or back to stance, swing to heel strike or directly back to stance, and heel strike to stance). Data from the FSRs and from the gyroscope were used to define the transitions. They verified their algorithm by comparing the data with results from a commercial motion analysis system using optical motion analysis (a Vicon 370 from Oxford Metrics Ltd.). In addition to testing their algorithm with walking and running speeds ranging from 0.5 to 12 km/hour, they challenged it with non-walking motions: sliding of the feet, standing up, sitting down, and shifting weight during standing. Their classification algorithm achieved a 99% detection rate for normal subjects and a 96% detection rate for subjects with impaired gait, as compared with the commercial system, with a detection delay of less than 90 ms. These results demonstrate that on-shoe systems with gyroscopes and FSRs are able to achieve comparable results to commercial optical systems.

More recent work [14] has resulted in an insole-only system where the gyroscope and a microcontroller have been embedded in the insole. Using the results from their previous work, the system was used on two subjects with incomplete spinal injury resulting in drop-foot. The system was used to trigger functional electrical stimulation (FES), and they were able to demonstrate a functional benefit of using it, for both subjects, while walking horizontally, uphill, downhill, and while sitting and standing.

This device was developed in order to detect two specific phases (stance, swing) and two specific events (heel strike, heel off) which occur during gait, in order to accurately trigger the electrical stimulation by neuroprosthetics used for walking assistance. As such, it only measures rotation of the foot in the sagittal plane. However, the system described within this thesis will be capable of quantifying motion of the foot in three axes, so it will include two additional gyroscopes, as well as three axes of accelerometers.

2.1.2 Off-Shoe Approaches

In addition to research into on-shoe devices, there has also been work in developing different types of instrumented laboratory spaces. One method of deriving more information about the foot is to instrument the surface on which the subject walks. This approach can be used both in the gait lab, as well as in the clinical setting. Cutlip *et al* [15] have developed an instrumented walkway 4.6 m in length, and have demonstrated the ability to calculate correct values for step period, stance duration and swing duration. Their system is also capable of calculating step length and stride velocity, although it was more accurate at low speeds, and less accurate when the subjects walked more quickly. Giacomozzi and Macellari [16] have developed a "piezo-dynamometric platform" which can be used instead of a force platform. They have shown their system to be highly accurate at calculating the center of pressure of the foot. The ability to measure the pressure distribution at the floor interface allows users of this system to walk without any hardware attached to their shoes or feet. However, these types of systems constrain the walking distance of each trial to the length of the measuring platform. Also, these types of systems do not provide any information about the motion of the foot above the platform.

2.1.3 Gait Recognition Systems

A number of research platforms have been developed to recognize gait without instrumenting the subject. Analysis of videotaped subjects is of particular interest, and has received significant funding from DARPA for the "HumanID at a Distance" program, for potential use as a biometric identifier; earlier work on video analysis of gait was done at the MIT Media Lab by Jim Davis [17]. At the University of Southampton, UK, researchers have been able to recognize subjects from videos, with results better than 80% for walking gait, and better than 90% for running gait, in a study with 20 subjects [18]. Researchers at the MIT AI Lab have achieved similar results, with better than 84% recognition using videos of walking gait, over 25 subjects [19]. At Georgia Tech, work is in progress both using video analysis [20], as well as a separate initiative using radar to analyze the gait cycle [21]. In addition the Aware Home project at Georgia Tech has resulted in a "Smart Floor" that includes ten tiles, each supported by four industrial load cells. The ground reaction force profiles measured across the tiles were capable of correctly identifying subjects 90% of the time, from a sample population on the order of ten people [22]. A group at the University of Oulo, Finland, also used a pressure-sensitive floor to recognize gait; with three successive footsteps, they were able to recognize gait correctly 89% of the time, for a sample of eleven people [23].

2.1.4 Commercial Systems

A variety of shoe interfaces have been developed commercially, with a wide range of applications. Taptronics developed a dance interface with a pair of piezoelectric tap detectors at the toe and the heel [24]. Force sensors have been used by ProBalance [25] for analysis of the golf swing, while inertial sensors have been used by Acceleron [26], Reebok (the Traxtar) [27], FitSense [28], and other companies for other athletic applications (primarily for runners). An example of a runner-specific shoe is the Raven from Vectrasense, a running shoe that detects whether the user is running or walking, and adjusts an air bladder within the frontal area of the shoe, such that the air bladder is filled for running to provide more cushioning [29]. A product poised to become available in late 2004 is the "1" from Adidas, a running shoe with an on-board microcontroller, Hall effect sensors to measure the change in heel compression, and a motor to adjust tension in a stainless steel cord in the heel to achieve the user's desired heel compression [30] [31].

For medical applications, Tekscan and Clevemed, among others, have developed insoles which measure pressure distribution [32] [33]. NCSA's Cyberboots use a pressure sensor array in an overshoe to provide walking interaction in a virtual reality environment [34]. In addition, MiniSun markets "The IDEEA LifeGait System", which uses the outputs of accelerometers placed on various parts of the body with "artificial intelligence" algorithms to determine a number of parameters relating to gait and motion [35].

Of all these products, FitSense and Acceleron have developed systems most closely related to this research [26] [36] [37]. Acceleron has developed a sensor which attaches to the laces or within the insole of a shoe, and measures linear acceleration in three axes, transferring the data wirelessly and in real-time; they have obtained two patents on their technology. In the patents, they describe the use of accelerometers and rotation sensors in conjunction with an electronic circuit which carries out math calculations. They detail the methodology and the equations they use to calculate the distance, speed, and height jumped. In the later patent [37], they describe using radio frequency to send data from the sensors to a wristwatch or a remote device; this patent also describes the inclusion of a GPS device for direction and location information; details about the accuracy of the measurements and calculations were not available. The FitSense FS-1 system similarly attaches to the laces of a shoe, and transmits distance and speed to a watch (it also has an optional heart rate monitor); it has a reported accuracy of 98% [28]. While these systems accomplish some of the goals of this research project, neither system has the extent of sensors used in this research, and neither attempts to fully describe the gait in a manner that can be used as a clinical supplement to the motion analysis laboratory, and could be developed for a recreation sports product.

2.1.5 Expressive Footware: Instrumented Insole and Multiple Shoe-Based Sensors

The work in this thesis developed from the Expressive Footware project developed by Dr. Joseph Paradiso, and students in the Responsive Environments Group at the MIT Media Lab [38]. The Expressive Footware project resulted in a pair of running shoes that were
each equipped with a wireless sensor board and an instrumented insole. Each insole measured dynamic pressure at the heel, bidirectional bend of the insole, the height of each foot above a conducting mat on the floor, and had three FSRs: two placed roughly under the medial and lateral metatarsal heads (to allow the dancer easy control by leaning left or right), and one outside the shoe, mounted at the toe. Each sensor board was permanently attached to the lateral side of the shoe, and contained a gyroscope for the angular rate of the foot about the vertical axis, a three-axis compass to determine the orientation of the foot relative to the Earth's local magnetic field, two axes of acceleration (the two axes in the plane of the sensor card), and three axes of shock acceleration. Finally, an integrated sonar receiver on each sensor board, in conjunction with four sonar transmitters on the floor, provided the position of each foot in the plane of the floor. This system was built for control, not for measurement; the sensor outputs were not saved for analysis, but were used to directly control real-time musical outputs, generated by a computer that interpreted the basestation data stream with an elaborate rule base.

This highly instrumented shoe was worn by dancers and the outputs of the sensors were used to interactively control music. It was completely wireless, with all hardware located directly on the shoe, and provided real-time control of the musical mappings. It reached high acclaim in the dance community, and was recognized with the *Discover Award for Technical Innovation* in 2000. It is shown in Figure 2.1, with the shoe hardware in the foreground, and sensor outputs in the background.



Figure 2.1 Expressive Footware and its sensor outputs

After using this device extensively in interactive dance, Prof. Paradiso became interested in further developing it as a medical tool for quantifying the motion of the foot. This interest, combined with interest from collaborators at the MGH Biomotion Lab led to the initiation of this thesis.

2.2 Current State of the Art

Clinical gait analysis is currently carried out in two very different ways. One is visual observation, and the other is analysis in a motion laboratory.

2.2.1 Observational Gait Analysis

Observational gait analysis (OGA) consists of a well-trained physician or physical therapist assessing patients by watching them walk, either in real-time or on a videotape. This method requires no specialized equipment other than a video camera and no cost beyond the clinician's time and training; however, it is entirely qualitative.

A study on the reliability of OGA in children with lower-limb disabilities examined the ratings on a three point scale, as rated by three experts observing fifteen subjects on video-tape; the raters agreed on fewer than 7 ratings out of 10 [39]. Another study looked at the reliability of analyzing knee motion of three different subjects with gait changes due to rheumatoid arthritis, on videotape, as rated by fifty-four licensed physical therapists; this study found only slight to moderate agreement between the raters [40].

A recent effort reviewed fourteen studies (including the two mentioned above) that investigated the reliability of OGA, and found that the majority of studies concluded that qualitative observation of gait has poor to moderate reliability [41]. The authors also evaluated OGA, using eighteen physical therapists who routinely use OGA to assess changes in gait following a stroke. The therapists were shown a video, and asked to evaluate ankle power generation by rating each subject on a 22-point scale; the video included an audible tone at heel strike. The subjects were also evaluated using reflective markers and a camera-based motion analysis system. Under these highly systematized evaluation conditions, the therapists were able to demonstrate moderate to high reliability in their ratings, as compared with the results from the motion analysis system.

2.2.2 Overview of Clinical Gait Analysis

Comprehensive gait analysis is generally used for the assessment of a patient with a movement disorder. There are as many as five components to the gait analysis [42]:

- 1. *Videotape Examination*: to observe gait abnormalities [in slow motion or freeze-frame]
- 2. *Temporo-Distance Parameters*: cadence, stride length, speed [may be measured manually]
- 3. *Kinematic Analysis*: measurement of movement [usually measured with cameras, LEDs, IR]
- 4. *Kinetic Measurement*: forces between foot and ground [usually measured with force plate]
- 5. *Electromyography*: electrical activity of muscles [surface or fine wire electrodes]

All components are not necessarily used, especially when a motion analysis lab is utilized, as often the bulk of the gait analysis is performed employing the data from the kinematic analysis and kinetic measurements. Data from these can also be used to calculate the temporo-distance gait parameters. Electromyography (EMG) is used less often than the other techniques; surface electrodes may have trouble sensing deep muscles and are less accurate than the data obtained from fine wire electrodes, which can be painful to the patient [43].

While this is a very accurate method of measuring all the parameters of gait, it requires expensive equipment, and a dedicated lab space, usually a minimum of 10 meters x 10 meters. This size means that subjects cannot walk very far before stopping and turning around. Investigators have found that, in general, a minimum of two trials are needed and better results are achieved when data from multiple trials are averaged, since data from a single trial are too variable to rely on alone [44]. In addition, the subject must step directly

on the force plate in order to obtain an accurate measurement; however, asking the subject to aim for the force plate may result in an alteration of the subject's gait, called "targeting."

The following sections discuss different methods used for kinematic analysis; the technology has been driven in part by the computer graphics and animation industries. For example, a group at Laboratoire d'Electronique de Technologie de l'Information (LETI), France, is investigating the use of three axis sensors (with accelerometers and magnetometers) for the purpose of using these small sensors in wearable clothing to improving motion capture for 3D virtual worlds [45].

2.2.3 Optoelectronic Systems

An optoelectronic system involves placing light emitting diodes (LEDs) on the subject. The LEDs are turned on sequentially by a computer, and viewed by a camera. Because the computer triggers the LEDs, there is no question about which LED is viewed by the camera at a given time point. However, reflection off the floor, or other surfaces reflective to infrared, such as human skin, can reduce the accuracy of the system.

The system in use at the Massachusetts General Hospital (MGH) Biomotion Lab uses a Selspot II system (Selective Electronics, Partille, Sweden) to serially sample up to 64 infrared LEDs, arranged in arrays, at a rate of 153 Hz. The LED arrays are placed on eleven body segments (bilaterally: feet, shanks, thighs, arms; and, the pelvis, trunk, and head). The TRACK kinematic data analysis software package is used to generate photostereogrammetric reconstruction of the 3-D positions of the LEDs and to define the six degree of freedom kinematics of the arrays [46]. Within the viewing volume¹, this system is capable of accurately defining the 3D positions of each body segment to within 1 mm, and the three orientations to within 1 degree, though actual results during testing may vary. With the technology currently in use, this system requires the subject to be wired

^{1.} The viewing volume is the area of the room visible to the cameras that sample the LED output; in the Massachusetts General Hospital Biomotion Lab, the viewing volume has a width just under 2 m along the direction of forward gait.

("tethered") to the computer, though it is likely possible to convert the system to a wireless system.

In addition, two Kistler piezoelectric force plates (Kistler Instruments Type 9281A, Winterthur, Switzerland) are used to acquire ground reaction forces; this system has an accuracy of $\pm 1\%$ of full scale; as set in the MGH Biomotion Lab (BML), this corresponds to ± 10 N of vertical force, and ± 5 N of shear force, for forces and frequencies encountered during gait (the unloaded force plate is recalibrated to a load of 0 N after each gait trial) [47] [48]. A photo of a subject instrumented with the MGH Biomotion Lab equipment is shown in Figure 2.2.



Figure 2.2 A subject at the MGH Biomotion Lab

2.2.4 Videographic Systems

Systems using videography with reflective markers are the most frequently used system in motion analysis labs. This type of system involves placing markers which are highly

reflective on the subject. The markers are illuminated and viewed by cameras; illumination is generally achieved with infrared lights or an infrared strobe located near the camera. This type of system allows the subject to walk untethered, however, not all markers may be illuminated a given time point. This results in the need for significant post processing to sort and identify the markers.

Two major manufacturers of such systems are Vicon Motion Systems and Motion Analysis Corporation. Vicon systems can be set up with as many as 24 cameras; the top of the line M2 camera has 1280 x 1024 resolution (with a digital CMOS sensor), and can capture up to 1000 frames per second [49]. Motion Analysis Corporation has more than 600 systems installed in motion laboratories worldwide. Its premier system, the Eagle Digital, can be set up with as many as 64 cameras, and also has a 1280 x 1024 CMOS sensor. At this resolution it can capture 480 frames per second [50].

2.2.5 Electromagnetic Systems

Electromagnetic systems involve having a stationary transmitter which emits a magnetic field, and instrumenting the subject with electromagnetic coils, which detect this field. A benefit of this type of system is that there are no "line of sight" requirements, as the relatively low-frequency magnetic field lines easily penetrate human tissue and non-conductive objects. However, the receivers must be within the range of the transmitter. At this point, electromagnetic systems are not widely used in gait analysis, most likely because the systems currently available only track a small number of points. In addition, the electromagnetic field is vulnerable to distortion by magnetically susceptible materials in the vicinity of the system. However, these systems are currently used in other areas of motion research, such as hand or head tracking, and may be of interest for gait analysis when they have the ability to track a greater number of points.

The two best known systems of this type are made by Polhemus and Ascension Technology Corporation. Polhemus uses an AC magnetic field; its FASTRAK® system is advertised as having an accuracy of 0.03 inches RMS for position and 0.15 degrees RMS for orientation, and a resolution of 0.0002 inches and 0.025 degrees per inch distance from the transmitter (4-6 feet are recommended, but up to 10 feet is possible). It can track up to four sensors per transmitter, and up to four transmitters can be used at once, providing the ability to track sixteen sensors [51] [52]. Ascension Technology Corporation uses a pulsed DC magnetic field; its Flock of Birds® system can track up to four sensors, and is advertised as having an accuracy of 0.07 inches RMS for position, and 0.5° RMS for orientation, with a resolution of 0.02 inches and 0.1° when 12 feet from the transmitter. Alternatively, Asencion's MotionStar Wireless® system can track up to twenty sensors, and is advertised as having an accuracy of 0.6 inches RMS for position, and 1.0° RMS for orientation, with a resolution of 0.1 inches and 0.2° when 10 feet from the transmitter. [53].

2.3 Clinical Need

The current clinical methods of analyzing gait fall at two extremes - on one end is observational gait analysis, which is inexpensive but qualitative. At the other end is analysis in a motion laboratory, which is quantitative but expensive. In both methods, the subject is very aware of being observed and analyzed, which is likely to affect the gait of the subject.

An on-shoe system could provide the benefits of both methods without the drawbacks. It could be far less expensive than the motion lab, and could provide quantitative output about the gait. If unobtrusive, it could measure the gait of the subject while the subject is unaware of being tested, over an extended period of time.

In addition, an on-shoe system could ultimately provide unique features. Since it is mounted on the shoe of the subject, the patient can be sent home to monitor gait throughout the day, in a variety of environments, and in a variety of situations. In addition, it could analyze the gait in real-time, which would allow it to provide the wearer feedback of various types (e.g. musical, tonal, visual, tactile, electro-stimulation), which could be useful for physical therapy or gait training. There are many types of patients who may benefit from such real-time feedback; the subject testing in this thesis included investigation of the gait of subjects with Parkinson's disease (PD). Finally, an on-shoe system could be used in localities (e.g. rural areas or third-world countries) where patients do not have access to a motion lab facility.

2.3.1 Utility of On-Shoe Device in Subjects with Parkinson's Disease

This instrumented shoe is likely to be useful for a wide range of gait conditions; in order to begin evaluation of subjects with altered gait, the subject testing for this thesis included five patients with Parkinson's disease (see Appendix A.2 for more information on Parkinson's disease). Future interests for this system include the ability to use this system to analyze the gait of PD subjects in their home environment, as well as to provide real-time auditory feedback to subjects with PD.

Changes in Gait Due to Parkinson's Disease

Hausdorff *et al* [54] investigated changes in gait variables in subjects with PD. Using their system described in Section 2.1.1 (see page 31), they recorded data from control subjects, PD subjects, as well as subjects with Huntington's disease. They found a statistical difference in the speed (1.35 m/sec for controls, 1.00 m/sec for PD) and in the "double support time," which is the duration both feet supported the patient simultaneously (305 msec for controls, 376 msec for PD). These are both parameters that can be easily measured by our device; these are important to keep in consideration when designing the study with PD subjects.

Changes in gait parameters between walking and shuffling were examined by Wertsch *et al* [5] using their system described in Section 2.1.1 (see page 30). Predictably, they found that the average velocity and stride length to be lower during shuffling (0.51 m/sec and 0.63 m) than during walking (1.29 m/sec and 1.55 m). They found that peak pressures were lower at all fourteen sensor sites (seven per foot) during shuffling, while foot-to-floor contact duration was increased at all fourteen sensor sites. The decreases in peak pressures ranged from 7.0% at the fifth metatarsal to 63.2% at the great toe, with a decrease of 41.6% in peak pressures summed over the entire foot. The increase in contact

duration ranged from 22% at the fifth metatarsal to 76.9% at the heel. The findings in their study may be of use in analyzing the data which will be collected from PD subjects.

A Clinical Need to Evaluate PD Subjects Outside the Motion Lab

Morris *et al* [2] have evaluated the biomechanics and motor control of gait in subjects with PD. Their paper includes a tabular listing of eighteen studies involving subjects with PD, and lists the medication state of the patients. Morris *et al* recommend a change from the current trend of lab based studies investigating PD subjects in straight line walking, towards studies in their homes and communities with more complex gait activity.

Effect of Rhythmic Cues on Subjects with Parkinson's Disease

A recent study found that auditory rhythm and the resulting physical response shows great potential as a therapeutic method for rehabilitating patients who have movement disorders [55]. The authors initially investigated the effect of "rhythmic auditory stimulation" (RAS), provided to thirty-one subjects with PD for three weeks. With RAS at 10% faster than each subject's baseline cadence, significant improvement was found in mean gait velocity, cadence, and stride length, both for the twenty-one subjects on medication, and the ten subjects off medication; in subjects on medication, the mean gait velocity increased by 36%, and in subjects off medication, the mean gait velocity increased by 25% [56]. The authors followed this with a twelve week study of twenty-one subjects; the subjects were pretested, used RAS, using audio tapes, for 30 minutes daily for three weeks, and were post-tested at seven weekly follow-ups, and a final post-test during the twelfth week (RAS was not used after the first three weeks). The gains in stride length, cadence, and velocity were maintained for 3-4 weeks after the training period, but most values returned to pretest values by the fifth follow-up week [57].

Another group studied the effects of "musical therapy" (MT), as compared to physical therapy (PT). Thirty-two subjects with PD were randomly split into two groups of sixteen: one group received weekly sessions of MT for three months, the other received weekly sessions of PT for three months. The MT sessions involved a variety of active music par-

ticipation, including choral singing, voice exercises, and rhythmic movement, and the PT sessions involved a variety of motions designed to improve balance and gait, including passive stretching and specific motor tasks.

The study used standard scales for rating the subjects: the Unified Parkinson's Disease Rating Scale: Motor Subscale (UPDRS-MS) was used with the Wilcoxon signed-rank test to evaluate the significance of changes in movement. The Unified Parkinson's Disease Rating Scale: Activities of Daily Living (UPDRS-ADL) and the Parkinson's Disease Quality of Life (PDQL) were used with the Mann-Whitney U test to evaluate the significance of changes in quality of life. The subjects in the MT group demonstrated significant improvement (p<0.0001) in UPDRS-MS scores, with particular improvement (p<0.0001) of bradykinesia (slowness of movement). Subjects in the PT group did not show significant improvement in either the UPDRS-MS scores or in bradykinesia. In addition, only subjects in the MT group showed significant improvement in UPDRS-ADL (p<0.0001) and in PDQL (p<0.0001). However, only subjects in the PT group had significant (p<0.0001) improvement in rigidity. In addition, at follow-up two months after the end of the study, the improvement in rigidity found in the PT group had persisted, while the parameters improved by MT had returned to their baseline values [58].

These initial studies investigating the use of auditory feedback for persons with PD suggest that such techniques may provide a method to improve motion, and even quality of life. The system described in this thesis could open up new avenues of therapy by allowing auditory feedback to be tailored in real-time to the movement of the subject, and by providing the subjects access to therapy in their home environments.

2.3.2 Studying Gait Outside of the Motion Lab

As mentioned in Section 2.3.1 (see page 45), subjects with PD would greatly benefit from a system capable of quantitatively analyzing gait outside of the motion lab. In addition, such a system could be very useful in areas where a motion lab is not accessible.

From the limited research available regarding gait analysis in third world countries or rural areas, it is clear that an inexpensive and quantitative method of studying gait is needed. For a study investigating recovery of gait after a stroke in Soweto, South Africa, a paper survey was used to ask patients about their recovery. In this survey, the ability to catch a taxi in Soweto was used as a measure to assess the patient's gait handicap [59].

A study in Germany investigated changes in gait in healthy subjects and in subjects with PD, with forty-three subjects recruited from an urban area (Berlin), and forty-seven subjects recruited from a rural / semi-urban area (Innsbruck and surrounding Tyrol). To evaluate gait, a system was used which involved attaching threads with Velcro straps to each foot at the second metatarsal head [60]. The threads were attached to a pulley system, and rotations of this system were measured by an optical recording device as the subject walked (the thread length allowed the subject to walk up to 10 m). All subjects were evaluated in a quiet environment with a gray colored walkway. This system provided measurements of parameters such as stride length, stride duration, and cadence. The study found that within each group (urban or rural), subjects with PD walked with a slower cadence than their healthy counterparts; additionally, subjects in the urban group (including those with PD) walked with a faster cadence than the subjects in the rural group [61].

A system which is not confined to evaluating gait within the motion lab would have great benefit in allowing better testing and evaluation of subjects who do not live near motion lab facilities. Judging from the lack of papers investigating the gait of subjects in rural areas, as well as the low levels of sophistication of the methods employed, such a system would open up a new venue of research. and greatly benefit patients who have no access to fully-equipped motion labs.

Chapter 3

HARDWARE DESIGN

The focus of this thesis was the design and implementation of the on-shoe system used for the measurement of gait. Section 3.1 discusses sensor selection, and Section 3.2 describes the physical implementation. Section 3.3 describes the function of each sensor in further detail. Section 3.4 describes additional electronics used in the system, and Section 3.5 summarizes the overall design of the "GaitShoe" system.

3.1 Sensor Selection

The first step in designing the GaitShoe was to select the appropriate sensors, with the goal of creating a highly instrumented system capable of sensing many parameters which characterize gait. As discussed in Section 1.4, several important parameters of gait were identified: heel strike timing, toe off timing, dorsi-/plantar- flexion, stride length, and stride velocity. Additional parameters of interest include global rotations of the foot, distance moved and velocity in the vertical and side-to-side axes, the pressure distribution underneath the foot, and orientation of the feet relative to each other.

Timing Parameters and Pressure Distribution

To assess the timing parameters and pressure distribution, force sensitive resistors (FSRs) and polyvinylidene fluoride (PVDF) strips were selected to be placed underneath the foot.

A FSR is a sensor whose electrical resistance decreases as the applied load increases. Two FSRs were placed underneath both the first and fifth metatarsal heads, and two more were placed medially and laterally underneath the heel pad. While use of only four force sensitive resistors will not provide a full picture of the force distribution beneath the foot, the number is sufficient to provide a general picture of medial vs. lateral force, and heel vs. metatarsal force. The assessment of heel vs. metatarsal force provides information to be used in determining stance time, and thus heel-strike and toe-off timing.

The PVDF strips are piezoelectric sensors, which were configured to provide an output corresponding to dynamic pressure. They were chosen for their fast response time, and were selected to be located directly beneath the heel and the great toe in order to provide additional information about heel-strike and toe-off timing.

Other force or pressure sensors were considered. In particular, fine-grain printed arrays of FSRs, such as Tekscan's F-Scan® system provide extensive information about the pressure distribution underneath the foot [32]; however, the high cost of Tekscan's proprietary system rendered it a prohibitive choice.

Flexion

Two bi-directional bend sensors were selected for use in analyzing flexion during gait. The resistance of the bend sensors changes as the sensor is bent. One of the bend sensors was located at the back of the heel, and held next to the shin by an ankle bracelet to provide information about plantar flexion and dorsiflexion. The second bend sensor was located in the insole approximately centered on the metatarsals, to provide information about flexion at the metatarsals.

Other methods of measuring dorsi-/plantar- flexion were considered, such as capacitance between foot and shin, or ultrasound measurements between the foot and the shin. Ultimately, a back-to-back pair of resistive bend sensors were selected due to their unobtrusiveness and ease of implementation. Fiber optic bend sensors such as those made by Measurand[®] were also considered because of their high accuracy and resolution, but were prohibitively expensive [62].

Distances, Velocities, and Orientations

Three gyroscopes and two dual-axis linear accelerometers were chosen to be placed at the back of the shoe. By orienting the gyroscopes and accelerometers such that the individual sensing axes are aligned along three perpendicular axes, the angular velocity and linear acceleration can be measured in three-dimensions. Hence, velocity and stride length can be obtained respectively from single- and double-integration, with respect to time, of the acceleration component corresponding to forward motion of the foot. Displacements and velocities in the other two axes can be similarly acquired. The gyroscopes provide information about the rotation of the foot, which can similarly be integrated once with respect to time to provide the angle. The use of all six measurements can therefore be used to analyze the orientation of the foot; a device capable of all six measurements is called an "inertial measurement unit" (IMU).

Late in the project, two additional types of sensors were implemented: an electric field sensor and an ultrasound sensor. The electric field sensor was added to investigate the utility of using a more direct method of measuring the height of the foot above the floor, via capacitive loading [63]. Using multiple electric field sensors would allow the height of the foot to be measured at discrete locations, such as at the heel and the toes. The ultrasound sensor was added to provide a method of measuring the distance and relative orientation between the two feet. In addition, an ultrasound sensor could also be used to measure the height of the foot above the floor.

Summary

Table 3.1 summarizes the parameters of interest, along with the type of sensor(s) selected for each parameter, and the corresponding sensor output.

Parameter	Sensor	Sensor Output
Heel-strike timing and toe-off timing	FSRs, and PVDFs	<i>FSRs:</i> Resistance change corresponding to applied force across the sensor, resulting from change in displacement of the sensor. <i>PVDFs:</i> Voltage change corresponding to dynamic pressure across the sensor.
Dorsi-/plantar- flexion	Bend sensors	Resistance change corresponding to flexion angle, resulting from strain of the sensor.
Stride length and stride velocity	Accelerometers	Voltage change corresponding to acceleration; sin- gle integration of forward acceleration yields velocity, double integration yields distance (inte- gration occurs after correcting for gravitational component).
Orientation	Gyroscopes	Voltage change corresponding to angular velocity; single integration yields angle of rotation.
Stance width	Ultrasound	Time of flight corresponding to distance.
Displacement side-to-side	Accelerometers	Voltage change corresponding to acceleration; double integration of lateral acceleration yields distance.
Height of foot above floor	Accelerometers, electric field, and ultrasound	Accelerometer: Voltage change corresponding to acceleration; double integration of vertical acceleration yields distance. <i>Electric field:</i> Capacitance corresponding to distance. <i>Ultrasound:</i> Time of flight of reflections, corresponding to distance.

3.2 Physical Implementation

In designing the GaitShoe hardware, the top-level functional requirements, as listed in Section 1.3, were considered:

- 1. Effect no change in gait.
- 2. Characterize the motion of both feet.
- 3. Be untethered.
- 4. Allow the subject to use his or her own shoes.

The most important requirement is the first, so in order not to cause changes in the gait, the hardware had to be small, compact and lightweight. Minimizing the weight is important because it has been shown that subtle effects on the gait occur when the lower-extremities are loaded with weights which are on the order of 1-2% (or greater) of the body weight [64]. Therefore, by keeping the mass of the final prototype under 300 g, it is not expected that the adult subjects would experience any change in gait due to the weight of this system; 300 g is 1% of 30 kg (66 pounds); subject testing was carried out on adults, all of whom weighed more than 45 kg (100 pounds).

Both shoes were instrumented, which satisfied FR2. To satisfy FR3, each shoe had its own power supply, and a wireless transceiver¹, based on radio frequency (RF), was used to transmit the data to a basestation connected to a laptop.

To meet FR4, the hardware was designed to be readily attachable to shoes and removable without causing any damage; and to comply with FR1, the attachment to the shoes was designed to not interfere with walking.

The design of the hardware needed to accommodate the sensors that must be located beneath the foot, all of the electronics (including additional sensors), an antenna for the wireless transmission, and the power supply. These requirements resulted in the design of the GaitShoe system, as shown in Figure 3.1.

The GaitShoe system was comprised of two shoe modules and a basestation. Each shoe module consisted of an instrumented insole placed beneath the foot, and an attachment which mounted to the back of the shoe. The instrumented insoles contained the force sensitive resistors, the polyvinylidene fluoride strips, one bend sensor, and part of the electric field sensor; the other bend sensor was connected to the insole, but placed behind the shin and held in place with an ankle strap, and an additional part of the electric field sensor was placed underneath the shoe. The shoe attachments contained a "stack" of printed circuit boards containing the IMU sensors, general electronics, the antenna, and the power supply, as well as the electronics and the hardware for the ultrasound sensor [65]. The

^{1.} Another option for wireless data collection was to store the data on-board and download it later; however, wireless transmission was selected since it allowed the data to be analyzed in real time.

basestation received the data from both shoes, and transmitted the data to a computer via the serial port. Each part of the GaitShoe system is described below.



Figure 3.1 Schematic of the GaitShoe system

3.2.1 The Stack

The electronics used a stacking platform initially developed by Ari Benbasat [66], and redesigned to meet the requirements of the GaitShoe [65]. The idea was to make several small printed circuit boards which can be stacked together, resulting in a compact volume, rather than one large printed circuit board. The GaitShoe stack is shown in Figure 3.2.

Each individual circuit board contains sensors and electronics that meet a specific function; each board will be briefly discussed below, and the sensors and electronics are discussed in detail in the following sections.



Figure 3.2 The stack

The Main Board

The Main board contained the microcontroller, wireless transceiver, antenna connection, and the connection for the inputs from the power board. It is shown in Figure 3.3 (attached to the Power board). The Main board controlled the collection and transmission of the data from all of the sensors.



Figure 3.3 Photo of the front and back of the Main board

The IMU Board

The inertial measurement unit (IMU) board contained the two dual-axis accelerometers, and the three gyroscopes. The accelerometers and gyroscopes were oriented such that the board was capable of measuring angular velocity and linear acceleration about three axes. It is shown in Figure 3.4.



Figure 3.4 Photo of the front and back of the IMU board

The Tactile Board

The Tactile board is shown in Figure 3.5. It contained all the electronics for the force sensitive resistors, the bi-directional bend sensors, the polyvinylidene fluoride strips, and the electric field sensors. The electric field sensors connected to a header via co-axial cable, and plugged into the small receptacle at the side of the board. The rest of the sensors were connected to a header via ribbon cable, and plugged into the large receptacle at the top of the board.



Figure 3.5 Photo of the front and back of the Tactile board

The Ultrasound Boards

The Ultrasound boards, designed by undergraduate Steven Dan Lovell, contained the electronics for the ultrasound sensor. As shown in Figure 3.6, there were two versions, one each for the left and the right shoes. The ultrasound sensor measured the distance between and the angle between the two feet. The left ultrasound board contained an ultrasound transmitter and the corresponding electronics, and the right ultrasound board contained the connections for two ultrasound receivers and the corresponding electronics.



Figure 3.6 Photos of the front and back of the transmit Ultrasound board for the right foot (right photos), and the front and back of the receive Ultrasound board for the left foot (left photos)

The Power Board

The power board was developed to provide each GaitShoe system with an on-board power supply. It contained connections for a +9 V battery, an on-off switch, indicator lights, a fuse, voltage regulators for +3.3 V, +5 V, and +12 V, and a connector to transmit the ground and power lines to the main board. It is shown in Figure 3.7.



Figure 3.7 Photo of the front and back of the Power board

3.2.2 Insole Design

An insole was designed to accommodate the several sensors which were connected to the electronics via ribbon cable, as described in Section 3.2.1: The Tactile Board. The insole, shown in Figure 3.8 contained the four FSRs (yellow), the two PVDF strips (green), and one of the bi-directional bend sensors (magenta). The insole provided a straightforward method to position the sensors in the correct locations beneath the foot (see also Figure 3.1).



Figure 3.8 Photograph of an insole sensor

The second bi-directional bend sensor was connected to the electronics via the same ribbon cable. It was placed between the back of the shoe and the shin, and attached to the ankle with a nylon ankle-strap, as shown in Figure 3.9. The nylon ankle-strap was fitted around the subject's ankle, and held the bend sensor against the ankle, so to minimize buckling.



Figure 3.9 Bend sensor in ankle strap

When the electric field sensor was used, a ground plane was placed on top of the insole, and the driven shield of the electric field sensor was placed on the bottom of the insole; the sensors of the electric field sensor were attached to the bottom of the shoe. The ground plane, driven shield and electric field sensors used the smaller header to connect, as described in Section 3.2.1; a cross section of the shoe showing the layout of all these parts is shown in Figure 3.10.



Figure 3.10 Cross section showing the electric field sensor components

To protect the insole sensors from the humid conditions of the shoe, the sensors were enclosed between two 0.02 inch sheets of clear Type 1 polyvinyl chloride (PVC) heavy duty film sheets. PVC sheets are rated to withstand humid and warm environments for up to three years. Insoles could be made in multiple sizes; for testing, one size was used, and sensors were moved to adjust for each subject's foot size.

3.2.3 Shoe Attachment

A shoe attachment was designed to hold the stack of printed circuit boards, the power board, and the antenna. It was designed such that the bulk of the volume was located behind the heel, so as to have a minimum effect on the gait. The attachment was made from 0.125 inch polyethylene terephthalate glycol (PTG) sheets. PTG is thermoformable, machinable, and shatter-resistant.



Figure 3.11 Photos of three versions of the shoe attachment

Three versions of the shoe attachment, shown in Figure 3.11, were designed, both to accommodate changes in the stack hardware and to improve the mechanical attachment. On the far left is the first prototype, which had three large screws at the top to fasten the attachment to the shoe. The middle photo is the second prototype, which had two screws at the left and the right, and the power board was rotated to the side to decrease the length of the attachment. The final prototype featured longer plastic hooks at the location of the two set screws to help keep the attachment stable on the shoe, included a mounting point for the antenna, and improved alignment between the main stack and the power board. Both the second prototype and the final prototype had a plastic loop at the bottom of the attachment, through which fishing line was threaded, looped under the shoe, and tied to the laces, to reduce motion of the attachment during heel strike. The left and right attachments of the final prototype are shown in Figure 3.12 (directions for fabricating the attachments are in Appendix D.3).

The attachment was designed for walking shoes. The plastic loops pulled the flexible shoe material of the walking shoes away from the foot, so that the plastic loops did not rub against the foot. The top of the shoe attachment was at the same height as the back of the shoe, which helped to keep the shin from hitting the electronics while walking. As most walking shoes have a large heel, the height of the shoe attachment was short in relation to the shoe height, which resulted in a large clearance between the bottom of the shoe attach-

ment and the floor, and helped to keep the attachment from hitting the floor while walking. The mounting points for the antennas were on the outside of each foot, in order to not affect the clearance between the two feet.



Figure 3.12 Final prototype of the GaitShoe attachment

The attachments are shown on a pair of shoes in Figure 3.13. The fishing line which held the bottom of the attachment against the shoe and was tied through the eyelets is visible on the side of the shoes. The co-axial cable connected to the electric field sensors on the bottom of the shoe is visible on the left shoe.



Figure 3.13 Photo of the GaitShoe system on two shoes

The fishing line helped to diminish vibration of the shoe attachment as a result of heel strike. Figure 3.13 shows the output of the accelerometers when the heel of an empty shoe is hit against the floor. The accelerometers demonstrate an excitation frequency around 20 Hz, which is quickly damped.



Figure 3.14 Accelerometer response to impact at heel

3.2.4 The Basestation

The final component of the GaitShoe system was the basestation. The basestation consisted of a metal box, with an antenna mounted externally. Inside the metal box was a main board, a power board, and a programming board (described in Appendix D.1.5) containing a MAX233 serial converter chip, to send the data to a laptop or desktop computer via the serial cable. The basestation is shown in Figure 3.15; on the outside of the box are a power switch, indicator lights, and a female DB-9 connector for the serial cable. In addition, a BNC connector provided a method of input from the MGH Biomotion Lab equipment via an optoisolated trigger connected to an input pin on the microcontroller.



Figure 3.15 Photo of the basestation

3.3 Sensor Specifications

This section discusses each sensor, including an overview of the working principle of the sensor, and the signal conditioning and implementation used in the GaitShoe system. Schematics of each sensor implementation are included and analyzed, and a table summarizing relevant parameters of each sensor (as provided by the manufacturers) is included; sensitivities, zero offsets, and cutoff frequencies refer to the output after any conditioning electronics shown in the schematics.

Bandwidth Requirements

The average step rate of adults is just under 120 steps per minute, which corresponds to a stride frequency of 1 Hz. The harmonic content of seven leg and foot markers was analyzed, and 99.7% of the signal power was found to be contained below 6 Hz [67]. In addition, it has been shown that if data is collected at 24 Hz, there will be negligible errors in the kinetic and energy analyses for normal gait speeds [68]; however, for kinematic analysis, a higher sampling rate, such as 50 Hz or above, may be necessary to capture all the information [69]. Thus, the sensors in the GaitShoe were all selected and set such that the low-pass cutoff frequencies (the 3 dB bandwidths) were greater than 25 Hz.

3.3.1 Accelerometer

The accelerometer selected for use in this application was the ADXL202E, a dual axis linear accelerometer from Analog Devices. The ADXL202E is a small, low-power, microelectro-mechanical system (MEMS) accelerometer, and acceleration measurements have a full-scale range of $\pm 2g$.

Two of these dual-axis accelerometers were used, oriented¹ perpendicularly to each other. This resulted in a single measurement along each of two axes, and in duplicate measure-

^{1.} The relative orientation between the two accelerometers, in the plane defined by the axis in one accelerometer corresponding to the vertical axis, and the axis in the second accelerometer corresponding to the horizontal axis, was determined using the gravitational vector, as described in Section 4.4.2.

ments along the third axis; one of the measurements along the duplicate axis was discarded.

Working Principle

The ADXL202E is constructed using a surface micro-machining process. The proof mass is suspended above the substrate by polysilicon springs, and can move in two perpendicular axes. Along each of the four sides of the square proof mass are eight sets of fingers which are positioned between plates affixed to the substrate; each finger and pair of plates results in a differential capacitor. When subjected to acceleration, the proof mass moves from its neutral position. Changes in the differential capacitances correspond to deflection of the proof mass due to acceleration [70].

This type of accelerometer measures both dynamic acceleration (resulting from shock, vibration, linear motion, or other types of motion), and static acceleration (resulting from gravity). Thus, for analysis of acceleration due only to linear motion, the orientation of the accelerometer must be determined so that the gravity contribution can be subtracted from the total output.

Implementation

The ADXL202E provides the acceleration output as either a digital duty-cycle or an analog voltage. The analog output was used in the final design¹ of the IMU board. Figure 3.16 shows the schematic of the circuitry used for one of the accelerometers, Figure 3.17 shows the placement of the components for both ADXL202E sensors on the IMU board, and Table 3.2 lists relevant parameters for this implementation. The bandwidth was limited externally by adding 47 nF capacitors, C_1 and C_2 , at the X_{filt} and Y_{filt} pins, resulting in a cutoff frequency of 100 Hz. A 0.1 µF capacitor, C_3 , was placed between the V_{dd} pin and ground to decouple the 3.3 V power supply. A ferrite bead, R_6 , was placed between the

^{1.} An earlier implementation used the duty-cycle output, but timing the duty-cycle with the microcontroller was very time-intensive (the shortest duty-cycle available was 1 ms), so the analog output was used in the final version to enable fast data acquisition (the other sensor outputs are all analog, as well).

power supply and the V_{dd} pin to further reduce digital noise (as recommended when a microcontroller shares the power supply; a 100 Ω resistor can also be used if a ferrite bead is unavailable). The resistor, R_5 , at pin 2 was provided for the duty cycle converter and was set to 125 k Ω (a resistor under 10 M Ω is recommended to keep the duty cycle converter running, even when the analog output is used).



Figure 3.16 Schematic of the ADXL202E



Figure 3.17 Photo of the IMU board, with the ADXL202E components highlighted

The X_{filt} and Y_{filt} pins have an output impedance of 32 k Ω ; since they cannot directly drive a load, the output of each pin was buffered by a non-inverting op-amp follower. The relationship between the voltage, V_{filt} , on pin X_{filt} , and the output voltage from the op-amp, V_{out} , is described by Eq. 3.1, where V_{ref} is the voltage supplied to the inverting input

through R_I . The two resistors, R_I and R_2 , were both set to 10 k Ω , for a gain of 2. The reference voltage, V_{ref} , was set to 1.6V, in order to keep the output centered around 1.6 V. The 1.6 V was provided through use of a voltage divider connected to the 3.3 V power supply; the 3.3 V was divided via a 200 k Ω potentiometer, RV_I , and buffered by a non-inverting op-amp with a gain of 1.

$$V_{out} = \left(1 + \frac{R_2}{R_1}\right) \cdot \left(V_{filt} - V_{ref}\right) + V_{ref}$$
(3.1)

Capacitor C_5 , in parallel with R_2 , reduces extraneous noise, with a cutoff frequency of 160 Hz. The same is true for the output from pin Y_{filt} . Finally, the power pin on the opamp has a 0.1 µF capacitor, C_4 , to minimize noise from the power supply; the voltage divider which provides the 1.6 V also has two 0.1 µF capacitors, C_7 and C_8 , to reduce noise in the supply line [71].

Parameter	Value
Measurement range	±2g
Sensitivity	Typical: 377.5 mV/g, min: 313.0 mV/g, max: 452.5 mV/g
Zero offset	Typical: 1.65 V, min: 1.0 V, max: 2.3 V
Cutoff frequency	100 Hz
Resonant frequency	10 kHz
Nonlinearity	0.2% full scale
Temperature sensitivity	2.0 mg/°C from 25°C
Noise floor	Typical: 200 $\mu g/\sqrt{Hz}$ rms, max: 1000 $\mu g/\sqrt{Hz}$ rms
RMS noise	Typical: 2.5 mg
Peak-to-peak noise estimate	95% probability: 10.1 mg
Quiescent supply current	Typical: 0.6 mA, Max: 1.0 mA
Power draw	Typical: 1.98 mW
Shock survival	Up to 1000g while unpowered, up to 500g while powered
Package size	5 mm x 5 mm, 2 mm tall
Package weight	<1.0 grams

TABLE 3.2 Relevant parameters of the Analog Devices ADXL202E accelerometer [71]

3.3.2 Gyroscopes

Two types of MEMS vibrating rate gyroscopes were used in this application: the Murata ENC-03J gyroscope, and the Analog Devices ADXRS150 gyroscope. The ADXRS150 is a yaw gyroscope, meaning it measures rotation about the axis perpendicular to the plane of the sensor. In contrast, the ENC-03J measures rotation about the long axis in the plane of the sensor. In order to measure rotation about three axes on one flat circuit board, two ENC-03J gyroscopes were placed nominally perpendicularly¹ to each other, with the ADXRS150 placed in the same plane.

Working Principle

Both types of gyroscopes measure angular velocity through use of a vibrating element and the Coriolis effect. When rotation is applied to a vibrating element, conservation of angular momentum results in a secondary oscillation orthogonal to both the axis of vibration and the axis of rotation (this is the Coriolis effect). The magnitude of this secondary oscillation is proportional to the magnitude of the rate of rotation [72].

In the Murata ENC-03J, the vibrating element is a piezoelectric ceramic prism, which is supported such that it can move freely within the plane normal to its long axis. The prism is driven at its resonant frequency by a piezoelectric transducer. When rotation occurs about the long axis of the prism, the resulting secondary oscillation is measured by a second piezoelectric transducer. This type of system is susceptible to external vibrations. To minimize resonant coupling when two of these sensors are placed in close proximity to each other, Murata sells an "A" and a "B" version, which have slightly different resonant frequencies [73].

In the Analog Devices ADXRS150, the vibrating element is a polysilicon structure. This structure consists of a mass tethered to an inner frame, which is tethered to the substrate. The mass is allowed to move freely only along the axis of vibration, and the inner frame is

^{1.} The relative orientation between the gyroscopes was not explicitly determined.

allowed to move freely only along the axis normal to the axis of vibration. The inner frame has fingers positioned between plates affixed to the substrate. When rotation occurs about the axis orthogonal to the plane of the polysilicon structure, the resulting secondary oscillation between the inner frame and the substrate can be measured by the changes in capacitance between the fingers on the inner frame and the plates on the substrate. In addition, the ADXRS150 has two of these structures, placed adjacent to each other, but operating in anti-phase. By examining the differential signal from the two structures, the conditioning electronics contained within the chip are able to reject common-mode signals unrelated to the angular rate, such as external shocks and vibrations [74] [75].

Implementation of the Murata ENC-03J Gyroscope

Two Murata ENC-03J sensors were used, one type A and one type B. The ENC-03J sensor is a dual-inline-pins (DIP) style package, and has been replaced by ENC-03M which is a surface-mount part. The Murata data sheets indicate that the ENC-03M is a direct replacement for the ENC-03J, with the same specifications other than size and weight (the hardware developed for this thesis includes a footprint for the ENC-03M within the footprint for the ENC-03J, as discussed in Appendix D.1.1; the ENC-03J was used in all testing).



Figure 3.18 Schematic of the ENC-03J



Figure 3.19 Photo of the IMU board, with the Murata ENC-03J components highlighted

The schematic for the implementation of one of the ENC-03J sensors is shown in Figure 3.18. Figure 3.19 shows the placement of the components for both ENC-03J sensors on the IMU board, and the relevant parameters are listed in Table 3.3. The output was buffered by an op-amp follower, in an inverting configuration. The relationship between the output voltage of the sensor, V_{sens} , and the output voltage from the op-amp, V_{out} , is described by Eq. 3.2, where V_{ref} is the voltage supplied to the inverting input through R_1 . While the op-amp was used in the inverting configuration, the output was always positive since V_{ref} , the zero-offset of the sensor provided by the ENC-03J, kept the output centered. Resistor R_1 was set to 10 k Ω and resistor R_2 was set to 15 k Ω , for a gain of 1.5. Capacitor C_2 , in parallel with R_2 , reduced extraneous noise, with a cutoff frequency of 106 Hz (the sensor itself has a cutoff frequency of 50 Hz). The power pin on the op-amp had a 0.1 μ F capacitor, C_1 , to minimize noise from the power supply [73].

$$V_{out} = -\frac{R_2}{R_1} \cdot (V_{sens} - V_{ref}) + V_{ref}$$
(3.2)

Parameter	Value
Measurement range	± 300°/sec
Sensitivity	0.37 mV/°/sec
Zero offset	1.65 V
Cutoff frequency	50 Hz
Nonlinearity	± 5% full scale
Temperature sensitivity	± 20% at -5 or +75°C
Current consumption	Max: 5 mA
Power draw	Max: 16.3 mW
Package size	15.44 mm x 8 mm, 4.3 mm tall (not including pins)
Package weight	<1.0 grams

TABLE 3.3 Relevant parameters of the Murata ENC-03J gyroscope [73]

Implementation of the Analog Devices Gyroscope

The Analog Devices ADXRS150 used in the GaitShoe system was an early demo version, and was a sixteen pin DIP package. The ADXRS150 is now available for purchase as a surface mount part, and has nearly identical specifications to the demo part used, other than size and weight (Appendix D.1.1 shows the pin-to-pin mappings between the two versions). All specifications and operating parameters in this chapter correspond to the demo version.



Figure 3.20 Schematic of the ADXRS150



Figure 3.21 Photo of the IMU board, with the ADXRS150 components highlighted

The implementation of the ADXRS150 is shown in Figure 3.20, Figure 3.21 shows the placement of the components for the ADXRS150 on the IMU board, and Table 3.4 lists the relevant parameters. The ADXRS150 required a 5 V power supply, and was the only

71

sensor in the system to use 5 V. Capacitors C_1 , C_2 , and C_3 were set as specified (22 nF, 22 nF, and 47 nF, respectively) in the data sheet, and had to be located close to their respective pins, as these capacitors were used in the charge pump that provided 16 V for the electrostatic resonator. Capacitors C_4 and C_5 were each set to 0.1 μ F, as specified, and also needed to be placed as near their respective pins as possible; these capacitors were used to minimize noise injection from the charge pump into the supply voltage. Capacitor C_6 was set to 0.1 µF, as recommended, to set the cutoff frequency before the final amplification stage to 400 Hz (±35%, due to tolerances of two internal resistors). This low pass filter served to eliminate high frequency artifacts prior to the final amplification. Capacitor C_7 was set to 0.1 µF, and resistor R_1 was set to 100 kΩ, in order to set the bandwidth and adjust the measurement range of the sensor. Resistor R_1 was in parallel with an internal resistor, $R_{out internal}$, resulting in an effective $R_{out} = 64.3 \text{ k}\Omega$. This resulted in a cutoff frequency for this sensor of 25 Hz. By reducing R_{out} by approximately a third (as compared to $R_{out internal}$), the sensitivity was decreased by approximately a third, and the measurement range was increased by approximately a third. The output of the sensor was divided via a potentiometer, R_2 , such that the resulting output was centered around 1.65 V, with a full-scale range from 0 - 3.3 V. Finally, the output was buffered by a non-inverting op-amp follower with a gain of 1, and capacitor C_8 was set to 0.1 µF to reduce noise into the opamp from the power supply [75].

3.3.3 Force Sensitive Resistors

Two sizes of force sensitive resistors (FSRs) manufactured by Interlink Electronics were used, FSR-402, which has a circular sensing area with a diameter of 12.7 mm, and FSR-400, which has a circular sensing area with a diameter of 5 mm. Two of the FSR-400 sensors were placed underneath the heel pad, one medially and the other laterally. One of the FSR-402 sensors was placed underneath the first metatarsal head, and a second FSR-402 sensor was placed underneath the fifth metatarsal head.

Parameter	Value
Measurement range	±420 °/sec
Sensitivity	Typical 2.9 mV/°/sec, min: 2.8 mV/°/sec, max: 3.2 mV/°/sec
Zero offset	Typical: 1.65 V, min: 1.4 V, max: 1.8 V
Cutoff frequency	25 Hz
Resonant frequency	16 kHz
Nonlinearity	0.1% full scale
Temperature sensitivity ^a	Min: 4.2 mV/°/sec, max: 4.9 mV/°/sec
Rate noise density	$0.05^{\circ}/\text{sec}/\sqrt{\text{Hz}}$
Quiescent supply current	Typical: 5.0 mA, Max: 8.0 mA
Power draw	Typical: 25 mW
Shock survival	Up to 30,000g while unpowered, up to 500g while powered

TABLE 3.4 Relevant parameters of the Analog Devices ADXRS150 gyroscope [75]

a. Greatest deviations from initial value at 25°C to worst case value at T_{min} or T_{max}.

Working Principle

A force sensitive resistor is a type of sensor which experiences a decrease in electrical resistance when force is applied orthogonally to the active area of the sensor. Though less accurate than a load cell, FSRs are generally inexpensive, and when manufactured from polymers, the typical thickness is on the order of 0.25 mm [76].

The FSRs manufactured by Interlink Electronics are polymer-based sensors, and consist of three layers. The lowest layer is a flexible substrate which has been coated with a print-able semi-conductor material. The middle layer is a spacer adhesive, with material only along the outline of the part, providing an open region at the active area of the device. The top layer is a flexible substrate, printed with interdigitating electrodes and two printed leads which connect to solder tabs. The active area of the sensor is the area containing the electrodes. FSR-400 uses a 0.10 mm layer of polyethersulfone for the top and bottom layers, with a 0.05 mm layer of acrylic for the spacer. FSR-402 uses a 0.13 mm layer of polyetherimide for the top and bottom layers, with a 0.15 mm layer of acrylic for the spacer. Polyethersulfone is a transparent substrate which has excellent temperature resistance,
moderate chemical resistance, and good flexibility. Polyetherimide, on the other hand, is a semi-transparent substrate which has excellent temperature resistance, excellent chemical resistance, and limited flexibility [77].

The semi-conductor material on the lowest layer provides an electrical connection between the sets of interdigitating electrodes. Without any force applied, the adhesive provides an air gap between the semi-conductor and the electrodes, so the resistance across the sensor is high. When force is applied across the active area, the electrodes are pressed into the semi-conductor, which reduces the resistance across the sensor.

If the sensor is bent, local changes along the line of bending may result in a decrease in resistance, which would cause a false reading of applied force. Changes in temperature and humidity can also affect the output of the FSR.

Implementation

A voltage divider was used to measure the change in resistance of the FSRs. The circuit implemented is shown in Figure 3.22. Figure 3.23 shows the placement of the components for all four FSRs on the Tactile board, and relevant parameters of the FSRs used are listed in Table 3.5. The same values of resistors were used for both the FSR-400 and the FSR-402 sensors. To lower the impedance of the signal, the output of the voltage divider was followed by a bipolar junction transistor (BJT), in the typical *npn* configuration, used as an emitter follower; there was a 0.6 V voltage drop across the BJT. As shown, a 1 k Ω resistor, R_1 , was used in the voltage divider, and a 10 k Ω resistor, R_2 , was placed between the emitter and the ground. The relationship between the FSR output, V_{out} , and the resistance of the FSR, R_{FSR} , is described by Eq. , where R_1 is 1 k Ω and V_{cc} is 3.3 V.

$$V_{out} = \left(\frac{R_1 + R_{FSR}}{R_{FSR}}\right) \cdot V_{cc} - 0.6V$$
(3.3)



Figure 3.22 Schematic of the FSR



Figure 3.23 Photo of the Tactile board, with the FSR components highlighted

Parameter	Value
Force sensitivity range	<100 g to >10 kg, depending on mechanics
Pressure sensitivity range	$<0.1 \text{ kg/cm}^2$ to $>10 \text{ kg/cm}^2$, depending on mechanics
Part-to-part repeatability	$\pm 15\%$ to $\pm 25\%$ of established nominal resistance ^a
Single part repeatability	$\pm 2\%$ to $\pm 5\%$ of established nominal resistance ^a
Cutoff frequency	500 Hz
Device rise time	1-2 msec
Resolution	0.5% full scale
Current consumption	FSR-400: 0.2 mA, FSR-402: 1.3 mA
Power draw	FSR-400: 0.66 mW, FSR-402: 4.3 mW
Lifetime	>10 million actuations
Package size	FSR-400: 7.5 mm x 38.1 mm, 0.30 mm thickness FSR-402: 18.3 mm x 54.1 mm, 0.46 mm thickness

TABLE 3.5 Relevant parameters of the Interlink FSR-400 and FSR-402 [77]

a. With a repeatable actuation system.

3.3.4 Bend Sensors

The bend sensors used were the FLX-01, manufactured by The Images Company. Each FLX-01 is 11.4 cm long and 0.64 cm wide, with a thickness of 0.5 mm. One bi-directional bend sensor was located in the insole to measure the flexion at the metatarsal-phalangeal joint, and the other was placed between the shin and the back of the shoe to measure the dorsi-/plantar- flexion of the foot.

Working Principle

The bend sensors work in a manner similar to the FSRs: the resistance through the sensor changes as the sensor is actuated. However, instead of having interdigitating electrodes, the electrodes are printed linearly along the length of the sensor. When the bend sensor is bent in the direction which lengthens the spacing between the printed electrodes, the resistance increases (bending in the direction which shortens the spacing has a negligible effect on the resistance). When unbent at 0°, the nominal resistance of the sensor is 10 K Ω , and when bent to 90° in the direction of sensitivity, the resistance is on the order of 30-40 K Ω . Since each individual bend sensor measures bend in only one direction, two sensors were used in pairs, such that their sensitive bending directions would complement each other, resulting in a measuring range of $\pm 180^\circ$, as shown in Figure 3.24. A differential circuit was used to combine the outputs, in order to provide an output corresponding to the bidirectional bend.



Figure 3.24 Bend sensors shown back to back

Implementation

The differential circuit implemented to combine the individual uni-directional bend sensor outputs into a bi-directional bend output is shown in Figure 3.25, Figure 3.26 shows the placement of the components for both bi-directional bend sensors on the Tactile board, and relevant parameters of the FLX-01 are listed in Table 3.6. A voltage divider was used

to measure the change in resistance of each individual bend sensor, with resistors R_1 and R_2 each set to 20 k Ω . The outputs from each voltage divider were then combined through an op-amp set as a differential amplifier. The relationship between the bi-directional bend output, V_{out} , and the resistance of the bend sensors, R_A and R_B , is described by Eq. 3.4, where R_1 is 10 k Ω and V_{cc} is 3.3 V (note $R_3 = R_1$, $R_4 = R_2$, and $R_7 = R_8$). Since R_2 is 100 k Ω and R_8 is 220 k Ω , there is a gain of 2.2 across the op-amp. Assuming the bend sensors are well-matched when flat, the output is centered around 1.65 V, with the output greater than 1.65 V when bend sensor B is bent in its sensitive direction, and less than 1.65 V when bend sensor A is bent in its sensitive direction. Capacitor C_1 , in parallel with R_3 , reduces extraneous noise, with a cutoff frequency of 154 Hz.

$$V_{out} = \frac{R_8}{R_2} \cdot \left(\frac{R_B}{R_B + R_1} - \frac{R_A}{R_A + R_1}\right) \cdot V_{cc} + \frac{V_{cc}}{2}$$
(3.4)



Figure 3.25 Schematic of the bi-directional bend sensor



Figure 3.26 Photo of the Tactile board, with the bi-directional bend sensor components highlighted

TABLE 3.6 Relevant parameters of The Images Co. FLX-01 [78]

Parameter	Value
Angle sensitivity range	0° to 90°
Cutoff frequency	154 Hz
Package size	11.4 cm x 0.64 cm, 0.5 mm thickness

3.3.5 Polyvinylidene Fluoride Strips

Two polyvinylidene fluoride strips, part LDT0 made by Measurement Specialities, were used to measure the dynamic force applied across the sensor. One was placed under the heel and the other was placed under the great toe.

Working Principle

Polyvinylidene fluoride (PVDF) is a manufactured polymer that can be poled to become a a piezoelectric material. Piezoelectric materials have somewhat of a crystalline structure (PVDF is a semi-crystalline homopolymer) which generates an charge difference when subjected to stress. While a piezoelectric material is electrically neutral at rest, stress deforms the crystalline structure such that the dipoles align and a polarity develops along one axis, resulting in a net charge expressed across the structure; this is a reversible physical phenomenon [79].

The LDT0 has a thin film of PVDF laminated between two electrodes; forces applied to this sensor result in stress across the piezoelectric material. The electrical equivalent of this sensor is a voltage source in series with a capacitor, with a shunting leakage resistor.

The capacitance of the LDT0 is on the order of 400 pF, and the leakage resistor is very large (on the order of $10^{12} - 10^{14} \Omega$), which means that the sensor has a very high output impedance at low frequency. Piezoelectric sensors are typically interfaced with matching charge amplifiers, or voltage amplifiers with high input resistances [76] [80].

Implementation

The circuit implemented for the PVDF strips provides an output corresponding to dynamic force, and is shown in Figure 3.27. Figure 3.28 shows the placement of the components for both PVDF strips on the Tactile board, and relevant parameters of the PVDF strips used are listed in Table 3.7.



Figure 3.27 Schematic of the PVDF strips



Figure 3.28 Photo of the Tactile board, with the PVDF strip components highlighted

The output from the PVDF strip was connected to the gate of an *n*-channel junction fieldeffect transistor (JFET). The JFET was used as a source follower, with a source resistor, R_3 , set to 10 k Ω ; JFETs have extremely low leakage current, which results in a high input impedance, as necessary for buffering the PVDF strip. When the PVDF strip was bent by a force, it generated a voltage difference across its capacitance, which biased the JFET, and caused current to flow from the drain to the source, proportionally increasing the voltage at the "PVDF output," as labeled in Figure 3.27. The PVDF strip was shunted by two resistors in series, R_1 and R_2 , each set to 10 M Ω for a total¹ resistance of 20 M Ω . This shunt attenuated low frequency drift from charge buildup on the piezo strip, and this attenuation resulted in an output signal that corresponded to dynamic changes in the force applied across the sensor. The (first-order, high-pass) cutoff frequency for the system was 20 Hz, and was determined by resistors R_1 and R_2 (20 M Ω total) and the LDT0 capacitor (~400 pF). The shunt capacitor, C_1 (330 pF), provided a frequency-independent attenuation of the piezo's signal.

TABLE 3.7 Relevant parameters of the Measurement Specialties LDT0 PVDF strip [80]

Parameter	Value
High-pass cutoff frequency	20 Hz
Package size	25.0 mm x 13.0 mm, 0.15 mm thickness

3.3.6 Electric Field Sensor

An electric field sensor was developed during the latter part of the project, using an electric field imaging device manufactured by Motorola, part MC33794DH, for occupant detection in automotive seat applications [81]. This sensor was used in testing to measure the height of the heel above floor via capacitive coupling. Another electrode was later configured to measure the height of the toes/metatarsals above the floor.

Working Principle

An electric field sensor measures the capacitive coupling between an electrode at an oscillating potential and other electrodes or grounds [63]. When an object moves into the elec-

^{1.} The use of two resistors allowed for more flexibility in setting the total resistance, as surface mount resistors in the high $M\Omega$ range were only available in a small number of discrete values.

tric field created by the oscillating electrode, it increases the capacitive loading of the electrode and changes the intrinsic capacitive coupling between the electrode and its environment. For measuring the height of the foot above the floor, both the transmitting electrode and the ground were placed on the bottom of the shoe. When the foot is "high" off the floor, all of the electric field lines emanating from the sensing electrode will shunt to ground via C_o . As depicted in Figure 3.29, the system (and the measurements presented in Section 4.8) consisted of two electrodes: an active sensing electrode partly surrounded by another electrode at local ground. When the foot is far off the floor, the capacitive loading is dominated by the intrinsic coupling between the sensing electrode and the ground electrode (C_o), as dictated by their geometry. As the foot nears the floor, however, the conductive and dielectric property of the flooring ($R_F \parallel C_F$) becomes increasingly significant, through the series chain of C_{FE} to $R_F \parallel C_F$ to C_{FG} and eventually dominates the intrinsic capacitance C_o when the foot is in contact with the floor. By measuring this change, the height of the foot above the floor can be estimated.



Figure 3.29 Schematic showing electrode coupling to both the ground and the floor

A second electrode configuration, developed after subject testing, consisted of circular electrodes on the bottom of the shoe, with a driven shield layer in the insole and a ground layer above the driven shield (the insole with the other sensors was located between the driven shield and the ground layer), as depicted in Figure 3.30. The driven shield effectively prevents coupling between the sensing electrode and the ground electrode, resulting in a larger measurement output range.



Figure 3.30 Schematic showing the second electrode configuration coupling to the floor

The MC33794DH drives the electrodes by generating a sine wave at 120 kHz, and measuring the capacitive coupling by detecting the load current. The sine wave has very low harmonic content, to reduce interference with other systems, and the frequency of the sine wave is adjustable via an external resistor. Up to nine electrodes can be driven by the MC33794DH; one electrode is selected at a time, by a 4 bit digital address, and the unselected electrodes are all grounded. To measure the capacitance on the selected electrode, the load current is converted to a DC level, filtered by an external capacitor, and multiplied and offset, resulting in an output range from 1.0 V to 4.0 V.

The MC33794DH also includes a shield driver, which provides a signal that follows the sinusoidal signal on the selected sensing electrode. This signal can be used to drive shielding electrodes that prevent coupling to nearby grounds. It can also be used to drive the shield on a co-axial cable connecting a sensing electrode. This effectively nulls the intrinsic capacitance of the co-axial cable, keeping measurement of the capacitance of the electrode minimally affected. In addition, a driven shield layer, constructed of an additional PVC insole lined with copper tape, and connected to the driven shield, was placed beneath the GaitShoe insole, to shield the sensing electrode from loading by the insole sensors.

Implementation

The circuit implemented for the MC33794DH is shown in Figure 3.31, Figure 3.32 shows the placement of the components on the Tactile board, and relevant parameters of the



MC33794DH are listed in Table 3.7. Due to its automotive legacy, this sensor required a +12 V input, and was the only component to require +12 V.

Figure 3.31 Schematic of the electric field sensor



Figure 3.32 Photo of the Tactile board, with the electric field components highlighted

The configuration of the electrode, as used to measure the height above the floor for the last few subjects during subject testing, is shown in Figure 3.33. The capacitive tape along the perimeter of the shoe is connected to the ground of the +9 V battery, and the inner strip is connected to electrode E2 (the driven shield within the shoe shown in Figure 3.10 was not used with this electrode configuration). The second configuration, developed after subject testing, including two sensing electrodes, is shown Figure 3.34.

The various capacitors and resistors were set as recommended in the data sheet. Resistor R_{33} was set to 39 kΩ, which set the frequency of the sine wave to 120 kHz. Capacitor C_{11} was the external capacitor which filtered the receiver multiplexer signal, and was set to 10 nF, which allowed in a 99% settling time for the detector output in under 5.0 ms¹.



Figure 3.33 Example of an electrode set up on the bottom of a shoe to measure heel height



Figure 3.34 Example of two electrode set up on the bottom of a shoe to measure heel and toe height.

Capacitors C_9 and C_{10} were set to 10 pF and 100 pF respectively; these capacitors served as external reference capacitors, which allowed a method for determining the absolute capacitance on a selected electrode. Capacitor C_8 was set to 47 pF, C_{12} was set to 0.1 µF, and C_7 was set to 4.7 µF, all for filtering noise from the power lines. The "ground plane" pad connected to the circuit board ground at pin 20 was placed underneath the MC33794DH, and soldered to its heat sink. The heat sink was designed into this chip for its other applications (e.g., as a lamp driver); the capacitive sensing function consumes very little power, and does not actually require the heat sink.

^{1.} All sensors on each shoe are sampled at 75 Hz, or, every 13.4 msec. The microcontroller turned each electrode on, and paused before measuring the output, in order to allow the output to settle.

Jumper pads $RJ_1 - RJ_6$ were used to allow the microcontroller to set the multiplexer and select the electrodes. To select two of the electrodes connected to the small receptacle shown in Figure 3.32 (e.g. for use with electrodes under the heel and the metatarsals), RJ_1 and RJ_3 were connected to pull each high and low via the local power supply and ground, respectively, RJ_2 and RJ_4 were left open, and RJ_5 and RJ_6 were connected to the micro-controller. Using the microcontroller, RJ_6 was pulled low, and RJ_5 was pulled low to select the first electrode, and high to select the second electrode. Jumper pad RJ_7 was connected to provide the output signal directly to the microcontroller.

The driven shield was used to drive the co-axial shield on the cables connecting the electrodes to the header pins. The driven shield was also used, in lieu of a ground plane, on the printed circuit board in the areas surrounding the header connections and the lines connecting the headers to the MC33794DH chip, to further shield the electric pins from any stray capacitance or noise on the board. In addition, the driven shield signal was routed to capacitive tape which formed a shield beneath the insole, to prevent any noise or load capacitance from the insole sensors from interfering with the sensing electrodes on the bottom of the shoe.

Parameter	Value
Measurement range	Typical: 10 pF - 100 pF
Sensitivity	Typical: 30 pF / V
99% settling time	Max: 5 ms
Cutoff frequency	Max: 200 Hz
Electric field frequency	120 kHz
Package size	14 mm x 16 mm; 3 mm tall

TABLE 3.8 Relevant parameters of the Motorola MC33794DH [82]

3.3.7 Ultrasound Sensor

An ultrasound sensor was developed during the latter part of this research, by Mr. Lovell. The implementation and sample results are available in Appendix E. The ultrasound sensor was configured to measure the distance and the angle between the two shoes.

Working Principle

Ultrasound waves are mechanical acoustic waves at frequencies higher than can be heard by human ears (humans can hear waves up to at most 20 kHz). Ultrasound sensors typically have a transmitter, which generates the waves, and a receiver, which is excited when waves are encountered. Ultrasound waves travel at the speed of sound, so distances can be determined by measuring the amount of time between the start of transmission and the first incidence of reception. The speed of sound in dry air, at 20° C, is 331 m/s; this number does vary with temperature and humidity [76].

Earlier tests using two closely spaced receivers to detect phase difference (as in monopulse sonar [83]) provided poor results, hence a large separation between receivers was used with a time-difference-of-arrival method.



Figure 3.35 Ultrasound sensor for distance and angle between two shoes

To measure the distance and relative position between the shoes using the time-differenceof-arrival, a transmitter was placed on the right shoe, and two receivers were placed on the left shoe, as shown in Figure 3.35. On the right shoe, the transmitter, shown in green, generated ultrasound waves at 40 kHz. On the left shoe, the ultrasound electronics measured the time which elapsed before each of the receivers, shown in red, were first hit by the ultrasound waves. These time measurements led to an estimate of the distances from the transmitter to each of the receivers (A and B), and since the distance between the two receivers (C) is known, θ can be determined from the law of cosines, as described in Eq. 3.5. If desired, the distance from the transceiver to an arbitrary point (at known position in relation to A or B) and the corresponding angle from the horizontal, ϕ , can be determined from the geometrical relationship between the points.

$$\cos\theta = \frac{A^2 + B^2 - C^2}{2 \cdot AB}$$
(3.5)

3.4 Additional Components

The additional hardware and circuits used in this system are described in this section. Detailed part numbers and ordering information for all parts are listed in Appendix D.2.

3.4.1 Microcontroller

The microcontroller selected for the GaitShoe was the C8051F206, manufactured by Cygnal Integrated Products, Inc. The features of this 48 pin chip include its small size (8 mm x 8 mm, 1.20 mm thick), its on-board analog-to-digital convertor (ADC), and its low operating current (under 9 mA, typical). The analog-to-digital convertor features 12-bit resolution multiplexed in 32 channels, and has a net throughput of up to 100 kilosamples/sec (ksps). The chip was clocked at 22.118 MHz, and therefore has just over 22 millioninstructions/sec (mips); the supply voltage provided was +3.3 V.

Figure 3.36 shows the implementation of the Cygnal C8051F206, as used in the GaitShoe system, and Figure 3.37 shows the placement of the components on the Main board. A six

pin JTAG programming header provided the ability to easily update the code on the chip as needed. Pin 14 must be connected to ground when the Cygnal chip is first turned on, in order to reset the ADC, so a manual switch was provided for this purpose. Pins 11 and 31 were connected to the 3.3 V power supply, each with 0.1 μ F capacitors to minimize noise. Five jumper pads ($RJ_1 - RJ_5$) were provided for future use with multiple stack boards; these can be used to give each of 32 boards a unique identification number.

A voltage reference was included on the board, shown connected to pin 7. This can be used to provide the C8051F206 with a reference voltage which is separate from the main power line; however, the use of this part was not implemented in testing.



Figure 3.36 Schematic of the Cygnal C8051F206



Figure 3.37 Photo of the main board, with the Cygnal C8051F206 components highlighted

3.4.2 Radio Frequency Transceiver

A critical requirement of the design of the GaitShoe was that it be an untethered system. Since the ability to use the GaitShoe for real time feedback was of strong interest, a wireless transceiver was required for data transmission.

Transceiver Hardware and Electronics

The wireless transceiver selected for use in this application was the DR3000-1 module, manufactured by RF Monolithics, Inc. The DR3000-1 module contains RF Monolithics' TR1000 amplifier-sequenced hybrid (ASH) transceiver on a daughter board, and was designed specifically for short-range wireless applications in the radio frequency (RF) range. The daughter board helps isolate the local ground from any noise on the main ground line, and contains most of the capacitors, resistors and inductors used to operate the TR1000. The TR1000 operates at 916.5 MHz, and transmits data at 115.2 kilobits/sec¹ (kbps), using amplitude-shift keyed (ASK) modulation. Surface-acoustic-wave (SAW) filtering is employed by the receiver (in two stages) to reject out-of-band noise and by the transmitter to suppress² output harmonics [85].

Figure 3.38 shows the schematic used for the DR3000-1, and Figure 3.37 shows the placement of the components on the Main board. Signals CTR0, CTR1, and TXEN were set by the microcontroller, to set the state (transmit or receive) of the transceiver. The microcontroller also sent signal TX, which contained the data to be transmitted. Signal RX was the data received by the transceiver, and was sent to the microcontroller. The NAND gate was implemented to ensure that pin 5 on the DR3000-1 was held low when the transceiver was operating in receive mode. It was necessary to add a 22 μ F capacitor across pins 7 and 8

^{1.} Transmission at 115.2 kbps was selected so that the data could be converted to serial data, and transmitted to the computer over the serial line.

^{2.} SAW filters use the piezoelectric effect; a transducer at the input of the device converts electric waves into surface acoustic waves. The acoustic waves then excite a half-wavelength resonant acoustic cavity within the transducer, and this energy is coupled to a second resonant acoustic cavity (also of dimension corresponding half-wavelength) transducer at the output, where the signal is converted back to electric waves. Various methods of coupling are employed; SAW filters are commonly used in the telecommunications industry to suppress output harmonics [84].

(Vcc and GND) in order to minimize noise from the power line [85]; this increased the range of the system by from approximately 3 m to more than 15 m, during use in the MIT Media Lab.



Figure 3.38 Schematic of the RF Monolithics DR3000-1



Figure 3.39 Photo of the Main board, with the components of the RF Monolithics DR3000-1 highlighted

The antenna used for most of the testing of the system¹ was a helical whip 1/4-wave antenna manufactured by Linx Technologies, Inc.; it has a female RP-SMA connector, and is omni-directional. It was connected to the DR3000-1 daughter board via a connector manufactured by Linx Technologies, which has a male RP-SMA connector with an 8.5"

^{1.} A solid core wire cut to 8.2 cm in length (the 1/4 wavelength for 916.5 MHz) was used with the first three subjects. However, the wire was continually hit by the equipment used by the MGH Biomotion Laboratory, which caused it to bend back and forth and eventually shear off. The helical whip antenna was found to work nearly as well as the wire, and was more stable physically. Though it is recommended that it be used with a proximity groundplane, a groundplane was not implemented in this version of the system; doing so could improve the reliability of the wireless transmission.

length of coaxial cable attached with heat shrink and strain relief. The voltage standing wave ratio (VSWR) of this antenna is typically under 1.9.

RF Issues

RF Monolithics estimates that in the 916.5 MHz band, with transmission at 115.2 kbps, the operating distance will be up to 65.3 m in free space or as low as 8.1 m in dense cubical office space. Additionally, it is important to note that UHF radiation is well absorbed by the human body, particularly above 750 MHz, which will always be a confounding factor in optimizing such a wireless system.

The testing of the GaitShoe was carried out at the MGH Biomotion Laboratory (BML), which contained a large quantity of electrical equipment. In addition, the equipment of other residents in the building was unknown (for instance, some wireless phones operate in the 900 MHz band), and temporary magnetic resonance imaging (MRI) and positron emission topography (PET) laboratories are often set up outside the building. These factors were likely contributors to the low transmission range achieved during testing: the maximum range of the GaitShoe, while in use at the BML, was on the order of 3-5 m from the basestation. In contrast, in use at the MIT Media Lab, the maximum range was on the order of 15 m or more. However, the basestation was placed in the middle of the room at the BML which provided sufficient length for collecting data during the entire gait trial.

The 12-bit output from the ADC was converted to two balanced (equal numbers of zeros and ones) bytes, using the algorithm described in Appendix F.1. Error checking algorithms, such as cyclic redundancy check (CRC), were not implemented, but are recommended for future work with the GaitShoe.

3.4.3 Power

Several different voltages were needed in the GaitShoe system. While most of the hardware used +3.3 V, the Analog Devices gyroscope (ADXRS150) required +5 V, the Motorola electric field imaging device (MC33794DH) required +12 V, and the ultrasound sensor required +9 V. To meet these requirements, and to allow the GaitShoe system to be untethered, a power board was developed, containing a replaceable +9 V battery and voltage regulators. The voltage regulators were three different fixed voltage output versions of the Analog Devices ADP1111 switching regulator. For fixed outputs of +3.3 V and +5 V, the ADP1111 operated in step-down mode, and for the fixed output of +12 V, the ADP1111 operated in step-up mode. In steady-state operation, the hardware on each GaitShoe attachment drew a current just under 50 mA, while the basestation drew just under 15 mA



Figure 3.40 Schematic of the Power Board

The schematic showing the circuits on the power board is shown in Figure 3.40. As implemented, R_1 , R_2 , and R_3 , were each set to 0 k Ω (the resistor pads were included to provide

the option to limit the current; this was unnecessary). The inductor values were selected based on allowing the battery to reach a minimum voltage of 7V. Inductors are available in a finite range of values, so the closest lower value was selected. This resulted in used (and calculated) values of: $L_1 = 27 \mu$ H (30 μ H), $L_2 = 18 \mu$ H (19 μ H), and $L_3 = 82 \mu$ H (98 μ H). The direct battery output was used to provide the +9 V for the ultrasound sensor. To minimize noise, each voltage line had a ceramic 0.1 μ F capacitor, C_8 , C_9 , C_{10} , and C_{11} , as well as an electrolytic capacitor, C_4 , C_5 , C_6 , and C_7 ; the electrolytic capacitors were set to 220 μ F on the +3.3 V and +5 V lines and 470 μ F on the +12 V and +9 V lines.

The power board also contained a MAX666 chip, which was used to detect when the battery level becomes low. The MAX666 pulls pin 7 high when the battery level is above the threshold set by resistors R_{10} and R_{11} and low when the battery level falls below the threshold. With $R_{10} = 2.7 \text{ M}\Omega$ and $R_{11} = 12 \text{ M}\Omega$, the threshold was set at 7 V. The level of pin 7 was provided as an output, and was connected to the +9 V battery source via a light emitting diode (LED), L_2 , such that the LED was lit when the battery level fell below the threshold of 7 V. This provided a visual indication that the battery level was low. A second LED, L_1 , was located between the +9 V battery source and ground, which provided a visual indication that the power board is turned on.

Finally, the power board also contained a resettable fuse, in series with the on-off switch, and before the rest of the hardware. Although a +9 V battery cannot source enough current for a shock to be felt through the skin, a fire or burn hazard from a short circuit remains a possibility, hence, a fuse was included as a safety precaution. The fuse used was part MF-SM030, a positive-temperature coefficient (PTC) polymer fuse manufactured by Bourns, Inc. PTC polymers change from a low resistance to a very high resistance state in response to a spike in current above the trip current level; PTC polymer fuses are resettable by power cycling the device (and removing the cause of the high current). The MF-SM030 has a trip current of 0.6 mA.

3.4.4 Other Components

An analog multiplexer from Analog Devices, part ADG608, was used for the collection of data from multiple sensors. The ADG608 switches between each of eight inputs to a common output. Its features include a high switching speed (typical transition time to switch inputs is 120 nsec), low power dissipation (1.5 μ W, max), and a low on resistance (30 Ω , max).

All op-amps used in the system were the Analog Devices OP262. The OP262 is a dual opamp, which has a low offset voltage (50 μ V, typical), low supply current (.5 mA, typical), high slew rate (10 V/µsec, typical) and low noise (0.5 μ V peak-to-peak, typical). Nearly all were operated at +3.3 V, except for the op-amps used in the ultrasound sensor circuits, which were operated at +9 V.

In the basestation, the RS-232 driver used to convert the streaming wireless data to serial data was a MAX233 chip, manufactured by Maxim. This chip runs on +5 V, and provides the higher RS-232 voltage with a charge pump, hence it does not require any external components.

A number of connectors were used in the GaitShoe system. The most critical selection was the choice of connectors between the stacked printed circuit boards. These provided both electrical interconnections, and some mechanical stability. The 26 total interconnections were split between the two headers (14 for one, 12 for the other), which were located at opposite corners. The connectors were Molex Milli-Grid shrouded headers and mating receptacles, and were rated for 100 insertion cycles (reasonable for prototyping)¹. The net names on the interconnections are shown in Figure 3.41.

^{1.} The footprint of these connectors and the foot print of the RF Monolithics DR-3000-1 board (the largest component) determined the basic size of the board: 4.5 cm x 3.7 cm, the smallest size which could accommodate these three components. A smaller single connector with 26 pins wa originally used, which allowed for a smaller board footprint, however, this connector was not robust enough when mounted on the back of a shoe.



Figure 3.41 Header interconnection net names

3.5 The GaitShoe System

The design of the GaitShoe has resulted in a wearable wireless system which can measure a vast number of parameters about gait. A photo of the GaitShoe hardware on a pair of shoes is shown in Figure 3.42, and the structure of the GaitShoe system is summarized in a high-level block diagram in Figure 3.44. Figure 3.44 shows the (uncalibrated) outputs across all the sensors used during the subject testing.



Figure 3.42 GaitShoe hardware



Figure 3.43 High level block diagram of the GaitShoe system



Figure 3.44 Sensor outputs across all sensors for both feet

Chapter 4

SENSOR ANALYSIS

This chapter discusses the pre-processing applied to the GaitShoe data, the analysis techniques and assumptions used, and presents an analysis of the output of all the sensors, including calibration of the most relevant sensors. The outputs of the calibrated sensors in the GaitShoe system were analyzed to generate clinically relevant gait parameters; the validation of these results by comparison to data collected simultaneously by the system at the Massachusetts General Hospital Biomotion Lab is discussed in Chapter 5.

Calibration of the sensors was necessary because sensor outputs may vary, depending upon, inter alia, manufacturing tolerances of sensors and other components in the circuits with the sensors. Although manufacturers provide specifications ("specs") for sensors, in general they are not sufficiently accurate. The sensitivity, which relates the output of the sensor to the standard units, must be determined for any sensor for which a quantitative output is desired. In addition, sensors such as the gyroscopes and accelerometers measure parameters that have positive and negative polarities. However, the actual sensors all have an output voltage greater than zero, so the zero offset for each of these sensors must be determined¹. The uncalibrated sensor outputs have been normalized to a scale of 0 to 1 by the maximum 12-bit analog-to-digital converter (ADC) value (i.e. the uncalibrated outputs were divided by 4095).

^{1.} In a sensor which has a linear output, the zero offset corresponds to the y-intercept, and the sensitivity corresponds to the slope of the line.

4.1 Data Processing

Before the parameters derived in this chapter were used to calibrate and analyze the sensor outputs, a number of processing steps were applied to the data.

4.1.1 Truncation

The GaitShoe collects data continuously, while the MGH Biomotion Lab (BML) collected data for seven seconds during subject testing for validation of the GaitShoe. The GaitShoe data were truncated so that it started a couple seconds before the subject started walking, and concluded just before the subject stopped or turned at the end of the room. To determine the truncation¹, the raw data were loaded, and the outputs from several sensors on both feet were plotted, as shown in Figure 4.1.



The starting and stopping points were selected by investigation of plotted data. The initiation of gait was easily determined, because the subjects were instructed to stand still with

^{1.} See timetrunc.m, an m-file available in Appendix F.2.

both feet on the floor before starting the gait trial; the starting point was selected approximately 2 seconds before the first changes seen in the sensors.

The stopping point was more difficult to determine. Subjects were told to stop when they reached a point a few feet away from the wall, and could either pause or continue walking by turning to their left. Because of this variation, the stopping point was set either when the subject slowed or turned, whichever occurred first. All four FSR outputs and both PVDF outputs, for both feet, were plotted for visualization of a slowing gait. For instance, in Figure 4.1, the left FSR output shows a broadened output during the last two steps, corresponding to a slowing gait. Two gyroscope outputs were plotted for both feet: the z-gyro, which measures the angular velocity corresponding to changes in the pitch of the foot, and the y-gyro, which measures the angular velocity corresponding to changes in the yaw of the foot. Turning on the heel of the foot results in changes picked up by the y-gyro only, this can be seen well in the left foot output in Figure 4.1 (this event is marked in the top graph of Figure 4.1). The stopping point was set just before either of these events occurred, and was always set mid-stance (load on the FSRs) for one foot, and just after toe-off (no-load on the FSRs) of the other foot.



Figure 4.2 Truncated section of the raw output

The resulting truncated data are shown in Figure 4.2. All of the truncated gait trials for each subject were stored as a Matlab data file. The bottom graphs in both Figure 4.1 and Figure 4.2 is the output of the BML transistor-transistor logic (TTL) trigger, which was used to align the time scale of the GaitShoe with the time scale of the BML data (described in Appendix D). The initiation of the trigger flag corresponds to time zero on the BML system.

4.1.2 Time Adjustment

Stored with the data were two columns of information about the time points corresponding to each row of sensor outputs: the time as provided by the laptop computer clock, "comptime," and a timestamp provided by the clock and the microcontroller in the basestation, "basetime." The comptime stream was continuous, but was only accurate only to about 50 msec. The basetime stream was accurate to 0.1 msec, but it reset every 0.4096 seconds. The comptime was used to make initial observations of the data, such as for truncation, without needing to convert the basetime. It was also used as a moderately accurate reference during the basetime conversion; this reference was needed if the wireless connection failed and packets of data were not received by the basestation. The basetime was converted¹ to a continuous time stream using the comptime as a reference. The trigger output, shown on the bottom graph in Figure 4.2, was used to set time zero, corresponding to the BML time scale.

4.1.3 Data Adjustment

The final step² before calibration was to adjust the data³, by reordering the sensor columns, and by checking all data for outliers.

^{1.} See timeadjuster.m, an m-file available in Appendix F.2.

^{2.} See getshoedataorder.m, an m-file available in Appendix F.2, which ran timeadjuster.m and dataadjuster.m

^{3.} See dataadjuster.m, an m-file available in Appendix F.2.

Column	Sensor Name	Abbreviation
1	FSR, under the medial heel	FSR-HM
2	FSR, under the lateral heel	FSR-HL
3	FSR, under the medial (first) metatarsal	FSR-MM
4	FSR, under the lateral (fifth) metatarsal	FSR-ML
5	PVDF, under the heel	PVDF-H
6	PVDF, under the great toe	PVDF-T
7	Bend, in the insole	Bend-I
8	Bend, at the ankle	Bend-A
9	Gyroscope, measuring about the z-axis	Gyro-Z
10	Accelerometer, measuring along the z-axis	Accel-Z
11	Gyroscope, measuring about the y-axis	Gyro-Y
12	Accelerometer, measuring along the y-axis	Accel-Y
13	Gyroscope, measuring about the x-axis	Gyro-X
14	Accelerometer, measuring along the x-axis	Accel-X
15	Electric Field Sensor, under the heel	EF-H
16	reserved for future use	
17	reserved for future use	
18	reserved for future use	
19	reserved for future use	
20	reserved for future use	
21	Time	
22	Outlier information	

TABLE 4.1 Final sensor order

Sensor Order

The data were reordered such that the columns of the data corresponded to the order of sensor outputs as listed in Table 4.1 (rows corresponded to time points). This was done mainly to set a standard order of sensor data across all subjects, since different versions of microcontroller code transmitted the sensor data in different orders. It was also used to group similar sensors together. The five columns reserved for future use were left empty for both the planned additional sensors, such as an electric field sensor under the metatar-

sals, and the ultrasound sensors, as well as any future sensors of interest (if more than five additional sensors were added to the system, additional columns could, of course, be added as well).

Outliers

As described in Section 3.4.2, the wireless transmission involved encoding the twelve bit sensor reading as two balanced bytes. The basestation received these data and used the serial line to send the computer the two balanced bytes, where they were decoded to reconstruct the twelve bit sensor reading. However, if a single bit (or 3, 5, or 7 bits) was not received correctly at the basestation, the two bytes could not be decoded to the original twelve bit number. If the decoding failed, the sensor output at that time point was flagged¹.

In addition, if two (or 4, 5, or 8) bits were not received correctly at the basestation, the resulting byte was still balanced, so it was able to be decoded and reconstructed as a twelve bit number; however, this reconstructed value was incorrect. Errors of this type resulted in outlying data points², which were identified using the first difference of the sensor data. For each point, the standard deviation of the first differences over a total of 75 points (which corresponds to about a minute's worth of data) were calculated. When evaluated, most points were located in the middle of the 75 points, but at either end of the data, a block of the first 75 or last 75 points were used.

A point was identified as an outlier if both of the following criteria were met: 1) the magnitudes of the first differences between the point and each of its neighbors were greater than the standard deviation times a threshold (the value of the threshold is discussed below), and, 2) the signs of those two first differences were opposite. The first criteria indicated that the value of the point changed rapidly with respect to its neighbors, and the

^{1.} The computer program converted the twelve-bit output (range, 0-4095) to a normalized output with a range from 0-1; data which could not be decoded was set equal to 2.

^{2.} See findoutliers.m, an m-file available in Appendix F.2.

second criteria indicated that the point was not along a valid trajectory of rapidly increasing (or decreasing) points. Each sensor data vector was evaluated twice, since outliers with a very large first difference could skew the mean standard deviation within the 75 point window for outliers which occurred earlier.



Figure 4.3 Outlier identified in bend sensor output

The FSRs, bend sensors, and electric field sensors all measured parameters which did not change rapidly, so a threshold of three standard deviations was used. Data from a bend sensor is shown in Figure 4.3, with the first differences plotted in the lower portion of the graph. One of the data points visually stands out as an outlier. The circle around this data point in the upper graph indicates that it met the criteria for being labeled an outlier. The horizontal lines on the bottom graph correspond to the value of the three standard deviations of the local first difference.

The accelerometers, gyroscopes, and PVDF strips were all high-bandwidth devices and measured parameters that changed very quickly, so a higher threshold of five standard deviations was used. Though many of the data points in the accelerometer output change very rapidly, only one met the criteria to be labeled outlier, as can be seen from the horizontal lines on the lower graph in Figure 4.4.



Figure 4.4 Outlier identified in accelerometer output

When a sensor data point was identified as either an odd-bit error from the flag, or as an even-bit error from the outlier evaluation, it was replaced by fitting a line¹ between the two neighboring points. Column 22 was used to store which data points were replaced, by using a base-18 number, where the position in the base-18 number corresponding to the column number was set to 1 if the sensor data point was determined to be an outlier, and set to 2 if the sensor data point was flagged as unable to be decoded (or left as zero if the data was unaltered).

Another wireless transmission problem was that some data packets were not received by the basestation. Each shoe was instructed to update every 0.0134 seconds, which corresponds to a data transmission rate of approximately 75 Hz. However, due to issues with the wireless transmission, as discussed in Section 3.4.2, there were occasional "dropped packets," where the time between data packets for a single shoe was more than 0.0134 seconds. An additional step² was performed on all of the data, where for any series of three dropped packets or fewer, data were generated for the missing time points by fit-

^{1.} Polyfit.m, a standard Matlab function, was used with the polynomial degree set to 1 to fit the line, and polyval.m, also a standard Matlab function, was used to calculate the value at the time of the outlying sensor data point.

^{2.} See gapfiller.m, an m-file available in Appendix F.2.

ting a line in the same manner as was used for the outliers. In this case, the corresponding row had a value of 0.1 placed in column 22.



Figure 4.5 Bend sensor data with a number of adjustments

By using column 22 to store information about data replacement and data generation, it was easy to recover the status of every data point. A sample of bend sensor data which had all three types of data adjustments is shown¹ in Figure 4.5. Each resulting matrix of data, for the left and right feet over a single gait trial were stored in a Matlab data files for each subject.

4.1.4 Calibration and Analysis

To calibrate² the data, the truncated and adjusted data were loaded into Matlab. All rows with a non-zero value in column 22 were removed, because of concerns about the accuracy of the line-fit (the use of a spline-fit for data is discussed later in this chapter; this was done on an as-needed basis).

The accelerometers, gyroscopes, and bend sensors were calibrated using the sensitivities and zero offsets, and the FSRs were calibrated using the sensitivity functions; the determination of these parameters will be discussed in this chapter. The PVDFs were scaled to a

^{1.} See plotadjdata.m, an m-file available in Appendix F.2.

^{2.} See gencalibrations.m, an m-file available in Appendix F.2, was used to apply the calibrations to the data

centered around zero. The electric field sensor was only used on a few of the last subject, and so the electric field sensor output was not included in the calibrated data matrix; the data order is listed in Table 4.2. In addition, the sum of the four calibrated FSR outputs was calculated, and stored in column 20. The resulting calibrated data were stored in a Matlab data file.

Column	Data	Abbreviation
1	Calibrated FSR, under the medial heel	FSR-HM
2	Calibrated FSR, under the lateral heel	FSR-HL
3	Calibrated FSR, under the medial (first) metatarsal	FSR-MM
4	Calibrated FSR, under the lateral (fifth) metatarsal	FSR-ML
5	Centered PVDF, under the heel	PVDF-H
6	Centered PVDF, under the great toe	PVDF-T
7	Calibrated Bend, in the insole	Bend-I
8	Calibrated Bend, at the ankle	Bend-A
9	Calibrated Gyroscope, measuring about the z-axis	Gyro-Z
10	Calibrated Accelerometer, measuring along the z-axis	Accel-Z
11	Calibrated Gyroscope, measuring about the y-axis	Gyro-Y
12	Calibrated Accelerometer, measuring along the y-axis	Accel-Y
13	Calibrated Gyroscope, measuring about the x-axis	Gyro-X
14	Calibrated Accelerometer, measuring along the x-axis	Accel-X
15	Pitch, determined from Gyro-Z	
16	Velocity in X _{Room}	
17	Displacement in X _{Room}	
18	Velocity in Y _{Room}	
19	Displacement in Y _{Room}	
20	Sum of calibrated FSR outputs (columns 1-4)	FSRsum
21	Time	

TABLE 4.2 Order of calibrated and analyzed data

As will be discussed in this chapter, the z-gyroscope was analyzed to determine the foot pitch, and the pitch was stored in column 15. The x-accelerometer and y-accelerometer were used to determine the velocity and displacement along the x- and y- axes of the

room; the x-velocity and x-displacement were stored in columns 16 and 17, and the x-velocity and x-displacement were stored in columns 18 and 19. The final order of the calibrated and analyzed data is detailed in Table 4.2.

4.2 Analysis Model

4.2.1 Coordinate Systems

Analysis of the data requires the understanding of two coordinate systems, shown in Figure 4.6. The first coordinate system corresponds to the global reference frame of the room. The second corresponds to the local body frame, where the sensors are located and collect their measurements. Determination of room-based parameters such as orientation or position (for instance, corresponding to stride length) requires a transformation to the global reference frame.

For very simple motions, this transformation is straightforward. For example, a simple rotation about a single gyroscope axis is transformed to an angle of orientation within the fixed reference frame by integrating the corresponding gyroscope signal. Similarly a simple translation consisting only of motion along a single acceleration axis is transformed to a displacement within the fixed reference frame by double integration of the corresponding accelerometer signal. Of course, most motions of interest will consist of rotations about multiple axes, vector translations, and the two axes will be at some arbitrary position and orientation with respect to each other, requiring more complex transformations to obtain measurements of interest in the fixed reference frame of the room.



Figure 4.6 Frames of reference used in evaluation

4.2.2 Inertial Measurement Evaluation

The GaitShoe system contains three gyroscopes and three accelerometers, which form a strapdown "inertial measurement unit" [86], by collecting information about the threedimensional motion of the foot through space. These measurements can be integrated with respect to time and combined to determine the position and orientation of the foot.

Evaluation of Complex Rotations and Translations

As discussed above, complex rotations and translations require careful evaluation to transform the motion in the local body frame back to the fixed reference frame. In particular, because orientation space wraps onto itself, and rotations do not commute, complex rotations in 3D space cannot be determined by examining only the rotations about each axis independently. For instance, in a given reference frame, rotation about the y-axis of $+30^{\circ}$ results in the same orientation as a rotation about the y-axis of -330° ; in addition, successive rotations about the x-axis and z-axis can result in the same final orientation. The two most commonly used methods of evaluation to deal with this issue are *Euler angles* and *quaternions. Euler angles* represent complex rotations as the product of three successive rotations about three orthogonal axes. *Quaternions* represent rotations via a three element vector, Q_v and a real number Q_r , where the magnitude of the rotation is $2acos(Q_r)$, and the rotation is about the axis described by Q_v . *Quaternions* can be transformed to *Euler angles*, and vice versa.

The Kalman filter is an important mathematical tool used to combine *quaternions* or *Euler angles* with translations in order to obtain the position and orientation. The Kalman filter was developed to provide a method of estimating a process with feedback from measurements [87]. The basic Kalman filter has two types of equations: time update equations, which predict the state of the process prior to the next time point; and, measurement update equations, which use the feedback from measurements at the time point to correct the estimation of the state. The basic Kalman filter is useful for estimating the state of a linear process; for estimating the state of a three-dimensional non-linear motion, the
extended Kalman filter, which linearizes about the current mean and covariance, is used (this initial analysis of the data did not include the implementation of a Kalman filter, though one is recommended for future work) [88] [89] [90].

Integration Considerations

Applying integration to gyroscope and accelerometer outputs results in a long-term increase in error of the position and orientation, because noise errors accumulate during integration via a random walk process; the root mean square of this drift is proportional to the square root of the time of the integration [91]. This results in a second source of error for the accelerometers. Because the accelerometers measure both dynamic and static acceleration, the contribution of gravity must be correctly subtracted before integration to determine position due to dynamic acceleration. As discussed in Section 4.4.2, this contribution of gravity is calculated by integrating the gyroscopes to determine the orientation of the foot. Errors in the integration of the angular velocity therefore contribute to errors in the calculation of the gravity contribution, which is subtracted prior to integration of the accelerometer signal.

These effects can be minimized by periodic recalibration of position and orientation, or by integrating over short time intervals. Because the subjects in this study of the GaitShoe were all walking, the gait included a state when each foot was flat on the floor. This state occurs between heel strike and toe off, and can be confirmed with other sensors, such as the four FSRs in the insole. This was exploited to improve the integration of the gyroscope and accelerometer signals: it was used to set the bounds of integration over a single stride, as well as to recalibrate the orientation about the z-axis and x-axis (both 0° when the foot is flat on the floor), and the position in the y-axis (0 cm). Though the positions in the z-axis and x-axis and the orientation, resetting the integration reduces errors. Healthy gait typically has a step rate close to 120 steps per minute, or 60 strides per foot per minute. As shown in Figure 4.7, stance time is typically 60% of the gait cycle, which results in integration over about 0.40 seconds; this short integration time should help reduce the random

walk noise accumulated during integration. For persons with gait which does not include foot-flat, such as toe-walkers, a different method will be required for recalibration. For instance, the bend sensor in the insole could be used to determine the orientation of the gyroscope every time the toes were on the floor, as detected by the FSRs located underneath the metatarsals.



Figure 4.7 The gait cycle

Analysis Assumptions

The initial analysis of the GaitShoe sensor output was simplified by assuming that the data collected involved linear motion in the $X_{GS}-Y_{GS}$ plane only (no translation in Z_{GS}), and rotation about the Z_{GS} axis only (the $X_{GS}-Y_{GS}$ plane remained parallel to the $X_{room}-Y_{room}$ plane). In other words, it was assumed that motion involved only changes portray-able on flat paper, such as in Figure 4.7.

The subject testing involved collecting data during walking, with the subject walking in a straight line only (as discussed in Section 4.1.1, the data were truncated before the subject turned upon reaching the end of the room). While this assumption is unlikely to be completely true for any of the subjects, it is a reasonable approximation to reality, particularly

for subjects with normal gait. Figure 4.8 shows the outputs of the three gyroscopes and three accelerometers from one of the gait trials during subject testing. Though there is acceleration in Z_{GS} , the magnitude is smaller than the accelerations measured in X_{GS} and Y_{GS} . In this sample, the standard deviation of the acceleration in Z_{GS} is 2.2 m/s², compared to greater than 5 m/s² in X_{GS} and Y_{GS} , and the spread between the largest positive and largest negative acceleration in Z_{GS} is 20 m/s², compared to greater than 45 m/s² in X_{GS} and Y_{GS} ; thus, the accelerations in X_{GS} and Y_{GS} are at least double those in Z_{GS} . Similarly, the angular velocity about Z_{GS} , is significantly larger in magnitude than the angular velocities measured about X_{GS} and Y_{GS} . In this sample, the standard deviation of the angular velocity about Z_{GS} is 164.5 °/s, compared to less than 35 °/s about X_{GS} and Y_{GS} is 781.2 °/s, compared to less than 270 °/s in X_{GS} and Y_{GS} ; thus, the angular velocities the largest positive and largest negative angular velocity about Z_{GS} are less than 270 °/s in X_{GS} and Y_{GS} ; thus, the angular velocity about Z_{GS} are less than 270 °/s in X_{GS} and Y_{GS} .



Figure 4.8 Comparison of IMU outputs during walking gait

For this initial analysis of the GaitShoe system, these assumptions were reasonable, without requiring the complex mathematics required to represent true 3D motion. Thus, the motion analysis consisted of integrating the output of the z-gyroscope to obtain the pitch of the foot about Z_{room} , and transformation of the single and double integration of the output of the x- and y-accelerometers (after incorporating the pitch of the foot) to obtain the velocity and displacement along the X_{room} direction.

Methods of Integration

Two methods of integration were used in the analysis, a linear integrator, and Simpson's integration with a spline-fit. For the latter, a spline¹ was fit to a subset of the data and used to evaluate the integral numerically, using adaptive Simpson quadrature².

The linear integrator used the assumption that the data collection rate of $\Delta t = 0.0134$ sec between data updates was a small enough interval that the change in the gyroscope or accelerometer output could be considered linear between the two time points, as shown in



Figure 4.9 Illustration for linear integration

The equation of the line between two time points can be defined by its slope, m,

$$m = \frac{(\omega_i - \omega_{i-1})}{(t_i - t_{i-1})},$$
(4.1)

and its y-intercept, b,

$$b = \omega_{i-1} - m \cdot t_{i-1}. \tag{4.2}$$

^{1.} See spline.m, a standard matlab function was used for the spline-fit.

^{2.} See quad.m, a standard matlab function was used to integrate using adaptive Simpson quadrature.

Then, the integral of the angular velocity between time point t_{i-1} and t_i is:

$$(\Theta_i - \Theta_{i-1}) = \int_{t_{i-1}}^{t_i} (m \cdot \tau + b) d\tau, \qquad (4.3)$$

which reduces to

$$\Theta_i = \frac{1}{2} \cdot m \cdot (t_i^2 - t_{i-1}^2) + b \cdot (t_i - t_{i-1}) + \Theta_{i-1}.$$
(4.4)

Equations 4.1, 4.2, and 4.4 provided a simple method¹ of linearly integrating the data.

4.3 Gyroscopes

All three gyroscopes were calibrated; the calibrated z-gyroscope output was integrated to determine the pitch of the foot.

4.3.1 Calibration

For the gyroscopes, two types of information were required: the zero offset, to center the output around zero; and, the sensitivity, to convert the output to units of $^{\circ}$ /s.

Zero Offset

The zero offset of the gyroscopes was simply the output of the gyro when the hardware was at rest. The results are summarized in Table 4.3.

Sensitivity

The sensitivity of the gyroscopes was determined by rotating each gyroscope about its sensitive axis, through a range of constant angular velocities.

A precision gearhead motor from Globe Motors (#409A582) was used to rotate the gyroscopes. This motor is a high torque permanent magnet motor with a no-load rating of

^{1.} See linear_integrator.m, an m-file available in Appendix F.2.

3000 rpm at +12 V. As purchased, the assembly is geared down such that the gearhead output shaft has a no-load rating of 25 rpm at +12 V, with 400 ounce/inches of torque; the motor functions from 6-12 V. During calibration, the polarity of the inputs were reversed in order to switch the direction of rotation, and the voltage input was varied to adjust the angular velocity. A flat metal turntable was attached to the gearhead output shaft, and the stack hardware was mounted directly to the metal turntable. The rotational velocity was measured with an optical tachometer.



Figure 4.10 Photo of the turntable used for the gyroscope calibration

For each gyroscope, the stack hardware was placed on the turntable shown in Figure 4.10, and taped securely into place, with the sensitive axis of the gyroscope undergoing calibration visually aligned over the center point of the turntable. Data were collected for 20 to 30 seconds each at several discrete angular velocities, and the clipped means¹ were calculated² for each set of data. A line was fit to the clipped mean data points and the zero offset value (which corresponds to 0° /s); the slope of the fit line is the sensitivity of the sensor. The results are summarized in Table 4.3, including the coefficient of correlation of the line fit, and the line fit for gyroscopes on IMU-1 is shown in Figure 4.11. Note that the Murata

^{1.} The mean, calculated after discarding the top 10% and bottom 10% of the data.

^{2.} As discussed in Section 4.1.3, the GaitShoe data may contain some incorrect data points, as a result of the wireless transmission protocol. For data collected about gait, these points are identified by looking at the first difference; here, any spurious points were eliminated by using the clipped mean.

gyroscope (axes Y and Z) specification indicated a sensitivity of 3.05×10^{-4} [output/(°/s)] and a zero offset of 0.5 [output]. The Analog Devices gyroscope (axis X) specification indicated a typical sensitivity of 8.93×10^{-4} [output/(°/s)], a maximum sensitivity of 9.82×10^{-4} [output/(°/s)] and a zero offset of 0.5 [output].



Figure 4.11 Line fits for determining the sensitivity of the gyroscopes (shown for IMU-1)

Gyroscope	Sensitivity ^a		Zero Offset
	Slope	Coeff. of Corr.	
IMU-1 X	9.25 x 10 ⁻⁴	1.000	0.492
IMU-1 Y	3.21 x 10 ⁻⁴	0.999	0.462
IMU-1 Z	3.30 x 10 ⁻⁴	1.000	0.423
IMU-2 X	9.33 x 10 ⁻⁴	1.000	0.508
IMU-2 Y	3.32 x 10 ⁻⁴	1.000	0.446
IMU-2 Z	3.23 x 10 ⁻⁴	1.000	0.432

TABLE 4.3 Gyroscope sensitivities and zero offsets

a. Sensitivity units are [output/(°/s)]; all other units are the uncalibrated output (scaled from 0-3.3 V to 0 to 1

4.3.2 Analysis of the Pitch

The pitch of the foot was determined by integrating the z-gyroscope output. The sign of the pitch follows the convention used at the MGH Biomotion Lab (BML), where in facing a subject such as the one pictured in Figure 4.7, a positive rotation about the z-axis corresponds to a rotation from the y-axis to the x-axis (left hand rule). The z-gyroscope output was integrated over a single stride. A variety of approaches for integration were developed.

The integration bounds for the gyroscope were determined from the times of heel strike and toe off. The midpoint between the heel strike time and the toe off time was calculated and was used for the starting and stopping point of each integration.

Direct Linear Integration

The first method¹ of determining the pitch used the calibrated z-gyroscope data and linear integration. A sample is shown in Figure 4.12, where the integration bounds are indicated with a red dot. The problem with this approach is that the integration sometimes fails to return the angle to 0° after a single stride. In Figure 4.12, the first integration is significantly off, and the second and fourth integrations are slightly off, while the third integration looks very reasonable.

There are two reasons which usually account for the failure to return the angle to 0° at the end of the stride: dropped packets; and drifting z-gyroscope output. In the first integration, there is a fairly large gap of dropped packets, which results in a miscalculation of the angle at the first time point following the gap, and a shift of the subsequent angles. This problem can be most readily addressed by improving the wireless implementation to prevent packet loss or by providing a method of on-board data storage. For this work a number of methods using spline-fits were developed to generate the missing data.

^{1.} See gyroz_linint.m, a m-file available in Appendix F.2.



Figure 4.12 Sample of direct linear integration of the z-gyroscope

The z-gyroscope output corresponding to the fourth integration interval is shown in Figure 4.13, The output of the z-gyroscope is not zero during stance in this sample. This is because during calibration of the gyroscope, the zero-offset determined in Chapter 4 was used, but gyroscope output actually drifts very slightly over time. To compensate for this, an iterative method was developed.



Figure 4.13 Z-gyroscope output where angular velocity is non-zero during stance

Iterated Linear Integration

To compensate for the slight drifting of the gyroscope's zero-offset, the second method¹ of determining the pitch linearly integrated the z-gyroscope data, but checked the final value of each integration between each set of integration bounds. An "endlimit" value was defined as 0.1% of the full-scale across all integrations. In Figure 4.12, the full-scale is approximately -50° to 60° , so the endlimit value would be 0.1% of 110°, or 0.11°.

For a given pair of integration bounds, if the final value had a magnitude greater than the endlimit, a small "nudge" value was added to the calibrated z-gyroscope data between these integration bounds. The magnitude of the nudge value was equal to the difference between the endlimit and the final value after integration, and the sign of the nudge value was opposite from the sign of the final value after integration. The linear integration between these integration bounds was repeated. This process was repeated until the final value was within the endlimit.



Figure 4.14 Sample of linear integration of the z-gyroscope, with iteration

^{1.} See linear_integrator_endadjust.m and gyrozlinint_withspline.m, m-files available in Appendix F.2.

The results of this method are shown in Figure 4.14. All four of the integrations, including the first with dropped packets, now return smoothly to 0° .

Iterative Linear Integration with Spline-fit and Iterative Simpson's Integration

The third method¹ added an additional step to compensate for any gaps of dropped packets. After the iterative linear integration was complete, the data were checked between each set of integration bounds for gaps with more than four packets dropped in a row. If such gaps were found, a spline was fit to the calibrated z-gyroscope data. Then, the spline was used with Simpson's integration to determine the angle corresponding to the time points where packets were dropped (an iterative integration was used, similar to the one described above).

The results of this method are shown in Figure 4.15. There were three gaps of significant length, two during the first iteration, and one during the final iteration. The results of the spline-fit fill in the missing data points, but do not significantly affect the overall calculation of the angles.



Figure 4.15 Sample of linear integration of the z-gyroscope, with iteration and spline-fit

^{1.} See simpsonsintegrator_endadjust.m and gyrozlinint_withspline.m, m-files available in Appendix F.2.

Iterative Linear Integration with Post-Integration Spline-Fit

The final method¹ was developed as an alternative to the previous method; it was computationally expensive to do the spline-fit and iteration, particularly since the overall results were not changed. In this method, after the z-gyroscope was linearly integrated with iteration as described above. As necessary, any area of interest in the pitch was fit with a spline for evaluation.

For evaluating the performance of the pitch calculation with the BML results (see Chapter 5), the minimum and maximum values of the pitch, and the times at which they occurred, were compared. The maximum and minimum pitch during the gait cycle are labeled in Figure 4.16.



Figure 4.16 Pitch of the foot during the gait cycle

Each minimum and maximum of the calculated pitch were located, and a spline was fit to the pitch data over nine points (with the extremum at the center; this corresponds to 0.1 sec of data if no packets are dropped). The spline was fit with time points every 3.34 msec, which is four times as many data points as are collected if no packets are dropped. The results of this method are shown in Figure 4.17.

The maximum or minimum of each spline-fit (and the corresponding times) were determined, and compared to the maximum and minimums (and times) of the foot pitch information from the BML data. Figure 4.18 shows the BML data plotted with the integrated pitch. The shape of the pitch iteratively integrated from the angular velocity closely resembles the shape of the BML foot pitch

^{1.} See gyroz_postspline.m, an m-file available in Appendix F.2.



Figure 4.17 Sample of linear integration of the z-gyroscope, with iteration and post-integration spline-fit



Figure 4.18 Final results, compared to BML data

The root mean square error (RMS error) between these two curves was calculated. As shown in Figure 4.19, a spline was fit to the linearly integrated data at timepoints corresponding to the timepoints of the BML data. The RMS error calculated between these two curves was 3.4° .



Figure 4.19 Comparison of BML pitch to GaitShoe pitch

4.4 Accelerometers

All three accelerometer outputs were calibrated; the calibrated x- and y- accelerometer outputs were used to determine the velocity and displacement of the foot in the X_{room} and Y_{room} coordinates.

4.4.1 Calibration

Three types of information about the accelerometers were required: the zero offset, to center the output around zero; the sensitivity, to convert the output to units of m/s^2 ; and the orientation of the accelerometers relative to the foot, to interpret the sensor output accurately.

Zero Offset and Sensitivity

Determining the sensitivity of the accelerometers was very straightforward, by using gravity. Naturally, the gravitational acceleration vector, **g**, is stable, accurate, and readily available. By rotating the sensor such that the axis of interest was orthogonal to the earth's surface, and then rotating the sensor 180°, measurements of +1 g and -1 g were easily obtained. The sensitivity, $V_{sensitivity}$ (units: output/ m/s²), of the acceleration measurement is described by Eq. 4.5, where V_{Ig} is the measurement at +1 g and V_{Ig} is the measurement at -1 g.

$$V_{sensitivity} = \frac{1}{2g} \cdot (V_{1g} - V_{-1g})$$
 (4.5)

The zero offset, $V_{zero-offset}$ (units: output), was set as the midpoint between the +1 g and the -1 g measurements, as described by Eq. 4.6.

$$V_{zero - offset} = \frac{1}{2} \cdot (V_{1g} + V_{-1g})$$
(4.6)

The positioning of the accelerometers with respect to the gravity vector was done by hand. The hardware was slowly rotated about each of the three axes; this allowed each accelerometer to sweep through the gravitational acceleration vector, **g**, twice. As shown in Figure 4.20, rotation about the x-axis resulted in the y-axis and z-axis accelerometers sweeping through **g**; similarly, rotation about the z-axis resulted in the x-axis resulted in the x-axis and y-axis accelerometers sweeping through **g**, and rotation about the y-axis resulted in the x-axis resulted in the x-axis and z-axis accelerometers sweeping through **g**.



Figure 4.20 Demonstration of accelerometer calibration

Rotating by hand provided the ability to slightly modify the orientation of the accelerometers during rotation, in order to place the accelerometer axes parallel to **g**, and maximize the sensor reading. The two axes within an accelerometer are specified to be perpendicular to each other to within 0.01° , and are parallel to the sides of the accelerometer to within 1° [71]. The accelerometer on the face of the IMU measures along the y-axis and the zaxis, and the accelerometer located along the top of the board measures along the x-axis. Both accelerometers were soldered to the board by hand, so while the x-axis is expected to be close to perpendicular to the y-axis and the z-axis, the exact angle was unknown.

The data were low-pass filtered¹, with second-order Butterworth coefficients² and a lowpass cutoff frequency of 2 Hz, and the sampling frequency was 75 Hz. The rotation by hand was performed slowly, with the goal of keeping the rotation at a constant speed, as the goal was to measure only gravitational acceleration. The filtering was used to remove any small acceleration changes resulting from hand jitter. The values of V_{Ig} and V_{-Ig} for each calibration were determined by finding the maximum and minimum values of each output (the uncalibrated output of all sensors has been scaled from 0 to 1). A sample graph of the filtered accelerometer outputs is shown in Figure 4.21.



Figure 4.21 Sample accelerometer calibration for determining sensitivity and zero-offset.

The calibration was carried out seven times, on each IMU³, throughout the subject testing. The values of V_{1g} and V_{-1g} for each IMU were determined from the average of the results from all seven calibrations. The sensitivity and zero offset for each axis of the accelerom-

^{1.} Matlab function filtfilt.m, a backwards and forwards filter, was used.

^{2.} Matlab function butter.m was used to generate the Butterworth coefficients.

^{3.} Two IMUs were used during testing of the GaitShoe system, one per foot; this document refers to the two IMUs as IMU-1 and IMU-2.

125

eters on each IMU were then determined from Eq. 4.5 and Eq. 4.6, using the mean values of V_{1g} and V_{-1g} ; results are summarized in Table 4.4, and shown in Figure 4.21. Note that the Analog Devices specification indicated a typical sensitivity of 0.11 [output/m/s²], and a maximum sensitivity of 0.14 [output/m/s²]; typical zero offset was indicated as 0.5 [output], with a maximum of 0.7 [output].

Accelerometer	+1 g Output		-1 g Output		Sensitivity ^a	Zero Offset
	Mean	Std Dev	Mean	Std Dev		
IMU-1 X	0.676	0.001	0.425	0.001	0.13	0.551
IMU-1 Y	0.694	0.002	0.442	0.002	0.13	0.568
IMU-1 Z	0.619	0.005	0.375	0.002	0.12	0.497
IMU-2 X	0.716	0.002	0.467	0.002	0.13	0.591
IMU-2 Y	0.684	0.002	0.433	0.004	0.13	0.558
IMU-2 Z	0.750	0.003	0.507	0.003	0.12	0.628

 TABLE 4.4
 Accelerometer sensitivities and zero offsets

a. Sensitivity units are [output/(m/s²)]; all other units are the uncalibrated output (scaled from 0-3.3 V to 0 to 1].

4.4.2 Orientation

Determining the orientation of each accelerometer in space was critical because the acceleration due to gravity must be correctly subtracted from the total acceleration signal during the analysis of gait data.

For simple rotation about the z-axis, the orientation of the foot can be determined by integrating the angular velocity about z to obtain the pitch of the foot. Then, the pitch orientation of the accelerometer is determined by subtracting the angle of inclination of the accelerometer with respect to the foot from the pitch, to allow the contribution of the gravity vector to be appropriately subtracted from the total acceleration measured. The orientation of the accelerometer with respect to the foot is different for each subject, as the size and shape of shoe influence the inclination of the GaitShoe attachment on the shoe. When a subject stands still, with both feet flat on the floor (the foot flat on the floor corresponds to a pitch of 0°), each accelerometer measures only the fraction of gravitational acceleration along its sensitive axis. Therefore, as shown in Figure 4.22, the angle α_x can be determined as described by Eq. 4.7., where A_x is the acceleration measured by the x-accelerometer with the foot flat on the floor.



$$\alpha_x = \operatorname{asin} \frac{A_x}{\dot{g}} \tag{4.7}$$

Figure 4.22 Determination of the orientation of the accelerometers

The angle α_y can be similarly determined; the difference between these two angles provides the relative orientation between the x- and y-accelerometers in the x-y plane.

4.4.3 Velocity and Stride Length Analysis

Velocity and stride length were determined by single and double-integration, respectively, of the acceleration along the X_{room} axis, using the output of the x- and y-accelerometers. In addition, the vertical velocity and vertical displacement were determined by single and double-integration, respectively, of the acceleration along the Y_{room} axis. The accelerome-

ters are fixed to the back of the shoe via the GaitShoe hardware, and their orientation with respect to the room and the gravity vector changes as the subject walks.

Determination of Dynamic Acceleration

The acceleration measured by the x-accelerometer can be resolved into two components, corresponding to the dynamic acceleration from foot motion, and the static acceleration due to gravity, $Ax_{dynamic-dyn}$, and Ax_{static} respectively. Figure 4.23 shows these components, and the relationship between the gravity contribution and the static component, given the pitch orientation of the x-accelerometer, Φ_x .



Figure 4.23 Dynamic acceleration along X_{room} only

In order to calculate the contribution of Ax_{static} component, the orientation of the x-accelerometer with respect to the room, Φ_x , must be determined. At time point t_i , $\Phi_x(t_i)$ is calculated from the sum of the pitch at the same time point, t_i , $\Theta_x(t_i)$, and the orientation of the x-accelerometer with respect to the horizontal, α_x :

$$\Phi_x(t_i) = \alpha_x + \Theta_x(t_i), \qquad (4.8)$$

where α_x is calculated as described in Eq. 4.7 on page 126 (the sign of α_x has been determined by the right-hand rule, so Θ_x , which was determined using the left-hand rule to correspond to the BML data, is multiplied by -1).

The dynamic acceleration measured by the x-accelerometer at time point t_i , $Ax_{dynamic}(t_i)$, is calculated by subtracting¹ the contribution of gravity from the total acceleration:

$$A_{x_{dynamic}}(t_i) = A_x(t_i) - \dot{\boldsymbol{g}} \cdot \sin \Phi_x(t_i).$$
(4.9)

The dynamic acceleration measured by the y-accelerometer is determined using the same method, but substituting the appropriate y-variables.

Determination of Appropriate Acceleration for Integration

The total acceleration vector experienced by the foot, *Afoot*, can be resolved into two components corresponding the reference frame of the room, Ax-room_{dynamic} and Ay-room_{dynamic}. These both contribute the dynamic component measured by the x-accelerometer, $Ax_{dynamic}$, as determined above, and are shown in Figure 4.24



Figure 4.24 Dynamic acceleration with both X_{room} and Y_{room} components

^{1.} Although Einstein's Equivalency Principle states that dynamic and static acceleration cannot be determined from measurement of acceleration alone, given the orientation of the object (e.g. Θ_x), the static component resulting from gravitation acceleration can be determined and subtracted from the total vector acceleration. However, any error in the orientation will contribute to the error in the dynamic component.

The most accurate way to determine the actual dynamic acceleration along X_{room} , which is the axis of interest for the velocity and the displacement calculations, is to combine the outputs of both the x-accelerometer and the y-accelerometer. The y-accelerometer has a dynamic component, $Ax_{dynamic}$, that also measures components of both Ax-room_{dynamic} and Ay-room_{dynamic}, as shown in Figure 4.24. Using the output of both the x- and y-accelerometers, Ax-room_{dynamic} and Ay-room_{dynamic} can be determined from:

$$A_{\text{x-room}_{dynamic}} = \frac{A_{x_{dynamic}} \cdot \sin \Phi_y - A_{y_{dynamic}} \cdot \sin \Phi_x}{\cos \Phi_x \cdot \sin \Phi_y - \cos \Phi_y \cdot \sin \Phi_x}, \qquad (4.10)$$

and

$$A_{y-\text{room}_{dynamic}} = \frac{A_{x_{dynamic}} \cdot \cos \Phi_y - A_{y_{dynamic}} \cdot \cos \Phi_x}{\sin \Phi_x \cdot \cos \Phi_y - \sin \Phi_y \cdot \cos \Phi_x}.$$
(4.11)

These equations were used to integrate Ax-room_{dynamic} and Ay-room_{dynamic} twice, to determine the stride length, and the vertical displacement of the foot.

In addition, a second method of estimating velocity and displacement along the x-axis was evaluated. Because the subjects walked in (essentially) a straight line, which was (essentially) aligned with X_{room} axis, the total dynamic acceleration along X_{GS} , $Ax_{dynamic}$ was also integrated twice to estimate stride length. This corresponded to the path length traced out by the foot during walking. While this result does not correspond precisely to the room coordinates, it is a simple method of approximating these parameters of interest, and does not use the output of the y-accelerometer.

For future work with applications for the more usual case of non-straight walking (such as turning corners), the full three axes of acceleration and angular velocity will need to be analyzed and combined. In addition, it may be useful to have three two-axis accelerometer, such that each axis has two (redundant) measurements.

Integration Methods

Following determination of the dynamic components of the outputs of each of the accelerometers, a single linear integration determined¹ the velocities, and a subsequent linear integration determined the displacements, as shown in Figure 4.25 and Figure 4.27. After each integration, when the foot was flat on the floor and not accelerating, the velocity was reset to zero.

The integration bounds for both the x- and y-acceleration were determined by the characteristics of $Ax_{dynamic}$, using the z-gyroscope integration bounds as starting points. The lower integration bound was determined by starting at the first quarter-point between a pair of z-gyroscope integration bounds, and moving back in time toward the first z-gyroscope integration bound until the magnitude of $Ax_{dynamic}$ was less than 0.2 m/s². If no value met this condition, the lower bound was set at the time point where the magnitude of $Ax_{dynamic}$ was a minimum, across the points between the first z-gyroscope integration bound and the first quarter-point.

The upper integration bound was determined by moving forward in time from the midpoint between two subsequent z-gyroscope integration bounds towards the second z-gyroscope integration bound until the magnitude of $Ax_{dynamic}$ was either less than 0.2 m/s² or the value of $Ax_{dynamic}$ was greater than 0 m/s². The latter condition was for the instances where a large positive acceleration was detected. This condition, which appeared in the fourth integration bound in Figure 4.25, most likely corresponds to a strong heel-strike, so it is appropriate to cease integration, since the stride is completed upon heel strike. If neither of these conditions were met, the upper bound was set at the time point where the magnitude of $Ax_{dynamic}$ was a minimum, across the points in between the second z-gyroscope integration bound and ten points prior. Both the lower and upper integration bounds are indicated in Figure 4.25 and Figure 4.27.

^{1.} See accel_integrator.m and linear_integrator.m, m-files available in Appendix F.2.



Figure 4.25 Integration of the acceleration in X_{room} and X_{shoe}

The x-displacement is shown plotted with the corresponding BML data in the lower graph in Figure 4.25. The BML data, which has a different 0 m origin than the GaitShoe, has been shifted to align with the GaitShoe results at a time point during stance (here, at approximately 0.8 sec). For evaluating the performance of the displacement calculation with the BML results (see Chapter 5), the stride lengths were compared, by calculating the difference in displacement from one stance to the next.

Figure 4.26 shows a spline fit to the linearly integrated data at timepoints corresponding to the timepoints of the BML data. In this example, the RMS error between the BML X-displacement and the GaitShoe X_{room} -displacement was 10.3 cm, while the RMS error between the BML x-displacement and the GaitShoe X_{shoe} -displacement was 14.2 cm.



Figure 4.26 Comparison of BML X_{room} -displacement to GaitShoe X_{room} - and X_{shoe} -displacement

The y-displacement is shown plotted with the corresponding BML data in the lower graph in Figure 4.27. Since the foot LED array is mounted above the foot, during stance the value of the BML Y-displacement is on the order of 5 cm, so the BML data has been shifted to align with a displacement of 0 m a time point during stance (here, at approximately 0.8 sec). For evaluating the performance of the displacement calculation with the BML results (see Chapter 5), the peak Y-displacement and the time point at which it occurred were compared. Again, as shown in Figure 4.28, a spline was fit to the linearly integrated data at timepoints corresponding to the timepoints of the BML data. In this example, the RMS error between the BML Y-displacement and the GaitShoe Y_{room}-displacement was 3.9 cm.



Figure 4.27 Integration of the acceleration in Y_{room}





Figure 4.28 Comparison of BML Y_{room} -displacement to GaitShoe Y_{room} -displacement

The GaitShoe's X_{room} -displacement, X_{shoe} -displacement, and Y_{room} -displacement all had maximum differences compared to the corresponding BML displacements on the order of 10-15 cm. For the Y-displacement, this results in a larger percent change; the RMS error of the Y-displacement is smaller because the GaitShoe's Y_{room} -displacement is reset to zero at each stance, which results in a large number of data points which have only a very small difference from the BML Y-displacement.

These results will likely be greatly improved by improving the GaitShoe pitch calculation, and, in addition, by incorporating all of the gyroscope and accelerometer data into the analysis.

4.5 Force Sensitive Resistors

Both types of force sensitive resistors (FSRs) were calibrated; the calibrated outputs for each shoe were added together, and the sum was used in determining heel strike and toe off timing.

4.5.1 Calibration

Use of the FSRs in the analysis required a calibration of their output to units of applied force in kg. The zero offset, representing the output of the FSR with no applied force, was not required in this analysis, because the FSRs were likely to be slightly pre-loaded during subject testing. Because the FSRs are located under the foot inside a shoe, pre-load can result when the laces are tied snugly. In addition, the equipment worn on the foot for acquisition of the MGH Biomotion Lab's data involved a large strap which wraps around the foot, which can cause further pre-loading of the FSRs. Therefore, the load on the FSRs when the foot is in the air is not expected to be 0 kg, so as long as the sensitivity is determined at small values close to 0 kg, the zero offset is not required.

Unlike the output of the accelerometers and the gyroscopes, the output of the FSRs is nonlinear, due in part to the choice of conditioning electronics¹. In addition, the manufacturer's specifications report that the single part repeatability is from $\pm 2\%$ to $\pm 5\%$ of the nominal resistance. Also, over time and with use, the adhesive layer in the FSRs may break down and contribute to an increased non-linearity.

Sensitivity

The FSRs were characterized through the application of a series of forces to several FSRs. The tests were carried out using a TA-XT Texture Analyser from Stable Micro Systems, with both a 5 kg and a 30 kg load cell.

As described in Section 3.3.3, the output of the FSR can be biased if it is bent. There are two ways an FSR could be bent, either by a simple folding motion, or by being indented; the likelihood of either will depend both on the location of the FSR beneath the foot, as well as the shoe of the subject (e.g. the amount of flexibility behind the FSR). The two larger FSRs are located beneath the metatarsals, at the first and the fifth metatarsal heads. As the subject rolls off the foot preceding toe-off, the sensor portion of the FSR may experience some bending across its sensing area, particularly in flexible running shoes, which are designed to allow a great degree of flexion in the insole. Also, since the size of the metatarsal heads is on the same order as the size of the FSRs, a large amount of force applied through the metatarsal heads could result in an indentation of the FSRs into the insole, particularly if the insole is very flexible, as found in most running shoes. On the other hand, the two small FSRs are located beneath the heel pad, and are less likely to experience much bending, both because there is not typically any bending underneath the heel, and because the area of the heel pad is much larger than the sensing are of the FSRs under the heel, and so bending due to indentation is less likely.

Therefore, the output of the FSR will depend not only on the applied force, but on the method of application, which is likely to vary slightly from subject to subject, depending on the shape of their feet, and the type of shoe.

^{1.} As described in Section 3.3.3 and Eq. 3.3, the output of a given FSR is proportional to $(R_1 + R_{FSR})/R_{FSR}$, which results in non-linear output. For future work, an implementation using the FSR at the input to a current-to-voltage converter, as shown to the right, is recommended, as the output of the FSR would be inversely proportional to R_{FSR} , with V_{out} = - $(R_1/R_{FSR})\cdot V_{ref}$. Direct proportionality could be obtained by using R_{FSR} as the feedback resistor.



An alternate FSR implementation

To mimic these conditions during calibration, the following set-up was used: the FSRs were placed on a typical running shoe insole, which allowed some indentation, as would occur in a normal walking shoe. The applicators are shown in Figure 4.29, were wooden dowels covered with a thin layer of a compressible material, cut to the same diameter, and glued to the bottom. This was designed to mimic the metatarsal head and the calcaneous behind a layer of skin and tissue. The testing set-up is shown in Figure 4.30. The large indentor was the same diameter as the sensing area of the FSR-402. The small indentor was the same diameter as the physical area of the FSR-400; in calibrating the FSR-400, the force applied during calibration was converted to a pressure (applied across the indentor area), and converted back to the force measured across the area of the sensing area of the FSR-400.



Figure 4.29 Force applicators (above) and FSRs (below)

Six of each type of FSRs (small: FSR-400, large: FSR-402, both from Interlink Electronics [77]) were tested, four which had been used in the insoles during patient testing, and two new unused sensors. The TA-XT Texture Analyser was used to apply successive forces every 0.25 kg from 0.25 kg to 10 kg, measured with a 5 kg load cell, for both the FSR-402 and the FSR-400 sensors. For the FSR-402 sensors only, the TA-XT Texture Analyser was used to apply successive forces every 1 kg from 10 kg to 30 kg, measured with a 30 kg load cell. The TA-XT Texture Analyser continually recorded the actual applied force, and was re-calibrated after each series of successive forces. Each of the successive forces applied by the TA-XT Texture Analyser, and the corresponding outputs of the FSRs were determined and correlated.



Figure 4.30 Test set-up for calibration of FSRs

Figure 4.31 shows a sample of the calibration data from a used FSR-402 sensor. The upper left graph shows the FSR-402 output during calibration, and the lower left graph shows the pressure applied by the TA-XT Texture Analyser across the indentor during calibration. The right graph shows the calibration results, with the FSR-402 output plotted vs. the applied pressure, with a second x-axis showing the equivalent applied force units.



Figure 4.31 Sample FSR-402 calibration data

For each type of FSRs, a single curve was fit to the calibration data for multiple FSRs, rather than fitting individual curves for each FSR, because it was expected that the output of each FSR during calibration may not directly correspond to the output during subject

138

testing. As mentioned previously, FSR output may degrade over its lifetime use, and the calibration was carried out after subject testing was concluded. In addition, a few of the FSRs were replaced during subject testing, hence the replaced FSRs could not be calibrated. Though two new FSRs of each type were calibrated, the output of the new FSRs was not used for the curve fit. Since the output of all the used FSRs grouped together, despite a few FSRs being replaced during testing, it was expected that the best representation of FSR output during testing would be determined from the output of the used FSRs only. Also the FSRs are thought to have some temperature and humidity sensitivity, which could certainly affect their performance within the environment of the shoe. Finally, while the indentors were designed to mimic the conditions inside the shoe, each subject's foot is certainly unique, which may result in some minor variation in the output.

The calibration results for all six tested FSR-402s and the calibration data for the FSR-402 provided in the FSR data sheet [77], are shown in Figure 4.33. To capture the non-linear relationship between the FSR output and the applied force, first, second, and third order polynomials in exponentials were fit¹ to the calibration data from the four used FSR-402s, as shown in Figure 4.32. The third-order polynomial in an exponential was empirically found to provide a good fit to the data, and is also plotted in Figure 4.33.



Figure 4.32 Various line-fits to the FSR-402 calibration data

^{1.} Matlab function polyfit.m was used to fit a third-order line to the applied force and the natural logarithm of the FSR output; polyfit.m returns p, polynomial coefficients, and s, for estimating the error in the linefit.

The relationship between the FSR-402 output, V_{F402} , and the applied force in N, F, is described by Eq. 4.12.

$$F = 9.8 \cdot e^{\{-8.7 \cdot V_{F402}^{3} + 14.2 \cdot V_{F402}^{2} - 10.6 \cdot V_{F402} + 2.4\}}$$
(4.12)

Figure 4.33 also shows two additional limits: the mean value of the minimum FSR-402 output measured across all trials during subject testing (0.20), and the absolute minimum FSR-402 output over all trials (0.01). While the absolute minimum was much lower than the mean minimum output, fewer than 15% of all FSR-402 trials (145 out of 1080) had a minimum value under 0.10.



Figure 4.33 FSR-402 calibration curve

The FSR-400 calibration curves are shown in Figure 4.34. Again, four used FSRs and two new FSRs were tested (calibration data for the FSR-400 was not provided in the FSR data sheet). The compressible material on the bottom of the indentor delaminated during the testing of the last used FSRs; this FSR is shown in a different color in Figure 4.34.

As with the FSR-402s, a third order exponential was fit to the calibration data from the three FSR-400s with the indentor in the proper configuration. The resulting curve is shown in Figure 4.34, and the relationship between the FSR-400 output, V_{F400} , and the applied force in N, *F*, is described by Eq. 4.13.



Figure 4.34 FSR-400 calibration curve

$$F = 9.8 \cdot e^{\{-10.5 \cdot V_{F400}^{3} + 21.9 \cdot V_{F400}^{2} - 21.6 \cdot V_{F400} + 6.8\}}$$
(4.13)

Again, Figure 4.34 shows two additional limits: the mean value of the minimum FSR-400 output measured across all trials during subject testing (0.60), and the absolute minimum FSR-402 output over all trials (0.47). Fewer than 2.6% of all FSR-400 trials (28 out of 1080) had a minimum value under 0.50.

Figure 4.35 shows the 95% confidence intervals, which were calculated¹ on the line-fit of the log of the applied force for the curve-fit, and converted to units of applied force in N, for both FSR types. Since the sensitivity curve is an exponential, it only approaches zero, but as discussed above, the FSRs are likely to have a small amount of pre-load. The calibration started at 0.25 kg, to define the output when a small force was applied.

The 95% confidence intervals were calculated for each FSR from the absolute minimum value seen during testing (0.47 for the FSR-400, 0.01 for the FSR-402) to 0.8, with an increment of 0.005; the mean across these values was 14.64 N for the FSR-400, and 14.95 kg for the FSR-402.

^{1.} Matlab function polyval.m was used to obtain error estimates, using "s", a structure returned from polyfit.m for estimating the error in the line-fit.



Figure 4.35 95% confidence intervals for FSR sensitivity curves

This error is rather large, and is likely due at least in part to the non-linearity of the FSR response. Different conditioning electronics, such as those discussed at the beginning of this section, that would result in a more linear FSR response should be implemented before using the FSRs for numerical analysis of the force distribution. In the analysis for this work, the FSRs were calibrated and summed together for use in determining heel strike and toe off timing.

4.6 Bend Sensors

4.6.1 Calibration

For the bend sensors, two types of information were required: the zero offset, to center the output around zero; and, the sensitivity, to convert the output to units of degrees (°).

Zero Offset

The zero offset of the bend sensors was simply the output of each bi-directional bend sensor while lying flat; the zero offsets are summarized in Table 4.5.

Sensitivity

Each bend sensor was characterized individually; the native resistance of the individual bend sensors varies widely, making the output of each pair of bend sensors unique. The

test setup is shown in Figure 4.36. Pairs of 9 V batteries were taped together and used as a weighted clamp to position the bend sensors. A printed graph with lines radiating to mark angles from -50° to $+50^{\circ}$ was taped to the table to guide the calibration process. One pair of batteries was taped to the table such that they created a gap parallel to the 0° line, with the right edge aligned with the (0,0) mark on the graph. The bend sensor was threaded through this pair of batteries, and positioned such that the midpoint of the bend sensor was aligned with the (0,0) mark on the graph. A second pair of batteries was placed over the bend sensor, about an inch away from the pivot point, and was used for positioning the bend sensor.

For each of the four bend sensors, a static calibration test was carried out, with data collected at 10° increments from 0 to $+50^{\circ}$, and from 0 to -50° . First, the bend sensor was aligned at the increment, and a small weighted box was place on each side of the positioning pair of batteries, to hold the bend sensor securely in place. One of the FSRs was pressed to indicate the start of data collection, and after 20-30 seconds, a second FSR was pressed to indicate the end of data collection. Then, the bend sensor was moved to the next increment, and the process was repeated.



Figure 4.36 Method of calibration of bend sensors



A sample of the data from the calibration of one of the bend sensors is shown in Figure 4.37 (the FSR plots are included to indicate the times of data collection; rapid changes in the bend sensor output, such as seen around 115 sec corresponds repositioning of the bend sensor in between the data collection). The output appears to change linearly from 0° through $+30^{\circ}$, but at angles higher than 30° , the output nears the maximum output of 1, and starts to saturate. In addition, there is some drift present: the output at 0° at the end of the calibration trial is different than at the start (a slight change in positioning could contribute to less than 1° of this change, but the full change is closer to approximately 5° , suggesting another process is contributing). In particular, it is likely that the electrical tape used to hold the two uni-directional bend sensor output is likely to be affected by the radius of curvature at the bending point.

For each bend sensor, the clipped means¹ were calculated on the output at each of the measured different angles, and on the data from the 0° position at the start of each trial (the data from the 0° position at the end of each trial was not used). A straight line was fit² to

^{1.} Again, as discussed in Section 4.3.1, spurious points were eliminated by using the clipped mean.

^{2.} Matlab function polyfit.m was used.

the clipped mean data points; for data which saturated at the upper limit, only the first clipped mean value greater than 0.95 was used, so as not to skew the line-fit away from the non-saturated data (the calibration was carried out only once¹). The clipped mean points, and the resulting line-fit are both shown in Figure 4.38 for all four sensors. The sensitivity (slope of the line-fit) and the zero offset are shown in Table 4.5.



One sensor (shown in red in Figure 4.38) actually had a uni-directional output, due to a deficient individual bend sensor, and saturated at the upper bound. This sensor was located in the right insole, and the saturation of this sensor was not an issue during subject testing of normal gait, as flexion of the insole is only possible in one direction; negative bend corresponded to insole flexion. The left insole sensor also had some saturation, but only at the higher range of positive bend, so it was not an issue during subject testing either.

^{1.} As the rest of this section discusses, the bend sensor did not perform well; though the calibration would be more accurate if the bending routine was repeated multiple times, the inaccuracies in the bend sensor discussed below in Section 4.6.2 and Section 4.6.3 were likely more influenced by the placement of the bend sensor and the drift issue discussed above. Future work with the GaitShoe should briefly investigate whether the influence of the placement and the drift can be alleviated, and if so, a better calibration routine should be devised at that time. But it is likely that far better results would be achieved by switching to a different type of sensor, or by placing a second IMU on the shin.
Sensor		Sensitivity [degrees/output]	Zero Offset [output] 0.54 0.59		
	Left, Ankle	87.7	0.54		
	Left, Insole	95.4	0.59		
	Right, Ankle	124.5	0.39		
	Right, Insole	104.8	1.00		

 TABLE 4.5
 Bend sensor sensitivities and zero offsets

The bend of the ankle bend sensors corresponds to plantar flexion and dorsiflexion; positive bend corresponded to dorsiflexion and negative bend corresponded to plantar flexion. The two sensors located at the left and right ankles also had some saturation, however, the outputs of both sensors are linear across the range of typical angles encountered. A healthy gait cycle typically has a maximum dorsiflexion close to 10° and a maximum plantar flexion of 10-20° [105].

To investigate the issue of drift further, the calibration was run on a bend sensor with two brand new uni-directional bend sensors, with and without tape covering. The results are shown in Figure 4.40; a black dashed line has been plotted over the bend sensors, corresponding to the value of the bend sensor at time zero. Again, the FSRs were used to indicate the start and stop of each calibration point.



Figure 4.39 Bend calibration routine, on a new sensor with and without tape

These results do not show the drift present in the earlier calibration plots, indicating that the drift is likely to be due to aging of either or both the bend sensor and the electrical tape used to protect the bend sensors and to force the two uni-directional bend sensors to follow the same motion (the bend sensors used in the subject testing were assembled with the tape in January 2003 at the latest). Further work is necessary to determine which aging process is the major contributor to the drift. If the electric tape, other methods of protecting the bend sensors (and forcing the same motion) could easily be found, such as slipping a plastic sleeve over the sensors. However, if the bend sensor itself is degrading as it ages, the use of this part may need to be rethought as it would be inconvenient and expensive (each individual bend sensor is \$10, for a total cost of \$80 across both feet) to replace.

4.6.2 Placement of the Bend Sensors

One of the bend sensors was used to check the variation in the output due to a change in the location of the pivot point, as shown in Figure 4.40. The insole bend sensors were not repositioned before each subject, but were located such that their midpoints were roughly aligned with the metatarsals, since flexion in the insole occurs about the metatarsals. The ankle bend sensors were positioned between the heel of the foot and the back of the shoe such that the midpoint was roughly aligned with the back of the shoe. In normal use, the ankle attachment (shown in Figure 3.9) would be mounted on the subject's ankle to hold the ankle bend sensor in place. However, testing was carried out with the subject simultaneously wearing the MGH Biomotion Lab equipment, and so as not to unduly interfere with the gait of the subject, the ankle attachment was mounted on the MGH Biomotion Lab shin attachment rather than directly against the ankle. This resulted in the ankle attachment being located further away from the edge of the shoe, particularly on taller subjects, so the bend sensor had to be pulled out of the shoe in order to be enclosed completely by the ankle attachment. This resulted in the bend sensor pivoting about a point between the end of the sensor and the midpoint.

Figure 4.40 demonstrates that the sensitivity of the output remains similar over the three different pivot points: the midpoint, and the 1/4 and 3/4 points along the sensor's length. However, there is some variation in the zero offset value, which is likely due to the drift seen in Figure 4.37, as these tests were performed in succession.



Figure 4.40 Bend sensor, calibrated with different pivot points

4.6.3 Plantar Flexion and Dorsiflexion

The bend sensor at the ankle was used to measure the angle of plantar flexion and dorsiflexion. However, as discussed above, the bend sensors have a baseline drift in the output. In addition, the ankle attachment, which was designed to hold the bend sensor next to the ankle to prevent buckling, was placed higher relative to the back of the shoe, during subject testing. This resulted in the ankle bend sensor being more likely to buckle during data collection. Also, plantar flexion and dorsiflexion were determined from the BML data from the pitch of the foot array and the pitch of the ankle array, resulting in the angle between the shank and the foot. There are two difficulties in using the BML data to validate the calibrated bend sensor data. First, the shank and foot arrays are visible to the camera detection system at different (but overlapping) times, which results in a shortened amount of data available as only the overlapping data can be used to determine the pitch. The second difficulty is that the BML data occasionally contains outlying data points; because two different data vectors are combined, there are two sources of error: these must be removed, and thus further shorten the data available for comparison. 148

The bend sensor output for an ankle bend sensor, and the combined output of the BML foot pitch and shank pitch, resulting in plantar flexion / dorsiflexion curve measurement for the corresponding foot are shown in Figure 4.41 (the bend sensor data shown is the result of a spline fit to the bend sensor calibrated output, using the time points corresponding to the BML data).



Figure 4.41 Plantar Flexion and Dorsiflexion from the GaitShoe (left) and the BML (right)

Although the shape of the calibrated bend sensor output generally follows the shape of the BML plantar flexion / dorsiflexion curve, the RMS error between the two curves is 6.3° . With a full range across the BML plantar flexion / dorsiflexion of less than 30° , this represents an error of more than 20%. The sharp changes in the GaitShoe ankle bend output (e.g. shortly after 1 sec and around 2.25 sec) are most likely due to the sensor buckling as a result of the poor location of the shoe attachment, as the direction of flexion reverses.

A sample of the calibrated insole bend data is shown in Figure 4.42, for the same gait trial as the calibrated ankle bend data shown in Figure 4.41. The minima correspond to the foot motion preceding toe off. This is a parameter that the BML does not measure: the BML foot array is mounted in the center of the foot and cannot distinguish between pure rotation of the foot, and flexion of the foot with the forefoot flat on the ground.



Figure 4.42 Calibrated insole bend sensor output

Because of these issues, the bend sensor signals were not further validated with the BML data. Future work with the GaitShoe using this bend sensor should involve a different design of the ankle attachment, or evaluation with a system that allows the bend sensor to be held next to the ankle as it would under conditions where only the GaitShoe is collecting data. Alternatively, other methods of determining plantar flexion / dorsiflexion should be considered, such as the use of fiber optic sensors, or placement of a second IMU on the ankle, which would provide the full gyroscope and accelerometer information, and would allow the position of the shin and foot to each be determined independently, and then combined to find relative positions, such as plantar flexion / dorsiflexion.

4.7 PVDF Strips

The polyvinylidene fluoride (PVDF) strips were included for their rapid dynamic response, and were expected to be useful in determining heel strike and toe off timing. Unfortunately, the output varied greatly, not only from subject to subject, but between the left and right feet of a given subject, and even from step to step within a given subject. The PVDF outputs from five females with healthy gait are shown in Figure 4.43; data from the left shoe are on the left, and data from the right shoe are on the right, and the outputs from the PVDF output stript.



was scaled from the twelve bit output such that the output ranged from 0 to 1, and was centered around zero.

Figure 4.43 Various PVDF output plots

Some of the variation seen across Figure 4.43 may be caused by the physical design of the shoe worn by the subject: the PVDF strip is traditionally used to measure vibration, and the subjects were asked to wear running shoes, which are designed to minimize impact on the foot via compressible components; in addition, the PVDF strip output is likely sensitive to motion of the foot within the shoe. Thus, some variation may be due to the PVDF strip picking up the dynamics of shoe as well as the dynamics of the foot moving within the shoe, rather than the dynamics of gait. To investigate the effect of the shoe type, data were collected on the same subject wearing three different types of shoes: a running shoe (manufactured by Reebok), a walking shoe (manufactured by Ecco), and a stiff clog (manufactured by Dansko); the shoes are shown in Figure 4.44.



Figure 4.44 Reebok running shoe, Ecco walking shoe, and Dansko clog.

Figure 4.45 shows the uncalibrated¹ data for each of three shoe types. The data are all from the right foot, and the same insole was used in all three trials. The back edge of the insole was aligned with the back of each shoe and taped into place, to position the PVDF strips in the same location relative to the subject's foot. Visually, the PVDF strips have a stronger similarity within a shoe-type than between shoe-types, particularly the toe PVDF strips, suggesting that the type of shoe worn during testing may indeed contribute to changes in the PVDF output.



Figure 4.45 Comparison of PVDF output in three different types of shoes

^{1.} To display both PVDF output clearly on one graph, the PVDF-heel output has been shifted by a value of one.

The PVDF strips were selected to be placed at the toe and the heel were because they were expected to have a faster response than the FSRs. Figure 4.46 shows the uncalibrated (centered around zero) PVDF outputs and the sum of the calibrated FSR outputs for comparison, as well as the heel strike time and toe off time determined by the BML. The heel PVDF output does appear to rise slightly in value one time step before the FSRsum output rises (and before the BML heel strike time); however, the PVDF sensor is located centrally beneath the heel, and closer to the rear edge of the heel, while the FSRs at the heel are located medially and laterally, so this very slight anticipation by the PVDF sensor may be due only to a better location within the insole. In addition, as seen in Figure 4.43 and Figure 4.45, the heel PVDF output tends to have multiple peaks, due both the dynamic nature of the PVDF output, as well as the likely response to movement of the foot within the shoe, whereas the FSRsum has a direct correspondence to the actual force distribution beneath the foot. The toe PVDF output has already returned to the baseline at the BML toe-off time, making it no more useful than the FSRsum.



Figure 4.46 Comparison of PVDF response to FSRsum response

This PVDF output certainly contains interesting information, and, the potential change in response in different types of shoes may be useful for future work in evaluating orthotics

or different types of shoes for a given patient. However, it was selected for this work for use in determining heel strike and toe off; the FSRs appear to quite useful, not only in determining heel strike and toe off, but in providing information about the force distribution underneath the foot. The analysis of heel strike and toe off time, as described, did not employ the use of the PVDF output, but rather relied on the FSRsum, and information from the accelerometers and gyroscopes.

For future work with the GaitShoe, the PVDF strips should be replaced with FSRs, which will allow the FSRs to be placed directly under the heel pad and under the great toe, resulting in a more complete force distribution profile, and possibly improving the determination of heel strike and toe off times.

4.8 Electric Field Sensor

The electric field sensor was developed after subject testing had started, so it was included during the testing of the last five subjects to investigate its utility for measuring the height of the foot above the floor. The purpose was both to evaluate the implementation, as well as to investigate the signal output. Electrodes for the electric field sensor were created by placing copper tape on the bottom of the shoe. Figure 4.47 shows the uncalibrated output of the electric field sensor on five different subjects¹. The output level which corresponds to the zero height at the initiation of gait (e.g. the output while the subject stands with both feet flat on the floor, before starting the gait trial, around time -6 sec in the graphs in Figure 4.47) is indicated.

The physical implementation required placing copper tape on the bottom of the shoe to construct the sensor, and co-axial cable to connect the sensor to the circuit board (the driven shield signal was connected to the outer layer of the co-axial cable). The copper tape on the bottom of the shoe lasted throughout the each subject's testing. The co-axial

^{1.} These subjects were the five subjects on which the electric field sensor was used; data were not collected from the right foot sensor on subjects A and B; these subjects are not the same as those in Figure 4.43, above

154

cable has a diameter of 3 mm; for most athletic shoes, the co-axial cable can be placed in a tread groove to avoid adding this small thickness under one side of the shoe, however, this connection may need to be redesigned for use with other types of footwear.



Figure 4.47 Various electric field sensor output plots

In addition, as seen in Figure 4.47, the output of the electric field sensor was greatly dependent on the implementation of the electrodes, which varied depending upon the size of the subject's shoe. the range of the output varies greatly over all eight implementations. Also, the actual output of the electric field sensor is likely to vary depending on the construction of the floor. Interestingly, most subjects generally pass the force plate between +1 sec and +4 sec; in the graphs shown in Figure 4.47, the electric field sensor output generally dips below its initial zero height during this time, as the metal force plate provides increased capacitive loading.

Figure 4.48 presents data collected during one of the gait trials from Subject A. The top graphs show the height of each foot, as measured by the MGH Biomotion Lab (BML). The height is measured by the foot array, which is located above the foot; thus, the height is never 0 cm, because the lowest height seen during stance (approximately 10 cm on the left foot, and approximately 5 cm on the right foot) corresponds to the actual height of the

array above the floor. The bottom graphs show the output of the electric field sensors on the left and right feet. The electric field output reaches its maximum value once the foot is lifted above the floor to a distance where the sensor can no longer couple into the floor. By visual comparison with the BML height graphs, both electric field sensors appear to have a range up to approximately 3-5 cm, over a third of the output range. The electric field sensor was located on the heel of the shoe, while the foot arrays were located mid-foot; this results in the electric field sensors anticipating the changes seen by the foot arrays.



Figure 4.48 Comparison of the electric field sensor output to foot height.

Before using the electric field sensor as a method of determining the height of the foot above the floor, further investigation into the design of the electrode is needed. For instance, as discussed in Section 3.3.7, after subject testing was completed, a second design was developed, using a round electrode made out of copper tape, with the driven shield inside the shoe. This sensor was not tested at the Biomotion Lab, but it does appear to have a range greater than 5 cm. In addition, two electrodes can be used underneath the foot, to provide measurements of height at two different locations. The output of the two electrodes on the left and right feet are shown in Figure 4.49. Again, the value of the output depends on the electrode configuration. The sensors located at the heels both clearly measure a decrease in distance above the ground before the sensors located at the toes (e.g. at 1.75 sec in the left foot heel sensor, and at 2.4 sec in the right foot heel sensor), while the sensors located at the toes are clearly measuring an increase in distance above



the ground after the sensors located at the toes (e.g. at 1 sec in the left foot toe sensor, and at 1.75 sec in the right foot toe sensor).

Figure 4.49 Sample output from the second electrode design

Once the electrode design has been finalized, methods of preparing the electrode such that it is consistent from subject to subject should be investigated. For instance, rather than affixing copper tape, electrodes could be pre-cut from a thin metal foil, which would be taped to the bottom of the shoe. Finally, data should be collected on different floor types, from wood floors and sidewalks to concrete floors with and without rebar, to generate a family of calibration curves and allow calibration of the electric field sensor signal.

4.9 Heel Strike and Toe Off Timing

Timing of heel strike and toe off were determined using several of the GaitShoe sensors. Initial values for each were determined using the sum of the four FSRs, "FSRsum", which was calculated when the FSRs were calibrated, as described in Section 4.1.4. These initial values were used to determine the integration bounds for the z-gyroscope. The final values of the toe off timing were determined from the pitch, and alternative values for the heel strike timing were determined from the x-accelerometer integration bounds.

FSRsum Determination of Heel Strike and Toe Off Timing

The four FSRs were coarsely distributed underneath the foot, with two underneath the heel, medially and laterally, one underneath the first metatarsal head, and one underneath the fifth metatarsal head. Though this cannot provide a complete picture of the force distribution underneath the foot, the information can still be of use.



Figure 4.50 Comparison of the sum of the four calibrated [N] FSR outputs to the force plate output

The calibrated FSR output with units of force [N] (e.g. the total force applied across the FSR), the sum of the four calibrated FSR outputs, and the BML force plate output are shown in Figure 4.50. The FSRs only cover a small percentage of the total weight-bearing area underneath the foot; as such, they only measure a portion of the total force. Thus the shape of the summed FSR output in the middle graph is different than the bottom graph of the force plate measurement, because the summed FSR output is subsumed by the total force output.

For further comparison, data collected using the Tekscan F-Scan system [32] on two different subjects was examined. One subject had a peak pressure under the heel area of 32 psi, a peak pressure under the area of the first metatarsal head of 52 psi, and a body weight of 144.8 pounds, while the other subject had a peak pressure under the heel area of 43 psi, a peak pressure under the area of the first metatarsal head of 33 psi, and a body weight of 162.6 pounds [92]. These are similar ranges as the calibrated FSR data, shown in Figure 4.51 with units of pressure [psi] (e.g. the pressure applied across the FSR sensing area), though Figure 4.51 presents data from a subject with a body weight of 253 pounds.



Figure 4.51 Comparison of the sum of the four calibrated FSR outputs in pressure units of psi.

Figure 4.50 demonstrates that the larger FSR-402s (located under the metatarsals) measure more force, while Figure 4.51 shows that even though the smaller FSR-400s (located under the heels) measure less force, the pressure under the heel, for this subject, is higher than underneath the metatarsals.

The change in the shape of the FSRsum, with either force or pressure units, as compared to the typical "m"-shape measured by force plate is a result of the very discrete measurements taken by the four FSRs. However, this does not render the FSR measurements use-less: in both Figure 4.50 and Figure 4.51, the shape of the FSRsum is similar across the three steps shown, however, there were substantial differences in the weight distribution between the first and fifth metatarsal heads. In the first step shown, more weight is on the

fifth metatarsal, in the second step, more weight is on the first metatarsal, and in the third step, the weight is distributed fairly evenly across the first and fifth metatarsal heads. This is information which cannot be obtained from a standard force plate.

In the BML, timing of heel strike and toe off are determined by examining the force plate output; when it first crosses a certain threshold, that time point is identified as heel strike, and when it re-crosses that threshold, that time point is identified as toe off.

This general approach¹ was adopted for deriving the heel strike and toe off times from the FSRsum, as shown in Figure 4.52. In order to improve the resolution, a spline was fit to the FSRsum with time points every 1 msec; the ideal data rate of the GaitShoe was 75 Hz, which corresponds to a complete data packet every 13.4 msec. The "no-load cutoff" was calculated from the mean of the minimum value of the FSRsum and the clipped mean of the FSRsum. The no-load cutoff was used as the limit below which the FSRsum value could correspond to a no load situation, such as during swing. The times of the FSRsum points (original data, not the spline-fit) that occurred just before and just after (FSRsum_{no-load} and FSRsum_{noload+}, respectively) the FSRsum crossed the no-load limit were identified; these are marked in Figure 4.52.



Figure 4.52 Determination of heel strike and toe off from FSRsum

^{1.} See hstos_fsrsum_spline.m, an m-file available in Appendix F.2.

To locate heel strike and toe off, the first difference of the spline-fit, $D(FSRsum_{spline-fit})$ was calculated, as shown in Figure 4.52. The maximum peak of $D(FSRsum_{spline-fit})$ corresponded to the peak loading during heel strike, and was located using the vicinity determined by FSRsum_{noload+}. The heel strike time was set as the first spline-fit time point prior to the maximum peak where the condition $D(FSRsum_{spline-fit}) < \Delta 0.005$ kg was met. This condition was set after inspection of many gait trials; as shown in Figure 4.53, $\Delta 0.005$ kg was greater than the small changes of FSRsum registered during swing, and indicated the start of the rapid increase in force corresponding to heel strike. This was non-objective method of finding the point corresponding to initial loading of the FSRs; for future work with the GaitShoe, the loading profile of the FSRs should be more fully investigated, perhaps by using a separate system where its time scale precisely aligned with the GaitShoe's time scale (as discussed in Appendix F.1, in the subject testing data, the time scale of the BML could only be aligned with the GaitShoe's time scale within 13.4 msec).

Similarly, the minimum value of $D(FSRsum_{spline-fit})$ corresponded to the peak unloading leading up to toe-off, and was located using the vicinity determined by FSRsum_{noload}. The toe off time was estimated at the first spline-fit time point following the minimum where the magnitude of $D(FSRsum_{spline-fit})$ was less than $\Delta 0.005$ kg. As shown in Figure 4.53, $-\Delta 0.005$ kg approximately corresponded to the unloading of the FSRsum due to toe-off and initiation of leg-swing.



Figure 4.53 First difference of spline-fit, zoomed-in

Because there are two FSRs located under the heel, this method was expected to perform well in determining the heel strike. However, there are no FSRs located underneath the great toe, so this method was expected to estimate the toe off time early. Thus, these heel strike and toe off times were used to determine the integration bounds for the z-gyroscope, and the toe off time was subsequently re-evaluated.

Maximum Pitch Determination of Toe Off Timing

The final toe off times were determined by the location of the minimum pitch. As shown in Figure 4.16, the maximum pitch occurs at toe-off; as the foot rolls off the foot, the pitch increases, and once the toe is off the ground, a rapid acceleration at the start of leg-swing occurs, and results in a decrease in the pitch. Once evaluation of the z-gyroscope was complete, the times corresponding to the maximum pitch values during each stride were determined using the post-spline-fit method described in Section 4.3.2. These toe off times were compared to values for toe-off obtained from the BML analysis.

X-Accelerometer Upper Integration Bound for Alternate Heel Strike Timing

As described in Section 4.4.3, the upper integration bound for the x-accelerometer was set when the magnitude of $Ax_{dynamic}$ was either less than 0.2 m/s², or when the value of $Ax_{dynamic}$ was greater than 0 m/s². This second condition was expected to correspond to an especially strong heel strike. Each of the upper integration bounds were investigated, and if the bound corresponded to a positive spike in the $Ax_{dynamic}$ output, the time point was saved as an alternate measurement of heel strike timing. Where available, these alternate heel strike times were compared to values for heel strike obtained from the BML analysis, as were the heel strike times calculated from the spline-fit of the FSRsum.

Chapter 5

GAIT PARAMETER VALIDATION

The outputs of several sensors in the GaitShoe system were analyzed to generate clinically relevant gait parameters. The data were collected during subject testing (described in Appendix B), and was evaluated by comparison to data collected simultaneously by the Massachusetts General Hospital Biomotion Laboratory (BML). This chapter discusses the results of the comparison with BML data. The examples presented while describing the analysis model and the analysis techniques all use data from the same gait trial.

Testing of the fifteen subjects resulted in 270 total trials of gait. The GaitShoe pitch, stride length, and vertical displacement results for all of these trials were compared to the data collected simultaneously from the MGH Biomotion Lab (BML). The validation was carried out by comparing quantities of interest from each of these results: the maximum and minimum values of pitch and the times at which they occurred, the total stride length, and the maximum vertical displacement and the time at which it occurred. The magnitude differences and the percent differences, using the BML as the reference value where available, were determined.

The GaitShoe heel strike and toe off times were compared to heel strike and toe off times determined by a physical therapist by inspection of the BML force plate data. For each subject, approximately three heel strike times and three toe off times of each times were determined for each foot, for a total of 86 comparisons. The magnitude difference was calculated for the heel strike times and the toe off times.

5.1 Pitch

The maximum and minimum values and times of the GaitShoe pitch were determined¹ with a post integration spline-fit², as described in Section 4.3.2. The BML data were evaluated for outlying data points, which were removed, and the maximum and minimum values and times of the remaining BML data were determined. Figure 5.1 shows the GaitShoe pitch, BML pitch, and all extrema.



Figure 5.1 Comparison of GaitShoe pitch to BML foot array pitch

Figure 5.1 was included to demonstrate the issues which arise in the BML data. The lower left graph in Figure 5.1 shows that, like the GaitShoe data (see Section 4.1.3), there are occasional outliers in the BML data; these were excluded³ for the purposes of determining the maximum and minimum points of the BML foot pitch. In addition, there are often unusual data at the start and/or end of the BML data, usually due to only a portion of the array being visible to the cameras. These effects can be seen at the start of both BML foot pitch graphs in Figure 5.1: the lower left graph has an unusual dip before the true pitch

^{1.} See pitch_compare_postspline.m, and pitchpeakfinder.m, m-files available in Appendix F.2.

^{2.} See gyroz_postspline.m, an m-file available in Appendix F.2.

^{3.} See findmghoutliers.m, an m-file available in Appendix F.2.

minimum at the start, and the lower right graph has an unusually large magnitude of pitch at the start; thus the first 16 and last 16 BML data points, corresponding to approximately 0.1 sec, were also excluded.

During determination of the extrema for both the GaitShoe data and the BML data, the number of dropped packets (in the BML data, dropped packets are the result of removal of outlying data points) in the vicinity of the extrema were stored, as well as the size of the largest number of successive dropped packets over the entire data series. Extrema were not included in the comparison if either the GaitShoe or the BML data had a series of dropped packets longer than 0.2 sec during the gait trial, or if either had 2 packets dropped in the vicinity of the extrema. This resulted in 279 comparisons of maximum pitch and time, and 241 comparisons of minimum pitch and time. The results are summarized in Figure 5.1, and histograms for each are shown in Figure 5.2. The percent change was not calculated for the time values, since the BML data, as seen in the lower right graph of Figure 5.1, did not always contain two successive pitch maximums or two successive pitch minimums from which to calculate a reference time.

Comparison	Difference		Percent Change [%]		Samples
	Mean	Std Dev	Mean	Std Dev	
Shoe Pitch Max minus BML Pitch Max	1.9°	6.7°	4.22	10.4	279
Time of Shoe Pitch Max minus Time of BML Pitch Max	04 sec	.02 sec			279
Shoe Pitch Min minus BML Pitch Min	-4.9°	5.1°	29.1	20.5	241
Time of Shoe Pitch Min minus Time of BML Pitch Min	01sec	.02 sec			241

TABLE 5.1	Pitch	Validation	Results
-----------	-------	------------	---------



Figure 5.2 Histograms of the pitch validation results

5.2 Distance

The stride length (horizontal displacement) was determined by the GaitShoe via doubleintegration of the acceleration along the X_{room} axis, and, for comparison, along the X_{shoe} axis. The vertical displacement of the foot was determined via double-integration of the acceleration along the Y_{room} axis. These displacements were compared to the BML data samples if the BML data included the time corresponding to a lower integration bound and an upper integration bound (e.g. the BML data encompassed an entire swing phase); this resulted in 363 total samples available for comparison. For the stride length, the total displacements were compared, and for the vertical displacement, the peak displacements and the corresponding times were compared. Comparisons were omitted if either the GaitShoe or the BML data had a series of dropped packets longer than 0.2 sec during the gait trial, or if either had a total of more than 10 packets dropped during the swing phase used in the comparison. This resulted in 166 comparisons of stride length, and 166 comparisons of peak vertical displacement and time. The results are summarized in Table 5.2; histograms for stride length are shown in Figure 5.3 and for peak vertical displacement and time in Figure 5.4 (the vertical axes are the same on all histogram plots; again, the percent change was not calculated for the time of peak vertical displacement, since an independent stride time was not always available from the BML data). Of note, while the magnitude of the mean difference of peak vertical displacement is only slightly larger than the mean difference of stride length, since the vertical displacement of the foot during stride is an order of magnitude smaller than the stride length (on the order of 10 cm as compared to more than a meter), the percentage change in peak vertical displacement is unacceptably large, at 72.7%. This is most likely mainly due to small errors in the pitch that have affected the determination of the acceleration in the Y_{room} axis.

Comparison	Difference		Percent Change		Samples
(All GaitShoe minus BML)	Mean	Std Dev	Mean	Std Dev	
Stride Length Double linear integration of Ax-room _{dynamic}	6.1 cm	15.4 cm	5.4%	12.9%	166
Stride Length Double linear integration of Ax-shoe _{dynamic}	-4.0 cm	20.1 cm	-1.8%	17.3%	166
Peak Vertical Displacement	6.9 cm	3.3 cm	72.7%	45.8%	166
Time of Peak Vertical Displacement	-0.03 sec	0.06 sec			166

TABLE 5.2	Displacement	Validation	Results
------------------	--------------	------------	---------



Figure 5.4 Histograms of the vertical displacement validation results

5.3 Heel Strike and Toe Off

The heel strike and toe off times determined by the GaitShoe were compared to times determined by a physical therapist at the MGH Biomotion Lab, by inspection of the force plate data. A total of 86 samples including both heel strike and toe off time were provided from the for validation; of these, nine were excluded, by inspection, because of obvious errors (such as a BML heel strike occurring when the GaitShoe FSRs did not have any load); these should be investigated further to find the reason for the errors. Two additional samples were excluded from toe off analysis because the IMU became loose towards the end of testing of one of the early subjects, and so the gyroscope output was not available.

As explained above, two methods of determining heel strike were evaluated. The first method used only the FSRsum to detect loading corresponding to heel strike; the second method used a spike in the x-accelerometer data corresponding to a strong floor impact where available and the FSRsum method if the x-accelerometer was not jolted. The toe off time was evaluated using the time of occurrence of the maximum pitch. The results are summarized in Figure 5.3, and histograms for each are shown in Figure 5.5.

Comparison	Differer	Samples	
	Mean	Std Dev	
Shoe Heel Strike minus BML Heel Strike FSRsum method only	-14.4	21.5	77
Shoe Heel Strike minus BML Heel Strike FSRsum and x-accelerometer method	-6.7	22.9	77
Shoe Toe Off minus Toe Off	-2.9	16.9	75

TABLE 5.3 Heel Strike and Toe Off Validation Results



Figure 5.5 Histograms of the heel strike time and toe off time validation results

5.4 Discussion

The overall results from this initial analysis are encouraging, though a more detailed analysis may be needed to improve the results.

The GaitShoe's calculation of maximum pitch correlate well with the BML reference pitch, with a mean difference of 1.9° (standard deviation 6.7°), which corresponded to a mean percent change of 4.22% (standard deviation 10.4%). The difference in time between the maximum pitch points in each group was 40 msec. However, the calculation of minimum pitch did not correlate quite as well. The mean difference was -4.9° (standard deviation 5.1°), but because the magnitude of the minimum pitch is typically on the order of $10-15^{\circ}$, this corresponded to a mean percent change of 29.1% (standard deviation

20.5%). The minimum pitch suffers from cumulative integration errors, as it occurs toward the end of the stride.

The stride length results are very encouraging. Surprisingly, the mean difference is better when the stride length is determined via integration along the X_{shoe} axis than via integration along the X_{room} axis; however the standard deviation is much larger along the X_{shoe} axis. As compared to the BML stride length, the stride length calculated along the X_{shoe} axis has a mean difference of -4.0 cm (standard deviation 20.1 cm) and percent difference of -1.8% (standard deviation 17.3%), while as compared the stride length calculated along the X_{room} axis has a mean difference of 6.1 cm (standard deviation 15.4 cm) and percent difference of -5.4% (standard deviation 12.9%). These results are encouraging considering the large number of assumptions made in the analysis; by incorporating the full analysis of all three gyroscopes and all three accelerometers, both the mean difference, and the standard deviation are likely to decrease. The results of stride length calculated along the X_{shoe} axis suggests that if smaller and less expensive system were required for certain applications analyzing straight-line gait, a single-axis accelerometer could be used with a single gyroscope.

Though the difference for the vertical displacement as compared to the BML results is of similar magnitude to the stride length difference, the percent change is much larger. A typical stride length is longer than a meter, while vertical displacement is generally under 10 cm. Thus, while the vertical displacement as compared to the BML results has a mean difference of 6.8 cm (standard deviation 3.8 cm), the percent difference is -72.7% (standard deviation 50.0%). The time of the peak displacements as compared to the BML results has a mean difference of -.03 sec (standard deviation 0.06 sec). Unlike the stride length, these results appear to have suffered from the large number of assumptions made in the analysis, but will hopefully improve once the analysis is broadened to include all three gyroscopes and accelerometers. In addition, errors in the pitch are likely to strongly contribute to the large errors in the vertical displacement, as well as the result that the stride length calculated along the X_{shoe} axis is as accurate as or better than the stride length calculated X_{room} . Determining the dynamic acceleration requires the orientation of the accelerometers with respect to the room, Φ_x and Φ_y , which correspond to the pitch. However, in addition, determination of the accelerations along the Y_{room} and X_{room} axes from the dynamic acceleration uses relationships involving the sines and cosines of Φ_x and Φ_y . Thus, errors in the pitch affect the calculations along the Y_{room} and X_{room} axes more than the calculation along the X_{shoe} axis. Improvements in the pitch calculation are therefore likely to improve the results of the stride length and vertical displacement along the Y_{room} and X_{room} axes. In particular, a hardware change to use the Analog Devices gyroscope to measure rotation about the z-axis, rather than the Murata gyroscope, may sufficiently improve the pitch calculation; as discussed in Section 3.3.2, the Analog Devices gyroscope has two vibrating structures operating in anti-phase to reduce common mode signals unrelated to angular velocity (such as external shocks or vibrations).

The results for the heel strike time and toe off time are particularly interesting. Both methods of calculating the heel strike time anticipate the time determined by the BML; the method using the FSRsum has a mean difference of -14.4 msec (standard deviation 21.5 msec) as compared to the BML time, while the method using the x-accelerometer in combination with the FSRsum has a mean difference of only -6.7 msec (standard deviation 22.9 msec). These results suggest that the it is possible that the automated analysis of the GaitShoe results in detection of heel strike earlier than the human interpretation of the force plate data. However, the time scales of the GaitShoe system and the BML system are only aligned within 13.4 msec (see Appendix F.1), such that the GaitShoe time may register up to 13.4 msec ahead of the BML time; this may account for the shift between the detection on the two systems. Further evaluation, perhaps with a different type of validation equipment, and certainly with an improved method of aligning the time scales, is recommended to confirm these results. The toe off times are the most well correlated parameter in comparison with the BML system, with a mean difference of only -2.9 msec (standard deviation 16.9 msec). However, these results are moderately surprising, given that the maximum pitch was used to detect toe off (as the maximum pitch occurs right around the time of toe off), and the pitch results demonstrated that the time of detection of maximum pitch by the GaitShoe had a mean shift of 40 msec from the BML maximum pitch. It would be interesting to add FSRs underneath the great toe to see how the toe off time detected by an FSRsum including more FSRs compared to the BML and maximum pitch generated toe off times.

Chapter 6

PATTERN RECOGNITION ANALYSIS

The ability of the GaitShoe system to provide information capable of identifying gait patterns was investigated by using standard pattern recognition techniques with the data collected during subject testing. Two classification problems were investigated; the primary goal was to distinguish the gait of subjects with Parkinson's disease from the gait of subjects with normal gait. Of secondary interest was recognizing individual subjects from their gait. This chapter discusses the techniques used, the selection of the subject data that was classified, the selection of the feature set, hypotheses, training and testing data, and the results of the classification.

6.1 Pattern Recognition Techniques

Several classic techniques were applied to data sets extracted from the subject testing, in order to investigate which technique was best at classifying this type of data. These techniques are described briefly here; all employ supervised learning, where a labeled training set was used to train the classifier, and a separate labeled test set (containing samples not in the training set) was used to evaluate the classifier.

6.1.1 Classification and Regression Trees (CART)

CART is a decision tree tool, distributed by Salford Systems software. It was developed from original work by Breiman, Friedman, Olshen, Stone at the Stanford University and

the University of California at Berkeley [93]. It uses binary recursive partitioning; each node is split into exactly two nodes, and the process is repeated with each child node. The CART software analyzes the data through a set of rules which determine the following: node splitting (to create the tree), tree completion, tree pruning, terminal node classification.

To determine the split of each node, the CART conducts a brute force search by calculating all possible splits for all features. For continuous data (as opposed to categorical data), such as was generated from the GaitShoe data, a split is a numerical point which forms all or part of the boundary that separates two classes. For example, in gait data there might be a split in stride length at 1.4 m, such that stride lengths greater than 1.4 m correspond to male gait, while stride lengths shorter than 1.4 m correspond to female gait. The brute force search finds the split for each of the features for all samples in the data set, and then selects the best split for each node using a set rule. The Gini rule was the default in CART; the implementation of the Gini rule in CART was set to separate classes of data one at a time, by trying to isolate the classes in order of importance (most to least) [94]. By default in CART, importance is defined by the number of samples in the class, with larger classes more important (classes can be weighted; this option was not used). Thus, the split that provides the highest separation of the largest class was selected.

The splitting process is repeated on each resulting node until the tree is completed, as determined by the default settings: splitting was stopped when either fewer than ten samples were in a node or when all samples in the node were of a single class.

Once the tree is completed using the training set, CART uses the testing set of data to prune the split nodes to select the "best" tree, determined by the default "standard error rule," which picks the tree with the minimum cost. The cost is calculated by the rate at which the samples in the test set are misclassified, and is determined for both the full tree and for all possible sub-trees, with a penalty for large trees (because the testing set is used to select the final tree, this results in an overly optimistic estimation of error rates). This method is "overly optimistic" in that the final decision for the tree is determined by the testing set, rather than determined independently of the testing set.

One of the great benefits of CART is that the tree provides information about the most useful parameters in the feature set, as the splits at each of the nodes represent the features which best classify the data. In addition, the data set can have any number of classes, and the software completes the classification in seconds. There were two limitations to the software: the interface is not readily automatable, making analysis of several data sets tedious, and CART does not directly report or store which data were misclassified [94].

6.1.2 Bayes Decision Theory & Naïve Bayes

Bayes decision theory is a classification method based on probabilities. It assumes that the classification problem can be described in probabilistic terms, and that all the underlying probability values are known or can be reasonably approximated.

In order to calculate the posterior probability, $P(\omega_j/x)$, that *x* is in class ω_j , the following probability values¹ must be obtained or estimated: 1) the state-conditional probability density function, $p(x/\omega_j)$, which is the likelihood that *x* is observed, given class ω_j ; and, 2) the prior probability, $P(\omega_j)$, of class ω_j , which is the probability that any observation is class ω_j (all prior probabilities sum to one). Bayes rule, shown in Eq. 6.1, states that the posterior probability that sample *x* is in class ω_j is equal to the likelihood of *x* multiplied by the prior probability of class ω_j and divided by the evidence, p(x), defined in Eq. 6.2. The evidence is a scale factor to guarantee that the posterior probabilities over all classes sum to one. Bayes rule was published posthumously, in 1763, by mathematician Rev. Thomas Bayes [95].

$$P(\omega_j | x) = \frac{p(x | \omega_j) P(\omega_j)}{p(x)}$$
(6.1)

^{1.} P represents a probability, and p represents a probability density

$$p(x) = \sum_{j=1}^{n} p(x|\omega_j) P(\omega_j)$$
(6.2)

For example, Bayes rule would state that the probability that a subject with a stride length of 1.3 m is a female is equal to the likelihood that female subjects have a stride length of 1.3 m multiplied by the probability that the subject is female, and divided by the evidence. The evidence would be equal to the sum of the numerator plus the likelihood that male subjects have a stride length of 1.3 m multiplied by the probability that the subject is male.

For the two class problem, both posterior probabilities, $P(\omega_1/x)$ and $P(\omega_2/x)$, are calculated and *x* is classified as class ω_1 if $P(\omega_1/x) > P(\omega_2/x)$; otherwise, *x* is classified as class ω_2 , where the data sample is either a single feature *x*, or a vector *x* of several features.

When x is a vector of several features, information about the interdependence of the features is required to estimate the prior probabilities. Naïve Bayes is a technique which simplifies the estimation of the prior probabilities by assuming that class-conditional independence exists. This assumes that all features are independent from each other, so the prior probability can be simply calculated from the prior probabilities of each individual feature, as shown in Eq. 6.3.

$$p(\omega_j | \mathbf{x}) \propto \prod_{i=1}^d p(x_i | \omega_j)$$
(6.3)

Though the class-conditional independence assumption rarely holds in practice, Naïve Bayes classifiers are easy to implement, and often perform reasonably well. In addition, multiple classes can be considered as easily as two classes [96].

Use of Bayesian approach is usually to update the prior probabilities as new information becomes known; in this case, the analysis is carried out once. Since the Naïve Bayes assumptions are unlikely to apply to the gait feature data, which are certainly not independent, this was not expected to provide the best classification, but was included to see how well a simple technique could perform.

For this work, an implementation of Naïve Bayes in a program called Weka was used. Weka is a collection of machine learning algorithms for data mining tasks, written in open-source Java, and developed at the University of Waikato in New Zealand. The probabilities were estimated using the (default) normal distribution, and the classification was complete in a few seconds [97].

6.1.3 Support Vector Machines (SVM)

SVMs are a classification method based on a linear learning machine (a learning algorithm that uses linear combinations of the input variables) and were first introduced in 1992 by Vapnik and co-workers, at the Computational Learning Theory conference [98].

SVMs use linear discriminant functions, which means that the form of the underlying function is known (or assumed), and the training data are used to estimate the values of the parameters of the classifier. Unlike Bayes methods, SVMs require no knowledge or assumptions of the underlying probability distribution of the samples.

The essential function of SVMs is to find the optimal hyperplane that separates the labeled samples into two categories. The optimal hyperplane is defined to be the one that has a maximal distance to the closest training samples (see Eq. 6.4, below). This distance is called the margin, and the closest training samples are called the support vectors. The support vectors are the most informative samples; the hyperplane could be recreated from their information alone.

A larger margin corresponds to a SVM classifier with better generalization properties, in the sense that the distance between the hyperplane and the support vectors corresponds to the separation between the two data classes. The complexity of a SVM classifier is indicated by the number of support vectors; more support vectors indicate a more complex classification. Training data are not expected to be linearly separable in the feature space. Thus SVMs use a "kernel" to map the feature to a higher dimensional space where separability is more likely. In this case, it is useful to replace the dot product with a kernel function, to map the data to a higher dimensional space where it may be linearly separable by a hyperplane. Many functions can be used as a kernel function. While use of kernels can be a powerful technique for separating data, use of kernels can result in a feature space that overfits the training data, but does not generalize well to new data despite separating the training data well.

In practice, the generalized optimal hyperplane is found by maximizing the margin and minimizing the classification error in this new space. For example, let x_i be the original features, then given a hyperplane defined by $w_1x_1 + w_2x_2 + ... + w_Nx_N = b$, the generalized optimal hyperplane is found by minimizing

$$\Phi(w,\xi) = \frac{1}{2} \|w\|^2 + C \cdot \sum_{i}^{N} \xi_i, \qquad (6.4)$$

where ξ_i is some error function, and C is a weighting factor to weight the importance of maximizing the margin vs. minimizing classification error. By using Lagrange undetermined multipliers, and the Kuhn-Tucker construction, this optimization can be carried out by maximizing

$$L(\alpha) = \sum_{i}^{N} \alpha_{i} - \frac{1}{2} \sum_{k,j}^{N} \alpha_{j} \alpha_{k} z_{j} z_{k} K(x_{1}, x_{2}), \qquad (6.5)$$

subject to the constraints

$$\sum_{k=1}^{N} z_k \alpha_k = 0 \quad \alpha_k \ge 0, \, k = 1, \dots, n,$$
(6.6)

given the training data. $K(x_1, x_2)$ refers to the kernel used, and the α_k are the undetermined Lagrange multipliers, in z space.
Many implementations of SVMs are available [99]; the Matlab implementation by Steve Gunn was used [100] [101]. This classifier was easily automated within Matlab, and in addition, the classifier ran quickly, generally taking under 20 seconds. The data were treated as linearly separable in the native feature space (the dot product was not replaced with a specialized kernel).

SVMs do not provide information about which features are most informative. In addition, in most implementations, including the one used, SVMs can only classify two categories (it is possible to use SVMs to classify more than two categories, by serial analyses, but this is considerably more complex, and was not used) [96] [101] [102].

6.1.4 Neural Networks

An artificial neural network is an adaptive learning method that represents the data through the use of a parallel network of nodes ("neurons") arranged in layers and connected by weighted links. The link weights are determined by the training data.

A neuron with a single input is shown in Figure 6.1, left. The neuron is simply a model for processing the scalar input p via multiplication by weight w and addition of bias b. This sum, net input n, is the input to function f, which determines the output a. The function f can take any form, from a hard limit function, where a = 1 if $n \ge 1$ and a = 0 if n < 0, to a linear function, where $a \propto n$, to a sigmoid function, such as $a = 1/(1+e^{-n})$. This particular sigmoid function results in an output a that has a range of 0 to 1, for any value of n between plus and minus infinity. Sigmoid functions are often referred to as "squashing" functions because they compress the input into a predefined range; a key feature of all sigmoid functions is that they are differentiable.

The neural network is built by composing layers consisting of multiple neurons, where the input is usually a vector of multiple features. Figure 6.1, right, shows a vector input p and a single layer; the weights form a matrix w and the biases form vector b. Neural networks are not limited to a single layer: multiple layers can be added serially, by using the outputs

of a the first layer as inputs to the next layer. The final layer is called the output layer, and all other layers are called hidden layers. For pattern recognition, the neural network is set up such that the final layer has as many neurons, and thus as many outputs, as there are classes.



Figure 6.1 Examples of a single input neuron and a multiple input neural layer

Backpropagation is a frequently used method to improve the accuracy of neural network. In backpropagation, an effective error is calculated for every neuron, and this error is used to adjust the weights and biases; the sigmoid function is commonly used in backpropagation. To train a network, the training data is first applied to the network, in order to generate the output vectors, *a*. Next, the error is calculated, by comparing *a* to the labels on the training data; generally, a sum-of-squares error or mean-squared error is used, such that the error is proportional to $\sum_{s=1}^{s} \sum_{i=1}^{l} (a_{i,s} - b_{i,s})^2$, where I is the total number of neurons, and S is the total number of samples in the training data. This type of error is useful analytically, because the errors for each output $a_{i,s}$ remain independent, and are simply summed together to calculate the overall error.

There are many variations of the error adjustment algorithm used in backpropagation. The basic implementation adjusts the weights and biases in the direction of the negative of the

gradient, which corresponds to the direction in which performance degrades most quickly, such as

$$\dot{\bar{x}}_{k+1} = \dot{\bar{x}}_k - \alpha \dot{\bar{g}}_k, \tag{6.7}$$

where \dot{x}_k is a vector containing the current weights and biases, \dot{g}_k is the current gradient, and α is the learning rate. The learning rate is a parameter that can be set to control how much the weights and biases are updated, where a number closer to zero corresponds to a slower learning rate; typical values are between 0.05 and 0.75 (α is always positive).

There are two approaches for updating the weights and biases: batch learning and on-line learning. In batch learning, \tilde{g}_k is calculated from the sum of the individual gradients of the errors for all samples in the training set, given \tilde{x}_k , and the weights and biases are updated across all samples. Here, the subscript *k* refers to a single pass through all the training set; such a pass is called an epoch. Batch learning updates the weights and biases slowly, so many epochs are usually required to minimize the error. However, \tilde{g}_k is a good approximation of the true gradient.

In on-line learning, a sample is selected from the training set, the gradient \hat{g}_k is calculated for that sample, and the weights and biases are immediately updated. Here, the subscript *k* refers to a sample. In a single epoch, on-line learning updates the weights and biases as many times as there are samples, whereas batch learning updates only once (typically, the samples are selected in a random order each epoch, so as to avoid cyclic effects). However, the individual \hat{g}_k gradients are essentially noisy estimates of the true gradient, so this approach no longer approximates the true gradient. Thus, at individual samples, the updates may result in a larger overall error. However, although the path may not be direct, overall, the error will decrease; however, the randomness in on-line learning results in a jitter about the minimum. This generally prevents the network from getting stuck at a local minimum, which can happen in batch learning, because the jitter allows the on-line learning algorithm a method to move out of a local minimum. Another parameter which is often incorporated into backpropagation is a parameter called momentum, μ . The purpose of momentum is to prevent a network from being stuck in a local minimum, by adjusting the weights not only in response to the local gradient, but also in response to recent trends in the error. It can be incorporated by including a fraction of the previous weight change, $\Delta \hat{x}_{k-1}$, such as

$$\dot{\bar{x}}_{k+1} = \dot{\bar{x}}_k - \alpha \dot{\bar{g}}_k + \mu \Delta \dot{\bar{x}}_{k-1}, \qquad (6.8)$$

where the momentum, μ , is between 0 and 1 [96][103][104].

For this work, the implementation in Weka was used (Matlab 6.5 also contains an extensive neural network toolbox). The Weka neural network uses backpropagation to train, with sigmoid functions at the nodes. The default values for the training parameters were used: the learning rate was 0.3, the momentum was 0.2, the number of hidden layers was set to one half the sum of the number of features plus the number of classes, and the number of epochs was 500. There was also a validation threshold, set to 20, which limited the number of times in a row the error could get worse before training was stopped, and a reset flag, set to true, which allowed the weights and biases to be reset and the training restarted with a lower learning rate, if the network diverged from the training output. The final values of the weights and biases were determined with the training data, and subsequently evaluated by the testing data. This implementation completed the training in under a minute [97].

6.2 Data Sets and Classifications

Since the primary goal was to investigate the use of pattern recognition techniques to distinguish Parkinsonian gait from normal gait, using data from the GaitShoe, it was important to select the data sets carefully so as not to bias the results. Factors which contribute to natural variations in gait were examined, and the subject data available was evaluated for occurrence of these factors. The data classes were selected in an effort to minimize differences between classes due to known causes of variations in gait, as discussed below.

6.2.1 Gait Variation

Several factors which contribute to variation in gait were considered. Over a general population of people under age 65, there is a standard deviation of 10% across gait parameters such as velocity or stride length. This variation across the population is due to a number of factors, such as gender, age, and subject height [105].

Gender

There are statistical differences between the gait of males and of females. These are summarized in Table 6.1. In general, males have a longer stride length, but take fewer steps per minute, as compared to females. Overall, these two factors combine to give males a slightly faster velocity than females [105].

TABLE 6.1 Gender differences in gait [105]

	Males	Females
Mean stride length [m]	1.46	1.28
Mean cadence [steps/min]	111	117
Mean velocity [m/min]	82	77

Age

In healthy subjects with normal gait¹, age does not have a measurable effect on gait until the population includes subjects older than 60 years. For a group of subjects aged 60 to 65 years, the mean velocity was found to decrease by 3%, while for a group aged 60 to 80 years, the mean velocity decreased by 9% [105].

Leg Length

In adults, there is a weak correlation between stride length and leg length during walking gait (a stronger correlation exists during running gait). A study which grouped 120 male

^{1.} When arthritis and other disabilities which affect gait are included, studies of gait in older adults have shown a 14% deviation in velocity [105].

adults into equal groups of short, medium, and tall heights (leg length generally corresponds to height) found only a 4% change between the mean stride lengths in each group [105].

6.2.2 Subjects and Data Sets

The fifteen subjects (detailed subject information is available in Appendix B.2) were divided into five subgroups, summarized in Table 6.2 by presence of PD, gender, and age. As discussed above, gender and age may affect parameters of gait such as stride length and velocity. With large numbers of subjects, the number of males and females would likely be balanced, however, in this small study, only three out of the ten subjects with normal gait were male. This resulted in a concern that differences between male and female gait might affect the classification.

Group	Parkinson's disease	Gender	Lowest Age [years]	Highest Age [years]	Number of subjects
1	No	Female	24	28	5
2	No	Male	25	30	3
3	No	Female	48	54	2
4	Yes	Female	53	65	3
5	Yes	Male	64	76	2

TABLE 6.2 Summary of subject groupings

In addition, the subjects with PD all had an age greater than 50 years; in particular, the male subjects with PD were 64 and 76, while the male subjects with normal gait were all 30 years old or younger. However, the seven females with normal gait had two clusters of ages: five subjects in their mid-twenties, and two 48 and 54 years old. The three females with Parkinson's disease were in their fifties and sixties, so the female subjects with normal gait had two subjects who could be considered age-matched to the female subjects with PD. Thus, to not bias the classification results, the groups selected for data sets included only the ten female subjects.

In regards to the effect of leg length¹, the ten females had a mean height of 1.62 m, and a standard deviation of 0.03 m (the tallest subject was 1.68 m, and the shortest was 1.57 m). It was expected that, given the small spread of heights across the ten females, leg length was unlikely to contribute significantly to any changes between the groups.

The groups selected for the classification are summarized in Table 6.3. Class "NLFY" ("young" females with normal gait) had a total of 92 trials across five subjects, class "NLF" (age-matched females with normal gait) had a total of 37 trials across two subjects, and class "PDF" (age-matched females with Parkinson's disease) had a total of 64 trials. This sums to a total of 193 trials across all ten subjects.

Moniker	Group	Number of Subjects	Total Trials
NLFY	1	5	92
NLF	2	2	37
PDF	4	3	64

TABLE 6.3 Classifications of the ten female subjects

6.3 Feature Set

A key decision when building a classifier is the selection of features (the techniques described above all classify samples via a vector of single features about the sample; none evaluate the sample based on a data-time series). Clearly, the success of the classifier will greatly depend on whether the features encapsulate the distinguishing characteristics of the data classes.

The GaitShoe provides such a vast amount of information that selection of features can be viewed as an ongoing problem, with the features used here as an initial solution. The feature set used is detailed in Table 6.4. The feature set was developed to include a variety of

^{1.} Though leg length was recorded via the BML calibration routines, body height was directly accessible from the subject information and was used for this comparison.

information about the gait, but without unduly influencing the outcome of the classification. Parameters traditionally expected to change with age, such as stride length and velocity were not included; however, the training process of a classifier should eliminate dependence on parameters that are not related to age, if age is not the classification goal, so parameters, such as foot pitch, that may be correlated with stride length and velocity were included.

Table 6.4 has five columns: the first numbers the features, the second lists the data type, the third explains the type of analysis used to extract the feature from the data, the fourth lists the metric used, and the fifth is a description of what the feature represents. Three types of analyses were applied: the mean of the data, the standard deviation of the data ("std dev"), and the clipped standard deviation (the standard deviation of the data with the top and bottom 10% of data removed; "std dev (clipped)").

Two metrics were used, "L and R combined" and "L to R ratio." The metric "L and R combined" was a combination of the data from the left and right feet, where the analysis was applied to a vector consisting of a concatenation of all the data from the left foot and all the data from the right foot (e.g. the right foot data vector was appended to the left foot data vector). The purpose of this metric was to extract an overall feature from both feet. Initial work included the left foot and the right foot data as separate features, but it was found that left foot and right foot features were often interchangeable, so they were combined for the final feature set.

The metric "L to R ratio" was the absolute value of minus one plus the ratio of the analysis applied to the data from the left foot divided by the analysis applied to the data from the right foot. The purpose of this metric was to get a measure of any asymmetry between the two feet (healthy gait is expected to be symmetric). One was subtracted from the ratio, and the absolute value was applied, so as to lump all the asymmetries together (this resulted in a slight skewing of asymmetries in which the value for the right foot is greater than the value for the left foot, but was not expected to affect the results significantly).

TABLE	6.4	Feature	set

	Data Type	Analysis	Metric	Description
1	FSRsum/bodyweight	Std dev (clipped)	L and R combined	Walking energy variation
2	FSRsum/bodyweight	Std dev (clipped)	L to R ratio	Walking energy asymmetry
3	StepF/bodyweight	Mean	L and R combined	Step energy amplitude
4	StepF/bodyweight	Mean	L to R ratio	Step energy asymmetry
5	StepF/bodyweight	Std dev	L and R combined	Step energy variation
6	Gyro-x	Std dev (clipped)	L and R combined	Yaw variation
7	Gyro-x	Std dev (clipped)	L to R ratio	Yaw asymmetry
8	Gyro-y	Std dev (clipped)	L and R combined	Roll variation
9	Gyro-y	Std dev (clipped)	L to R ratio	Roll asymmetry
10	Gyro-z	Std dev (clipped)	L and R combined	Pitch variation
11	Gyro-z	Std dev (clipped)	L to R ratio	Pitch asymmetry
12	Accel-x	Std dev (clipped)	L and R combined	Forward motion variation
13	Accel-x	Std dev (clipped)	L to R ratio	Forward motion asymmetry
14	Accel-y	Std dev (clipped)	L and R combined	Upward motion variation
15	Accel-y	Std dev (clipped)	L to R ratio	Upward motion asymmetry
16	Accel-z	Std dev (clipped)	L and R combined	Sideways motion variation
17	Accel-z	Std dev (clipped)	L to R ratio	Sideways motion asymmetry
18	Maximum pitch ^a	Mean	L and R combined	Shuffle index
19	Maximum pitch	Mean	L to R ratio	Asymmetric shuffle
20	Maximum pitch	Std dev	L and R combined	Shuffle variation
21	Minimum pitch	Mean	L and R combined	Shuffle index
22	Minimum pitch	Mean	L to R ratio	Asymmetric shuffle
23	Minimum pitch	Std dev	L and R combined	Shuffle variation
24	Percent stance time	Mean	L and R combined	Shuffle duration
25	Percent stance time	Mean	L to R ratio	Shuffle duration asymmetry
26	Percent stance time	Std dev	L and R combined	Shuffle duration variation

a. This parameter was ultimately excluded, as it possibly was age-dependent in this sample.

Features 1-5 involve the summed output from the FSR sensors. Features 1 and 2 involve the sum of the FSRs ("FSRsum") divided by the subject's body weight, and features 3, 4, and 5 involve the integration of the sum of the FSRs between heel strike and toe off ("StepF") divided by the subject's body weight.

Features 6 through 17 are the clipped standard deviations of the calibrated outputs of the three gyroscopes and the three accelerometers, with both metrics applied. The bend sensor outputs and the pvdf strip outputs were not used at all for the feature set, because of the likelihood that there was intra-subject variation due to the fitting of these sensors, as discussed in Section 4.6 and Section 4.7 respectively, rather than resulting from changes in the gait of the subject.

The remaining features, like features 3-5, involved quantities that were derived from sensor outputs during the gait parameter analysis. Features 18 through 23 involve the pitch of the foot, which comes from the integration of one of the gyroscopes, and as discussed in Section 4.3.2, the characteristic pitch of the foot has a maxima and a minima during each stride (see Figure 4.16 on p. 120). Features 18, 19 and 20 use the maximum pitch values, and features 21, 22 and 23 use the minimum pitch.

However, feature 18, the mean of the maximum pitches in the left and right feet, was excluded when it was seen that it may have had a correlation with age in this sample. Initial work with CART revealed that when the older of the two NLF subjects was used as the test set, feature 18 was used as a splitting criteria, and the test set was grouped with the PDF group.

This feature, and feature 20, the mean of the minimum pitches in the left and right feet, are shown for the data samples of each subject, plotted against the subjects' ages in Figure 6.2. While it is impossible to say conclusively whether either of these features have some age-dependency, the mean maximum pitch does seem to be significantly less for the subjects older than 50 years, while the minimum pitch has a lesser degree of separation by age. Though it generally trends upwards, the mean minimum pitch of the subjects older

than 50 overlaps with those younger than 50 by more than 10 degrees out of a full range of 30 degrees (approximately 30% overlap), while the mean maximum pitch of the two groups overlap by under 5 degrees out of a full range of 40 degrees (less than 13% overlap). While the training of the pattern recognition system should exclude age-related parameters if not classifying by age, it was decided to exclude the mean maximum pitch as a feature (both of these parameters will be interesting to investigate further, with larger subject groups).



Figure 6.2 Plots of maximum and minimum pitch, by age

The final features, 24 through 26, use the calculations of heel strike and toe off timing to determine the percentage of the step each foot spends in stance ("percent stance time").

In feature generation from the data which were not continuous time series, there were often only a few measurements available in each sample, for a few reasons. There were usually five to seven footsteps per sample, so the number of total measurements was small. The integrated FSR sum and the pitch extrema measurements were likely to be adversely affected by dropped packets, so these data were analyzed for dropped packets in the region of each of the individual measurements for that sample. If there were a significant number of dropped packets, as listed in Table 6.5, the measurement was not used in determining the feature. In evaluating the percent stance time, which was calculated by

Percent Stance Time =
$$\frac{T_{TO_k} - T_{HS_k}}{T_{HS_{k+1}} - T_{HS_k}},$$
(6.9)

the last step was not used, in order to exclude any variance due to slowing at the end of the gait trial. This meant that samples that had only four footsteps total only had two usable measurements of percent stance time, since calculation of percent stance time required two successive heel strikes.

Data	Exclusion Criteria
Integrated FSR sum	10 dropped packets between heel strike and toe off
Maximum and minimum pitch	5 dropped packets in vicinity of each extrema

TABLE 6.5 Measurement exclusion due to dropped data

Because of all these factors, a few of the samples have only one or two measurements for either the left foot or the right foot or both, which meant the standard deviation could not be calculated for that foot. Thus, the ratio of the standard deviation was not used as a feature for the analysis of these data; there were no samples that had only two or fewer measurements total for the left and right feet together, so the standard deviation of the combined left and right feet measurements was used.

Again, this feature set of 25 features is an initial attempt to characterize the gait. There may be other features which are more informative.

6.4 Hypotheses

The subjects used during gait testing for validation of the sensor outputs provided the opportunity to investigate changes in gait of subjects with Parkinson's disease, as compared to gait of normal subjects.

Three hypotheses were investigated. The first hypothesis was that the NLFY and NLF groups were likely to have a high degree of confusion, because the only known difference between the two groups was the age difference. For the three techniques that could handle more than two classes (CART, Naïve Bayes, and Neural Nets), Hypothesis 1 was tested with the three class problem of NLFY, NLF, and PDF. Since the SVM implementation

used could only handle two classes, three two class problems were tested: Hypothesis 1.1, NLFY and NLF; Hypothesis 1.2, NLF and PDF; and, Hypothesis 1.3, NLFY and PDF.

Assuming that Hypothesis 1 would be supported, the second hypothesis was that the PDF group would be highly separable from NLFY and NLF grouped together. Hypothesis 2 is a two class problem, NLFY/NLF and PDF, and was tested using all four techniques.

The third hypothesis was that the feature set was likely to contain enough information to classify individual subjects. Hypothesis 3 had ten classes, corresponding to each of the ten subjects, and was tested using CART, Naïve Bayes and Neural Nets (the SVMs were not evaluated, but could have been tested with each single subject classified against the other nine).

6.5 Training and Testing Groups

Careful selection of training and testing groups is an important step in the testing of classification problems. Three approaches were used for Hypotheses 1 and 2; these approaches are summarized in Table 6.6. A fourth technique, called cross validation, was used for Hypothesis 3.

	Moniker	Number	Testing Sets	Training Sets
1	Leave out entire subjects	N _s	An entire subject	All other subjects
2	Leave out entire gait types	3	 Free gait Distracted gait Paced gait 	 Distracted gait, paced gait Free gait, paced gait Free gait, distracted gait
3	Modified leave one out.	9	One of each gait type for each subject	All remaining samples

TABLE 6.6 Training and testing groups

The first training and testing group involved using each subject as a testing set. The number of training sets and testing sets in this group was equal to the total number of subjects, N_s , in the classification problem; N_s =10 for all tests, except for the SVM two-class tests

194

for Hypothesis 1 (Hypothesis 1.2: $N_s=7$, Hypothesis 1.3: $N_s=5$, Hypothesis 1.4: $N_s=8$). This grouping was selected to investigate the robustness of the classifier, to see whether a classifier of, for example, "normal" gait could classify any subject without requiring a sample of that subject's gait during training.

The second training and testing group involved using a type of gait as the test set. As described in Appendix B.1, the gait trials for each subject were classified as "free gait," where the subject was told to walk as though she was taking a brisk walk through the park, "distracted gait," where the subject was given a task which was designed to provide distraction, and "paced gait," where a metronome was set at 120 clicks per minute, and the subject was asked to walk at a pace of one step per click. Because the purpose of the "distracted gait" was to draw out gait abnormalities in the subjects with PD, and the purpose of the "paced gait" was to improve the gait in the subjects with PD, these groupings were selected to see whether distracted PD gait was less likely to be misclassified and paced PD gait was more likely to be misclassified. The number of training sets and testing sets in this group was equal to three, for each of the three types of gait.

The third training and testing group was a modified version of "leave one out." Leave one out is generally considered one of the more accurate methods for analyzing classification results when only a relatively small amount of training data is available, because it maintains separation between the training and testing sets, while maximizing the (total) number of samples available to the training set [106]. In true "leave one out," the test set is a single sample from all of the data, and the training set is the rest of the data; the classification is trained as many times as the total number of samples. Because three of the techniques used were not readily automatable (only the SVM, a Matlab package, could easily be set to run through all the training and testing sets, providing all results after completion), it was not deemed feasible¹ to do this across all 193 samples for this initial evaluation of the

^{1.} This should not be viewed as a long-term limitation; after a technique is identified as one which works well with this data, modifications to these implementations, or even new implementations, can certainly be developed.

various techniques. A modified version was developed, which involved filling the test set with three samples from each subject, with one sample of each gait type (free, distracted and paced). Each subject had between 4 and 9 trials of each gait type, so nine sets of testing and training data were set up. An algorithm¹ was developed to randomly place the trials into testing and training groups; for subjects which had fewer than nine trials of a particular gait type, the existing trials were randomly selected to fill all nine training and testing groups. The overall error is calculated from the sum of the errors for each of the nine groups. These groups were tested using all four techniques; the same groups were used for all techniques and all hypotheses.

Evaluation of Hypothesis 3 used a technique called cross validation for evaluation. For both CART, and the Weka implementation of Naïve Bayes and Neural Nets, a 10-fold cross evaluation was used (this simplified evaluation of this hypothesis, since these three techniques were not readily automatable). In a 10-fold cross validation, the samples are split into ten subsets of approximately equal size, with each of these ten subsets each used as a test set. The training is completed ten times and the overall error is completed from sum of the errors for the ten groups. This is quite similar to the modified leave one out used in Hypotheses 1 and 2, except, with the implementations used, there is no way to ensure that each of the ten subgroups has an equal representation of subjects, so it may be less accurate than the modified leave one out method.

6.6 Results

This section presents the results of each of the three hypotheses, using the various training and testing sets, and the four different techniques. The results are also detailed in Appendix C.2, including all the individual results for the nine groups used in the "modified leave one out" training and testing groups (this section presents the summed results).

^{1.} Genleaveoutids.m, an m-file available in Appendix F.2, was used to generate the testing and training groups.

An additional classification with Neural Nets, and useful features indicated through use of CART are also presented.

6.6.1 Hypothesis 1

Hypothesis 1 was tested with three classes, using CART, Naïve Bayes, and a Neural Net, and with three sets of two classes, using SVMs. This hypothesis was designed to show that the two groups of females with normal gait were very similar. The hypothesis investigated the separation between the three classes, to see whether the NLF group were more similar to the NLFY group than the PDF group (e.g. the features were not classifying based on age-related parameters), and to support combining the NLF and NLFY in Hypothesis 2.

Three classes

The results for the "modified leave one out" training and testing sets, and tested on CART, Naïve Bayes, and a Neural Net, are shown in Tables 6.7, 6.8, and 6.9. The tables show the summed classification over the nine training/testing groups, the percentage of correct classifications, and the percentage classified as either NLFY or NLF. The results show that all three techniques are remarkably good at classifying the three classes of data. In the CART and Neural Net results, there is more confusion between the NLFY and the NLF classes than with the PDF class. The degree of confusion is not very high, but that is likely due to the fact that the "modified leave one out" training sets contain enough information about each subject to build the classifiers well. The Naïve Bayes results show close to even confusion between all three classes, but still with acceptable classification rates.

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	122	10	3	90.4	97.8
NLF	6	47	1	87.0	98.2
PDF	0	3	78	96.3	3.7
Overall:			erall:	91.5	

TABLE 6.7 CART Hypothesis 1 results, using modified leave one out

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	129	2	4	95.6	97.0
NLF	2	50	2	92.6	96.3
PDF	0	3	78	96.3	3.7
Overall:				95.2	

TABLE 6.8 Naïve Bayes Hypothesis 1 results, using modified leave one out

TABLE 6.9 Neural Net Hypothesis 1 results, using modified leave one out

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	131	2	2	97.0	98.5
NLF	3	50	1	92.6	98.2
PDF	0	0	81	100.0	0.0
Overall:			verall:	97.0	

CART was also used to test this hypothesis using "leave one subject out" training and testing groups. The results, summed over the individual training and testing groups, are shown in Table 6.10, and shown by subject in Table 6.11. From Table 6.10, it is clear that there is significantly more confusion between the NLFY and NLF classes, than between either NLFY and PDF, or between NLF and PDF. While nearly a third of the NLFY samples were classified incorrectly, and nearly 90% of the NLF samples were classified incorrectly, only 5 out of 26 of the misclassified NLFY samples were misclassified as PDF, and only 1 out of 33 of the misclassified NLF samples were misclassified as PDF. These results demonstrate that when the classifier is tested with subject data not included in the training set, subjects with normal gait are more similar to each other, regardless of age.

Close to 30% of the PDF samples were misclassified (in Table 6.10), but a closer inspection of the results by subject, in Table 6.11, reveals that all of these misclassifications were samples from a single PDF subject. One of the limitations of this initial work is the small number of subjects available for evaluation. Subjects PDF-A and PDF-B appear to have similarities between their gait, because when either of these are used as the test sample, the correct classification rate is 100.0%. However, the lower correct classification rate of PDF-C (or, rather, the misclassification of PDF-C as NLF and NLFY) suggests that this subject does not have similar feature values when compared with PDF-A or PDF-B. The effect of a small number of subjects in a group can also be seen in the results for the NLF, group, as only 4 samples of NLF-A and no samples of NLF-B were classified as NLF when each subject was the testing group.

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	66	21	5	71.7	94.6
NLF	32	4	1	10.8	97.3
PDF	4	13	47	73.4	26.7
Overall:			verall:	60.6	

TABLE 6.10 CART Hypothesis 1 results, by class, using leave one subject out

TABLE 6.11 CART Hypothesis 1 results, by subject, using leave one subject out

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY-A	16	0	0	100.0	100.0
NLFY-B	8	9	0	47.1	100.0
NLFY-C	10	9	2	47.6	90.5
NLFY-D	18	0	0	100.0	100.0
NLFY-E	14	3	3	70.0	85.0
NLF-A	15	4	0	21.1	100.0
NLF-B	17	0	1	0.0	94.4
PDF-A	0	0	20	100.0	0.0
PDF-B	0	0	23	100.0	0.0
PDF-C	4	13	4	19.1	81.0
		Ov	verall:	60.6	

CART was also used with the testing groups consisting of a single type of gait (free gait, distracted gait, or paced gait), and the results for each of the three testing groups are

shown in Tables 6.12, 6.13, and 6.14. These testing groups were selected to investigate the effects of distraction and pacing on the gait of the PDF subjects. The distracted gait did have only a single misclassified sample, as compared to three with the free gait test set, suggesting that the distractions may have been successful at eliciting abnormalities in the gait of the PDF subjects. However, all of the PDF paced gait were correctly classified, suggesting that the pacing did not provide restorative feedback. An unusual result is the misclassification of seven samples of NLFY free gait as PDF; this may be a result of the technique used, as this was not seen in similar tests using the SVM, discussed below.

 TABLE 6.12
 CART Hypothesis 1 results, using "free gait" as the test set

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	29	2	7	76.3	81.6
NLF	4	8	1	61.5	92.3
PDF	1	2	18	85.7	14.3
Overall:		76.4			

TABLE 6.13 CART Hypothesis 1 results, using "distracted gait" as the test set

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	27	0	3	90.0	90.0
NLF	1	14	0	93.3	100.0
PDF	1	0	24	96.0	4.0
Overall:		92.9			

TABLE 6.14 CART Hypothesis 1 results, using "paced gait" as the test set

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	19	2	3	79.2	87.5
NLF	0	9	0	100.0	100.0
PDF	0	0	18	100.0	0.0
Overall:		90.2			

Two Classes

The results for each of the two class versions of Hypothesis 1, using "modified leave one out," and tested on SVMs, are shown in Tables 6.15, 6.16, and 6.17. The tables show the summed classification over the nine training/testing groups and the percentage of correct classifications.

The SVMs are reasonably good at classifying the different groups of data. As expected, there is some confusion between the NLFY and the NLF groups, as seen in Table 6.15. However, there is also confusion between the NLFY and the PDF groups, as seen in Table 6.17, and the most misclassification across all three versions is of NLF subjects as PDF, as seen in Table 6.16.

		7 1	,
	NLFY	NLF	% correct
NLFY	130	5	96.3
NLF	4	50	92.6

Overall:

TABLE 6.15 SVM Hypothesis 1.1 results, using modified leave one out

TABLE 6.16 SVM Hypothesis 1.2 results, using modified leave one out

95.2

	NLF	PDF	% correct
NLF	46	8	85.2
PDF	1	80	98.8
	(Overall:	93.3

TABLE 6.17 SVM Hypothesis 1.3 results, using modified leave one out

	NLFY	PDF	% correct
NLFY	132	3	97.8
PDF	3	78	96.3
	(97.2	

To further evaluate these misclassifications, it is useful to look at the results from the "leave one subject out" tests, shown in Tables 6.18, 6.19, and 6.20. These results show that a sample in the PDF group was nearly equally likely to be misclassified as an NLF subject (50% misclassified), as it was to be misclassified as an NLFY subject (43.7%). As seen in the CART results earlier in this section, the classification rates of the NLF group, which had a total of two subjects were very low; 91.9% were misclassified as NLFY and 73% were misclassified as PDF. This second result is particularly interesting for the support of Hypothesis 1. Table 6.11 and Table 6.18 indicate that the gait of NLF-A and of NLF-B each have more similarities with the group of five NLFY subjects than they have similarities with each other. However, Table 6.19 indicates that when the only choice for classification of NLF-A or NLF-B is NLF, trained on only the opposite subject, or PDF, a higher percentage are classified correctly.

TABLE 6.18 SVM Hypothesis 1.1 results, using leave one subject out

	NLFY	NLF	% correct
NLFY	80	12	87.0
NLF	34	3	8.1
		Overall:	64.3

TABLE 6.19	SVM Hypothesis	1.2 results, using	leave one subject out
-------------------	----------------	--------------------	-----------------------

	NLF	PDF	% correct
NLF	10	27	27.0
PDF	32	32	50.0
	(Overall:	41.6

TABLE 6.20 SVM Hypothesis 1.3 results, using leave one subject out

	NLFY	PDF	% correct
NLFY	89	3	96.7
PDF	28	36	56.3
		80.1	

SVMs were also used with testing groups consisting of one type of gait, and the full results are included in Appendix C.2. Of particular interest were Tables 6.21 and 6.22, which show the results of the PDF subjects compared with NLF and with NLFY, with the paced gait subset. Again, none of the PDF samples were misclassified, which suggests that the pacing may not have been effective at restoring the gait of the PDF subjects (there were PDF misclassifications with the distracted gait testing set and the free gait testing set).

Also of interest is Table 6.23. Above, the CART testing of Hypothesis 1 with the free gait testing set showed the unusual result of 7 NLFY samples misclassified as PDF. However, with SVM, only one of the NLFY samples was misclassified as PDF with the same testing set, suggesting that the CART result was due to the technique, rather than the actual data samples.

	NLF	PDF	% correct
NLF	8	1	88.9
PDF	0	18	100.0
		Overall:	96.3

TABLE 6.21 SVM Hypothesis 1.2 results, using "paced gait" as the test set

TABLE 6.22 SVM Hypothesis 1.3 results, using "paced gait" as the test set

	NLFY	PDF	% correct
NLFY	21	3	87.5
PDF	0	18	100.0
		Overall:	92.9

TABLE 6.23 SVM Hypothesis 1.3 results, using "free gait" as the test set

	NLFY PDF		% correct
NLFY	37	1	97.4
PDF	1	20	95.2
-		Overall:	96.6

6.6.2 Hypothesis 2

The results for Hypothesis 2, using "modified leave one out," and tested on CART, SVMs, Naïve Bayes, and a Neural Net, are shown in Tables 6.24, 6.25, 6.26, and 6.27. The tables show the summed classification over the nine training/testing groups, and the percentage of correct classifications. The results for all classifications are very good, with correct classifications better than 95% for both categories using all four techniques. The standout technique is the Neural Net, which had 100% correct classifications for all PDF samples, and misclassified a mere 2 (1.1%) of NLF and NLFY samples.

	NLF/NLFY	PDF	% correct
NLF/NLFY	181	8	95.8
PDF	1	80	98.8
		Overall:	96.7

TABLE 6.24 CART Hypothesis 2 results, using modified leave one out

ut
U

	NLF/NLFY	PDF	% correct		
NLF/NLFY	181	8	95.8		
PDF	2	79	97.5		
Overall: 96.3					

TABLE 6.26 Naïve Bayes Hypothesis 2 results, using modified leave one out

	NLF/NLFY	PDF	% correct		
NLF/NLFY	181	8	95.8		
PDF	4	77	95.1		
		Overall:	95.6		

TABLE 6.27 Neural Net Hypothesis 2 results, using modified leave one out

	NLF/NLFY	PDF	% correct
NLF/NLFY	187	2	98.9
PDF	0	81	100.0
		Overall:	99.3

The results for Hypothesis 2, using "leave one subject out," and tested on CART, SVMs, and a Neural Net are shown in Tables 6.28, 6.29, and 6.30. These results are cautionary: though the combined NLF/NLFY group still has reasonable rates of classification, particularly with CART, the PDF group has significantly lower rates of correct classification with both techniques. As seen previously, this is likely due to the small number of subjects (only three PDF), so an important next step will be to collect data from more subjects.

 NLF/NLFY
 PDF
 % correct

 NLF/NLFY
 123
 6
 95.3

 PDF
 21
 43
 67.2

 Overall:
 86.0

 TABLE 6.28
 CART Hypothesis 2 results, using leave one subject out

TABLE 6.29	SVM Hypothesis 2	results, using	leave one subject out
	21	, U	

	NLF/NLFY	PDF	% correct		
NLF/NLFY	114	15	88.4		
PDF	33	31	48.4		
		Overall:	75.1		

TABLE 6.30 Neural Net Hypothesis 2 results, using leave one subject out

	NLF/NLFY		% correct
NLF/NLFY	122	7	94.6
PDF	18	46	71.9
		Overall:	87.0

Hypothesis 2 was also tested using the testing groups consisting of one type of gait, on CART and SVMs. The results, included in Appendix C.2, are similar to the results seen when these testing groups were used with Hypothesis 1.

6.6.3 Hypothesis 3

Hypothesis 3 was tested using ten-fold cross-validation, on CART, Naïve Bayes, and a Neural Net. The CART and Naïve Bayes results are in Appendix C.2, and the Neural Net results are shown in Table 6.31, including the summed classification over the ten subsets of training/testing groups, the percentage of correct classifications, and the percentage classified as either NLFY or NLF.

	NLFY A	NLFY B	NLFY C	NLFY D	NLFY E	NLF A	NLF B	PDF A	PDF B	PDF C	% correct	% NLFY or NLF
NLFY-A	15	0	0	1	0	0	0	0	0	0	93.8	100.0
NLFY-B	0	17	0	0	0	0	0	0	0	0	100.0	100.0
NLFY-C	0	0	20	0	1	0	0	0	0	0	95.2	100.0
NLFY-D	0	0	0	18	0	0	0	0	0	0	100.0	100.0
NLFY-E	0	0	0	0	18	1	0	0	0	1	90.0	95.0
NLF-A	0	0	0	0	0	19	0	0	0	0	100.0	100.0
NLF-B	0	0	0	0	0	0	18	0	0	0	100.0	100.0
PDF-A	0	0	0	0	0	0	0	20	0	0	100.0	0.0
PDF-B	0	0	0	0	0	0	0	0	23	0	100.0	0.0
PDF-C	0	0	0	0	0	0	0	1	0	20	95.2	0.0
Overall:							97.4					

TABLE 6.31 Neural Network Hypothesis 3 results, by subject, using cross-validation

Though all three techniques performed well, the Neural Network again had the best results. Only five samples out of 193 were misclassified with the Neural Network, and only one of those was classified as a subject of a different class. This high classification rate suggests that the GaitShoe may be highly capable of capturing the nuances of individual subjects' gait (see Section 6.6.5 for Neural Net results using fewer features).

6.6.4 CART Feature Information

As discussed earlier in this chapter, one of the most powerful aspects of CART is the transparency of the trees as to which features are the most informative. All of the decision trees resulting from the tests done on the three hypotheses were investigated, and the features which were most frequently used to split nodes are listed in Table 6.32, in decreasing order of frequency (the feature number corresponds to the number in Table 6.4); these reduced number of features will be used to train a Neural Net in the following section.

Feature Number	Feature Derivation	Feature Description
1	Standard deviation of FSRsum/bodyweight, L and R combined	Walking energy variation
21	Mean minimum pitch, L and R combined	Shuffle index
3	Mean StepF/bodyweight, L and R combined	Step energy amplitude
24	Mean percent stance time, L and R combined	Shuffle duration
10	Gyro-z variation, L and R combined	Pitch variation
20	Maximum pitch variation, L and R combined	Shuffle variation
8, 6	Gyro-y, -x variation, L and R combined	Roll, yaw variation
12, 14, 16	Accel-x, -y, -z variation, L and R combined	Linear motion variation

TABLE 6.32 Informative features, as identified by CART

Of particular interest is the fact that all of these features used the metric "L and R combined," which simply combined the data from the left and right feet into a single vector before the analysis was applied. This result suggests that only a single shoe may be necessary to capture the differences between the gait of subjects with PD and subjects with normal gait. Of course, certain pathologies, such as cerebral vascular accidents (strokes), known to result in asymmetrical gait would likely still benefit from data from both shoes, so that the metrics with the ratio between the left foot data and the right foot data could be calculated. However, for non-asymmetric pathologies, using only half of the GaitShoe system would greatly simplify the data collection process.



Figure 6.3 Subject data of two most informative features in CART analyses

The first two features listed in Table 6.32 occurred far more frequently than the other features. These two features, the mean minimum pitch, and the variation in the sum of the four FSRs, are plotted for all of the data samples in Figure 6.3. This graph shows an impressive separation of the PDF class from the NLFY and NLF classes. In addition, it shows considerable confusion between the NLFY and NLF classes, suggesting that these two features are not simply separating the PDF samples on the basis of an age-related parameter.

Feature 1 is plotted along the x-axis, and was derived by taking the clipped standard deviation of the sum of all four FSRs, for both the left and right feet data. This feature is a measure of the variance of the force measured underneath the foot (and normalized by body weight). The magnitude of the total force seen between the foot and the floor is typically 120% of body weight in normal walking gait, but reaches as much as 220% of body weight during running [105]. Therefore, a larger variance (when normalized by bodyweight) corresponds to a more "energetic" gait, so this feature can be referred to as a measure of "walking energy variation." Figure 6.3 demonstrates that the subjects with PD have a much less energetic gait.

Feature 21 is plotted along the y-axis, and was derived by calculating the mean of the minimum pitch seen during gait. As shown in Figure 4.16 (p. 120), the minimum pitch occurs just before heel strike, and corresponds to both the amount of leg swing, the gait velocity, and the amount of dorsiflexion of the foot. As both leg swing and dorsiflexion are likely to be greatly reduced when the feet are shuffled, this feature can be described as a "shuffle index." Figure 6.3 shows that subjects with PD are more likely to have a lower magnitude minimum foot pitch, or a higher "shuffle index".



Figure 6.4 Subject data of two most informative features in CART analyses, individual PD subjects

Figure 6.4 is the same graph as Figure 6.3, but with the individual PD subjects identified, and labeled by clinical treatment. Interestingly, the two PD subjects that had surgical interventions are closer to the NLFY and NLF groups, with the PD subject treated only with medication is further away. With only three subjects, these results cannot be extrapolated

to the PD population at large, however, it will be of great interest to see whether this result holds as more PD subjects, with various interventions, are tested and included in the classification.

6.6.5 Additional Neural Net Studies

The features identified by the CART, as described in Section 6.6.4, were used to train Neural Nets using reduced numbers of features, with 10-fold cross-validation.

For Hypothesis 1, when only the top two features, the mean minimum pitch ("shuffle index") and the insole force variation ("walking energy variation") were used, the overall classification was 86.5%, as shown in Table 6.33. However, this included a large number of misclassification between NLFY and NLF, which is not surprising given Figure 6.3, and only three misclassifications between NLFY/NLF and PDF.

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	85	6	1	92.4	98.9
NLF	17	19	1	51.4	97.3
PDF	1	0	63	98.4	1.6
Overall:			verall:	86.5	

TABLE 6.33 Neural Net Hypothesis 1 results, using 10-fold cross-validation, and the top two features

For Hypothesis 2, the results, shown in Table 6.34, were excellent when the top two features were used. The overall classification rate was 99%, with just a single sample from each class misclassified.

TABLE 6.34 Neural Net Hypothesis 2 results, using 10-fold cross-validation,
and the top two features

	NLF/NLFY	PDF	% correct
NLF/NLFY	128	1	99.2
PDF	1	63	98.4
		Overall:	99.0

The results for Hypothesis 3 are shown in Table 6.35, using the top six features, the mean minimum pitch ("shuffle index"), the insole force variation ("walking energy variation"), mean force per step ("step energy amplitude"), mean percent stance time ("shuffle duration"), z-gyroscope variation ("pitch variation"), and maximum pitch variation ("shuffle variation").

	NLFY A	NLFY B	NLFY C	NLFY D	NLFY E	NLF A	NLF B	PDF A	PDF B	PDF C	% correct	% NLFY or NLF
NLFY-A	15	1	0	0	0	0	0	0	0	0	93.8	100.0
NLFY-B	1	14	0	2	0	0	0	0	0	0	82.4	100.0
NLFY-C	1	0	19	0	0	0	0	0	1	0	90.5	95.2
NLFY-D	0	0	0	18	0	0	0	0	0	0	100.0	100.0
NLFY-E	0	0	1	0	17	2	0	0	0	0	85.0	100.0
NLF-A	0	0	0	0	3	16	0	0	0	0	84.2	100.0
NLF-B	0	0	0	1	0	0	17	0	0	0	94.4	100.0
PDF-A	0	0	1	0	0	0	0	18	1	0	90.0	5.0
PDF-B	0	0	0	0	0	0	0	1	22	0	95.7	0.0
PDF-C	0	0	0	0	0	0	0	0	0	21	100.0	0.0
									Ove	rall:	91.7	

TABLE 6.35 Neural Net Hypothesis 3 results, by subject, using 10-fold cross-validation,
and the top six features

The overall classification rate for Hypothesis 3 is 91.7%, with only two misclassification between NLFY/NLF and PDF. This is an interesting result, because there are fewer features than subjects (6 features, 10 subjects), as compared to the results in Section 6.6.3, where there were more features than subjects; one could argue that with sufficient number of features, any subject could be identified. However, this result demonstrates that, for this group of subjects, these six features were enough to classify these ten subjects with a classification rate better than 90%.

Finally, a Neural Net was trained on three different sets of "contrived" groups. With the small number of subjects, and with the strong results for Hypothesis 3, the question arises whether the strong classification rate between NLF/NLFY and PDF is simply the result of the Neural Net training on subjects, rather than on group similarity. Although Figure 6.3 indicates that there is actual separation between the NLF/NLFY and PDF groups, three "contrived" groupings were set up, each with one subgroup of seven subjects, and a second subgroup of three subjects. The groupings, shown in Table 6.36, were selected such that each subgroup of three subjects had one of the three PDF subjects, and such that the two NLF subjects were each in a subgroup of three, as well. In addition, because just three contrived groupings were set up for this quick investigation (rather than an exhaustive analysis of all combination of seven and three subgroups), the groups of three subjects were set up by inspection of the data tables from all the previous classification, such that subjects who had been misclassified as each other were included in the same subgroup.

TABLE 6.36	Contrived Groupings
------------	---------------------

	7 Subject Subgroup	3 Subject Subgroup
Group A	NLFY-A, NLFY-B, NLFY-C, NLFY-D, NLF-A, PDF-B	NLFY-E, NLF-B, PDF-C
Group B	NLFY-A, NLFY-B, NLFY-D, NLFY-E, NLF-B, PDF-A, PDF-C	NLFY-C, NLF-A, PDF-B
Group C	NLFY-B, NLFY-C, NLFY-E, NLF-A, NLF-B, PDF-B, PDF-C	NLFY-A, NLFY-D, PDF-A

The results are shown in Tables 6.37, 6.38, and 6.39, and have classification rates of 79.3%, 68.4%, 80.8%. Compared with the 99% classification result shown in Table 6.34, for groups NLFY/NLF and PDF, this suggests that the Neural Net is indeed able to train on group characteristics rather than on individual characteristics...

and the top two features, with "Contrived Groups A"						
Group A-1 Group A-2 % co						
Group A-1 (7 subjects)	118	16	88.1			
Group A-2 (3 subjects)	24	35	59.3			

Overall:

79.3

TABLE 6.37 Neural Net results, using 10-fold cross-validation,
and the top two features, with "Contrived Groups A"

	Group B-1	Group B-2	% correct
Group B-1 (7 subjects)	108	22	83.1
Group B-2 (3 subjects)	39	27	38.1
		Overall:	68.4

 TABLE 6.38
 Neural Net results, using 10-fold cross-validation, and the top two features, with "Contrived Groups B"

 TABLE 6.39
 Neural Net results, using 10-fold cross-validation, and the top two features, with "Contrived Groups C"

	Group C-1	Group C-2	% correct
Group C-1 (7 subjects)	124	15	89.2
Group C-2 (3 subjects)	22	32	59.3
		Overall:	80.8

6.7 Discussion

This chapter investigated the ability of four classic pattern recognition techniques to distinguish gait using features derived from the vast quantity of information measured by the GaitShoe.

The primary goal was to classify the gait of subjects with Parkinson's disease from the gait of subjects with normal gait, to see whether the GaitShoe sensor measurements encapsulated information about changes in gait between the two groups. This type of information could be used to evaluate treatment strategies for patients with Parkinson's disease.

The subjects were selected from the fifteen volunteers who were subjects for the validation of the GaitShoe described in Chapter 5. The goal of the subject selection was to create classes that differed only on the basis of the presence of Parkinson's disease. Only females were used, both to eliminate any question of gender differences, and because the females had a better match of ages. Though the ages ranged from early twenties to mid-sixties, the group of ten female subjects had a subset of two subjects with normal gait who were reasonably age-matched to the three subjects with Parkinson's disease. In addition, the females had a small variation in height, suggesting that leg length was unlikely to affect the outcome of the classification.

The first hypothesis examined three classes: NLFY, females with normal gait who were under 30 years old; NLF, females with normal gait who were 48 and 54 years old; and, PDF, females with Parkinson's disease who were over 54 years old. The goal of this hypothesis was to see if the features selected showed significantly more discriminating among PDF and NLF classes than among NLFY and NLF classes. This was well demonstrated by the CART and SVM models built with the training and testing sets where individual subjects were used as the testing set. Samples from the NLFY and NLF classes were far more likely to be misclassified as NLF or NLFY, respectively, than as PDF.

The second hypothesis used the positive result from the first hypothesis to combine all the NLFY and NLF subjects into one group, NLFY/NLF. The four techniques were then used to classify samples as either NLFY/NLF or PDF, and all techniques performed very well, with correct classifications better than 95%. The strongest result was from the Neural Net, which correctly classified all of the PDF samples, and only misclassified 2 (1.1%) of the combined NLF and NLFY samples. In addition, when trained using only the top two features, the mean minimum pitch ("shuffle index") and the insole force variation ("walking energy variation"), the Neural Net classification rate was 99%, with only two samples misclassified. These results suggest that the GaitShoe is quite capable of capturing changes in the gait due to Parkinson's disease.

The third hypothesis investigated whether subjects could be classified individually, using CART, Naïve Bayes and Neural Nets, all of which could handle multiple classes, to classify the ten different subjects. Again the Neural Net results were outstanding, with only five out of 193 samples misclassified. The Neural Net was trained on only six features, the mean minimum pitch ("shuffle index"), the insole force variation ("walking energy variation"), mean force per step ("step energy amplitude"), mean percent stance time ("shuffle duration"), z-gyroscope variation ("pitch variation"), and maximum pitch variation ("shuff-

fle variation"), and was able to classify the ten subjects with a classification rate of 91.7%. These results suggest that the features derived from the GaitShoe measurements are able to capture the individualities of each subject's gait.

The Naïve Bayes classification results tended to be the weakest. However, Naïve Bayes requires the assumption that the features were independent, which was certainly not true for the features used here, and likely had an impact on the classification. The SVM classification results were reasonable, but as use of SVMs are limited to two classes, the applicability to larger groups of gait subjects is limited, though multiple classes can be evaluated by developing SVMs with one of the classes evaluated against the others (further work with SVMs, including adjusting the settings may result in better SVM classification rates).

Though the CART classification results were reasonable, the real benefit of the CART analysis was in the identification of two features which provide excellent separation between the subjects with Parkinson's disease and the subjects with normal gait. As discussed above, the top CART-identified features were used to run additional Neural Net analyses to demonstrate that these subjects can be classified with a small number of features.

The Neural Net classification results were consistently the strongest, and demonstrate great promise for future use in using the GaitShoe system to classify both individual gaits as well as the gaits of groups of subjects.

The overall results from the pattern recognition analysis suggest that use of methods such as a Neural Net in combination with the features of the GaitShoe may have great benefit in analyzing gait.

Chapter 7

REAL-TIME THERAPEUTIC FEEDBACK

One application for the GaitShoe would use real-time analysis of the sensor outputs to provide feedback about the current gait, to allow the user to make adjustments. This could be useful for many areas, such as physical therapy, sports medicine, or athletic training, and the feedback could take a variety of forms: musical, tonal, visual, tactile. It could also be used to provide electro-stimulation at certain parts of the gait cycle [13], or to control an artificial leg. This chapter describes an initial investigation into using musical feedback controlled by a real-time analysis of the GaitShoe sensor data [107].

7.1 Overview

Music Therapy is an established field; however, it generally consists of patients listening to a specific type of music, or patients playing musical instruments [108]. Even so, the idea of using physiological measurements to control electronic music has been explored for some time, notably the work by David Rosenboom and Richard Teitelbaum in the late 1960s, involving the use of brainwaves, heartrates, EMGs, and skin conductivity to produce real-time musical biofeedback [109]. Other recent work involves interactive music and visuals set up as a meditation chamber, which responded to measurements of galvanic skin response, respiratory rate, and heart rate [110], or interactive music with causal mappings between free gesture and sound to encourage withdrawn mentally disabled and autistic children to become engaged [111].

The use of on-body sensors for sports applications is becoming more common, but audio feedback is generally limited to a simple beep, or a critique of a golf or batting swing from a talking virtual coach [112]. However, applications in dance, such as the Expressive Footware, as described in Section 2.1.5, have made use of motion-to-music mappings to allow the dancer to control the music heard during the performance [38].

Lack of applications in both the physical therapy and sports fields is likely due at least in part to the absence of readily available methods for gathering and analyzing relevant physiological data in real-time. Thus, the heavily instrumented GaitShoe could open the door to many new applications of musical feedback.

7.2 Rhythmic Auditory Stimulator

To explore the use of the GaitShoe for real-time feedback, the "Rhythmic Auditory Stimulator" (RAS) program [113] was developed in conjunction with Erik Asmussen, an undergraduate researcher collaborating with our group. Using insight gained during the gait analysis described in Chapter 5, the RAS provided three different types of sensing and feedback. Mr. Asmussen is pictured next to the RAS running in Figure 7.1.



Figure 7.1 Erik Asmussen and the Rhythmic Auditory Stimulator
7.2.1 The RAS system

Mr. Asmussen wrote the interactive RAS environment using the Max/MSP graphical programming language [114] on an Apple Computers powerbook. The GaitShoe basestation was connected to the powerbook via a Keyspan 19HS USB Serial Adapter [115]. Software requirements within Max/MSP limited the serial data to rate to 56.6 kbps rather than the usual 115.2 kbps, which reduced the data transfer rate of each shoe to approximately 30 Hz. The musical feedback generated with Max/MSP was output from the powerbook using Musical Instrument Digital Interface (MIDI). MIDI is a standard that provides a method of easy transmission of information corresponding to electronic music. A Midiman USB MidiSport [116] was connected to the USB port of the powerbook, and transmitted the MIDI output to an E-MU Proteus 2000 synthesizer [117], which output the sound to speakers. This allowed the system to respond in real-time, approximately 100 ms after the gait event of interest.

Although the RAS program can be configured to use any sensor outputs produced by the GaitShoe, the tests discussed here only involved using the FSRsum parameter derived from the sum of the four FSRs, "FSRsum," which is described in detail in Section 4.9, and the paired sums of the two medial FSRs and of the two lateral FSRs. Screenshots for three of the menus in the RAS system are shown in Figure 7.2.



Figure 7.2 Screenshots of the RAS system, showing the main menu (upper left), therapy configuration menu (upper right), and feedback control menu (lower).

7.2.2 RAS Feedback

Three types of sensing were developed for the RAS: pace sensing, force distribution sensing, and peak force sensing.

Pace Sensing

This application was designed to aid subjects with Parkinson's disease. As discussed in Section 2.3.1 (p. 44), previous work has shown that rhythmic cues at a pace slightly faster than the PD subject's normal pace helps to lengthen stride and increase mean gait velocity. Previous work, however, has just involved a passive metronome [55] [56], so the goal of this application was to actively sense the pace of the subject, and provide feedback only as necessary.

The FSRsum was used to determine heel strike time, and the current pace was determined by subtracting the previous heel strike time from the current heel strike time. The RAS contained a field to set the ideal pace, and the current pace approached the ideal pace to determine the feedback.

Two different modes of feedback were available; the first provided the user with very subtle rhythmic cues, which faded out when current pace was equal to the ideal pace. The second mode played a charming tune while the current pace was at the ideal pace, with a quiet drumbeat at the ideal pace in the background, and when the current pace diverged from the ideal, the feedback converted to a loud drum sound only. When the user returned to the ideal pace, the charming tune returned as a reward. The first mode explored simple cues only when needed, while the second mode always produced background music (inspired by the ubiquity of portable music players in today's society), with changes in correspondence with the feedback.

Force Distribution Sensing

This application was an initial investigation into changes in force distribution. A gait parameter of interest for runners is the degree of pronation and supination, as excessive pronation or supination can lead to injury if left uncorrected. This condition is typically treated by the placement of orthotics in the shoe, but feedback could allow the runner to try to correct the condition (for instance, if it occurs only as the runner becomes fatigued). Pronation is a complex motion involving changes in the motion of the ankle. However, excessive pronation or supination each have manifestations in the distribution of force underneath the foot; inspection of the underside of an older pair of running shoes usually reveals uneven wear patterns on the medial or lateral edge of the shoes, corresponding to over-pronation and supination, respectively.

The paired sums of the medial FSRs and the lateral FSRs during static standing were stored. During gait, the sum of the medial FSRs and the sum of the lateral FSRs were compared to the stored results, to see if an excessive amount of force was applied either medially or laterally. This mode of feedback played the user a charming tune in a major key when force distribution was determined to be within normal bounds. When excessive force was detected either medially or laterally, the tune transitioned from a major to a minor key, and if left uncorrected, became progressively dissonant.

Peak Force Sensing

The final application looked at the peak force detected by the FSRsum during stance. This application was designed to aid patients, for example, recovering from a broken leg, or after hip or knee replacement. During recovery, patients are told to apply a certain percentage of body-weight each week. The percentage gradually increases until the patient can walk normally again; generally, the patient is instructed to step on a scale to see what that week's weight limit feels like. Use of the GaitShoe system could provide the patient with real-time feedback as to the force applied on the recovering leg, and could remove an element of guess-work from the patient's recovery.

The RAS contained a field to set the acceptable threshold for force applied across the four FSRs. This mode of feedback was similar to the previous mode: when the force was under the threshold, the user heard the charming tune in a major key. If the threshold was

exceeded by a small amount, the tune transitioned to a minor key, and if exceeded by a large amount, the music became very dissonant.

7.3 Conclusions

This initial work demonstrated that the GaitShoe could be analyzed in real-time, and configured to provide real-time feedback to the user about a variety of changes in gait. There are many applications in both physical therapy and sports medicine which might benefit from this time of feedback. Videos of the three sensing modes described here are archived on-line, at http://www.media.mit.edu/resenv/GaitShoe/index.html. The next step for this work is to evaluate the feedback on patients; tests using the pace sensing for subjects with Parkinson's disease are planned.

Chapter 8

CONCLUSION

The development of the GaitShoe has resulted in a wireless wearable system with an unprecedented number of sensors designed to capture information that can characterize gait of both feet. The system costs under \$500 per foot in prototype quantities and the hardware for a single shoe weighs under 300 g. The hardware is readily fixed to a variety of typical walking shoes, and data can be continuously collected over a few hours.

The gait parameter analysis indicated that the GaitShoe can be further developed into a true wearable podiatric laboratory, which could be of great use in evaluating gait over longer periods of time than are available in motion laboratories, as well as allowing the evaluation to be carried out in a neutral environment, such as the subject's home. It would also allow the evaluation of subjects who are without access to a motion laboratory.

Relevant GaitShoe sensors were calibrated and analyzed to determine parameters of gait, which were validated by comparison with data collected simultaneously by the Massachusetts General Hospital (MGH) Biomotion Laboratory. The GaitShoe's determination of heel strike time regularly anticipated the time determined from analysis of the force plate output; the results suggest that the GaitShoe is capable of detecting heel strike before the force plate. The toe off times determined by the GaitShoe and from the force plate data were very similar. Placement of force sensitive resistors (FSRs) underneath the toe may result in even the ability to detect toe off timing more accurately than the force plate. This should be investigated further, by further calibration of the force sensors in order to deter-

mine appropriate thresholds corresponding to the initiation of loading of the FSRs; this needs to be done such that the time-scale of the GaitShoe can be precisely aligned with the time-scale of the calibration system. In addition, the four coarsely spaced FSRs provide a reasonable approximation to the force distribution measured by the force plate, and are capable of providing information about shifting weight patterns from stride to stride (information which is not available from a force plate).

A simplified analysis of the motion of the foot, using the x-accelerometer and the z- gyroscope, resulted in reasonable estimations of the pitch, velocity and stride length. However, a more complete analysis including the outputs of the x- and y-gyroscopes, and the outputs of the y- and x-accelerometers is expected to improve the results. In addition, the implementation of a Kalman filter is likely to further improve the outcome. Though future work should certainly make use of the full suite of gyroscopes and accelerometers, the results of the simplified analysis indicate that the GaitShoe is capable of reasonable estimations of orientation and displacement.

The bend sensor output generally has a shape corresponding to the expected plantar flexion and dorsiflexion curve. The output is likely to be more uniform with an improved method of positioning and/or retaining the sensor. Further evaluation should utilize an alternative reference system that allows the bend sensor to be easily held next to the ankle, and that has fewer errors. Alternatively, the use of a more repeatable bend sensor could be considered, such as one made from fiber optics.

The electric field sensor provides a method of determining the height of the foot above the floor. In particular, multiple discrete electric field sensors can be implemented, such as at the heel and the toe or metatarsals, which would provide additional information about the orientation of the foot. The recent development of an ultrasound sensor by Steven Dan Lovell provides a method of measuring the distance between the two feet and the relative orientation of the feet, as well as a (future) method to measure the height of the foot above the ground.

In addition, the overall insole design should be reconsidered for future work. The PVDF sensors were not used in the analysis, because of the variable output. The FSRs were very valuable, but a more precise pressure sensor or alternative implementations of the FSR might provide even better results. The electric field sensor connector must be replaced with one that can better accommodate the coaxial cable used to provide the direct signal shield around the connection to the electrode. The ultrasound sensor transmitter and receiver attachments need to be redesigned so that they are much more stable, and are not affected by dynamics of gait, such as the impact of heel strike.

The pattern recognition results suggested that a great future application for the GaitShoe may be the use of the GaitShoe's ability to derive meaningful features from the extensive sensor suite, and to use those features to recognize individual subjects as well as groups of subjects with a similar gait. In particular, Neural Nets appeared to be a very promising method for discriminating between both individual subjects and between groups of subject with normal gait, and groups of subjects with Parkinson's disease. The results should be confirmed with a broader study including larger numbers of subjects, as well as subjects of both genders. If the Parkinson's disease results remain strong once subjects are added, this technique may be able to be used to assess the effectiveness of the patient's medication regimen, or even to assess the impact of various treatments.

In addition, the use of Classification and Regression Trees (CART) provided insight into the most useful features for discriminating between the two groups. The standard deviation of the FSRsum, normalized by body weight ("walking energy variation") and the mean minimum pitch ("shuffle index") provided a excellent separation between the subjects with Parkinson's disease and the subjects with normal gait,. These results show that the subjects with Parkinson's disease are closer to a shuffle-gait than the normal subjects, and that their steps have less force than the steps of subjects with normal gait.

The GaitShoe system was incorporated with a program written by Erik Asmussen in the Max/MSP graphical programming language, and output MIDI (the standard for transmit-

ting electronic music) to a synthesizer to provide rhythmic auditory stimulation (RAS). The RAS implemented was real-time musical feedback corresponding to the detection of three gait conditions: stride pace, weight distribution, and total weight. The system worked well and provided interesting and engaging feedback; in particular, while filming videos of the GaitShoe and the RAS, people walking by could be observed moving their head and upper body in response to the catchy percussion rhythm used in the stride pace feedback. Future work should include testing the feedback on patients with relevant gait pathologies to evaluate the effectiveness of the feedback, and whether the feedback is interesting and engaging to those who would use it for physical therapy. This is an exciting area of future application for the GaitShoe, as it provides a new framework in which interactive real-time physical therapy can be investigated. This has many applications, from rehabilitating gait in patients such as those recovering from hip surgery who need to bear only a certain amount of weight on one leg, or those recovering from a stroke who need to relearn how to walk symmetrically, to investigating the gait of subjects with diabetic neuropathy to evaluate risk for ulceration, to countless applications in the sports medicine and sports training fields, such as detections of over-pronation or supination to provide runners with feedback to allow them to make changes in their running gait or to make decisions about when to stop a run if at risk for injury, to analysis.

Table 8.1 details a comparison between the GaitShoe and the MGH Biomotion Laboratory systems. While the GaitShoe is not yet a tool to use for evaluation of gait prior to surgical intervention, with the changes suggested in this chapter, it may be able to replace traditional motion laboratories for such clinical work as evaluation during design of orthotics and prosthetics. In particular, it will be an excellent tool for areas without access to motion laboratories. In addition, the GaitShoe has already been implemented in a simple real-time analysis and feedback system; a valuable possible application for the GaitShoe would be to use real-time analysis to provide electro-stimulation (for persons with spinal cord injury) or control of an artificial leg, at specific times during the gait cycle.

GaitShoe	MGH Biomotion Lab
More than 30 degrees of freedom across lower legs and feet.	24 degrees of freedom on lower legs and feet.
No data collection on rest of body.	42 degrees of freedom on rest of body.
Collects data continuously.	Collects 7 seconds of data.
Collects data "anywhere".	Collects data in BML lab only.
Wireless.	Tethered.
Can be analyzed in real-time.	Data processed after collection is completed.
Can be used to control real-time feedback.	Cannot be used for real-time control.
Total mass < 0.6 kg.	Total mass < 2 kg.
Cost < \$1K in prototype quantities.	Cost: Proprietary equipment \$1M, plus space and personnel costs for use of system (furnishing a lab with commercial equip- ment would be >\$250K)
Heel strike time determined with a $\sigma < 0.023$ sec and toe off time determined with a $\sigma < 0.017$ sec, as compared with BML.	Heel strike and toe off times determined within 0.007 sec.
Stride length determined with a σ < 16 cm, as compared with BML.	Distances, such as stride length, determined within 1 mm
Pitch determined with a $\sigma < 7^{\circ}$, as compared with BML.	Orientations, such as pitch, determined within 1°
Capable of classifying gait of groups and gait of individuals, on small subject sample.	

TABLE 8.1 Comparison between the GaitShoe and the MGH Biomotion Lab

The GaitShoe is a research tool that enables the analysis of gait in untraditional ways, such as over long periods of time and in the home environment or through use of pattern recognition, and provides a method for real-time feedback for use in such applications as sports medicine, electro-stimulation, or physical therapy.

Appendix A

MEDICAL INFORMATION

A.1 Terminology

Calcaneous: The largest foot bone, located in the heel of the foot.

Dorsiflexion: Flexion of the foot; forefoot motion upward, toward the ankle.

Gait: Manner of walking.

Heel strike: Time at which the heel first makes contact with the floor; indicates end of *swing* and start of *stance*; see Figure A.1.

Lateral: Away from the centerline of the body; for the right foot, the right edge is lateral.

Medial: Toward the centerline of the body; for the right foot, the left edge is medial.

Metatarsal heads: The distal ends of the metatarsal bones, at the point of articulation with the proximal phalanx of the corresponding toe (located at the "ball of the foot").

Plantar flexion: Extension of the foot; forefoot motion downward, away from the ankle.

Pronation: Complex motion of the ankle, resulting in the sole of the foot shifting medially; occurs during the start of stance to absorb shock from heel strike and to assist in balance.

Stance: Period in which the foot is in contact with the floor; this generally takes up about 60% of the stride cycle; see Figure A.1.

Step: Interval between two successive heel strikes for opposite feet; see Figure A.1.

Stride: Interval between two successive heel strikes of the same foot; see Figure A.1.

Supination: Complex motion of the ankle, resulting in the sole of the foot shifting laterally.

Swing: Period during which the foot is not in contact with the floor; this generally takes up about 40% of the stride cycle; see Figure A.1.

Toe off: Time at which the great toe is first no longer in contact with the floor; indicates end of *stance* and start of *swing*; see Figure A.1.



Figure A.1 The Gait Cycle

A.2 Parkinson's Disease

Parkinson's disease (PD) is a chronic and progressive movement disorder resulting from the loss of dopamine-producing neurons in the *substantia nigra* area of the brain. Dopamine is a chemical messenger which relays neurological signals for the coordination and controlled initiation of movement. With the loss of dopamine-producing neurons, dopamine levels fall, resulting in the symptoms of PD. The diagnosis of PD is made following a physical exam for common PD symptoms, and ruling out other conditions with similar symptoms. The exact cause of PD is not yet known, though there appears to be some genetic contribution. Environmental toxins may play a role as well.

The most common symptoms of PD are tremors of the limbs, jaw, and face, rigidity of the limbs and trunk, bradykinesia (slowed movement), and postural instability, resulting in impaired balance and coordination. Manifestation of these symptoms in gait result in short steps and a shuffling gait, called festination, in difficulty initiating gait, called freezing, in difficulty to turn, and in loss of balance.

Parkinson's disease is generally treated first with medications designed either to work by increasing dopamine levels (e.g. by providing precursors, or by activating the release of stored dopamine), or by activating the dopamine receptor directly; medications to slow progression of the disease are in development. Surgical interventions are available, generally to patients who are not satisfied with the results of medication-controlled treatment. Pallidotomy (to alleviate rigidity and bradykinesia) and thalamotomy (to alleviate tremors) are procedures in which small regions of the brain are permanently destroyed to alleviate symptoms. Deep brain stimulation implants, considered by some to be a safer and more effective surgical treatment, involve an electrode implanted in the brain to provide an electrical impulse to a targeted region to alleviate symptoms (the electrodes are connected via wires to an impulse generator placed under the subject's clavicle) [118] [119].

Appendix B

SUBJECT TESTING

This appendix describes the subject testing used to acquire data used in Chapter 5 for validation and in Chapter 6 for pattern recognition. The study design, including subject recruitment, consent forms, and the testing protocol, is discussed, and information about the subjects is included as well. Donna Moxley Scarborough, MS/PT, was the principal tester at the Massachusetts General Hospital (MGH) Biomotion Laboratory (BML).

B.1 Study Design

The subject testing involved placing the GaitShoe instrumentation on the subjects' own walking shoes, with the insole inside the shoe and the shoe attachment mounted to the posterior aspect of each shoe. The small antenna on the circuit board transmitted the sensor information to the receiving transmitter. All signals were digitized and saved on a laptop computer set within 30 feet of the shoes during data collection. The subject underwent simultaneous gait evaluation using the MGH Biomotion Laboratory's Selspot II data acquisition system. Each subject was asked to perform a series of locomotor tasks, while both gait evaluation systems simultaneously collected data. The gait parameters collected from the two systems was analyzed and compared to validate the analysis of gait parameters from the data acquired by the GaitShoe, as discussed in Chapter 5; in addition, the data were used for the pattern recognition study, as discussed in Chapter 6. The informa-

tion gathered did not offer any direct benefits to the subjects tested. After the subjects complete the protocol, they concluded their participation in the study.

B.1.1 Subject Recruitment

The subjects with healthy gait were recruited via e-mail and word of mouth. When subjects replied with interest in the study, they were provided with more details about the study, and an appointment was set up at the Biomotion Lab.

The subjects with Parkinson's disease were recruited by collaborators Drs. Stephen Parker and Leslie Shinobu of the MGH Department of Neurology, who performed initial screening of PD patients within their practice and described the research project to prospective subjects. If a subject verbally agreed to being contacted by phone from a Biomotion Laboratory study representative, Drs. Parker or Shinobu provided the subject's telephone number to the study's administrator who contacted prospective subjects via telephone to provide more details about the research project. With further agreement from the subject, an appointment was set up at the Biomotion Lab.

All subjects were adults who could understand and follow basic directions. Persons were excluded if they reported acute pain which prevented performance of their comfortable, typical movement. Similarly, persons were excluded if they have a unstable medical condition such as hypertension or diabetes mellitus.

B.1.2 Consent Forms

The protocols for this study was approved by both the MGH Institutional Review Board (IRB) and the MIT Committee On the Use of Humans as Experimental Subjects (COU-HES). The subjects all consented in accordance with the MGH IRB and the MIT COU-HES. The MGH IRB consent form for subjects with healthy gait is shown in Figure B.1, the MGH IRB consent form for subjects with difficulty walking is shown in Figure B.2, and the MIT COUHES consent form is shown in Figure B.3 (the two MGH forms were stamped with the IRB approval; the stamp is not visible in these figures). Subject testing took place from the March 25, 2003 through June 6, 2003; the Health Insurance Portability and Accountability Act of 1996 (HIPAA) went into effect April 15, 2003, so the final eleven subjects (tested after April 15, 2003) additionally signed the HIPAA paperwork.

<section-header><text><text><text><text><text><list-item><list-item><list-item><list-item><list-item><list-item><list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></text></text></text></text></text></section-header>	he)	Access Constant Section	Page 6
Reactive fragments and the second sec	Page 2	<section-header><text><text><text><text><text><text><text><text><text><text></text></text></text></text></text></text></text></text></text></text></section-header>	Dage S
<text><text><text><text><text><text><text><text><text><text><text><text></text></text></text></text></text></text></text></text></text></text></text></text>	Page 1	<text><text><text><list-item><list-item><list-item><list-item><list-item><list-item><list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></text></text></text>	Page 1

Figure B.1 MGH consent form for subjects with healthy gait



Figure B.2 MGH consent form for subjects with difficulty walking

 Research Consent Form
Massachusetts Institute of Technology
Imprint Patient ID Number
STUDY CONTACTS
The principal investigator for this study at MIT is Professor Joseph Paradiso. He can be reached at (617) 253-6215.
The principal investigator for this study at MGH is Donna Scarborough. She can be reached at (617) 726-3406 (weekdays 9am-4pm).
REQUEST FOR MORE INFORMATION
You may ask more questions about the study at any time. The investigator(s) will provide their telephone number so that they are available to answer your questions or concerns about the study. You will be informed of any significant new findings discovered during the course of this study that might influence your continued participation.
If during the study or later, you wish to discuss your rights as a research subject, your participation in the tudy and/or concerners about the study, are search-related injury with someone not directly involved in the study, or if you feel under any pressure to enroll in this study or to continue to participate in this study, you are aked to context the Committee on the Use of Humane as Experimental Subjects at MIT at (617) 253-6787, or the Human Research Committees at BWH at (617) 722-7200, at MGH at (617) 726-3493, or at the Protocol Administration Office at DFCI at (617) 632-3029.



GAT TESTING PROTOCOL FOR MCH BIOMOTION LAD Hathyse and Federalized Gath StableCE: with the procession of the stablece is the procession of the stablece is the stable is the stab	SUBJECT NAME
Health information provided from subject to tester during setup: (Cardiovascular system, signs of distress)	(VD) #
Signed informed concert? Signed HIPAA authorization?	For a second

Figure B.4 Subject Testing Protocol, page 1 (left) and page 2 (right)

B.1.3 Protocol

The protocol followed during subject testing is shown above in Figure B.4.

The subjects were asked to walk in a variety of ways. First, each subject was first asked to walk at his or her own pace (termed "free gait" for use in Chapter 6). Next, a number of calibration routines were carried out, including "chair rise", where the subject stood from a seated position. Following the calibrations, the subject started gait while within the viewing volume of the BML cameras, to collect data with the BML system about the initiation of gait, and then one to three "free gait" trials were carried out. In each of the following four trials, the subject was told to do a task (detailed in Figure B.3) designed to provide distraction ("distracted gait" in Chapter 6). Next, the metronome was turned on to 120 beats per minute, for two to three trials ("paced gait" in Chapter 6). Finally, lines were placed on the floor with a separation of approximately 1 m, and the subject was asked to step on the lines while walking (included with the "distracted gait" group in Chapter 6). Collection from the BML optical system concluded with "static standing", while the subject stood still with the feet 30 cm apart (this data was used to determine the orientations of the accelerometers with respect to the horizontal, for each subject). The testing concluded with data collected from the BML force plate and the GaitShoe while the subject walked forward to the center of the viewing volume, turned, and walked back to the starting point; this data was not analyzed for this thesis, but was collected for future work in analyzing gait which includes turns.

B.2 Subject Information

A total of sixteen subjects were recruited for the validation of the GaitShoe; they were provided with identification numbers¹ consecutively from 11 to 26. Gender, age, height,

^{1.} An additional ten subjects (01 to 10) were recruited for prototype testing and evaluation of the GaitShoe; the data from these initial subjects was not used in the final analysis presented in this thesis.

weight, and presence of Parkinson's disease are detailed in Table B.1 (in Chapter 6, the subjects are referred to by a coded name).

ID	Gender	Age	Height	Weight	Parkinson's disease
	[<u>M</u> ale/ <u>F</u> emale]	[years]	[m]	[kg]	[<u>Y</u> es/ <u>N</u> o]
11	F	24.9	1.6	48.2	Ν
12	F	28.2	1.6	59.1	Ν
13	М	25.3	1.8	75.0	Ν
14	М	27.5	1.8	115.0	Ν
15	F	48.2	1.6	52.3	Ν
16	F	28.6	1.6	50.0	Ν
17	F	54.0	1.7	54.5	Ν
18	F	26.8	1.6	66.4	Ν
19	F	27.4	1.7	55.9	Ν
20	F	53.8	1.6	63.6	Y
21	F	26.9	1.7	69.5	Ν
22	F	65.9	1.7	68.2	Y
23	М	64.9	1.8	94.5	Y
24	М	76.4	1.7	77.7	Y
25	F	65.4	1.6	52.3	Y
26	М	30.3	1.8	90.9	Ν

TABLE B.1 Information about volunteers for the subject testing

The data collected for Subject 21 were excluded from the analysis because the antennas on both the left and right GaitShoe attachments were in need of repair; this was not discovered until after data testing was underway, and resulted in very poor data transmission, with a data collection rate from the GaitShoe lower than 50 Hz for each foot, rather than the usual 75 Hz.

Thus, the final cohort of subjects included five subjects with Parkinson's disease (2 males and 3 females), and ten subjects with normal gait¹ (7 females and 3 females).

^{1.} One of these ten subjects was diagnosed with *myasthenia gravis* (a neurological disease usually affecting face muscles), however, no changes in gait were observed by the physical therapists at the BML, or in GaitShoe data used for the pattern recognition in Chapter 6. This subject was therefore not uniquely labeled, but included as a subject with normal gait.

Appendix C

PATTERN RECOGNITION INFORMATION

This appendix accompanies Chapter 6, and explains terminology commonly used in pattern recognition (see also Pattern Classification by Duda, Hart, and Stork [96]). This appendix also contains the complete results from the classification. Results are presented in the same order as in Chapter 6; many of the tables here were included in Chapter 6 as well (results from the additional neural net studies are only in Section 6.6.5).

C.1 Terminology

Bayes Decision Theory: See Appendix 6.1.2.

CART: Classification and Regression Trees (a type of decision tree); see Section 6.1.1. Figure C.1 shows a sample tree, with three classes, 123 total samples, and using four features to create the tree.



Figure C.1 Sample CART tree

Class: Descriptor for a group of samples, e.g. "apple" or "banana" or "orange".

Confusion Matrix: Table of results showing how the *sample* in the *testing set* were classified. Table C.1 shows a sample *confusion matrix* for a classification of apples, bananas, and oranges. Each row shows the results for a single *class*; in this example, 48 apples were correctly classified as apples, and 2 apples were misclassified as oranges, for a 96.0% classification rate for the apples. Similarly, all 50 bananas (100%) were classified correctly, and 40 oranges (80%) were classified correctly.

	Apples	Bananas	Oranges	% correct
Apples	48	0	2	96.0
Bananas	0	50	0	100.0
Oranges	10	0	40	80.0
			Overall:	92.0

TABLE C.1 Sample results presented in a confusion matrix

The overall classification rate is calculated by the sum of the correct classifications (the numbers along the diagonal), divided by the total number of *samples* in the *testing set*; in this example:

Overall Classification Rate =
$$\frac{48 + 50 + 40}{50 + 50 + 50}$$
 = 92.0%. (8.1)

Data Set: Collection of *samples*, including samples from all classes. The *data set* is typically split into a *training set* and a *testing set*.

Features: Continuous or categorical properties that (ideally) distinguish between the *classes*. For classifying apples, bananas, and oranges, categorical features might be color or shape, continuous features might be hue or mass. A feature such as "food type" would not be useful, as all three classes are types of fruit.

Leave One Out: A method of evaluating the classifier by having N *training* and *testing sets*, where N is the total number of *samples*, and each *testing set* consists of a single sample and its corresponding *training set* consists the N-1 *samples* not in the *testing set*.

Neural Networks: See Section 6.1.4.

Sample: A single piece of data, consisting of a class label and the features.

SVMs: Support Vector Machines; see Section 6.1.3.

Testing Set: Group of *samples* used to "test" the classifier; the classifier is applied to these samples to see what percentage of the testing set is correctly classified. If the *testing set* is used to adjust the parameters of the classifier (as it is with the CART software), the results are termed "overly optimistic;" it is good practice to use a separate "evaluation set" to avoid this.

Training Set: Group of *samples* used to "train" the classifier; these samples are used to determine the parameters of the classifier to distinguish between the *classes* of the samples within the training set.

C.2 Complete Results

C.2.1 Hypothesis 1

Hypothesis 1 compared group NLFY (females with normal gait, younger than 30 years old), with group NLF (females with normal gait, older than 48 years old), and group PDF (females with Parkinson's disease, older than 54 years old). The training and testing groups are described in Section 6.5.

The results for Hypothesis 1, using modified leave one out, are shown in Tables C.2, C.3, and C.4.

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	122	10	3	90.4	97.8
NLF	6	47	1	87.0	98.2
PDF	0	3	78	96.3	3.7
Overall:				91.5	

 TABLE C.2
 CART Hypothesis 1 results, using modified leave one out

TABLE C.3 Naïve Bayes Hypothesis 1 results, using modified leave one out

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	129	2	4	95.6	97.0
NLF	2	50	2	92.6	96.3
PDF	0	3	78	96.3	3.7
Overall:				95.2	

TABLE C.4 Neural Net Hypothesis 1 results, using modified leave one out

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	131	2	2	97.0	98.5
NLF	3	50	1	92.6	98.2
PDF	0	0	81	100.0	0.0
Overall:				97.0	

The results for Hypothesis 1, using "leave one subject out", are shown summed in Table C.5 and by subject in Table C.6.

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	66	21	5	71.7	94.6
NLF	32	4	1	10.8	97.3
PDF	4	13	47	73.4	26.7
Overall:				60.6	

TABLE C.5 CART Hypothesis 1 results, by class, using leave one subject out

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY-A	16	0	0	100.0	100.0
NLFY-B	8	9	0	47.1	100.0
NLFY-C	10	9	2	47.6	90.5
NLFY-D	18	0	0	100.0	100.0
NLFY-E	14	3	3	70.0	85.0
NLF-A	15	4	0	21.1	100.0
NLF-B	17	0	1	0.0	94.4
PDF-A	0	0	20	100.0	0.0
PDF-B	0	0	23	100.0	0.0
PDF-C	4	13	4	19.1	81.0
		0	60.6		

TABLE C.6 CART Hypothesis 1 results, by subject, leave one subject out

The results for Hypothesis 1, using "free gait", "distracted gait", or "paced gait" (gait types are described in Section B.1) as the test set, are shown in Tables C.7, C.8, and C.9.

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	29	2	7	76.3	81.6
NLF	4	8	1	61.5	92.3
PDF	1	2	18	85.7	14.3
		0	verall:	76.4	

 TABLE C.7
 CART Hypothesis 1 results, using "free gait" as the test set

TABLE C.8 CART Hypothesis 1 results, using "distracted gait" as the test set

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	27	0	3	90.0	90.0
NLF	1	14	0	93.3	100.0
PDF	1	0	24	96.0	4.0
		0	verall:	92.9	

	NLFY	NLF	PDF	% correct	% NLFY or NLF
NLFY	19	2	3	79.2	87.5
NLF	0	9	0	100.0	100.0
PDF	0	0	18	100.0	0.0
		0	verall:	90.2	

TABLE C.9 CART Hypothesis 1 results, using "paced gait" as the test set

As discussed in Chapter 6, SVMs were used to classify between two classes. Hypothesis 1.1 compared NLFY with NLF, Hypothesis 1.2 compared NLF with PDF, and Hypothesis 1.3 compared NLFY with PDF. The results for each, using modified leave one out, are shown in Tables C.10, C.11, and C.12.

TABLE C.10 SVM Hypothesis 1.1 results, using modified leave one out

	NLFY	NLF	% correct
NLFY	130	5	96.3
NLF	4	50	92.6
Overall:			95.2

TABLE C.11 SVM Hypothesis 1.2 results, using modified leave one out

	NLF	PDF	% correct
NLF	46	8	85.2
PDF	1	80	98.8
		93.3	

	TABLE C.12	SVM Hypothesis	1.3 results	, using	modified	leave one	out
--	------------	----------------	-------------	---------	----------	-----------	-----

	NLFY	PDF	% correct
NLFY	132	3	97.8
PDF	3	78	96.3
Overall:			97.2

The results for Hypotheses 1.1, 1.2, and 1.3, using "leave one subject out", are shown in Tables C.13, C.14, and C.15.

	NLFY	NLF	% correct
NLFY	80	12	87.0
NLF	34	3	8.1
Overall:			64.3

 TABLE C.13
 SVM Hypothesis 1.1 results, using leave one subject out

TABLE C.14 SVM Hypothesis 1.2 results, using leave one subject out

	NLF	PDF	% correct
NLF	10	27	27.0
PDF	32	32	50.0
		Overall:	41.6

TABLE C.15 SVM Hypothesis 1.3 results, using leave one subject out

	NLFY	PDF	% correct
NLFY	89	3	96.7
PDF	28	36	56.3
	Overall:		80.1

The results for Hypotheses 1.1, 1.2, and 1.3, using "free gait" as the test set, are shown in Tables C.16, C.17, and C.18.

	NLFY	NLF	% correct
NLFY	38	0	100.0
NLF	3	10	76.9
Overall:			94.1

TABLE C.16 SVM Hypothesis 1.1 results, using "free gait" as the test set

	NLF	PDF	% correct
NLF	12	1	92.3
PDF	4	17	81.0
		85.3	

TABLE C.17 SVM Hypothesis 1.2 results, using "free gait" as the test set

TABLE C.18 SVM Hypothesis 1.3 results, using "free gait" as the test set

	NLFY	PDF	% correct
NLFY	37	1	97.4
PDF	1	20	95.2
	(96.6	

The results for Hypotheses 1.1, 1.2, and 1.3, using "distracted gait" as the test set, are shown in Tables C.19, C.20, and C.21.

	NLFY	NLF	% correct
NLFY	29	1	96.7
NLF	1	14	93.3
		Overall:	95.6

TABLE C.19 SVM Hypothesis 1.1 results, using "distracted gait" as the test set

 TABLE C.20
 SVM Hypothesis 1.2 results, using "distracted gait" as the test set

	NLF	PDF	% correct
NLF	12	3	80.0
PDF	4	21	84.0
		Overall:	82.5

TABLE C.21 SVM Hypothesis 1.3 results, using "distracted gait" as the test set

	NLFY	PDF	% correct
NLFY	30	0	100.0
PDF	3	22	88.0
		Overall:	94.5

The results for Hypotheses 1.1, 1.2, and 1.3, using "paced gait" as the test set, are shown in Tables C.22, C.23, and C.24.

	NLFY	NLF	% correct
NLFY	22	2	91.7
NLF	1	8	88.9
	(90.9	

TABLE C.22 SVM Hypothesis 1.1 results, using "paced gait" as the test set

TABLE C.23 SVM Hypothesis 1.2 results, using "paced gait" as the test set

	NLF	PDF	% correct
NLF	8	1	88.9
PDF	0	18	100.0
	Overall:		96.3

TABLE C.24 SVM Hypothesis 1.3 results, using "paced gait" as the test set

	NLFY	PDF	% correct
NLFY	21	3	87.5
PDF	0	18	100.0
		Overall:	92.9

C.2.2 Hypothesis 2

Hypothesis 2 compared the combined group of NLFY and NLF (resulting in a group of all females with normal gait), with group PDF (all females with Parkinson's disease).

The results for Hypothesis 2, using modified leave one out, are shown in Tables C.25, C.26, C.27, and C.28.

	NLF/NLFY	PDF	% correct
NLF/NLFY	181	8	95.8
PDF	1	80	98.8
		Overall:	96.7

TABLE C.25 CART Hypothesis 2 results, using modified leave one out

TABLE C.26 SVM Hypothesis 2 results, using modified leave one out

	NLF/NLFY	PDF	% correct
NLF/NLFY	181	8	95.8
PDF	2	79	97.5
		Overall:	96.3

TABLE C.27 Naïve Bayes Hypothesis 2 results, using modified leave one out

	NLF/NLFY	PDF	% correct
NLF/NLFY	181	8	95.8
PDF	4	77	95.1
Overall:			95.6

TABLE C.28 Neural Net Hypothesis 2 results, using modified leave one out

	NLF/NLFY	PDF	% correct
NLF/NLFY	187	2	98.9
PDF	0	81	100.0
Overall:			99.3

The results for Hypothesis 2, using "leave one subject out", are shown summed in Tables C.29, C.31, and C.33, and by subject in Tables C.30, C.32, and C.34.

	NLF/NLFY	PDF	% correct
NLF/NLFY	123	6	95.3
PDF	21	43	67.2
		Overall:	86.0

 TABLE C.29
 CART Hypothesis 2 results, using leave one subject out

TABLE C.30 CART Hypothesis 2 results, by subject, using leave one subject out

	NLF/NLFY	PDF	% correct
NLFY-A	16	0	100.0
NLFY-B	17	0	100.0
NLFY-C	19	2	90.5
NLFY-D	18	0	100.0
NLFY-E	17	3	85.0
NLF-A	19	0	100.0
NLF-B	17	1	94.4
PDF-A	4	16	80.0
PDF-B	0	23	100.0
PDF-C	17	4	19.0
		Overall:	86.0

TABLE C.31 SVM Hypothesis 2 results, using leave one subject out

	NLF/NLFY	PDF	% correct		
NLF/NLFY	114	15	88.4		
PDF	33	31	48.4		
		Overall:	75.1		

	NLF/NLFY	PDF	% correct		
NLFY-A	15	1	93.8		
NLFY-B	17	0	100.0		
NLFY-C	21	0	100.0		
NLFY-D	18	0	100.0		
NLFY-E	20	0	100.0		
NLF-A	16	3	84.2		
NLF-B	7	11	38.9		
PDF-A	6	14	70.0		
PDF-B	12	11	47.8		
PDF-C	15	6	28.6		
		Overall:	75.1		

TABLE C.32 SVM Hypothesis 2 results, by subject, using leave one subject out

TABLE C.33 Neural Net Hypothesis 2 results, using leave one subject out

	NLF/NLFY	PDF	% correct
NLF/NLFY	122	7	94.6
PDF 18		46	71.9
		Overall:	87.0

TABLE C.34 Neural Net Hypothesis 2 results, by subject, using leave one subject out

	NLF/NLFY	PDF	% correct		
NLFY-A	16	0	100.0		
NLFY-B	17	0	100.0		
NLFY-C	20	1	95.2		
NLFY-D	18	0	100.0		
NLFY-E	15	5	75.0		
NLF-A	19	0	100.0		
NLF-B	17	1	94.4		
PDF-A	5	15	75.0		
PDF-B	0	23	100.0		
PDF-C	13	8	38.1		
		Overall:	87.0		

The results for Hypothesis 2, using "free gait", "distracted gait", or "paced gait" as the test set, are shown in Tables C.35, C.36, C.37, C.38, C.39, and C.40.

	NLF/NLFY	PDF	% correct
NLF/NLFY	43	8	84.3
PDF	2	19	90.5
		Overall:	86.1

 TABLE C.35
 CART Hypothesis 2 results, using "free gait" as the test set

TABLE C.36 SVM Hypothesis 2 results, using "free gait" as the test set

	NLF/NLFY	PDF	% correct
NLF/NLFY	51	0	100.0
PDF	2	19	90.5
		Overall:	97.2

TABLE C.37 CART Hypothesis 2 results, using "distracted gait" as the test set

	NLF/NLFY	PDF	% correct		
NLF/NLFY	38	7	84.4		
PDF	1	24	96.0		
	88.6				

TABLE C.38 SVM Hypothesis 2 results, using "distracted gait" as the test set

	NLF/NLFY	PDF	% correct		
NLF/NLFY	44	1	97.8		
PDF	3	22	88.0		
		Overall:	94.3		

TABLE C.39 CART Hypothesis 2 results, using "paced gait" as the test set

	NLF/NLFY	PDF	% correct
NLF/NLFY	31	2	93.9
PDF	1	17	94.4
		94.1	

	NLF/NLFY	PDF	% correct
NLF/NLFY	30	3	90.9
PDF	0	18	100.0
		Overall:	94.1

TABLE C.40 SVM Hypothesis 2 results, using "paced gait" as the test set

C.2.3 Hypothesis 3

Hypothesis 3 classified each of the ten subjects as individuals; the results, using cross-validation, are shown in Tables C.41, C.42, and C.43.

	NLFY A	NLFY B	NLFY C	NLFY D	NLFY E	NLF A	NLF B	PDF A	PDF B	PDF C	% correct	% NLFY or NLF
NLFY-A	15	0	0	1	0	0	0	0	0	0	93.8	100.0
NLFY-B	0	17	0	0	0	0	0	0	0	0	100.0	100.0
NLFY-C	0	0	20	0	1	0	0	0	0	0	95.2	100.0
NLFY-D	0	0	0	18	0	0	0	0	0	0	100.0	100.0
NLFY-E	0	0	0	0	18	1	0	0	0	1	90.0	95.0
NLF-A	0	0	0	0	0	19	0	0	0	0	100.0	100.0
NLF-B	0	0	0	0	0	0	18	0	0	0	100.0	100.0
PDF-A	0	0	0	0	0	0	0	20	0	0	100.0	0.0
PDF-B	0	0	0	0	0	0	0	0	23	0	100.0	0.0
PDF-C	0	0	0	0	0	0	0	1	0	20	95.2	0.0
									Ove	rall:	97.4	

TABLE C.41 Neural Network Hypothesis 3 results, by subject, using cross-validation
	NLFY A	NLFY B	NLFY C	NLFY D	NLFY E	NLF A	NLF B	PDF A	PDF B	PDF C	% correct	% NLFY or NLF
NLFY-A	14	1	0	1	0	0	0	0	0	0	87.5	100.0
NLFY-B	0	17	0	0	0	0	0	0	0	0	100.0	100.0
NLFY-C	0	0	16	0	4	1	0	0	0	0	76.2	100.0
NLFY-D	0	1	0	16	0	0	1	0	0	0	88.9	100.0
NLFY-E	0	0	4	0	13	3	0	0	0	0	65.0	100.0
NLF-A	0	0	2	0	3	13	1	0	0	0	68.4	100.0
NLF-B	0	2	0	0	1	0	12	0	0	3	66.7	83.3
PDF-A	0	0	0	0	2	0	0	17	1	0	85.0	10.0
PDF-B	0	0	0	0	0	0	0	2	21	0	91.3	0.0
PDF-C	0	0	1	0	0	0	1	0	0	19	90.5	9.5
									Ove	rall:	81.9	

 TABLE C.42
 CART Hypothesis 3 results, by subject, using cross-validation

TABLE C.43 Naïve Bayes Hypothesis 3 results, by subject, using cross-validation

	NLFY A	NLFY B	NLFY C	NLFY D	NLFY E	NLF A	NLF B	PDF A	PDF B	PDF C	% correct	% NLFY or NLF
NLFY-A	15	0	0	1	0	0	0	0	0	0	93.8	100.0
NLFY-B	0	17	0	0	0	0	0	0	0	0	100.0	100.0
NLFY-C	0	0	18	0	3	0	0	0	0	0	85.7	100.0
NLFY-D	0	0	0	17	1	0	0	0	0	0	94.4	100.0
NLFY-E	0	0	3	0	16	0	0	0	0	1	80.0	95.0
NLF-A	0	0	1	0	0	18	0	0	0	0	94.7	100.0
NLF-B	0	0	0	0	2	0	16	0	0	0	88.9	100.0
PDF-A	0	0	1	0	0	0	0	19	0	0	95.0	5.0
PDF-B	0	0	1	0	0	0	0	0	22	0	95.7	4.3
PDF-C	0	0	0	0	0	0	1	1	0	0	90.5	4.8
									Ove	rall:	91.7	

Appendix D

HARDWARE INFORMATION

This appendix contains information required for the building of the GaitShoe circuit boards and other hardware, and contains part numbers and purchasing information for all parts used.

D.1 Schematics and Board Layouts

D.1.1 Inertial Measurement Unit (IMU) Board, Rev. 5

The schematic for the IMU board (Rev. 5) is shown in Figure D.1. The board layouts with part placement information are shown in Figure D.2 (top side) and Figure D.3 (bottom side).

In Figure D.2, the two footprints for the two different versions of the Murata gyroscopes are visible. As discussed in Section 3.3.2, the ENC-03J was used in all testing, but the footprint for the surface mount ENC-03M was also included on the IMU board (the alternate pin connections can be seen in Figure D.1) in case the ENC-03J becomes no longer available.

Similarly, the Analog Devices gyroscope (the ADXRS150) was a demo DIP version and is not available commercially. It is now available in a 32-pin ball grid array surface-mount package; pin-to-pin mappings between this package and the layout on the board (e.g. the sixteen pin header would plug into the holes for the current DIP) are shown in Figure D.4.



Figure D.1 Schematic of the IMU board



Figure D.2 Layout of the top side of the IMU board



Figure D.3 Layout of the bottom side of the IMU board



As indicated, the capacitors should be included on the daughter board, to be located as close to the ADXRS150 as possible.

Figure D.4 Pin mappings for the commercially available ADXRS150

D.1.2 Tactile Board, Rev. 5

The schematic for the Tactile board (Rev. 5) is shown in Figure D.5. The board layouts with part placement information are shown in Figure D.6 (top) and Figure D.7 (bottom). Table D.1 lists the header pin to insole sensor mapping used for the GaitShoe insoles (all odd-numbered pins are connected to ground). Pins 1-4 can be used to connect to a ground plane, though the connection through the electric field sensor header can also be used.

Header Pins	Connection	Left Insole	Right Insole
1, 2, 3, 4	Ground		
5, 6	FSR	Lateral heel	Medial heel
7, 8	FSR	Fifth (lateral) metatarsal	First (medial) metatarsal
9, 10	PVDF	Calcaneous (heel)	Great toe
11, 12, 13, 14	Bend Sensor	Insole	Insole
15, 16	PVDF	Great toe	Calcaneous (heel)
17, 18	FSR1	Medial heel	Lateral heel
19, 20	FSR	First (medial) metatarsal	Fifth (lateral) metatarsal
21, 22, 23, 24	Bend Sensor	Ankle	Ankle

TABLE D.1 Insole sensor mapping



Figure D.5 Schematic of the Tactile board



Figure D.6 Layout of the top side of the Tactile board



Figure D.7 Layout of the bottom side of the Tactile board

D.1.3 Main Board, Rev. 5

The schematic for the Main board (Rev. 5) is shown in Figure D.8. The board layouts with part placement information are shown in Figure D.9 (top) and Figure D.10 (bottom).



Figure D.8 Schematic of the Main board



Figure D.9 Layout of the top side of the Main board

The six holes along the bottom edge of the Main board are for interfacing with the JTAG programmer, for uploading code to the microcontroller. The four holes along the left edge (looking at the top of the board) can be used to bypass the wireless transceiver and send the data directly to the computer. Both of these interface with the Programming board, described in Section D.1.5 below.



Figure D.10 Layout of the bottom side of the Main board

D.1.4 Power Board (Rev. 2)

The schematic for the Power board (Rev. 2) is shown in Figure D.11, and the board layouts with part placement information are shown in Figure D.12 (top) and Figure D.13 (bottom).



Figure D.11 Schematic of the Power board



Figure D.12 Layout of the top side of the Power board



Figure D.13 Layout of the bottom side of the Power board

D.1.5 Programming Board (Rev. 2)

The Programming board was designed by graduate student Ari Benbasat. The schematic for the Programming board (Rev. 2) is shown in Figure D.14. The board layouts with part placement information are shown in Figure D.15, showing traces on both the top and bottom are shown).



Figure D.14 Schematic of the Programming board



Figure D.15 Layout for the Programming board

D.2 Component Information

Information about the components used to build the GaitShoe is split across two pages, in Figure D.16 and Figure D.17. The part type, description, and a vendor (the websites for the vendors are listed in Figure D.2) and the part number for the vendor are listed, as well as the footprint and designator corresponding to the schematics of each board, and any relevant comments.



Figure D.16 Component information, part one.

Iments of instead of 4 (uH) of instead of 5 4uH) for securing power board to the attachment. for securing power board to the attachment.	to protect sensers in inside - by a width of 1 or 2 feet. to make attachment. d Instead of 5k pot.)	10M resistors used for subject testing, vary to adjust pvdf sensitivity.	ul for sating different pudf sensitivity.	ul for setting different pudf sensitivity.	to hold boards to the attachment. for securing power board to the attachment.	between boards - need to use multiple. between boards - ywwill need to cut these to length. n relief for the insole sensor header.	heeded if the elec field sensor is not used.
Com (User (Used ant Used	ant Used (Use	Two	extra Usefu	extra Usefu	ant Used ant Used ming2	ming2 ant Used ant Used Straii	Not r
Board power2 power2 power2 main5 tactile4 power2 power2 power2 power2 power2 attachme attachme	tactile4 attachme attachme tactile4 imu5 imu5 power2 tactile4 imu5	tactile4 tactile4 power2 tactile4	main5 main5 imu5 imu5 imu5 tactle4-e power2 tactle4- tactle4	main5 tactile4-e tactile4 imu5	attachme attachme programr	programr attachme attachme tactile4 power2	main5 power2 power2 power2 main5
Protel Designator 11 13 13 05. Os 05. Os 05. Os 03. Us 13. Us 01. Us, U7. U8	U1 R65, RV6 R61, R7 R1, R2, R3 R1, R2, R3 R1, R2, R1, R13, R13, R1, R21, R16, R17, R14, R15, R16, R13,	R2, R4, R6, R8, R11, R14 R9, R10, R12, R13 R11, 8 R4, R5, R2, R28, R30 R20, R22, R28, R30	2 88. R9 88. R9 87. R10. R12. R13 88. R9 819. R21. R27. R29 R13. R26. R31. R34 R26. R31. R34	R1 R9, R10, R12, R13 R24, R25, R32, R33 R18, R19 D1 D2 D3		1115	u4 u3 u1 u2 u4 u1 created by the author;)
Footprint 1008 1008 1008 1008 1008 1008 1008 100		603 603 603	603 603 603 603 603 603 603	603 1210 503 5010 500 500		SWITCH1270	50122-5 50-8 50-8 50-8 50-8 PR2000 (footprints in italics were
Vendor Vendor Digkey Digkey Digkey Makey Digkey Digkey Digkey Makaser Carr Makaser Carr Makaser Carr Makaser Carr	McMaster Carr McMaster Carr Digkey Digkey Digkey Digkey Digkey Digkey	Digikey Digikey Digikey Digikey	Dgkey Dgkey Dgkey Dgkey Dgkey Dgkey	Digikey Digikey Digikey Digikey	McMaster Carr McMaster Carr Digikey	Digikey Digikey McMaster Carr Digikey	Digkey Analog Devkes Analog Devkes Analog Devkes Digkey RF Monolithics
Part Number TKS3626CT TKS3626CT TKS3624CT TKS3624CT TKS3624LT MM8F5484LT1 67-1240 67-1240 67-1240 67-1240 67-1240 67-1240 712021F TCVW02FC	8/8/5/17 8707/112 MSP1006 FEC104CT FEC104CT 711-0.05CT 711-0.05CT P1.00KHCT	311-10M/GCT 311-10M/GCT P106CT P100KHCT	311-12.0KHCT P124KHCT 311-15.0KHCT 311-2.2KHCT P2.1MACT P2.1MACT P2.1MACT 311-220KHCT 311-220KHCT 311-220KHCT	P3.3KYCT P5.1MGCT 311-51.0KHCT P9810CT MRPS-130CT	91772A118 91772A106 209F	MAX233CPP 875K 94639A209 ASSR24 EG1847	ADM1111AR-3.001 ADM11111AR-12 ADM1111AR-5.3 ADM1111AR-5.3 MAX666CSA MAX666CSA DR3000-1
Description Inductor Inductor Inductor Inductor Inductor Inductor Inductor IED Inductor Nation Natio	Plastic Film Plastic Film Plastic Sheet Polymytidine Diffuoride Podentiometer Resistor Resistor Resistor	Resistor Resistor Resistor Resistor	Resistor Resistor Resistor Resistor Resistor Resistor Resistor Resistor Resistor Resistor Resistor Resistor	Resistor Resistor Resistor Resistor Schottku Dioda	Screws Screws Serial Connector	Serial Driver/Receiver Spacers Sparers, Strain Relief Svitch	Voltage keerence Voltage Regulator Voltage Regulator Voltage Regulator Voltage Regulator Wireless Transcelver
Part Type Bart Type 27 ult 27 ult 28 ult 27 ult 28 ult 20	mili sha sha sha sha sha sha sha sha sha sha	anonent	200 × × × × × × × × × × × × × × × × × ×	5 M 5 M Ferrite Bead	1 1/8" length, 4/40s 1/4" length, 4/40s DB9 Female	C RS-232 1/8" length 1 1/2" length 24 pin SWITCH_1270	DM4120 ADP1111-12 ADP1111-5.3 ADP1111-5.3 MAX666 DR3000-1

TABLE D.2 Vendor information

Company	Website Address
Analog Devices	www.analog.com
Android World	www.androidworld.com/prod47.htm
Digikey	www.digikey.com
Interlink Electronics	www.interlinkelec.com
McMaster Carr	www.mcmaster.com
ON Semiconductor	www.onsemi.com
RF Monolithics	www.rfm.com
Rochester Electronics	www.rocelec.com
The Images Co.	www.imagesco.com

D.3 Other Hardware Information

D.3.1 Insole Sensor Connections

The soldered connections to the sensors in the insole must be able to withstand the forces (normal and shear) encountered under the foot during gait. One method that has worked well was to cover the well-soldered connections with hot glue, let it the glue cool slightly, press it flat, and then trim off the excess. This is demonstrated on a pair of bend sensors.

First the bend sensors are soldered to the wires, as shown in Figure D.18.



Figure D.18 Bend sensors soldered to wire

Next, a "gob" of hot glue is applied to the solder connections, as shown in Figure D.19.

Figure D.19 Bend sensors with a "gob" of hot glue

After the hot glue cools slightly, it is pressed between fingertips to flatten it into a thin layer; the result is shown in Figure D.20.



Figure D.20 Hot glue flattened by fingers

Finally, the hot glue is trimmed, as shown in Figure D.21.



Figure D.21 Trimmed hot glue on bend sensor solder connection

D.3.2 GaitShoe Attachment

The GaitShoe attachment was made out of 0.125" polyethylene terephthalate glycol (PTG), which is a thermoformable and machinable material. The process of making the attachment is detailed in the photos below.

First, a pattern, such as shown (to scale) in Figure D.22, can be used to trace the pattern of the attachment on to the PTG, as shown in Figure D.23. The solid lines in Figure D.22 indicate the base pattern and the dashed lines indicate extra material to add for a right or left antenna attachment (the pattern does not need to be followed exactly).



Figure D.22 Sketch of the pattern for both the left and right GaitShoe attachments



Figure D.23 Pattern traced onto PTG for forming the GaitShoe Attachment



Next, the easiest way to cut the PTG is to use heavy-duty shears, as shown in Figure D.24.

Figure D.24 PTG cut to shape with heavy duty shears

After the PTG is cut, the holes should be drilled¹, and then the PTG can be formed using a heat gun and a rounded block (or other available implement²), as shown in Figure D.25, to shape it to the desired geometry.



Figure D.25 Heat gun and wooden block used to shape the PTG

^{1.} The final hole in the attachment connecting the battery enclosure to the attachment must be drilled after both pieces are shaped; the holes for the stack should be marked using the IMU of the stack which will be mounted on the attachment.

^{2.} For example, a the bottom of the attachment can be rolled around a small screwdriver to form the enclosure at the bottom of the attachment for the fishing line.



The battery enclosure can be formed similarly; a sample pattern is shown in Figure D.26.

Figure D.26 Sketch of the pattern for the battery enclosure

A left GaitShoe attachment and battery enclosure are shown separately from the back, and together from the front in Figure D.27.



Figure D.27 GaitShoe attachment and battery enclosure; viewed separately from the back (left), and viewed together from the front (right)

Appendix E

ULTRASOUND SENSOR

This appendix describes in further detail the ultrasound sensor developed by Steven Dan Lovell; photographs of the ultrasound circuit boards are included in Section 3.2.1, and the working principle is discussed in Section 3.3.7. Mr. Lovell contributed to the writing of this appendix.

Ultrasound transducers can turn electrical energy into mechanical energy and mechanical energy into electrical energy. They have a range of frequencies over which they are resonant, where the ratio of energy input to energy dissipated in the device itself is low. The mechanical energy they produce and sense is in the form of pressure waves, thus the signal they can send propagates at the speed of sound, making them suitable for determining distances between a transmitter and a receiver by measuring the time between the transmission of the signal and the reception of the signal.

E.1 Circuit Boards

The ultrasound sensor was implemented on two circuit boards: a transmit board that attached to the stack on the right shoe, and a receiver board that attached to the stack on the left shoe. Attached directly to the transmit board was a single daughter board to hold the transmitter. The receive board had two daughter boards that each held a receiver, and were connected to the receive board via wires. These are shown in Figure E.1.



Figure E.1 Photo of ultrasound hardware mounted on the GaitShoe hardware

The schematic of the main transmit board is shown in Figure E.2, and the schematic of the transmit daughter board is shown in Figure E.3.



Figure E.2 Schematic of the ultrasound transmit board.



Figure E.3 Schematic of the ultrasound transmit daughter board

The schematic of the main receive board is shown in Figure E.4, and the schematic of the receive daughter board is shown in Figure E.5.



Figure E.4 Schematic of the ultrasound receive board.



Figure E.5 Schematic of the ultrasound receive daughter board.

An omni-directional polyvinylidene fluoride (PVDF) ultrasound transducer manufactured by MSI, part 1005853-1, was used for the transmitter [120]. Two directional piezoelectric ceramic (PZT) ultrasound transducers (such as V-MA40A5R [121]) were used for the receivers.

The omni-directional transducer must be driven at high voltages (150 V-300 V) at 40 KHz to generate a signal that can be measured over a distance of a few feet. A step-up transformer, resonant at 40 KHz with the ultrasound transducer as a load, along with a push-pull amplifier that takes a 3.3 V input from the microcontroller and outputs 9 V, was used to drive the omni-directional transducer at 270 V.

The receivers were placed on signal conditioning daughter boards that could be placed away from the main ultrasound receiver board. The daughter boards had two stages of filtering: a medium gain single pole high-pass filter and a medium gain double pole bandpass filter that added DC bias. This signal was then fed back to the main receiver board where a comparator was used to determine when the signal crossed a threshold. The threshold is variable and set by a potentiometer. This threshold crossing is used to determine when the first arrival of the ultrasound signal occurs.

To determine the distance between the transmitter and the receiver, the time of flight (TOF) between the transmitter and receiver is measured. This TOF is multiplied by the speed of sound in air to determine the distance of separation between the transmitter and receiver. To determine the TOF, the transmission of the ultrasound signal and the start of timer on the receiver must occur simultaneously. For this, the wireless radio frequency (RF) transmission already employed for communication between the GaitShoe hardware and the basestation was used. The basestation issued the ultrasound transmission command to initiate the start of transmission and the start of the receiver timer; the length of time between the RF transmission from the basestation and the reception on the Main Stack board, and then the subsequent communication between the Start board and the ultrasound transmit and receiver boards was assumed to be the same for the transmit

and receiver boards. When the ultrasound transmit board received an ultrasound transmission command it drove a short pulse of 40Khz square wave to the ultrasound transducer. When the ultrasound receive board received an ultrasound transmission command it started a timer, counting the number of clock cycles. When the ultrasound receive board was signaled by the comparator that a threshold crossing occurred, the value of the timer (i.e. total number of clock cycles) was stored, and subsequently transmitted to the base station. This occurred twice, once for each receiver. The two distances can be post-processed to determine the angle at which the line the between the transmitter and middle of the two receivers is on relative to the line perpendicular to the two receivers (see Figure 3.35).

E.2 Microcontroller Code

The microcontroller code written by Mr. Lovell for the transmit board is shown in Figure E.6 and for the receive board is shown in Figure E.7.



Figure E.6 Microcontroller code for the ultrasound transmit board.



Figure E.7 Microcontroller code for the ultrasound receiver board.

E.3 Initial Results

The uncalibrated output from the two ultrasound sensors is shown in Figure E.8, along with the uncalibrated output of the four FSR sensors on both the right and left feet, to provide a reference to the gait cycle (data outliers caused by RF dropouts and undetected or mis-timed sonar burst were removed as described in Section 4.1.3).



Figure E.8 Sample data from the two ultrasound sensors

The ultrasound receivers was located on the left foot, and the ultrasound transmitter was located on the right foot. The bottom graph demonstrates that the ultrasound sensor outputs are proportional to the distance between the receivers and the transmitter (ultrasound sensor A is located behind the stack, while ultrasound sensor B is located near the metatarsals). Shortly after 1 sec, the left foot is in mid-stance, and the right foot is in mid-swing, and as the transmitter on the right foot passes each of the receivers, the sensors reach local minimums in turn: the rear sensor (A) first, and then the front sensor (B), as each is passed by the transmitter. Then, shortly after 2 sec, the right foot is in mid-stance and the left foot is in mid-swing, and the sensors again reach local minimums, but in the opposite order, as this time the front sensor (B) passes the transmitter and is followed by the rear sensor (A).

Similarly, the local maximums can be observed shortly after 1.5 sec and 2.5 sec as one foot starts heel strike and the other foot starts toe off, corresponding to the maximum distance between the two feet.

This initial observation of the ultrasound sensor demonstrates that it may be a useful implementation on the GaitShoe for acquiring additional information about gait. As indicated in Figure 3.35, the two distances measured by the ultrasound receivers can be combined to determine the distance and the relative position between the two feet. However, before fully integrating the ultrasound sensor in the GaitShoe, both daughter boards should be redesigned. As seen in Figure E.1, the daughter boards (in particular, the receive daughter boards) are rather bulky and obtrusively located; now that the sensor has been demonstrated to work, a redesign of the positioning and fastening of these boards is necessary. The attachment also needs to be robust enough to withstand active gait. In addition, the typical power draw of the stack without the ultrasound sensor is 45 mA. On the stack with the transmit board the power draw increases to 65 mA, and on the stack with the receive board to 70 mA; this may restrict the use of the ultrasound sensor to applications where the batteries can be changed frequently.

Finally, a plot showing data from all of the sensors on both feet is shown in Figure E.8; the sensor outputs are uncalibrated (but staggered along the y-axis for legibility), during slow gait (to get the ultrasound data with as few outliers as possible). One of the ultrasound sensor outputs is plotted with the left foot data, and the other is plotted with the right foot data. This graph shows the full suite of sensor output of the GaitShoe.



Figure E.9 Data collected from all sensors, during slow gait

Appendix F

CODE

F.1 Microcontroller Code

The code for the microcontrollers is based on code originally written by Ari Benbasat. It was modified by Steven Dan Lovell and the author. Figure F.1 contains the header file 206.h used by the microcontrollers, the basestation microcontroller code is in Figure F.2, and the stack microcontroller code, with comments indicating the lines which need to be switched for the left or right stack, is in Figure F.3 and Figure F.4.



Figure F.1 206.h Code



Figure F.2 Basestation microcontroller code

/* stack.c code for the left and right stack */	lu accelz2;
<pre>#include <stdio.h></stdio.h></pre>	<pre>bit in_cycle,pause,dirty;</pre>
#include "206.h"	lu bendl, bend2, fsr1, fsr2, fsr3, fsr4, fsr5, fsr6;
//Forward declarations void init(void);	lu capA, capB, cap2, cap3;
Vold CollectAncig(Vold); vold collectAccels(vold);	iu timestamp,checkedsit;
void trans3Nibbles(unsigned char,unsigned char); void setMux(unsigned char in):	unsigned char counter;
#define SET_REC CTR0=1;TEnable=0;	int i=0; //used to create operations that allow me to set breakpoints
<pre>#define SET_TRANS CTR0=0;TEnable=1;</pre>	void milliClock() interrupt 5 (
<pre>// Declare ports sbit CTR1 = P3^1;</pre>	
shit TEnable = $P30$; (/oht grupper = P27;	/*************************************
// Declare direct lines	Registers changed: SPIF. SPI In. counter
<pre>sbit DIR1 = P2^4; sbit DIR2 = P2^5;</pre>	Resources used: none
sbit DIR3 = P2^6; sbit DIR5 = P3^6;	Notes: *SPI interrupts occur when an SPI byte has finished transmitting
sbit DIR6 = P3^2;	*SPIF must be cleared by software in this interrupt or the MCU will hang at the SPI interrupt vector (interrupt 6, memory location 0x0033)
// SPI related declarations #define Ultrasound_NSS DIR2 bit. SPI financia. (used to indicate whether SPI transmission has concluded	/Juhon an CDT interview on CDT buts has finished transmitting
char SPI_In; char SPI counter = 0:	Ultrasound_NSS=1; //raise slave's NSS pin so it can clock in its received byte while(++SFI counter<10): //wait for slave to receive the NSS slave signal
char ULTRASOUND_START_CODE = 0x96; char ULTRASOUND_DATA_CODE = ~0x96;	SPI_counter=0; SPI_In = SPIODAT;
char ULTRASOUND_READY_CODE = 0xAA; char ultrasound_data[4];	SPIInterrupted = 1; SPIF=0;
enum SPIStates (TransData, HaveCounter, NoCounter); char SPIInterrupted=1; ####################################)
<pre>*/urine outsour 0 // Plug in lookup table idata unsigned char lookup(64) = (23.27.29.39.43.45.46.51.53.54.57.</pre>	SPIInt() initializes the MCU to run SPI interrupts
58,71,75,77,78,83,86,89,90,92,99, 101,102,105,106,108,113,114,116,	Registers changed: IE, SPIOCN, SPIOCFG, SPIOCKR, EIE1, PRT2MX, PRT2CF
135,139,141,142,147,149,150,153, 154,156,163,165,166,169,170,172,	Resources used: P2.0 (SCK), P2.1 (MISO), P2.2 (MOSI), P2.3 (NSS)
177,178,180,184,195,197,198,201,202, 204,209,210,212,216,225,226,228,232);	Notes: *If SPI interrupts are set then there should at the very least
#define MUXPORT 1 #define MUXPORT 7	be a "SPIF = 0;" statement in the interrupt to prevent the MCU from hanging at the SPI interrupt vector (interrupt 6, memory location 0x0033).
// Mux stuff spit X = P1^7:	// SPI is enabled as MASTER since SPICN^1 = 1
sbit A0 = P1^2; sbit A1 = P1^1;	IE = 0x80; //enable global interrups
<pre>sbit A2 = P1^0; sbit EN1= P1^3;</pre>	SPIOCN = 0x03; //enable SPI in MASTER mode SPIOCFG = 0x07; //SPI Frame size = 8bits, this is the default setting
sbit EN2= P1^4; sbit EN3= P1^5;	SPIOCKR = 0x09; //this sets SCK = SYSCLK/20, maximum value is SYSCLK/20 EIE1 = 0x01; //Enable SPI interrupts
SDIC ENGE P1"b;	PRT2MX = 0x01; //SPI pins SCK,MISO,MOSI,MSS available on P2 pins P2.0,P2.1,P2.2,P2.3 re
sbit ASK = P2^1; sbit HAVE = P2^2:	//These are Cygnal dev pins 25,26,23,24 repsectively PRT2CP = 0x05: //SCK_MOSI are output: MISO_NSS are input
// Variables	PRT2CF $\Delta = -0x0A$; PRT3CF $= 0x01$; //Use P3.0 as the output for the slave's NSS line input
lu gyrox; lu gyroy;)
lu gyroz; lu accelx;	void SPITransmit(char SPI_TX_BYTE) (
lu accelz1;	'SPITransmit sends a byte over the SPI bus
1	2
Bogisters changed:	while(delay);
Begisters changed: STIOMA, SPI_counter Resources used: IIItrasoured MSG) 22.0. (STIK) 22.1. (MTG) 22.2. (MOST)	<pre>while(delay); while (f(00EXXXT 4 0x80)); // Mait unit] eccentral crystal has // startedental</pre>
Registers changed: BYIGAT, B1. South B1. SFI - 7.2.5 (Utranound_MSS), P2.0 (SKK), P2.1 (MISO), P2.2 (MOSI) SFI - 7.2.5 (Utranound_MSS), P2.0 (SKK), P2.1 (MISO), P2.2 (MOSI) Steam the low for a few synchk (volume before it is recognized as low by the slave	<pre>while(dslay); while (!00EXXX & 0x80); // Wait until external crystal has // started. 00EXTO = 0x85; // Waith to external oscillator 00EXTO = 0x85; // Waith to external oscillator; amble</pre>
Registers changed: SPICOMT, SPI_counter RespIT = 72.5 (Ultranound_RSS), P2.0 (EXCK), P2.1 (MISO), P2.2 (MOSI) Sockes: *465 must be low for a few syscik cycles before it is recognized as low by the slave	<pre>while(dslay); while(!(OCCCON & 0x80)); // Wair until external crystal has</pre>
<pre>Begisters changed: SPIGNAT, SPI_counter Besources used: Most MSS must be low for a few syncht cycles before it is recognized as low by the slave "MSS must be low for a few syncht cycles before it is recognized as low by the slave """""""""""""""""""""""""""""""""</pre>	<pre>while(dslay); while(l(OCCCN & Gx80)); // Value until external crystal has</pre>
Registers changed, Bridge, 97-19. Bridge, 97-19. Bridge, 97-19. Bridge, 197-19. Bridge, 197-19. Bridge	<pre>while(delay); while(!(OGEXCM 4 0x40)); // Wait until extends crystal has OGECTM = 0x4C; // Dutch to axternal oscillator OGECTM = 0x48; // Dutable internal oscillator; mable // maising clock detector. while (!(OGEXCM 4 0x40)); // Wait until extends crystal has OGECTM = 0x08; // Dutch to external oscillator</pre>
<pre>Begisters changed: EVIDAT, EFICAT, EFICATION SOLES: Soles: Soles: Soles: Soles: Soles: Soles: Soles: Soles: Soles: Soles: Soles: Soles: Soles: Sole: So</pre>	<pre>while(delay); while((SGEXCM & 0x80)); // Mait until external crystal has OBCICM = 0x0C; // distract. OBCICM = 0x88; // Disable internal oscillator; mable obtained internal oscillator; mable // missing clock detector. while((OBCXCM & 0x80)); // Wait until external oscillator oBCICM = 0x88; // Saturd. OBCICM = 0x88; // Saturd.) (Satup function void introvidi</pre>
<pre>Registers changed: SUIDAY, SPI_counter Boilday, SPI_counter Boilday, SPI_counter Suit = 72.3 (Utrasound_MSS), 92.0 (ESCH), 92.1 (MOSO), 92.2 (MOSI) Mose:</pre>	<pre>while(dslay); while(!clo2CXCN & 0x80)); // Wait until external crystal has</pre>
<pre>Registers changed DFIDES, DF_counter Facources used:</pre>	<pre>while(dolay); while(foldsy); while(foldsy); oscillations, constructions, construction</pre>
<pre>Registers changed: Best changed: Boil 72.5 (Utrasound.MSS), P2.0 (ESCH) P2.1 (MISO), P2.2 (MOSI) SOIL 72.5 (Utrasound.MSS), P2.0 (ESCH) P2.2 (MOSI) Works and the low for a few synchk cycles before it is recognized as low by the slave """""""""""""""""""""""""""""""""</pre>	<pre>while(delay); while((IGGEXCM & 0x80)); // Wait until excessed crystal has OBCTCM = 0x8C; // Dutable internal oscillator OBCTCM = 0x88; // Dutable internal oscillator; mable // missing clock detector. while((IGGEXCM & 0x80)); // Wait until extend oscillator OBCTCM = 0x88; // Dutable internal oscillator) // Setup function void init(void) { descinit(); AGR = 0; serial come (using timer 1) // FUNCM [= UMAPERMARK; // Yurn on UMAPE CHORN = TALAW, INFERDARK; // YURN = TALAW, INFERDARK; //</pre>
<pre>Registers changed: EVIDAT, EFI- EVIDAT, EFI- EVIDAT, EFI- EVIDAT, EFI- EVIDAT, EFI- EVIDAT, EFI- EVIDATION CONTACT A Set synch k-volume before it is recognized as low by the slave """""""""""""""""""""""""""""""""</pre>	<pre>while(dslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while((ldslay); while(ldslay); while((ldslay); while(ldslay); while((ldslay); while(ldslay)</pre>
<pre>Registers changed: BJIOAT, SPLCONTET BJIOAT, SPLCONTET BJI = 72.5 (Ultrasound_NSB), P2.0 (BCK), P2.1 (MISO), P2.2 (MOSI) Botes:</pre>	<pre>while(dolay); while(loSCKN & 0x80)); // Variant of systal has // started. OUCCCN = 0x8; // Started. while((IOSCKN & 0x80)); // With to external contlator while((IOSCKN & 0x80)); // Main unit external control detector.) Sector // Started.) OSCICH = 0x89; // Sarted.) OSCICH = 0x89; // Sarted.) // Started.) // Sarted.) // Sarted.)</pre>
<pre>Registers changed:</pre>	<pre>while(delay); while(flogEXXF4 6xd0)); dCCTCH = boB; dCCTCH = boB; while(flogEXXF4 6xd0)); while(flogEXXF4 6xd0)); while(flogEXXF4 6xd0)); while(flogEXXF4 6xd0)); f) while(flogEXXF4 6xd0)); f) while(flogEXXF4 6xd0)); f) while(flogEXXF4 6xd0); f) while(flogEXXF4 6xd0); f) setup function while(flogEXXF4 flogEXF1) f) setup function while(flogEXXF4 flogEXF1) f) f) setup function while(flogEXXF4 flogEXF1) f) setup function while(flogEXXF4 flogEXF1) f) f) f) f) setup function while(flogEXXF4 flogEXF1) f) f) f) f) f) f) f) f) f) f) f) f) f)</pre>
<pre>Megisters changed: STORE is STORE is BT = 72.5 (Ultrasound_MSS), P2.0 (MSCK), F2.1 (MISO), F2.2 (MOSI) BT = 72.5 (Ultrasound_MSS), P2.0 (MSCK), F2.1 (MISO), F2.2 (MOSI) Ultrasound_MSS-(r) Ultrasound_MSS-(r) Just to slave that a transmission is begining Ultrasound_MSS-(r) Just to slave to result to MSS_slave signal STORAT = SFI_RTME; //transmit to slave Vid ultrasound_Athet() { Ultrasound_Athet() {</pre>	<pre>while(delay); while((IOSEXC) & 0x80)); // Wait until excessed crystal has OBCICS = 0x80; // Datable internal oscillator OBCICS = 0x80; // Datable internal oscillator; mable // missing clock detector. while ((IOSEXCN & 0x80)); // Wait until external oscillator oBCICS = 0x80; // Datable internal crystal has OBCICS = 0x80; // Datable internal crystal has OBCICS = 0x80; // Datable internal oscillator // Setup function void introvida/ (oscinit(); ADE = 0; serial come (using timer 1); // FORCK [= UMATTRANELS; // Urus on UMAT CHCORD = TIZDME; // UBCITZON; // Setup Descriticities DBC = TIZDME; // INSTRUMELS; // Urus on UMAT CHCORD = TIZDME; // INSTRUMELS; // Urus on UMAT CHCORD = TIZDME; // INSTRUMELS; // Using output set, TISES1(.org.0x60); // Using Descriticities setCh band tate have //TT = 1; // ADE to send TIZDME_TIME(1); // Setup ADC (Torm up 10)</pre>
<pre>Registers changed: BYIGAT, EF. BOILDAT, EF. BOIL BOIL BOIL BOIL STATUS</pre>	<pre>while(dslay); while(lslo2CCN & 0x80)); // '/ 'started.</pre>
<pre>Registers changed: DFLDE, DFLCounter DFLDE, DFLCounter DFLDE, DFLCounter DFLDE, DFLCESS, D</pre>	<pre>while(dalay); while(idotay); wh</pre>
<pre>Registers changed: guiness changed: guiness used: Br = 7.3.5 (Utransond_NES), P2.0 (MECK), F2.1 (MESD), F2.2 (MEET) Messaw the low for a few speck (cycles before it is recognized as low by the slave """""""""""""""""""""""""""""</pre>	<pre>while(delay); while(iOdEXXM & God0)); // // Mait until encement crystal has GGCTGM = GodC; // Distable iferential coeflictor; manile // Distable iferential coeflictor; // Distable iferential coeflictor; // Distable iferential coeflictor; // Distable iferential coeflictor; // Setup function rold init(void) // Setup function rold = init(void) // Setup distable; // Yern on UART rold = UARTHBAT; // Yern on UART rold = UARTHBAT; // Yern on UART rold = UARTHBAT; // Setup Distable iferential coefficient rold = Virthes / TilBITELAGAD; // Setup Distable iferential coefficient rold = Virthes / TilBITELAGAD; // Setup Distable iferential coefficient rold = Virthes / TilBITELAGAD; // Setup Distable iferential rold = Virthes / TilBITELAGAD; // Setup Distable rold = Virthes and rold base ACCCCF = SAD16[0]; // Turn on ACC ACCCH = SAD16[0]; // Setup ADC ACCCH = SAD16[0]; // Setup ADC ACCCH = SAD16[0]; // Setup ADC ACCCk and prescaler BEFCCH = VD; // Internal reference // Setup BET; // Setup ADC ACCCH = SAD16[0]; // Setup ADC ACCCk and prescaler BEFCCH = VD; // Internal reference // Setup ADC ACCCH = SAD16[0]; // Setup ADC ACCCk and prescaler BEFCCH = VD; // Internal reference // Setup ADC ACCCH = SAD16[0]; // Setup ADC ACCCk and prescaler BEFCCH = VD; // Internal reference</pre>
<pre>Registers changed: BJCGAT, gitcounter BJCGAT, gitcounter BJCGAT, gitcounter BJCGAT, gitcounter BJCGAT, gitcounter BJCGAT, gitcounter Utrassond, MESC, // //signal to slave that a transmission is begining Utrassond, MESC, // //signal to slave that a transmission is begining Utrassond, MESC, // //signal to slave that a transmission is begining BJCGAT = SFI_RATE; //transmit to slave BJCGAT = SFI_RATE; //transmit to slave JUCASSOND = SFI_RATE; //transmit to slave JUCASSOND = SFI_RATE; //transmit to slave JUCASSOND _ STATUTE: Utrassond, MESC, // // // // // // // // // // // // //</pre>	<pre>while(dslay); while(()OCCCM & 0x80)); // variable OCCCM = 0x60; // started. OCCCM = 0x60; // Started. while(()OCCCM & 0x80)); // will will extrand crystal has // Started.) oCCCM = 0x80; // variable () oCCCM = 0x80; // will will extrand crystal has // Started.) oCCCM = 0x80; // will will extrand crystal has // Started.) oCCCM = 0x80; // will will extrand crystal has // Started.) of started.) of</pre>
<pre>Registers changed: DFLDE, DFLCCALL EFICIENT, DFLCCALL For Total Section Section For Total Section Section</pre>	<pre>while(dalay); while(loGCXN & God0)); // // started. OUTCT = 040; OUTCT = 040; // Started. // St</pre>
<pre>Registers changed: survey to the second se</pre>	<pre>while(dolay); while((locKXX & God0));</pre>
<pre>Registers changed: group is a construct Br = 7.1.5(Uitrasond_MSS), P2.0 (BCK), P2.1 (MISO), P2.2 (MOSI) Witrasond_MSS.() ///igit to slave that a transmission is begining Uitrasond_MSS.() ///igit to slave that a transmission is begining Brite.res.() //igit to slave to sective the MSL_slave signal Brite.res.() //igit to slave the ultrasond hoard Brite.res.() //igit to slave the ultrasond start juitle(IFFIInterrupted) //igit to data transfer between the ultrasond reserve bard and the Dister Brite.res.() //igit to be data transfer between the ultrasond hoard Brite.res.() //igit to be data transfer between the ultrasond reserve bard and the Dister bard Brite.res.() //igit to be data transfer between the ultrasond reserve bard and the Dister bard Brite.res.() //igit to be data transfer between the ultrasond reserve bard and the Dister bard Brite.res.() //igit to be data transfer between the ultrasond reserve bard and the Dister bard Brite.res.() //igit to data transfer between the ultrasond reserve bard and the Dister bard Brite.res.() //igit to data transfer between the ultrasond reserve bard and the Dister bard Brite.res.() //igit to data bard Brite.res.() //igit to data bard Brite.res.() //igit to data bard</pre>	<pre>while(delay); while(!COSEXCM & DoRD); // Wait until external crystal has OBCECM = DORC; // SetText OBCECM = DORC; // Delate internal costilator; manble // Delate internal costilator; manble // Delate internal costilator; manble // Delate internal costilator; manble // Delate internal costilator; manble // SetUp function void introvidi // SetUp function void intervoid void void void void void void void</pre>
<pre>sequence changed: DFUDE, DF_conter: Benurce: sed: factor: sed: factor:</pre>	<pre>while(dalay); while(loCCCM + 0.050); // started. OUTCTM = 0.06C; OUTCTM = 0.06C; OUTCTM = 0.06C; OUTCTM = 0.06C; OUTCTM = 0.06C; OUTCTM = 0.06C; Mailer (100000 + 0.080); while ((IOCCCM + 0.080)); // Wailer will external crystal has // Batted to external called // Started OUTCM = 0.06C; // Started OUTCM = 0.00C; // Started CUTCM = 0.00C; // The 1; // Started CUTCM = 0.00C; // The 0.00C CUTCM = 0.00C; // Started CUTCM = 0.00C</pre>
<pre>Registers changed:</pre>	<pre>while(dalay); while(idalay); wh</pre>
<pre>Registers changed: more used: Bit 7.3.5 (Utrassound,MSS), P2.0 (MSCK), F2.1 (MISO), P2.2 (MSCI) Bit 2.3.5 (Utrassound,MSS), P2.0 (MSCK), F2.1 (MISO), P2.2 (MSCI) Will (</pre>	<pre>while(delay); while(iOdEXXM & God0)); GCTCH = bodC; GCTCH = bodCC</pre>
<pre>Registers changed: BYIE 7.2.5 (Utrascond_MSS), F2.0 (ESCK), F2.1 (MISO), F2.2 (MSEI) BYIE 7.2.5 (Utrascond_MSS), F2.0 (ESCK), F2.1 (MISO), F2.2 (MSEI) Utrascond_MSEO, () // signal to slave that a transmission is begining Utrascond_MSEO, () // signal to slave to twenty the MSE_slave signal BYIE 7.2 (Itrascond_MSEO, ICR), F2.1 (MISO), F2.2 (MSEI) BYIE 7.2 (Itrascond_MSEO, ICR), F2.1 (MISO), F2.2 (MSEI) BYIE 7.2 (Itrascond_MSEO, F2.2 (Itrascond_MSEO, F2.2 (MSEI) BYIE 1.2 (Itrascond_MSEO, F2.2 (Itrascond_MSEO, F2.2 (MSEI)) BYIE 1.2 (Itrascond_MSEO, F2.2 (Itrascond_MSEO, F2.2 (MSEI))</pre>	<pre>while(dalay); while(dalay); while(l()COCCM & 0x80)); // '/ started. () function = 0x85;) // started. () // started.</pre>
<pre>Registers changed:</pre>	<pre>while(dalay); while(idotXX) & 0x80)); // // started. OUTCT = 0x80; OUTCT = 0x80; // Started. // Started. //</pre>
<pre>Registers changed: Benorces used: Benorces used: Beno</pre>	<pre>while(dolay); while(iOoEXXX & God0)); GCCCCH = bod2; GCCCH = bod2; GCCCH = bod2; GCCCH = bod3; while(iOoEXXX & God0)); while(iOoEXXX & God0)); f(idiate internal coclinator; million (GCCCH = bod3; f(idiate internal coclinator) f(idiate inte</pre>
<pre>Registers changed; more in it. Represent the low for a few speck (yelse before it is recognized as low by the slave """"""""""""""""""""""""""""""""""""</pre>	<pre>while(dalay); while(dalay); while(()(OCCCM & 0x80));</pre>
<pre>sqpinzes changed:</pre>	<pre>while(dalay); while(dalay); while(i()GOECN & 0.080)); // // started. OUTCT = 0.087; OUTCT = 0.087; // Dutch to atternal conjliant // Started. // Star</pre>
<pre>% provide the set of the set</pre>	<pre>while(dalay); while(idalay); wh</pre>
<pre>% provide the set of the set</pre>	<pre>while(dolay); while(lockXX & God0);; () charged GCCCH = bod; GCCCH = bod; GCCCH = bod; GCCCH = bod; GCCCH = bod; GCCCH = bods; () blue (locXXX & God0); while (locXXX & God0); () lock = lock =</pre>
<pre>sqpiness changed:</pre>	<pre>while(dalay); while(dalay); while(l()COCCM & 0x80)); () started. () S</pre>
<pre>Registers changed:</pre>	<pre>while(dalay); while(dalay); while(i(OOCXN + 0.000)); // // startsd. OOCTCH = 0.00; OOCTCH = 0.00; while((I(OOCXN + 0.000)); // // Maing(Dala Control & Contro</pre>
<code-block><code-block><code-block><code-block><code-block><code-block><code-block><code-block></code-block></code-block></code-block></code-block></code-block></code-block></code-block></code-block>	<pre>while(dalay); while(iOoEXXM & God0));</pre>
<code-block><code-block><code-block><code-block><code-block><code-block><code-block></code-block></code-block></code-block></code-block></code-block></code-block></code-block>	<code-block><code-block><code-block><code-block><code-block><code-block></code-block></code-block></code-block></code-block></code-block></code-block>

Figure F.3 Stack microcontroller code, part one



Figure F.4 Stack microcontroller code, part 2
The balanced byte code for wireless transmission is within both the basestation and the stack code. The basestation issues commands for each stack to send its complete packet of data for all the sensors in turn, and the basestation also includes a timestamp set by its on-board clock. In addition, during subject testing, it checked a pin connected to the BML TTL trigger, to detect when time zero was set on the BML system. These timing issues are summarized in Figure F.5.



Figure F.5 Timing issues, as controlled by the basestation

F.2 Matlab Code

This section contains code for Matlab m-files used in this thesis; code located beneath the name of the file, and the order generally corresponds to the order in which these files were mentioned in the main text of the thesis. The color green is used for comments, red for text within the code, and the rest is black. To obtain these files electronically, email the author at <sjm@alum.mit.edu>.

timetrunc.m

rage 1 12, 2004 6:44:57 AM	May 12, 2004 6:44:57 Ab
ction [datatruncL,datatruncR,truncinfo]=timetrunc(vers, fig, saveit)	fig=1; saveit=0;
IMETRUNC(version,fig,saveit) is used to truncate shoe data, using the trigger	elseif sizeargs<3
signal. A window pops up allowing the user to select the data files of interest (text files are expected). The data is graphed, and the user selects a	saveit=U; end
imepoint before the trigger. The program than truncates the data from	
ne start of the trigger, and re-graphs the data. The user can than ccept, or truncate the end.	tpos=18;
the second we have also enclose data for classics and also and of data	gyroz_pos=1;
his is so that this function can be easily updated if the data order is	fsrs_start=7;
changed or the data quantity increases.	pvdfs_start_pos=11; trigger_pos=17;
Version=1: Basetime - 16, trigger - 17, computertime - 18.	basetime_pos=16;
Names are gsvNNN where NNN is the three digit subject number.	namelength=6;
one shoe iine is expected.	splots=4;
Version=2: Basetime - 16, trigger - 17, computertime - 18. Names are gsvNNN where NNN is the three digit subject number.	else % splots=8;
Two shoe files are expected. GETSHOEDATORDER 1	splots=7;
Version=3: Basetime - 16, trigger - 17, computertime - 18,	end interval=0.00668;
Names are gsvNNN where NNN is the three digit subject number.	elseif vers==3
iwo shoe files ale expected. GhishobbwiokbER 2	gyroz_pos=1;
Version=4: Basetime - 17, trigger - 18, computertime - 21.	gyroy_pos=2;
Two shoe files are expected. GETSHOEDATORDER ?	pvdfs_start_pos=9;
	trigger_pos=17;
ig is used to specify the figures the data will be plotted on (fig).	namelength=6;
If no value is specified, the figures will be set to 1 and 2.	<pre>% splots=8; splots=7;</pre>
aveit is used to control whether the data is saved or not (if you just	interval=0.00668;
ant to look at data, you may not want to save it). If saveit=1, the output will be saved to a mat file in c:\sim\Research\Data\trunc data\ and the mat	elseit vers==4 tpos=21;
ile name will be the first three letters of the data name plus trunc (i.e.	gyrox_pos=1;
.f data is sjm103.txt, the truncated file will be sjm trunc.mat and the truncated lata will be named sjm103tr). Saveit=0 or no value entered means the data	gyroy_pos=2; fsrs start=11;
rill not be saved.	pvdfs_start_pos=9;
running triggertrunc with data which has previously been truncated will	trigger_pos=18; basetime pos=17;
ause the first version to be overwritten.]	namelength=6;
	<pre>splots=8; splots=7;</pre>
c)2003 Stacy J. Morris - sjm@alum.mit.edu	interval=0.00668;
	erse error('That version number is non-existant!');
: * * * * * * * * * * * * * * * * * * *	end
args=who;	<pre>% % % % % % % % % % % % % % % % % % %</pre>
zeargs, cor)-size(arrargs);	% about the name for later saving. It also checks to make sure the data is
sizeargs<1	% correct, then saves it to variable "rawdata".
\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 v 12, 2004 6:44:57 AM	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A
\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 y 12, 2004 6:44:57 AM verse=1	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 y 12, 2004 6:44:57 AM verse=1 disp(load shoe data.') % For data ease, in vers=1, data is just called left shoe dat v verse</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=exist(fileplace,'file');
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 y 12, 2004 6:44:57 AM verresi disp('Load shoe data.') % For data ease, in vers=1, data is just called left shoe dat </pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=exist(fileplace,'file'); if checker == 2
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=exist(fileplace,'file'); if checker = 2 error('Sboetrunc: Invalid file name!'); end
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 y 12, 2004 6:44:57 AM verse=1 disp('Load shoe data.') % For data ease, in vers=1, data is just called left shoe dat </pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Eage May 12, 2004 6:44:57 A checker=exist(fileplace, 'file'); if checker == 2 error('hosterunc: invalid file name!'); end fileplacek=fileplace;
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 y 12, 2004 6:44:57 AM were: introd shoe data.') % For data ease, in verse1, data is just called left shoe dat v eff verse2 disp('load LETT shoe data.') lease, fileloc] = uigetfile('*.txt'); epiacetfileloc filename);</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=exist(fileplace,'file'); if checker == 2 error('bhetuue: invalid file name!'); end end end end
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=exist(fileplace,'file'); if checker = 2 error'Abserrunc: Invalid file name!'); error'Abserrunc: Invalid file name!'); error'abserrunc: Invalid file name!'); error abserrunce: Invalid file nabserrunc
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 y 12, 2004 6:44:57 AM varsesi disp('Load later disp('Loa</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=exist(fileplace, 'file'); if checker = 2 error('hostrunc: Invalid file name!'); end fileplaceR=fileplace; end rawdataR=dimeas(fileplaceR, '); messageR=(filemane * 'hosded'); juppersageR5;
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Dage 3 y 12, 2004 6:44:57 AM vers=1 dd ('Lood shee data.') % For data ease, in vers=1, data is just called left shee dat </pre> disp('Lood LET' shee data.') lename, fileloc] = uigetfile(''.txt'); epice=fileloc filename); ss,colf]=size(filename().in(colf-4)) 'tr'); % spt rid of .txt (and preserves whe mane abbr2/Lefilename(l.in(colf-4)) 'trmc'); % use general identifies to atore tr	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=exist(fileplace, 'file'); if checker = 2 error('hhetunc: invalid file name!'); end fileplacek=fileplace, 'i); messages[cfilenamek 'loaded']; disp(messagek]; end disp(messagek];
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 f12, 2004 field: fie</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=2 efficience: invalid file name!'); efficience*fileplace; end minodataMedimand(fileplace, ''); minodataMedimand(fileplace, ''); end minodataMedimand(fileplace, ''); end minodataMedimand(filepl
<pre>Asjm\Research\Matlab\THESIS\timetrunc.m Page 3 / 12, 2004 6:44:57 AM // 22, 2004 6:44:57 AM // 24:57</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=sel encoder=sel(fileplace, 'file'); if checker == 2 encoder=sel(implace, 'file'); encoder=sel(implace, 'file); encoder=sel(implace, 'file); implemenses(); end table=sel(implace, 'loosed'); displaceses(); end
<pre>ksjm\Research\Matlab\THESIS\timetrunc.m Page 3 pl2, 2004 6:44:57 AM vers=1 disp('Load hoe data.') % For data ease, in vers=1, data is just called left shoe dat v eif vers=2 disp('Load LET shoe data.') laname, fileloc] = uigstfile('*.txt'); epiace=[fileloc filename]; ss_coff=size(filename); ename abbril=[filename(l,l:(coif-4)) 'tr']; % upst rid of .txt (and preserves whe v et ata data filename(l,l:namelength) '_rrunc']; % use general identifies to store tr v data ename triggerinfo=[filename(l,l:(coif-5)) 'filey']; </pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=exist(fileplace, 'file'); if checker == 2 end fileplaceR=fileplace; end rawdstaW=dimerad(fileplaceR, '); messageM=(filemareR ')oaded'); disp(messageN; end this section plots three streams of data to the figure specified by fig
<pre>Asjm\Research\Matlab\THESIS\timetrunc.m Page 3 /12, 2004 6:44:57 AM vers=1 disp('Load hee data.') % For data ease, in versel, data is just called left shoe dat </pre> eff vers>2 disp('Load LEFT shoe data.') lename, fileloc filename); ename abbrllefilename(1:(colf-4)) 'tr']; % yet rid of .txt (and preserves whe f Lor D) filename(1:(colf-4)) '_trunc']; % use general identifies to store tr data	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=2 efficiences(fileplace,'file'); fileplacesFileplace; end rawdatabcdimread(fileplaces; end rawdatabcdimread(fileplaces; end rawdatabcdimread(fileplaces; end rawdatabcdimread(fileplaces; end rawdatabcdimread(fileplaces; end rawdatabcdimread(fileplaces; end rawdatabcdimread(fileplaces; end timetructure; end end end end end end end end
<pre>Asjm\Research\Matlab\THESIS\timetrunc.m Page 3</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checkerwaitffileplace, 'file'); if checker -= 2 error('sheetune: invalid file name('); end rrodstaN=dimesa(fileplacek, '); massage=[filename*.ioaded]; and imp(massage); and timetrisecondition(rawdata(); poos); grout-medita(i,group.poo); grout-medita(i,gro
<pre>Najm\Research\Matlab\THESIS\timetrunc.m Page 3 12, 2004 6:44:57 AM 12, 2004 6:44:57 AM vers=1 disp('Load thee data.') % For data ease, in vers=1, data is just called left shoe dat v entivers=2 disp('Load LETT shoe data.') lename, fileloc] = uigstfile(''.txt'); piace=[fileloc filename]; sa_coff]=size(inname); ename subtlet[filename(l,1:(colf-4)) 'tr']; % upst rid of .txt (and preserves whe v r t of n) ename subtlet[filename(l,1:(colf-4)) 'tr']; % upst rid of .txt (and preserves whe v r t of n) ename subtlet[filename(l,1:(colf-5)) 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename subtlet[filename(l,1:(colf-5)) 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)) 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)) 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)) 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)) 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-4)) 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)) 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)) 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)) 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)] 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)] 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)] 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)] 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-5)] 'truc']; % upst rid of .txt (and preserves whe v r t of n) ename triggerinfo=[filename(l,1:(colf-</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=stiffieplace, 'file'); if checker == 2 end fileplaces=fileplace; end rawdataM=dimeas(fileplace); ing messages[filemake* ') cade(); disp(messageN; end ******* * This section plots three streams of data to the figure specified by fig ******* * This section plots three streams of data to the figure specified by fig *******
<pre>Asjm\Research\Matlab\THESIS\timetrunc.m Page 3 y 12, 2004 6:44:57 AM versed disp('tood deno data.') % For data ease, in versel, data is just called left shoe dat </pre> ere eff(verse2 disp('tood dEff shoe data.') lename, fileloc] = uigetfile('*.txt'); epidace(fileloc filename); ename abbrlefilename(i:(colf-4)) 'tr'); % get rid of .txt (and preserves whe r t. or %) ename abbrlefilename(i.l:(colf-4)) 'tr'); % use general identifies to store tr data ename triggerinfo=(filename(i.l:(colf-5)) '_fing'); filename.triggerinfo=(filename(i.l:(colf-5)) '_fing'); filename.triggerinfo=(filename(i.l:(colf-5)) '_fing'); checker - 2 error('Sheetron: Invalid file name!');	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=sel:[mail:splace, 'file'); if checker = 2 end resegnet[filesame.'looded'); displacesed end resegnet[filesame.'looded'); displacesed end timeticsetion[the streams of data to the figure specified by fig timeticsedit(:gyros_pos); first=readtal(:gyros
<pre>Nsjm\Research\Matlab\THESIS\timetrunc.m Page 3 /12, 2004 6:44:57 AM vers=1 ddf'tasd shoe data.') % For data ease, in vers=1, data is just called left shoe dat </pre> disp('load LET shoe data.') lename, fileloc] = uigefile('*.txt'); epicarc(filename); ss,coll]wsire(filename); ss,coll]wsire(filename); ss,coll]wsire(filename); ri.ormolline(lineme); filename.thriggerinfo=(filename(1,1:colf-4)) 'tr'); % use general identifies to store tr data ename_thriggerinfo=(filename(1,1:colf-5)) '_fleg'); filename!(rcisjm\ceaserActiveLtarverce_data\' filename_abbr2L; ckerewist(fileplace,'file'); detal=data='';	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=sl(fileplace, file'); if checker = 2 error('bheetune: Invalid file name!'); end rawdsaBwdInesa(fileplacek, '); messages[(filenamek, 'logical'); disp(messages); end timetrimecondition(rawdstal(,tppa)); time specified by fig timetrimecondition(rawdstal(,tppa)); time vector gyroptrawdstal(,tprog.pos); time vector gyroptrawdstal(,tprog.pos); time vector gyroptrawdstal(,tprog.pos); timetrimecondit, timetrimecondity, the start of the s
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Ergg 3 y 12, 2004 6:44:57 AM yrsr=1 disp('Load thee data.') % For data ease, in vers=1, data is just called left shoe dat v erg vers=2 disp('Load LETT shoe data.') lename, fileloc] = uigetfile('*,txt'); ename abbrll=('ilename(1,1:(colf-4)) *tr'); % got rid of .txt (and preserves whe v r t or a) scoll;=sic(ilename(1,1:(colf-4)) *tr'); % use general identifies to store tr v disp'('Shoetrunc: Invalid file name!'); checker == 2 error('Shoetrunc: Invalid file name!'); asgul=['disubclic[ilename']; asgul=['disubclic[ilename</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=stif(fileplace, file'); if checker = 2 effor('Shoetrunc: Invalid file namet'); effices eff
<pre>Asjm\Research\Matlab\THESIS\timetrunc.m Page 3 y 12, 2004 6:44:57 AM versei disp('load bab data.') interverse disp('load lift' show data ease, in verse1, data is just called left show dat </pre> eff verse2 disp('load Lift' show data.') interverse disp('load Lift' show data.') iss_collpaine(file) = uigetfile('*.txt'); epiace(file) coll filename); manag abbrlet[ilename],1r(colf-4)) 'tr']; 'spt rid of .txt (and preserves whe erson disp('load Lift' show data.') iss_collpaine(file) = uigetfile(''.txt'); epiace(file) = uigetfile(''.txt'); ename_stabulet[ilename],1r(colf-4)) 'tr']; 'spt rid of .txt (and preserves whe erson ('Monterverse') filename['c:'typ/verserch/dita/trunc_dsta\' filename_abbr2L]; cherker = 2 error('Monterverse'); mataldinemead(fileplace, '); mataldinemead(fileplace, '); mataldinemead(fileplace, '); mataldinemead(file); p(masagel); 'tr' to T to T tettettettettettettettettettettettettet	C:\gjm\Research\Matlab\THESIS\timetrunc.m Eage May 12, 2004 6:44:57 A checker=sel(fileplace, 'file'); if checker == 2 error('Shoetuue: Invalid file name('); end randstabledimesd(fileplace); end randstabledimesd(fileplace); end if displaces/fileplace); end if displaces/fileplace); end if this setion plots thes streams of data to the figure specified by fig titlettittetttettittettittettitttettittetttitttettittettitt
<pre>Asjm\Research\Matlab\THESIS\timetrunc.m Dage 3 /12, 2004 6:44:57 AM versmi dig'toad there data.') % For data ease, in versmi, data is just called left shoe dat v eig versmi dig'toad LET shoe data.') lename, fileloc] = ujeftile('*.txt'); epiace(fileloce filename); ss.colf]=size(filename(i,l:(colf-4)) 'tr'); % upst rid of .txt (and preserves whe v ename abb/ll=[liename(i,l:namelength) '_runc']; % upst rid of .txt (and preserves whe v data ename abb/ll=[liename(i,l:namelength) '_runc']; % upst rid of .txt (and preserves whe v ename abb/ll=[liename(i,l:namelength) '_runc']; % upst rid of .txt (and preserves whe v ename abb/ll=[liename(i,l:namelength) '_runc']; % upst rid of .txt (and preserves whe v ename abb/ll=[liename(i,l:namelength) '_runc']; % upst rid of .txt (and preserves whe v ename abb/ll=[liename(i,l:namelength) '_runc']; % ename abb/ll=[liename(i,lienamelength) '_runc']; % ename abb/ll=[liename(i,lienamelength) '_runc']; % ename abb/ll=[liename(i,lienamelength) '_runc']; % ename abb/ll=[liename(i,lienamelength); % ename(i,lienamelength) '_runc']; % ename(i,lienamelength); % ename(</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checkeresit(fileplace, 'file'); if checker == 2 end fileplaces=fileplace; end revektaMedineeas(fileplace, '); massage=[filenaes* 'loaded']; disp(massage=[filenaes* 'loaded']; disp(massage=[filenaes* 'loaded']; disp(massage=[filenaes* 'loaded']; end timet=timecondition(ravdata[(,tpop)); time vector gyropl-ravdata[(,gyrop_pon); time vector gyropl-ravdata[(,trigger_pon); time vector gyropl-ravdata[(,trigger_pon); time vector gyropl-ravdata[(,trigger_pon); time vector gyropl-ravdata[(,trigger_pon); time vector gyropl-ravdata[(,trigger_pon); time vector groupl-ravdata[(,trigger_pon); time vector groupl-ravdata[(,trigger_pon); time vector groupl-ravdata[(,trigger_pon); time vector file_stata_stata]; time part
<pre>Asjm\Research\Matlab\THESIS\timetrunc.m Page 3 /12, 2004 6:44:57 AM /22, 2004 6:44:57 AM /24:57 AM /24:57 AM /25 AM</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checkeresit(fileplace,'file'); if checker = 2 error'?Botron: Invalid file name!'; error'?Botron: Invalid file name!'; error'Botron: In
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 y 12, 2004 6:44:57 AM were if verses if vers</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A c fecker=sci fileplace, file"); if checker == 2 error("Bheetune; invalid file name("); end rundstaN=dimena(fileplace); end rundstaN=dimena(fileplace); ind indpressayB0); end tittett t This section plots three streams of data to the figure specified by fig tittett t This section plots three streams of data to the figure specified by fig tittett t This section plots three streams of data to the figure specified by fig tittett t This section plots three streams of data to the figure specified by fig tittett t This section plots three streams of data to the figure specified by fig tittett t This section plots three streams of data to the figure specified by fig tittett t This section plots three streams of data to the figure specified by fig tittett t This section plots three streams of data to the figure specified by fig t tigepcreadatal(; firsg start(firsg start_pose)); t tigepcreadatal(; firsg start(firsg start_pose); t tigepcreadatal(; firsg start(firsg start_pose); t tigepcreadatal(; firsg start(firsg start_pose); t tigepcreadatal(; firsg start_pose); t tigepcreadatal
<pre>kyjm\Research\Matlab\THESIS\timetrunc.m</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checkeresit(fileplace, 'file'); if checker == 2 end fileplace*fileplace; end rudstaW-dimensi(fileplace); in passages[filename* ') cade rudstaW-dimensi(fileplace); end * This section plots three streams of data to the figure geneticed by fig * This section plots three streams of data to the figure geneticed by fig * This section plots three streams of data to the figure geneticed by fig * This section plots three streams of data to the figure geneticed by fig * This section plots three streams of data to the figure geneticed by fig * This section plots three streams of data to the figure geneticed by fig * This section plots three streams of data to the figure geneticed by fig * time-timecondition(raw/tadi/t,tpon))
<pre>Najm\Research\Matlab\THESIS\timetrunc.m Page 3 y12, 2004 6:44:57 AM versei disp('Load abso data.') eff(verse2) disp('Load LET shoe data.') ename, fileloc] = uigetfile('*.txt'); epitase(fileloc filename); ename_abbrlet[inename].in(colf-4)) 'tr'}; 'spt rid of .txt (and preserves whe v r L or 8) ename_abbrlet[inename[1.1:(colf-4)] 'tr']; 'spt rid of .txt (and preserves whe v data ename_abbrlet[inename[1.1:(colf-5)) '_fing']; filename[.'c:\symbol{shoe}.''); disp('Load LET'); ename_triggerinfo=(filename[1.1:(colf-5)) '_fing']; filename[.'c:\symbol{shoe}.''); dealefilename[.'c:\symbol{shoe}.''); dealefilename[.'c:\symbol{shoe}.''); dealefilename[.'c:\symbol{shoe}.''); dealefilename[.'c:\symbol{shoe}.''); dealefilename[.'c:\symbol{shoe}.''); dealefilename[.'c:\symbol{shoe}.''); dealefilename[.'c:\symbol{shoe}.''); dealefilename[.'c:\symbol{shoe}.''); dealefilename[.'c:\symbol{shoe}.''); filename[.c:['signeee(1.c:df-6)'''; filename[.c:['signeee(1.c:df-6)'</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=extifileplace, 'file'); if checker = 2 error('fhoeture: invalid file name!'); if iplicas=fileplace; end remediabedirised(fileplacek, '); adaptediabedirised(fileplacek, '); profilerisediat(i, profile inter, pagedirised inter, pagedirisediabedirisediabedirised inter, pagedirisediabedirised inter, pagedirisediabedirised inter, pagedirised inter, pa
<pre>Nsjm\Research\Matlab\THESIS\timetrunc.m Page 3 y 12, 2004 6:44:57 AM verse1 dig('Lood shee data.') \ For data ease, in verse1, data is just called left shee dat </pre> dig('Lood inf' of the data.') lename, filelog] = uigetfile(''.txt'); epicer(file)confiname); ss.pcil[prime(filename(l.):(colf-4)) 'tr']; tget rid of .txt (and preserves whe r Lo Ta ename_abbr2le[filename(l.):(colf-4)) 'tr']; tget rid of .txt (and preserves whe error('Sheetring' file); ckerrexist(fileplace, 'file); filename!'incend('); distoftingender, 'not in the test test test test test test test	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=sif(filplace, file'); if checker = 2 error('sheetrun: invalid file name('); end rendetaBeturesa(fileplace); end rendetaBeturesa(fileplace); injopessage[filename*'); displaceS=fileplace; end trissection plots three streams of data to the figure specified by fig timet=timecondition(rawdata(:,tpon)); time vector gyrout=rawdata(:,tping=rpon); time vector gyrout=rawdata(:,tping=rpon); time vector gyrout=rawdata(:,trigger_pon); time vector gyrout=rawdata(:,trigger_pon); if gyrout=rawdata(:,trigger_pon); if gyrout=rawdata(:,tringger_pon); if gyrout=rawdata(:,tr
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 y12, 2004 6:44:57 AM versei disp('Load bhc data.') * For data ease, in versel, data is just called left shoe dat < set set if verse2 disp('Load LET shoe data.') lename, fileloo: = uigetfile('*.txt'); epidese(fileloo filename); ename abble[tilename(1.i(colf-4)) *tr']; % thet rid of .txt (and preserves whe < rid of ansa eabble[tilename(1.i(colf-4)) *tr']; % thet rid of .txt (and preserves whe < rid of ansa eabble[tilename(1.i(colf-5)) *_fileg']; filename[tri][tilename[t,1:(colf-5)) *_fileg']; filename[tri][tilename[t,1:(colf-5)) *_fileg']; filename[tri][tilename[t,1:(colf-5)) *_fileg']; filename[tri][tilename[t,1:(colf-5)) *_fileg']; filename[tri][tilename[t,1:(colf-5)) *_fileg']; filename[tri][tilename[t,1:(colf-6)]; filename[t,1:(colf-6)]; filename[tri][tilename[t,1:(colf-6)]; filename[tri][tilename[t,1:(colf-6)]; filename[t,1:(colf-6)]; filename[t,1:(colf-6</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checkerestif(fileplace, 'file'); if checker -= 2 entror('Bhoetrunc: Invalid file name'); end rundstatedInneas(fileplace, 'i); massagke[filenamek' : loaded]; displacek*fileplace; end * This section plots three streams of data to the figure specified by fig itintiation is the streams of data to the figure specified by fig itintiation is the streams of data to the figure specified by fig itintiation is the streams of data to the figure specified by fig itintiation is the streams of data to the figure specified by fig itintiation is the streams of data to the figure specified by fig itintiation is the streams of data to the figure specified by fig itingscradual(:,ryory.pos); from the streams of data to the figure specified by fig itingscradual(:,ryory.pos); from the streams of data to the figure specified by fig itingscradual(:,ryory.pos); from the streams of data to the figure specified by fig itingscradual(:,ryory.pos); from the streams of data to the figure specified by fig itingscradual(:,ryory.pos); from the streams of data to the figure specified by fig itingscradual(:,ryory.pos); from the streams of data to the figure specified by fig itingscradual(:,ryory.pos); from the streams of data to the figure specified by fig dataplotters(form(:,r),rime,fig, 'raw data = isft',splots,1, 'b', ') dataplotters(form(:,r),rime,fig, 'raw data = isft',splots,1, 'b', ') dataplotters(form(:,r),rime,fig, 'ray,rimet,r); ', plots,1, 'b', ') dataplotters(form(:,r),rimet,fig, 'ray,r); ') dataplotters(form(:,r),rimet,fig, 'ray,r); ') dataplotters(form(:,r),rimet,fig, 'ray,r); ') dataplotters(form(:,r),rimet,fig, 'ray,r); ') dataplotters(form(:,r),rimet,fig, 'ray,r); ') dataplotters(form(:,r),r); ') streams of the streams of the strea
<pre>Asjm\Research\Matlab\THESIS\timetrunc.m Page 3 y12, 2004 6:44:57 AM wiczwi disp('load ibe data.') iname, fileloc] = uigetfile('*,txt'); ginzen[fileions[iname]; aname_shiple(fileiname]; aname_shiple(fileiname); aname_shiple(fileiname); aname_shiple(fileiname); aname_shiple(fileiname); aname_shiple(fileiname); fileiname[iname]</pre>	C:\gjm\Besearch\Matlab\THESIS\timetrunc.m Eage May 12, 2004 6:44:57 A c decker=satifilipiaes, "file"); if checker == 2 error("bhetrun: invalid file name("); end redetabledimesd(filepiaes, "); satisgape(filenesdetilepiaes; end invadeabledimesd(filepiaes, "); satisgape(filenesdetilepiaes; end invadeabledimesd(filepiaes, "); satisgape(filenesdetilepiaes; end invadeabledimesd(filepiaes; end invadeabledimesd(filepiaes; end invadeabledimesd(filepiaes; end invadeabledimesd(filepiaes; end invadeabledimesd(filepiaes; end invadeabledimesd(filepiaes; end invadeabledimesd(filepiaes; end invadeabledimesd(filepiaes; end figure(filepicesd); if wrs>cd trigspoc=s; end figure(filepicesd); if wrs>cd trigspoc=s; end figure(filepic); dtaplotter(figure(i,)); dtaplott
<pre>kyjn/Research/Matlab/THESIS/timetrunc.m Dage 3 y 12, 2004 6:44:57 AM vers=1 dig('load shoe data.') * For data ease, in vers=1, data is just called left shoe dat </pre> isi for vers>-2 dig('load LHT' shoe data.') i linamae, filelo] = uigefile(''.txt'); esplace(file)(filename(i,l:(colf-4)) 'tr'); ss.colf]=sis((filename(i,l:namelength) '_runc'); % use general identifies to store tr isiname_abb/21=[filename(i,l:(colf-5)) 'flag']; filename('cityim/ceaserth/datk/trunc_data\' filename_abb/21; checkr=-2 erro('baberone: invalid file name'); if displace(file)(filename(i,l:(colf-4)) 'tr'); % put rid of .txt filename('cityim/ceaserth/datk/trunc_data\' filename_abb/21; checkr=-2 filep_aos(file)[see('); filename('cityim/ceaserth/datk/trunc_data\' filename_abb/21; filename('cityim/ceaserth/datk(''); filenames('cityim/ceaserth/datk(''); filenames('cityim/ceaserth/datk(''); filenames('cityim/ceaserth/datk(''); filenames('cityim/ceaserth/datk(''); filenames(''); filenames(C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A checker=sif(filplace, file'); if checker == 2 end filplaces=filplace; end revektaMedInesa(filplace, '); sequencesses revektaMedInesa(filplace, '); inplaces=filplace; end 'Trissection plots three streams of data to the figure specified by fig time!=time:ondition(rawdtal(;,tpop)); time vector gyrop!-rawdtal(;,tpro;po); gyrop!-rawdtal(;,tpro;po); gyrop!-rawdtal(;,tpro;po); firstFirstMedInesa(filplace, time; time; time.time time!=time:ondition(rawdtal(;,tpro)); time vector gyrop!-rawdtal(;,tpro;po); firstFirstMedInesa(filplace, time; time; time; time time!=time:ondition(rawdtal(;,tpro)); time firstFirstMedInesa(filplace, time, filp(; time, time; time); firstFirstMedInesa(filplace, time); firstFirstMedInesa(filplace, time, filp(; time, time; time); firstFirstMedInesa(filplace, time, filp(; time); time; time); firstFirstMedInesa(filplace, time, filp(; time); time); firstFirstMedInesa(filplace, time, filp(; time); time); firstFirstMedInesa(filplace, time); firstFirstMedInesa(filplace, time); firstFirstMedInesa(filplace, time); firstFirstMedInesa(filplace, time); firstMedInesa(filplace, time); firstMed
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 y12, 2004 6:44:57 AM disp('load shoe data.') disp('load shoe data.') if('load iET shoe data.') i</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timetrunc.m May 12, 2004 6:44:57 A cfebcter=2 creating cfebcter=2 creating cfebcter=creating cfebcter=cfebcter=cfebcter=c</pre>
<pre>\sin\Research\Matlab\THESIS\timetrunc.m Page 3 y12, 2004 c:44:57 AM verse: verse:</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 204 6:44:57 A checker=satificiplace, file'); if checker = 2 error('theturn: invalid file name('); end rendesRetIneread(fileplaces, '); magesage(filename('); displaces=fileplace; end 'rendesRetIneread(fileplaces, '); magesage(file); displaces=fileplace; end 'rendesRetIneread(fileplaces, '); magesage(file); displaces=fileplace; end 'rendesRetIneread(fileplaces, '); magesage(file); end 'theturnet(file); fileplaces=fileplace; end 'theturnet(file); fileplaces=fileplace; end 'trippediati(;fire_place); firef=rendesI(;fire_place; file); firef=rendesI(;fire_place; file); firef=rendesI(;fire_place; file); firef=rendesI(;fire_place; file); firef=rendesI(;fire_place; file); firef=rendesI(;fire, file); firef=rendesI(;fire, file); firef=rendesI(;firef, file); firef=rendesI(;firef, file); firef=rendesI(;firef, file); firef=rendesI(;firef, file); firef=rendesI(;firef, file); firef=rendesI(;firef, file); firef=rendesI(;firef, file); firef=rendesI(;firef=file
<pre>\sjm\Research\Matlab\THESIS\timetrunc.m Page 3 y12, 2004 6:44:57 AM varsei disp('Load shoe data.') % For data ease, in versel, data is just called left shoe dat < set if verse-2 disp('Load LEFT shoe data.') lename, fileloc = uigetfile('*.txt'); epshare(fileiname); ename_shorle(filename); ename_shorle(filename); ename_shorle(filename); filename_triggerinfo=(filename(1,1:(colf-4)) 'tr'); % you rid of .txt (and preserves whe < set cata cata in the state of the state</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timetrunc.m May 12, 204 c::44:57 At checker=sat(filplace, file'); if checker == 2</pre>
<pre>Name and the second secon</pre>	<pre>C:\gjm\Besearch\Matlab\THESIS\timetrunc.m May 12, 2004 c:44:57 A checker=satifileplace, "file"); if checker==2 cord("Shoetune: invalid file name!"); end cndexe=satifileplace; end rendetabedimesa(fileplace, "); end fileplaces/fileplace; end filep</pre>
<pre>Ny 12, 2004 Page 3 (14,57 AM) Y 12, 2004 (14,57 AM) Y 12, 2004 (14,57 AM) Y 12, 2004 (14,57 AM) Y 12, 2004 (14,57 AM) Y 14, 200</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timetrunc.m May 12, 204 c:44:57 A checker=wait(filplace, "file"); if checker = 2 end filplace*filplace; end end end end end end end end end end</pre>
<pre>Najm\Research\Matlab\THESIS\timetrunc.m Page 3 y12,2004 6:44:57.AM virused disp('load shoe dat.') % For data ease, in versel, data is just called left shoe dat < wres if verse-2 disp('load LET' shoe dat.') lename, fileloc] = uigetfile(''.ttt'); epicer(file)ease(1,1:(colf-4)) 'tr'); % yet rid of .tt (and preserves whe < mresease_abs/l=[diename]; ename_thile_colfiename]; ename_thile_colfiename[1,1:(colf-5)) 'fing']; filename_t'c:\symptreserch\data\trunc_data\' filename_abs/21; checkerse1; filepiace, 'filesease(1,1:(colf-6)) 'tr'); % yet rid of .tt ename_thile_colfiename[1,1:(colf-5)) 'fing']; filename_t'c:\symptreserch\data\trunc_data\' filename_abs/21; checkerse1; filepiace, 'filesease(1,1:(colf-6)) 'tr'); % yet rid of .tt filename_t'c:\symptreserch\data\trunc_data\' filename_abs/21; checkerse1; filepiace, 'filesease(1,1:(colf-6)) 'tr'); % yet rid of .tt filename_t'c:\symptreserch\data\trunc_data\' filename_abs/21; checkerse1; filepiaces(1,1:(colf-6)) 'tr'); % yet rid of .tt filenameset[lenames(1,1:(colf-6)) 'tr'); % yet rid of .tt filenameset[lenameset[lenames(1,1:(colf-6)] 'tr'); % yet rid of .tt filenameset[lenameset[lenames(1,1:(colf-6)] 'tr'); % yet rid of .tt filenameset[lena</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timetrunc.m May 12, 2004</pre>
<pre>Nsjm\Research\Matlab\THESIS\timetrunc.m</pre>	C:\gjm\Besearch\Matlab\THESIS\timetrunc.m Eage May 12, 204 6:44:57 A cfstextrain(fileplace, 'file'); if checker = 2 error('fheetune: invalid file name('); eff rrodstabutinesd(fileplace); eff rrodstabutinesd(fileplace); eff rrodstabutinesd(fileplace); eff injpearsyst); eff timetrisecondition(raduate; !); public rodstabutinesd(fileplace); if This section plots three streams of data to the figure specified by fig timetrisecondition(raduate(; trops); public rodstabutinesd(fileplace); if restringenors); if restringenors(if riggerEff); eff dtsplotterf(tyropt.timed,fig,'raw data = iff',splots,li,'b', 1) dtsplotterf(tyropt.timed,fig,'raw data = iff',splots,li,'b', 1) timestimecodition(rawdatk(;tyros)); time wetter proti-maduatk(;tyros_startifies_start)); protis-maduatk(;tyros_startifies_start); if wetter i a adde
<pre>Najm\Research\Matlab\THESIS\timetrunc.m Page 3 /12, 2004 6:44:57 AM /22, 2004 6:44:57 AM /24:57 AM /24:57 AM /24:57 AM /25:00 /25:</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timetrunc.m May 12, 204 c::44:57 A checker=sif(iplace, file'); if checker == 2</pre>

Т

timetrunc.m, cont.

ay 12, 2004 6:44:57 AM	May 12, 2004 6:44:57 AM
<pre>dataplotterR(fsrsR(s,2),timeR,fig,'',aplots,1,f-1,'k', 1) dataplotterR(fsrsR(s,3),timeR,fig,'',aplots,1,f-1,'g', 1) dataplotterR(fsrsR(s,4),timeR,fig,'',aplots,1,f-1,'r', 1) dataplotterR(fsrsR(red),timeR,fig,'',aplots,1,f-1,'b', 1)</pre>	end if goodornot=999;return;end states to the states to the
<pre>dataplotterR(pvdfsR(:,2),timeR,fig,'',splots,1,7-1,'t', 1) dataplotterR(triggerR,timeR,fig,'',splots,1,splots,'b', 1)</pre>	<pre>% This section finds the start & end time points. starttimeptL=findtimepoints(time1, -1); endtimeptL=findtimepoints(time2,time1,1);</pre>
rowtR=length(timeR); nd & & & & & & & & & & & & & & & & & & &	<pre>if vers>=2 starttimeptR=findtimepoints(time1,timeR,-1);</pre>
This section prompts the user to enter the start times - i.e. just before trigger $t_{i} = t_{i} = t_{i} = t_{i}$	<pre>endtimeptR=findtimepoints(time2,timeR,l); end dim(t, t);</pre>
t starts walking/in '); (tisal=900;return, and las2=loguet()'n (Mere would you like the data to end?/n (pick a point just before subject \checkmark current in ');	<pre>disp('</pre>
op=1.1;	iter=1;
ter=1; hile iter<50 if time2<=time1	bottom=1; top=1.2; while iter<6
<pre>time2=time1+10; end</pre>	<pre>if iter==1 trigstart=time1; trigend=time2;</pre>
<pre>for i=!splots subplot(splots,i) axis([timel time2 bottom top]); end</pre>	<pre>else axis([time1 time2 bottom top]); time3=input(' Enter the time just before the trigger goes high. \n '); trigstarttime3-1.5;</pre>
<pre>goodornot=input('\n Is this what you wanted? \n To change start time ONLY, enter 1, t < change end time ONLY, enter 2, to change both enter 3, \n and if done, enter 0 (10 to z </pre>	trigend=time3+1.5; end
<pre>om out, -10 to zoom in.\n '); if goodcrnotm=999;return;red if goodcrnotm=0 if goodcrnotm=0 if indications(); if ind</pre>	<pre>for i=1:splots subplot(splots,1,i) axis([trigstart trigend bottom top]); cont</pre>
disp('Try again, start time must be after the beginning of both L and R times 🖌	<pre>tL1=findtimepoints(trigstart,timeL,-1);</pre>
<pre>elseif time2>timeL(rowtL) (vers>=2 & time2>timeR(rowtR)) disp('Try again, end time must be before the end of both L and R times!') else</pre>	<pre>tL2=tindtImepoints(trigend,timeL,1); [tryL,toss]=find(triggerL(tL1:tL2,1)==1);</pre>
break end aleaif geodernot=1	<pre>if vers>=2 tRl=findtimepoints(trigstart,timeR,-1); tRl=findtimepoints(trigend timeR, 1);</pre>
<pre>timel=input('\n Where would you like the data to start?\n '); elseif goodornot==2</pre>	<pre>(tryR, toss)=find(triggerR(tR1:tR2,1)==1); end</pre>
<pre>time2=input('\n Where would you like the data to end?\n '); elseif goodcrote=3 time1=input('\n Where would you like the data to start?\n '); time2=input('\n Where would you like the data to end?\n ');</pre>	<pre>if not(isempty(tryL)) & (vers==1 (vers>=2 & not(isempty(tryR)))) subplot(splots,l,trigspot) hold on; not(rime(til+ryt(l)-1,l),trigse(t(t)+tryt(l)-1,l),r');</pre>
<pre>elseif goodornot==10 bottom=-0.1; top=1.1;</pre>	<pre>text(timeL(tL1+tryL(1)-1,1),0.5,num2str(tL1+tryL(1)-1)); if vers>=2</pre>
<pre>elseif goodornot===-10 bottom=0.3; top=0.7; end</pre>	<pre>subplot(splots,1,splots); hold on; plot(timeR(tRl+tryR(1)-1,1),triggerR(tRl+tryR(1)-1,1),'r.') text(timeR(tRl+tryR(1)-1,1),0.5,num2str(tRl+tryR(1)-1));</pre>
:\sjm\Research\Matlab\THESIS\timetrunc.m Page 7 ay 12, 2004 6:44:57 AM	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 AJ
:\sjm\Research\Matlab\THESIS\timetrunc.m Page 7 ay 12, 2004 6:44:57 AM if iter=1 goodornot=0; else	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 AV for i=1:5 if triginfo(i,3)==1
<pre>:\sjm\Mesearch\Matlab\THESIS\timetrunc.m Page 7 ay 12, 2004 6:44:57 AM if iter=1 goodsrot=0; elsgoodsrot=input(\n Is this what you wanted ? \n To find trigger points manus & ye end end</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 AU for i=1:5 i slgnfs(i,3)==1 i slgnfs(i,3)==1 i slgnfs(i,5)100007 autoid*[sbertig]interig/10000]
<pre>\sjm\Research\Mallab\THESIS\timetrunc.m Page 7 ay 12, 2004 6:44:57 AM if iter=1 spoodcrat=0; spoodcrat=iput('\n is this what you wanted ? \n To find trigger points manua & yonter 1, else enter 0.\n '); end else spoodcrant=1; end end</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 Al for i=1:5 if stignifo(i,3)==1 ishostrig=trighto(i,5)=10000; autoid=(shostrig linetrig/10000) break end
<pre>\sim Research Matlab \THESIS \time trunc.m Page 7 ay 12, 2004 6:44:57 AM if icr==1 size odornot=0 goodornot=input(\n Is this what you wanted ? \n To find trigger points manua & y enter 1, else enter 0.\n '); end if goodornot=1; end if goodornot=0 if goodornot=0 if goodornot=0</pre>	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 Al for i=1:5 inhortig=triginfo(i,3)==1 inhortig=triginfo(i,5)=10000; autoid=bortig=linetrig/10000] end end end dig=1 b:
<pre>:\sjm\Research\Matlab\THESIS\timetrunc.m Dage 7 ay 12, 2004 6:44:57 AM if iter=1 goodornot=0; else onter 0.(n '); end else enter 0.(n '); end else discontel; end if goodornot=1; end if goodornot=0 trigstartL=tLiftryL(1)=1; if gresplate trigstartL=tLiftryL(1)=1; end end end end end end end end end end</pre>	<pre>c:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 AV for i=1:5</pre>
<pre>:\sjm\Research\Matlab\THESIS\timetrunc.m Dage 7 ay 12, 2004 6:44:57 AM if iter=1 goodornot=0 else goodornot=input('\n Is this what you wanted ? \n To find trigger points manua </pre> end end else goodornot=1; end if goodornot=0 triggtartL=tl+try(l)-1; if ques>2 triggtartL=tl+try(l)-1; else else disp(' ');disp('Click on trigger points until you find the first high trigger poi	<pre>c:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 Al for i=1:5</pre>
<pre>Najm\Research\Matlab\THESIS\timetrunc.m Page 7 ay 12, 2004 6:44:57 AM if itareni goodornot-input('\n 1s this what you wanted 7 \n To find trigger points manua & y enter 1, else enter 0.\n '); else goodornot=1; ed if goodornot=0; ed if goodornot=0; end if years=2 trigget=thethit:pt(1)=1; if yers=2 trigget=thethit:pt(1)=1; else int ing('); if yers=2 trigget=thethit:pt(1)=1; else int ing('); if yers=2 trigget=thethit:pt(1)=1; else int ing('); if yers=2 trigget=thethit:pt(1)=1; else int ing('); if yers=2 trigget=thethit:pt(1)=1; if ye</pre>	<pre>C:\sjs\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 A for i=1:5 if rightfo(i,3)==1 aboetrightrightfo(i,5):10000; autoid=sheatrightfo(i,5):10000; autoid=sheatrightfo(i,5):1000; autoid=sheatrightfo(i,5):100; autoid=she</pre>
<pre>1 kgjm\Research\Malab\THESIS\timetrunc.m Page 7 ay 12, 2004 6:44:57 AM if iter=1 if iter=1 ig modernet=0; end goodernet=1; end if goodernet=1</pre>	<pre>c:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 Af for i=1; for i=1;5 for i=1;5 if showtightTightfo(i,5)=: if showtightTightfo(i,5):10000; autoid=labortig linetrig/1000] break end end</pre>
<pre>:lsjmlResearch/Matlab\THESIS\timetrunc.m Page 7 ay 12, 2004 6:44:57 AM if Iter=1 if lecodornot=0; ig goodornot=iput('\n Is this what you wanted ? \n To find trigger points manus </pre> end else goodornot=1; else if goodornot=1; else if goodornot=1; else if gigit ''.jdisp'('Lick on trigger points until you find the first high trigger poi (c) if goodornot=0; if gigit ''.jdisp'('Lick on trigger points until you find the first high trigger poi (c) if grande; if gigit ''.jdisp'('Lick on trigger point of L. \n '); if wrande; triggtartL=input('\n Enter first high trigger point for L. \n '); end triggtartL=input('\n Enter first high trigger point for R. \n '); end triggtartL;	C:\sjm\Research\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 Af for i=1:5 if sheatig*trightfo(i,5)=00000 autoid*[sheatig] interig/10000] break end end end end sheatig*trightfo(i,5)=10000 break end end sheatig*trightfo(i,5)=10000 break end end end sheatig*trightfo(i,5)=10000 break end end end sheatig*trightfo(i,5)=10000 break end end end sheatig*trightfo(i,5)=10000 break end end end sheatig*trightfo(i,5)=10000 break end end sheatig*trightfo(i,5)=10000 break end end sheatig*trightfo(i,5)=10000 break end end sheatig*trightfo(i,5)=10000 break end sheatig*trightfo(i,5)=10000 break end if unoidyz=sheatig*trightfo(i,5)=10000 break end if unoidyz=sheatig*trightfo(i,5)=10000 break end end end end end end end end
<pre>ivajm\Research\Matlab\THESIS\timetrunc.m Page 7 ay 12, 2004 6:44:57 AM if inr==i goodconct=0; goodconct=0; end if goodconct=1; end if goodconct=1; if goodconct=1; if goodconct=1; end if goodconct=1; if goo</pre>	<pre>c:\sjm\Research\Matlab\THESIS\timetrunc.m Eage May 12, 2004 6:44:57 Af for i=1:5 for i=1:5 for i=1:5 for i=1:6 i shortig=trightfor(i,5)=10000; autoid=(shortig) interig/10001 med end end end shortig=input('\n Eater line number of first in L, 2 if first in R.\n) if statisg===0; if shortig=input('\n Eater line number of first trigger high n '); disp(' ')/dispd==g/'You entered ' numBer(shortig) ' for shortig, and ' numBer(') if shortig=input('\n Eater line number of first trigger high n '); disp(' ')/dispd==g/'You entered ' numBer(shortig) ' for shortig, and ' numBer(') if shortig=0; for () is this what you wanted ? \n To iterate enter 1; else if done, en to ',' if goodsrnowinget break end end end end for ',' if goodsrnowinget break end end end end end end end end end end</pre>
<pre>Najn Research (Matlab)THESIS\timetrunc.m Page 7 6:44:57 AM isonontet0; else goodornoteinput('\n Is this what you wanted 7 \n To find trigger points manus < ly entre 1, else enter 0.\n '); else goodornote1; end if goodornote1; end disp(');disp('Click on trigger points until you find the first high trigger poi </pre>	<pre>c:\sjn\Basearch\Matlab\THESIS\timetrunc.m Page May 12, 2004 6:44:57 M for i=1:5 for i=1:5 if rightfo(i,3)==1 aboting=trightfo(i,5):10000; action[shostrig]Instrightfo(i,5):10000; action[shostrig]Instrightfo(i,5):1000; action[shostrig]Instrightfo(i</pre>
<pre>% ksjm kæsearch Watlab \THESIS \timetrunc.m Page 7 ay 12, 2004 6:44:57 AM godornottel else godornottelput(`\n Is this what you wanted ? \n To find trigger points manus < ly entre in godornottelput(`\n Is this what you wanted ? \n To find trigger points manus < else godornottelp end if godornottelp end disp(`');disp('Click on trigger points until you find the first high trigger poi < e(s),''); disp('then hit escape.'); gomas; triggatartLeiput(`\n Enter first high trigger point for L. \n '); if vers>2 end end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n '); end it=triggatartLeiput(`\n Enter first high trigger point for R. \n ');</pre>	<pre>c:\sign\Basearch\Matlab\THESIS\timetrunc.m Page 1 May 12, 2004 6:44:57 M for i=1:5 if trightfo(i,3)==1 abortigerrightfo(i,5):10000; acticle(abortig)Instrig/TOD001 end end end end end digr('); attockys==spireskjend shottrig=input('\n Boty uwant to accept the autoid numbers? Enter 0 is yes, 1 to co ntime. (. h. '); if autoidys==spireskjend shottrig=input('\n Enter 1 if trigger high cours first in 1, 2 if first in R.\n ; linetrig=input('\n Enter 1 if trigger high cours first in 1, 2 if first in R.\n ; if autoidys==spireskjend shottrig=input('\n Enter 1 if trigger high cours first in 1, 2 if first in R.\n ; if instrig=input('\n Enter 1 if trigger high cours first in 1, 2 if first in R.\n ; if autoidys==spireskjend shottrig=input('\n Enter 1 if trigger high cours first in 1, 2 if first in R.\n ; if autoidys==spireskjend shottrig=input('\n Enter 1 if trigger high cours first in 1, 2 if first in R.\n ; if autoidys==spireskjend shottrig=input('\n Enter 1 if trigger high cours first in 1, 2 if first in R.\n ; if autoidys==spireskjend shottrig=input('\n Enter 1 if trigger high cours first in 1, 2 if first in R.\n ; if autoidys==spireskiend shottrig=input('\n I is this what you wanted ?\n To iterate enter 1, else if dome, en if autoidys==1 datatrunch=ravadataK(starttimeptI=dimeptK,litpos); end if abortig=lhottig linetrig=tarttimeptI=lif; truncinfo(=lhottig linetrig=tarttimeptI=lif; truncinfo(=lhottig linetrig=tarttimeptI=lif; truncinfo(=lhottig linetrig=tarttimeptI=lif; truncinfo(=lhottig linetrig=tarttimeptI=lif; truncinfo(=lhottig=linetrig=tarttimeptI=lif; truncinfo(=lhottig=linetrig=tarttimeptI=lif; truncinfo(=lhottig=linetrig=tarttimeptI=lif; truncinfo(=lhottig=linetrig=tarttimeptI=lif; truncinfo(=lhottig=tartI=tartimeptI=lif; truncinfo(=lhottig=tartI=tartI=tartI=tartI=tartI=tartI=tartI=tartI=tartI=tartI=tartI=tartI=</pre>
<pre>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>></pre>	<pre>c:\sjm\Research\Matlab\THESIS\timetrunc.m Page 1 May 12, 2004 6:44:57 AU for i=1:5 for i=1:</pre>
<pre>% kgjm\Research\Matlab\THESIS\timetrunc.m Page 7 6:44:57 AM if term=1 goodornot=0; else goodornot=1; end if goodornot=1; end if goodornot=1; end if goodornot=1; end if goodornot=1; end disp(');disp('Click on trigger points until you find the first high trigger poi < disp('); if with the terms of the trigger point for L. \n '); if uptartL=light(\n Enter first high trigger point for R. \n '); end triggtartL=input('\n Enter first high trigger point for R. \n '); end fightartL=input('\n Enter first high trigger point for R. \n '); end triggtartL=input('\n Enter first high trigger point for R. \n '); end titl=triggtartL; fitures>2 ttl=triggtartL=input('\n Enter first high trigger point for R. \n '); end titl=triggtartL=input('\n Enter first high trigger point for R. \n '); end titl=triggtartL=input('\n Enter first high trigger point for R. \n '); end titl=triggtartL=input('\n Enter first high trigger point for R. \n '); end titl=triggtartL=input('\n Enter first high trigger point for R. \n '); end titl=triggtartL=input('\n Enter first high trigger point for R. \n '); end titl=triggtartL=input('\n Enter first high trigger point for R. \n '); end titl=triggtartL=input('\n Enter first high trigger point for R. \n '); fir ====================================</pre>	<pre>c:\sjm\Research\Matlab\THESIS\timetrunc.m Page H May 12, 2004 6:44:57 AU for i=1:5 for i=1:</pre>
<pre>:logim\Research\Matlab\THESIS\timetrunc.m Page 7 ay 12, 2004 6:44:57 AM if itermi goodsnottoingut('\n is this what you wanted ? \n To find trigger points manua & iy enter 1, else enter 0.\n '); else end if goodsnottoingut('\n is this what you wanted ? \n To find trigger points manua & iy enter 1, else enter 0.\n '); end end end end end end end end end end</pre>	<pre>c:\sjm\Research\Matlab\THESIS\timetrunc.m Eage i May 12, 2004 6:44:57 Al for i=1:5 for i=1:</pre>
<pre>:\sjm\Research\Matlab\THESIS\timetrunc.m</pre>	<pre>c:\sjn\Research\Matlab\THESIS\timetrunc.m Eage 1 May 12, 2004 6:44:57 Al for i=1:5 for i=1:</pre>
<pre>:\sjm\Research\Matlab\THESIS\timetrunc.m</pre>	<pre>c:\sjn\Research\Matlab\THESIS\timetrunc.m Eage 1 May 12, 2004 6:44:57 Al for i=1:5 for i=1:</pre>
<pre>:(sjm\Research\Matlab\THESIS\timetrunc.m Page 7</pre>	<pre>max</pre>

getshoedataorder.m

<pre>munity.iumprime in the intervence is a large intervence intervence is a large intervence is a large intervence intervence is a large intervence intervence is a large intervence intervence intervence intervence is a large intervence int</pre>	ay 12, 2004 6:51:59 AM unction [shoedata ret1, shoedata ret2, qualinfo1, qualinfo2]=qetshoedataorder(subivers.	hasetimel=shoedatal(:,16);
<pre>start is a start is a start</pre>	oedatal, shoedata2, triggerflag, fig)	<pre>trigger1= shoedata1(:,17); comptime1=shoedata1(:,18);</pre>
	ETSHOEDATAORDER(subjvers, shoedatal, shoedata2, flagdata, fig) takes in shoe data, and \checkmark .turns the data in a revised	end
	reorder, with the time adjusted. Data is linearly interpolated at bad-data pts and duri \varkappa' time gaps with 3 or	<pre>[datalength2, datawidth2]=size(shoedata2); if notisempty(shoedata2) & datawidth2-=18</pre>
	fewer packets missing. Use the version number to specify the type of shoe data. If two 🖌 oes, shoedatal must be Left shoe, int_or_shoedata2 must be Right shoe.	error('Data is wrong size!') else
	Data is returned in this reorder: 1 far hm. 2 far hl. 3 far mm. 4 far ml. 5 nydf b. 6 nydf t. 7 hend i. 8 hend a.	<pre>basetime2=shoedata2(:,10); comptime2=shoedata2(:,17);</pre>
<pre>i : : : : : : : : : : : : : : : : : : :</pre>	9 gyroz, 10 accelz, 11 gyroy, 12 accely, 13 gyrox, 14 accelx, 15 conscripte-heel, 16 conscripte-toe.	end dataamount=15:
<code-block><code-block><code-block><code-block><code-block><code-block></code-block></code-block></code-block></code-block></code-block></code-block>	17 - future use, 18 - future use, 19 - future use, 20 - future use 21 time, 22 info about replaced data	elseif subjvers==2
	Subjvers:	reorder1=[9 11 13 12 10 14 7 8 5 6 3 1 2 4 15 16]; %left reorder2=[9 11 13 12 10 14 7 8 6 5 3 1 2 4 15 16]; %right
<pre>interface in this section</pre>	Version 1: Subjects 011-019	<pre>[datalength1,datawidth1]=size(shoedata1); if notisemptv(shoedata1) \$ datawidth1=18</pre>
<pre>set = 1 = def is a finite interestion</pre>	¥	error('Data is wrong size!') else
	o L = 18 columns in this reorder: gyroz, gyroy, gyrox, accely, accelz, accelx, fsr_hl, fsr_ml, fsr_mm, fsr_hm,	<pre>basetimel=shoedatal(:,16); triggerl= shoedatal(:,17);</pre>
	<pre>pvdf_h, pvdf_t, bend_i, bend_a, capacitive, basetime, trigger, computertime</pre>	<pre>comptimel=shoedatal(:,18); end</pre>
<pre>planety field field</pre>	(-no fsr_hl for 18, 17) o R - 18 columns in this reorder:	[datalength2, datawidth2]=size(shoedata2);
<pre>image: image: imag</pre>	gyroz, gyroy, gyrox, accely, accelz, accelz, accelz, rsr_ni, rsr_mi, rsr_mm, rsr_mm, pvdf_t, pvdf_h, bend_i, bend_a, capacitive, basetime, trigger, computertime l band funkv	error('Data is wrong size!') else
<pre>viii viii viii viii viii viii viii vii</pre>	v	<pre>basetime2=shoedata2(:,16); trigger2= shoedata2(:,17);</pre>
<pre></pre>	Version 2: Subjects 020-026 (cap data: 022-026) - all: new transmit, hence, new pvdf	<pre>comptime2=shoedata2(:,18); end</pre>
<pre></pre>	o L - 18 columns in this rearder:	<pre>dataamount=15; else error('please enter a valid version number')</pre>
<pre>s - f > f</pre>	gyroz, gyroy, gyrox, accely, accelz, accelx, bend_i, bend_a, pvdf_h, pvdf_t, fsr mm, fsr hm, fsr hl, fsr ml, cap h, basetime, trigger, computertime	end
<pre>Try mode for the for the</pre>	o R - 18 columns in this reorder: qyroz, qyroy, qyrox, accely, accelz, accelx, bend i, bend a, pvdf t, pvdf h,	%%Condition Time%************************************
<pre>kmpic/styles/style</pre>	fsr_mm, fsr_hm, fsr_hl, fsr_ml, cap_h, basetime, trigger, computertime	<pre>if subjvers==1 subjvers==2 [time1,time2,triggap]=timeadjuster([basetime1 trigger1 comptime1],[basetime2 trigger]</pre>
<pre>start start s</pre>	f margin<5; fig=0;emd	<pre>comptime2],triggerflag); end</pre>
<pre>remarks[1] 11 12 12 12 14 14 4 1 15 0 7 0 6 17 1 10; 12 10; 14 14 4 15 0 7 0 6 17 10; 10; 10; 10; 10; 10; 10; 10; 10; 10;</pre>	f subjvers==1	<pre>%% Re-reorder and "smooth" data %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%</pre>
<pre>[#straty]:(#ist is word; is is is word; is is</pre>	reorder1=[9 11 13 12 10 14 2 4 3 1 5 6 7 8 0 15]; %left reorder2=[9 11 13 12 10 14 2 4 3 1 6 5 7 8 0 15]; %right	<pre>[shoedata_ret1,qualinfo1]=dataadjuster([shoedata1(:,1:dataamount) time1],reorder1,fic);</pre>
<pre>int interview intervi</pre>	[datalength1, datawidth1]=size(shoedata1);	else datatoadj1=[];
<pre>is in introduction interview intervie</pre>	if notisempty(shoedatal) & datawidth1-=18 error('Data is wrong size!')	end
<pre>iv j2, 201</pre>		
<pre>mlimid (0,34) = (1 trigge); mlimid (0,34) = (2 trigge); mlimid (1,34) = (2 trigge);</pre>	:\sjm\Research\Matlab\THESIS\getshoedataorder.m Page 3 ay 12, 2004 6:51:59 AM	
	<pre>i\sjm\Research\Matlab\THESIS\getshoedataorder.m Page 3 ay 12, 2004 6:51:59 AM if fig=0 fig=0 fig=0 datacodj2=[]; d d</pre>	
	<pre>sly 12, 2004 Page 3 estimate and the state and the st</pre>	
	<pre>clysim\Research\Matlab\THESIS\getshoedataorder.m Page 3 ay 12, 2004 6:51:59 AM if fig==0 fig==0 fig==0 fig==0 addataodsid=[]; addataodsid=[]; d % quality info %% alinfol(1,3:4)=[1 trigge]; alinfol(1,3:4)=[2 trigge]; </pre>	
	<pre>clysim\Research\Matlab\THESIS\getshoedataorder.m Page 3 ay 12, 2004 6:51:59 AM if fig==================================</pre>	
	<pre>closinkResearch(Matlab\THESIS\getshoedataorder.m Page 3 ay 12, 2004 6:51:59 AM if fig=0 fig=0 fig=0 datacod;2=(;,dataadjuster([shoedat2(:,l:dataamount) time2],reorder2,fig < } datacod;2=(;,datacod;2=(;,dataadjuster([shoedat2(:,l:dataamount) time2],reorder2,fig </pre>	
	<pre>closin/Research/Watlab\THESIS\getshoedataorder.m Page 3 ay 12, 2004 6:51:59 AM if fig=0 fig=0 fig=0 datacadgle=1; datacadgl</pre>	
	<pre>:\sjm\Research\Matlab\THESIS\getshoedataorder.m Page 3 ay 12, 2004 6;51:59 AM if fig=0 fig=0; fig=0 fig=0; fig=0;</pre>	
	<pre>stylp: 2004 Page 3 sy 12, 2004 Page 3 fif fig=0 f</pre>	
	<pre>stylp: 2004 Page 3 ay 12, 2004 Control Co</pre>	
	<pre>Skgim/Research/Matlab\THEDIS\getshoedataorder.m Page 3 ay 12, 2004 6:51:59 AW if figue0 figue-6; ead (aboutint_ret2,qualinfo2)=dataadjuster([shoedata2(;,1:dataamount) tim2],reorder2,fig 4 adataadj2=[]; d skguplity info %% alinfo1((,3:4)=[2 triggap]; alinfo2(1,3:4)=[2 triggap];</pre>	
	<pre>ty 12, 204 Page 3 y 12, 204 ff fig=0 fig=-6; end isabel{second} fig=-6</pre>	
	<pre>tylp.2004 Page 3 y 12, 2004 Page 3 fif fig=0 fig=</pre>	
	<pre>> l3g 12, 204 Page 3 ag 12, 204 Csi159 AM if figue0 figue0 figue1 figue</pre>	
	<pre>:\sjm\Research\Matlab\THESIS\getshoedataorder.m Page 3 gy 12, 2004</pre>	
	<pre>% jm/Research/Matlab/THESIS/getshoedataorder.m Page 3 ay 12, 2004</pre>	
	<pre>% jm Research Matlab THESIS \get shoed at a order.m Page 3 ay 12, 204</pre>	
	<pre>% igim Research Watlab \TESIS \get boodstaorder.m Page 3 ay 12, 204</pre>	
	<pre>% igim/Research/Matlab/THESIS/getshoedataorder.m</pre>	
	<pre>% ig im/Research (Matlab\THESIS\getshoedataorder.m</pre>	
	<pre>% y 12, 204</pre>	
	<pre>% ju / 2. 204</pre>	
	<pre>% y 1,2 00</pre>	

Τ

timeadjuster.m

ay 12, 2004 7:03:20 AM	May 12, 2004 7:03:20 J
etime_trigger_rawtimeR,triggerflag)	^\$^\$^\$^\$^\$
[timeL,timeR,atir,returncheck]=timeadjuster(basetime_trigger_rawtimeL,basetime_trigger_ 🖌	<pre>%combine L & R alltimeinfo_unsorted=[btrL; btrR];</pre>
awtimeR,triggerflag) (c) 2003 Stacy J Morris ∽sjm@alum.mit.edu	alltimeinfo_sortbybasetime=sortrows (alltimeinfo_unsorted,1); alltimeinfo_sortbybasetimethenrawtime=sortrows (alltimeinfo_sortbybasetime,3);
to test: [testtimeL,testtimeR,triqqap,atir,returncheck]=timeadjuster(dataLtr(:,16:18),da 🎸	ati=alltimeinfo_sortbybasetimethenrawtime; atiL=length(ati);
sRtr(:,16:18), dataflag);	\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^
angngngng Inifia] pat-um	<pre>%pull out all rows marked with a 0 (also, pulls out "fake" row if one of the shoes has o data settered)</pre>
ht=.00668;int2=int*2;	atiOss=[]; atiNss=[];
engthL,tossj=size(basetime_trigger_rawtimeL);[lengthk,tossj=size(basetime_trigger_rawti ¥ sR);	if ati(i,4)==0
heshoemissing=0; f lengthL==0 & lengthR==0	atiOss=[atiOss;ati(i,:)]; else
error('TIMEADJ: L or R data must be non-empty!') seif lengthL==0	atiNss=[atiNss;ati(i,:)]; end
<pre>basetime_trigger_rawtimeL=[0 0 0];lengthL=1; oneshoemissing=1;</pre>	end [ati0ssL.toss]=size(ati0ss);[atiNssL.toss]=size(atiNss);
seif lengthR==0 basetime trigger rastimeR=(0,0,0);lengthR=1;	if atiOssL + atiNssL -= atiL;error('timeadj: mistake in subsets');end
oneshoemissing=2;	\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^\$^
<pre>rstate trigger_rawtimeL; btrL(:,1) = btrL(:,1) /10000;</pre>	Scheck multiple comptime section for mis-ordering
<pre>n=baselime_figger_nawcimer,birk(:,1)=birk(:,1)=10000; meL=[];timeR=[];</pre>	Dcomptimes=diff(comptimes);
vêvêvê	DCt0s=rind(Dcomptimes==0); DDct0s=diff(Dct0s);
riggerflag set up 5 notisempty(triggerflag) & triggerflag(1)==1	Dct0s(1,2)=0;
<pre>if btrL(round(triggerflag(2)),2)~=1 error('NEWTIMEADJ: problem 1 with triggerflag')</pre>	<pre>for i=2:length(Dct0s) if Dct0s(i,1)-Dct0s(i-1,1)==1</pre>
end btrL(round(triggerflag(2)),2)=-1;	Dct0s(i,2)=Dct0s(i-1,2)+1; end
<pre>self notisempty(triggerflag) & triggerflag(1)==2 if btrB(round(triggerflag(2)), 2)=1</pre>	end Det0s(:,2)=Det0s(:,2)+1;
error('NEWTIMEADJ: problem 1 with triggerflag')	<pre>Dct0s1s=find(Dct0s(:,2)==1);</pre>
<pre>btrR(round(triggerflag(2)),2)=-1; if anti-ampty(triggerflag(2)),2)=-1;</pre>	<pre>checktheserows=[]; for i=1=rot(Det0=1=)</pre>
error('NEWTIMEADJ: problem 3 with triggerflag')	if i==length(UctUS1s) if i==length(UctUS1s)
id Isrenenenenenenenenenenenenenenenenenenen	<pre>if Dct0s1s(i)==length(Dct0s) Dct0s_end=Dct0s(Dct0s1s(i))+1;</pre>
/n@^@^@^@ whoe specific info	<pre>else Dct0s_end=Dct0s(length(Dct0s))+1;</pre>
<pre>or i=1:lengthL;btrL(i,5:6)=[1 i];end or i=1:lengthR;btrR(i,5:6)=[2 i];end</pre>	end else
เลกสกรกรกรกิจกรกรกรกรกรกรกรกรกรกรกรกรกรกรกรก	Dct0s_end=Dct0s(Dct0s1s(i+1)-1)+1; end
initial evaluation of base time validity	<pre>checkrows=atiNss(Dct0s(Dct0s1s(i)):Dct0s_end,:); anyrollowers=find(checkrows(:.4)==-1);</pre>
rL=nta_cat(btrL);	<pre>if not(isempty(anyrollovers))</pre>
rigenta cat (btrB):	anyos-rind(checkrows(:,4)0);
turncheck=[btrl;btrR];	<pre>if not(isempty(any0s)) disp('fix this! (shouldn''t happen) - 1 - timeadj') C() cim/Research/Mitloh/THESTE/Fixendiuster =</pre>
turncheck=[btrL;btrR]; :\sjm\Research\Matlab\THESIS\timeadjuster.m Page 3 ay 12, 2004 7:03:20 AM	<pre>if not(isempty(any0s)) disp('fix this! (shouldn''t happen) - 1 - timeadj') C:\sjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20</pre>
<pre>turacheck=[birl;birk]; turacheck=[birl;birk]; turacheck=[birl;birk]; turacheck=[birl;birk]; ay 12, 2004 Product P</pre>	<pre>if not(isempty(any0s)) disp('fix this! (shouldn''t happen) - 1 - timeadj') C:\sjm\Research\Matlab\THECIS\timeadjuster.m Page May 12, 2004 7:03:20 merlion(datainfo back(),50/0.40%); if dtringher(1).50/0.40%); </pre>
<pre>turncheck=[birl;birk]; turncheck=[birl;birk]; typ=Research/Matlab\THESIS\timeadjuster.m Page 3 ay 12, 2004 7:03:20 AM atiMas(Dot0s(Dot0sis(i))-2:Dot0s(Dot0sis(i+1)-1)+1+2,:) end for a complement (backbook); [or a complement (backbook) backbook) complement (backbook)]; [or a complement (backbook) backbook); [or a complement (backbook) backbook]; [or a complement</pre>	<pre>if not(isempty(any0s)) disp('fix this! (shouldn''t happen) - 1 - timeadj') C:\cjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20 m=floor(datainfo_back(1,5)/0.40%); if datainfo_back(1,5)/0.40%); if datainfo_back(1,5)/0.40%; if datainfo_back(1,5</pre>
<pre>turncheck=[btrl;btrk]; (\sjm\Research\Wallab\THESIS\timeadjuster.m Page 3 ay 12, 2004 7:03:20 AM end atikss(bct0s(bct0s(a(i))-2:bct0s(bct0s(a(i))-1)+i+2;)) end if abs(bct0s(bct0s)) for k2:erl tind where the switch is</pre>	<pre>if not(isempty(anyCo)) disp('fix this!(zhouldn''t happen) - 1 - timeadj') disp('fix this!(zhouldn''t happen) - 1 - timeadj') C:\sjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20 rfloor(datainto_back(1,5)/0,4004)) rfloor(datainto_back(1,5)/0,4004)) datainto_back(1,5) 0(0,4004) datainto_back(1,5) 0(0,4004)) info_back(1,3)+n*.4096)/(.00668)) *sign(datainto_back(1,3)+n*.4096)/(.00668)) - ((datainto_back(1,3)+n*.4096)/(.00668)) </pre>
<pre>turnchecke[bttL;btrR]; (\sjm\Research\Matlab\THESIS\timeadjuster.m Page 3 y 12, 2004 7:03:20 AM atiNes(Dct0s(Dct0sis(i))-2:Dct0s(Dct0sis(i+1)-1)+1+2;;) end [crt.ps]pists(bcbcross); fi.fi.ab.cross((checkross(k,1)-checkross(k-1,1))/int)-((checkross(k,1)-checkr ¥ s(k-1,1))/int)>.1 clirowsk;</pre>	<pre>if not(isempty(any0s)) disp('fix this! (shouldn''t happen) - 1 - timeadj') disp('fix this! (shouldn''t happen) - 1 - timeadj') C:\sjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20 m=floor(datainfo_back(1,5)/0.4096); if datainfo_back(1,5)/0.4096); if datainfo_back(1,5)/0.4096); if datainfo_back(1,5)/0.4096); isom pack(1,5)/0.4096); is</pre>
<pre>turnchecke[bttL;btrR]; turnchecke[bttL;btrR]; turnchecke[bttL;btrR]; y 12, 2004 Page 3 y 12, 2004 Page 3 of the set of the s</pre>	<pre>if not(isempty(any0s)) disp('fix this! (shouldn''t happen) - 1 - timeadj') disp('fix this! (shouldn''t happen) - 1 - timeadj') C:\cjm\Research\Matlab\THEOIS\timeadjuster.m Page May 12, 2004 7:03:20 mofloor(datainfo back(1,3)(0,4096)) if datainfo back(1,3)(0,4096)) if datainfo back(1,3)(0,0066)) - ((datainfo back(1,3)+n*.4096)/(.00666)) - ((datainfo back(1,3)+n*.4096)/(.00666)) - (idatainfo back(1,3)+n*.4096)/(.00666))</pre>
<pre>iturncheck=[birljbrR]; iturncheck=[birljbrR]; iturncheck=[birljbrR]; atilSes(DetGs(DetGs(s(i))-2:DetGs(DetGs(s(i+1)-1)+1+2;)) end [crl,toss]=size(checkrows); for k=2icrl [ind where the switch is for k=2icrl [ind where the swi</pre>	<pre>if not(isempty(anyOs)) disp('fix this! (sholds''t happen) - 1 - timeadj') disp('fix this! (sholds''t happen) - 1 - timeadj') C(\cjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20 n=flor(datainfo_back(1,5)/0.4006)/ idstainfo_back(1,3)/0.7006(0)/ idstainfo_back(1,3)/0.7006(0)/ ifoback(1,3)/0.7006(0))*sig(datainfo_back(1,3)+n*.4006)/(.00660)) - ((datainfo_back(1,3)+n*.4006)/(.00660)) - ((datainfo_back(1,3)+n*.4006)/(.00660)) - ((datainfo_back(1,3)+n*.4006)/(.00660)) - (idstainfo_back(1,3)+n*.4006)/(.00660)) - (idstainfo_back(1,3)+n*.40060)/(.00660)) - (idstainfo_back(1,3)+n*.40060)/(.00660)) - (idstainfo_back(1,3)+n*.40060)/(.00660)) - (idstainfo_back(1,3)+n*.40060)/(.00660)) - (idstainfo_back(1,3)+n*.40060)/(.00660)/(.00660)) - (idstainfo_back(1,3)+n*.40060)/(.00660)/(.00660)/(.0060)/(.0060)/(.0060)/(.0060)/(.</pre>
<pre>turnchecte[bttl;btrR]; turnchecte[bttl;btrR]; turnchecte[bttl;btrR]; turnchecte[bttl;btrR]; turnchecte[bttl;btrR]; turnchecte[bttl;btrR]; turnchecte[bttl;btrR]; turnchecte[bttl;btrR]; turnchecte[bttl;btrR]; turnchect</pre>	<pre>if not(isempty(anyOs)) disp('fix this!(ikoolds''t happen) - 1 - timead)') disp('fix this!(ikoolds''t happen) - 1 - timead)') C:\sjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20</pre>
<pre>turnobeck=[btrL/btrR]; /sjm\Research\Matlab\THESIS\timeadjuster.m Page 3 y 12, 2004 7:03:20 AM atiNss(bct0s(bct0s(i))-2:bct0s(bct0s(i+1)-1)+1+2,:) acd, soss_ariss(bchckrows); for kwiterL Winn where the switch is if abs(round((chckrows); clicek;); clicek;; acd acd acd acd acd acd acd acd acd acd</pre>	<pre>if not(isempty(amyOs)) disp('fix this!(shouldn''t happen) - 1 - timeadj') disp('fix this!(shouldn''t happen) - 1 - timeadj') C:\sjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20 m=floor(datainfo_back(1,5)/0.4096)/ if datainfo_back(1,3)*n*.4096)/(.00668)) > (idat info_back(1,30*n*1)end info_bac</pre>
<pre>turnchecke[bttLybtrR]; turnchecke[bttLybtrR]; turnchecke[bttLybtR]; turn</pre>	<pre>if not(isempty(anyOs)) disp('fix this!(shouldn''t happen) - 1 - timeadj') disp('fix this!(shouldn''t happen) - 1 - timeadj') C:\ajm\Research\Matlab\THECIS\timeadjuster.m Page May 12, 2004 7:03:20 melloor(datainto_back(1,5)/0.40%0); if datainto_back(1,5)/0.40%0;); if datainto_back(1,5)/0.40%0;); if datainto_back(1,5)/0.40%0;); ison point without the sector of the se</pre>
<pre>\vgin\Research\Matlab\THEDIS\timeadjuster.m Page 3 y 12, 2004 7:03:20 AM atiMss(BetGs(DetGs[a(i))-2:DetGs(DetGs[a(i+1)-1)+i+2,:) end [crt_toss]=sire(checkrow); for lef2ret! Visid Water the solith is a(k-1,i)/int)-i(checkrows(k,1)-checkrows(k-1,1)/int)-i(checkrows(k,1)-checkr < solitorw(k) break end if enlroy1 %(if enlrow1, no need to reorder)) end if enlroy1 %(if enlrow1, no need to reorder)) end checktheserows[checktheserows/DetGs(DetGs[a(i))]; end and if enlroy1 %(if enlrow1, no need to reorder)) end if enlroy1 %(if enlrow1, no need to reorder)) end if enlroy1 %(if enlrow1, no need to reorder)) end if end if enlroy1 %(if enlrow1, no need to reorder)) end if end if enlroy1 %(if enlrow1) if end to reorder) end if endretheserows[checktheserows/DetGs(DetGs[a(i))]; end if endretheserows[checktheserows/DetGs(DetGs[a(i))]; end if endretheserows[checktheserows/DetGs(DetGs[a(i))]; endretheserows[checktheserows/DetGs[a(i)]; endretheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checktheserows[checkt</pre>	<pre>if not(isempty(anyOs)) disp('fix this!(sholds''t happen) - 1 - timeadj') disp('fix this!(</pre>
<pre>turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; adilss(btt3s[bct3s[cbc3s[i])-2:bct3s[bct3s[i]]-1)+i+2;;) adilss(btt3s[bct3s[cbc3s[i])-2:bct3s[bct3s[i]]-1)+i+2;;) adilss(btt3s[bct3s[bct3s[i]]-2:bct3s[bct3s[i]]-1)+i+2;;) adilss(btt3s[bct</pre>	<pre>if not(isempty(anyOs)) disp('fix this!(ikooldw''t happen) = 1 - timead)') disp('fix this!(ikooldw''t happen) = 1 - timead)') disp('fix this!(ikooldw''t happen) = 1 - timead)') closed to the set of th</pre>
<pre>turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtrR]; turnchecte[bttLybtR]; tur</pre>	<pre>if not(isempty(anyOs)) disp('fix this! (shouldn''t happen) - 1 - timead)') disp('fix this! (shouldn''t happen) - 1 - timead)') disp('fix this! (shouldn''t happen) - 1 - timead)') C:\sjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20 m=flord(atainfo_back(1,5)/0.4096); if datainfo_back(1,5)/0.4096); if datainfo_back(1,5)/0.4096; if datainfo_back(1,5)/0.4096;</pre>
<pre>turncheck=[bitLybrrR]; turncheck=[bitLybrrR]; turncheck=[bitLybrrR]; turncheck=[bitLybrrR]; turncheck=[bitLybrrR]; end stiNss(Dct0s(bct0sis(i))-2:bct0s(bct0sis(i+1)-1)+1+2,:) end if abs(round((bckrows); if abs(round((bckrows); if abs(round((bckrows); if abs(round(bckrows); if abs(bct0sis(i)):bct0s(end,i)=[bckckrows(rollrow:crL,:);bckckrows(k:1) if abs(bct0sis(bct0sis(i)):bct0s(end,i)=[bckckrows(rollrow:crL,:);bckckrows(l:r < if abs(bct0sis(bct0sis(i)):bct0s(bct0sis(i))]; end d d d tituttabaseros=[bckckbeserows/bct0s(bct0sis(i)]; end d d tituttabaseros=[chackbeserows/bct0s(bct0sis(i)]; end d tituttabaseros=[chackbeserows/bct0s(bct0sis(bct0sis(bct0sis(bct0sis(bct0sis(bct0sis(bct0sis(bct0sis(bct0sis(bct0sis(bc</pre>	<pre>if not(isempty(amyOs)) disp('fix this! (shouldn''t happen) - 1 - timeadj') disp('fix this! (shouldn''t happen) - 1 - timeadj') C:\sjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20 mefloor(datainto back(1,5)/0.4096); if datainfo back(1,5)/0.4096); if datainfo back(1,5)/0.4096); info back(1,5)/0.1096); info back(1,5)/0.4096); info back(1,5)/0.4096); if datainfo back nxrco0 & datainfo back(1,5)+m*.4096)/(.00668)) - ((datainfo back(1,5)+.4096); tionapper with pr iour point if datainfo back nxrco0 & datainfo back(1,5)+m*.4096)/(.00668)) - using (datainfo back(1,5)+.4096); tionapper with pr iour point if datainfo back nxrco0 & datainfo back(1,5)+m*.4096)/(.00668)) - using (datainfo back(1,5)+.4096); tionapper datainfo back(1,5)+.4096; tionapper</pre>
<pre>turnchect=[bttlybtrR]; t\sjm\Research\Matlab\THESIS\timeadjuster.m Page 3 yy 12, 2004 7:03:20 AM atilSaCotCotCotSis(i))=2:bctCosCotCotSis(i+1)=1+1+2;:) end [crt,toss]=sise(checkross); for twitterL titls Where the switch is n(t,1))/(t)=(checkross(t,1)-checkross(t-1,1))/(t)=(checkross(t,1)-checkr < n(t,1))/(t)=(checkross(t,1)-checkross(t,1))/(t)=(checkross(t,1)-checkr < n(t,1))/(t)=(checkross(t,1)-checkross(t,1))/(t)=(checkross(t,1)-checkross(t,1)-checkross(t,1)-checkross(t,1))/(t)=(checkross(t,1)-checkross(t,1)-checkross(t,1))/(t)=(checkross(t,</pre>	<pre>if not(isempty(anyOs)) disp('fix this! (sholdn''t happen) - 1 - timeadj') method (sholdn''t happen) - 1 - timeadj')</pre>
<pre>iturncheck=[birl/birk]; iturncheck=[birl/birk]; atiliss(bct0s(bct0s(i))-2:bct0s(bct0s(i))-1)+1+2;;) atiliss(bct0s(bct0s(i))-2:bct0s(bct0s(i))-1)+1+2;;) afir k2:crl tind where the switch is if rads(cond((chctores(k,1)-chcctore(k-1,1))/int)-((chcctrore(k,1)-chcctr < rs/s)) for k2:crl tind where the switch is if table(cond((chctores(k,1)-chcctore(k-1,1))/int)-((chcctrore(k,1)-chcctr < rs/s)) for k2:crl tind where the switch is if table(cond(chctores(k,1)-chcctore(k-1,1))/int)-((chcctrore(k,1)-chcctr < rs/s)) for k2:crl tind where the switch is if table(cond(chctores(k,1)-chcctore(k-1,1))/int)-((chcctrore(k,1)-chcctr < rs/s)) for k2:crl tind where the switch is if table(chctores(k,1)) for k2:crl tind where the switch is if table(chctores(k,1)) for the chctores(chcctores(k,1)) for the chctores(chcctores(k,1)) for the chctores(chcctores(k,1)) for the chc l shoe for switch(1)>1 for the for the chain(b pack(1)); for the chc l is for switch(1)>1 for the for the chain(b pack(1)); for the chc(l,1)>1 for the ch(l,1)>1 for the chc(l,1)>1 for the chain(b pack(1)); for the</pre>	<pre>if not(isempty(anyOs)) disp("fix this((houlds"'t happen) - 1 - timead)') disp("fix this((houlds"'t happen) - 1 - timead)') disp("fix this((houlds"'t happen) - 1 - timead)') control (the second of the se</pre>
<pre>turncheck=[bitLjbitR]; turncheck=[bitLjbitR]; turncheck=[bitLjbitR]; sy 12, 2004</pre>	<pre>if not(isempty(aryOs)) disp('fix this' (kholds''t happen) - 1 - timead)') disp('fix this' (kholds''t happen) - 1 - timead)') C:\tsjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004</pre>
<pre>turnchect=[birl/birk]; turnchect=[birl/birk]; turnchect=[birl/birk]; turnchect=[birl/birk]; atilkss(bct0s(bct0s(i))-2;bct0s(bct0sls(i+1)-1)+1+2;); end [cot,toss]=nis(checkros); fif.alrows(the set box nitch is fif.</pre>	<pre>if not(isempty(amyOs)) disp('fix this!(shouldn''t happen) - 1 - timead)') disp('fix this!(shouldn''t happen) - 1 - timead)') disp('fix this!(shouldn''t happen) - 1 - timead)') meThor(dataInto back(1,5)/0.4096); if dataInto back(1,5)/0.4096; if dataInt</pre>
<pre>turncheck=[birl/birk]; turncheck=[birl/birk]; turncheck=[birl/birk]; turncheck=[birl/birk]; atilise(birlof0s(Det0sls(i))-2:Det0s(Det0sls(i+1)-1)+1+2;;) ad [crt,toss]=sise(beckross); for k2:crt birls where the switch is [crt,toss]=sise(beckross(b,1)-checkross(k-1,1))/int)-((checkross(k,1)-checkr < for k2:crt birls where the switch is [crt,toss]=sise(beckross(b,1)-checkross(k-1,1))/int)-((checkross(k,1)-checkr < for k2:crt birls where the switch is [crt,toss]=sise(beckross(b,1)-checkross(k-1,1))/int)-((checkross(k,1)-checkr < for k2:crt birls where the switch is [crt,toss]=sise(beckross(birl)];</pre>	<pre>if not(issepty(amyOs)) disp('fix this((kholds''t happen) - 1 - timead)') disp('fix this((kholds''t happen) - 1 - timead)') disp('fix this((kholds''t happen) - 1 - timead)') reform (data info back(1,3)/0.400(3) reform (data info back(1,3)/0.400(3) reform (data info back(1,3)/0.400(3) reform (data info back(1,3)/0.400(3)) if (data info back(1,3)/0.400(3))) if (data info back(1,3)/0.400(3))) if (data info back(1,3)/0.400(3))) if (data info back(1,3)/0.400(3))) if (data info back(1,3)/0.400(3)))) if (data info back(1,3)/0.400(3)))) if (data info back(1,3)/0.400(3))))) if (data info back(1,3)/0.400(3)))))) if (data info back(1,3)/0.400(3)))))))))))))))))))))))))))))))))))</pre>
<pre>iturncheck=[birLjbrRk]; (tyjn\Research\Watlab\THESIS\timeadjuster.m Page 3 yy 12, 2004 7:03:20 AM atikss(bct0s(bct0s(s)) for k2:erl tim where the switch is if abs(cond((checkeos)) for k2:erl tim where the switch is if abs(cond(checkeos)) for k2:erl tim where the switch is if the conduction of the conduction of</pre>	<pre>if not(isempty(amyOD)) disp(fix this((koolds''t happen) - 1 - timead)') disp(fix this((koolds''t happen) - 1 - timead)') disp(fix this((koolds''t happen) - 1 - timead)') reflor(datain(_back(1,5)/(.4096)); if datain(_back(1,5)/(.4096); if datain(_back(1,5)/(.4096); if datain(_back(1,5)/(.4096); if datain(_back(1,5)/(.4096); isompoint w/rollower</pre>
<pre>turncheck=[bitLjbitK]; turncheck=[bitLjbitK]; turncheck=[bitK]; turncheck=[</pre>	<pre>if not(isempty(arg0s)) disp('fix this'(wholdn''t happen) - 1 - timedj') disp('fix this'(wholdn''t happen) - 1 - timedj') disp('fix this'(wholdn''t happen) - 1 - timedj') reflor(datainfo_back(1,5)/0.4096); if datainfo_back(1,5)/0.4096); if notisepty(1 = 1 = 1 = 1 = 1 = 1 = 1 = 1 = 1 = 1 =</pre>
<pre>turnedect=[bttl;bttR]; tivginkesearch/Matlab/THEDIS/timeadjuster.m Page 3 yi 12, 2004</pre>	<pre>if not(isempty(anyOs)) disp('fix this'(shouldn''t happen) - 1 - timedj') reflor(dataInD hack(1,5)/0.4090); if dataInD hack(1,5)/0.4090; if dataInD hack(1,5)/0.4090; if dataInD hack(1,5)/0.4090; if notIsempty(removerous) & last(removerous) may he previous row was removed, if notIsempty(removerous) & last(removerous) may he previous row was removed, if it itsef(1,5)/0.4000; if itsempty(removerous) & last(removerous) may he previous row was removed, if itsempty(removerous) & last(removerous) may he previous row was removed, if itsempty(removerous) & last(removerous) may he previous row was removed, if itsempty(removerous) & last(removerous) may he previous row was removed, if itsempty(removerous) & last(removerous) may he previous row was removed, if itsempty(removerous) & last(removerous) may he previous row was removed, if dataInfo hack(1,5)/0.5 for integrape if dataInfo hack(1,5)/0.5 for integrap</pre>
<pre>summediate[bitLybitR]; (\sjm\Research Matlab\THEGIS\timeadjuster.m Page 3 yy 12, 2004 7:03:20 AM atlss(bct0s(bct0s(bct0s(bct0s(bct0s(bct0s(ct1)-1)+1+2,:) end if atlss(bct0s(bct0s(bct0s(bct0s(bct0s(bct0s(ct1))))))) for be2/cct1 bind where the switch is if abs(ct0) content of the second of the switch is if abs(ct0) bct0s(bct0s(bct0s(bct0s(bct0s(bct0s(bct0s(bct0s(ct1))))))) titRetBct0s(bct0</pre>	<pre>if not(isempty(amyOD)) disp(fix this((kholds''t happen) - 1 - timed)') disp(fix this((kholds''t happen) - 1 - timed)') disp(fix this((kholds''t happen) - 1 - timed)') reflor(datain(_kholds)) reflor(datain(_kholds))) reflor(datain(_kholds)) reflor(datain(_kholds)) reflor(datain(_kholds)) reflor(datain(_kholds))) reflor(datain(_kholds)))) reflor(datain(_kholds))</pre>
<pre>turnedest=[bttlybtrR]; turnedest=[bttlybtrR]; turnedest=[bttlybtrR]; displacement Matlab\THESIS\timeadjuster.m Page 3 ny 12, 2004 7:03:20 AM end end end end end end end end end en</pre>	<pre>if not(isempty(aryOs)) disp('fix this'(wholdn''t happen) - 1 - timed)') disp('fix this'(wholdn''t happen) - 1 - timed)') disp('fix this'(wholdn''t happen) - 1 - timed)') refor (datain(> back(1,5)/0.4096)) if datain(> back(1,5)/0.4096); if datain(> back(1,5)/0.4096);</pre>
<pre>turncheck=[brtl;brtk]; (*djn\Research(Matlab\THEDIS\timeadjuster.m Page 3 yy 12, 2004 7:03:20 AM atilKs(CotOs(DotOsls(i))-2:DotOs(DotOsls(i+1)-1)+1+2;:) end [crt,tos]=sis(checkrows); for f=2:ert find there the minimum for the the minimum f</pre>	<pre>if not(isempty(aryOs)) disp('fix this' (shouldn''t happen) - 1 - timedd') disp('fix this' (shouldn''t happen) - 1 - timedd') control of the second of the</pre>
<pre>%ign\Research\Matlab\THESIS\timeadjuster.m Page 3 yul2, 2004 7:03:20 AM atlBs GeoSuBcetals())=2:0bc0s(DctDsls(i+)=1+1+2,:) end [crt,tcss]=size(chectrows); for k2:crt \find where the switch is if k=vcrt,rollrow-1, end if follrow1 \(if rollrow-1, on need to reorder)) atlBs Globa(DctDsls(i))=2:0bc0t(DctDsls(i+))/int)=((checkrows(k,1)-checkrows(k,1)-checkrows(k-1,1))/int)=((checkrows(k,1)-checkrows(k-1,1))/int)=(if checkrows(k,1)-checkrows(k-1,1))/int)=(if checkrows(k,1)-checkrows(k-1,1))/int)=(if checkrows(k,1)-checkrows(k-1,1))/int)=(if checkrows(k,1)-checkrows(k-1,1))/int)=(if checkrows(k-1,1))/int)=(if checkrows(k-1,1))/int)=(if</pre>	<pre>if not(seepty(anyOs)) disp('fix this((kholds''t happen) - 1 - timed)') disp('fix this((kholds''t happen) - 1 - timed)') disp('fix this((kholds''t happen) - 1 - timed)') for the set of t</pre>
<pre>iturncheck=[brfl;brfk]; iturncheck=[brfl;brfk]; iturncheck=[brfl;brfk]; iturncheck=[brfl;brfk]; atlbs(bct0s(bct0s(s(1))-2:bct0s(bct0s(s(1))-1)+1+2;;) atlbs(bct0s(bct0s(s(1))-2:bct0s(bct0s(s(1))-1)+1+2;;) brfk atlbs(bct0s(bct0s(s(1))-bct0ckrows(k-1,1))/lnt)-((cbcckrows(k,1)-cbcckr <</pre>	<pre>if not(seepty(anyOs)) disp("fix this((kholds"'t happen) - 1 - timedd)') disp("fix this((kholds"'t happen) - 1 - timedd)') disp("fix this((kholds"'t happen) - 1 - timedd)') reflor(datain6_back(1,5)/0,4096); if datain6_back(1,5)/0,4096); if datain6_back(1,5)/0,4096); if datain6_back(1,5)/0,4096); if datain6_back(1,5)/0,4096); isos point w/rellower</pre>
<pre>iturncheck=[brfl;brfk]; iturncheck=[brfl;brfk]; iturncheck=[brfl;brfk]; atiks(Dotbs(Dotbs(i))-2:Dotbs(Dotbs(i))+1+1+2,:) atiks(Cotbs(Dotbs(i))-2:Dotbs(Dotbs(i))+1+1+2,:) atiks(Cotbs(Dotbs(Dotbs(i))); for k2:crf \find where the patich is if abs(croad((Cotbs(cotbs(k,1))-checkrows(k-1,1))/int)-((checkrows(k,1)-checkr < acts)); for k2:crf \find where the patich is if abs(croad((Cotbs(cotbs(k,1))-checkrows(k-1,1))/int)-((checkrows(k,1)-checkr < acts)); for k2:crf \find where the patich is if abs(croad(Cotbs(cotbs(k,1))); acts and aff tollraw1 h(i) forlingw1, no need to records()); if (thisGotbs(Dotbs(i));Dotbs(amd,1)-(checkrows(cotbs(k,1))-checkrows(lr < acts)); for bocktheserows-[checktheserowsDotb(Dotbs(i))]; add add aff tollraw1 h(i) forlingw1, no need to records(); if athering back(i,2)(cotbs(i));Dotbs(amd,1)); add add aff tollraw1 h(i) forlingw1, no need to records(); if athering back(i,2)(cotbs(i));Dotbs(amd,1)); add add aff tollraw1 h(i) forlingw1, no need to records(); if athering back(i,2)(cotbs(i));Dotbs(amd,1)); add add aff tollraw1 h(i) forlingw1, no need to records(); if athering back(i,2)(cotbs(i));Dotbs(amd,1)); add add aff tollraw1 h(i) forlingw1, no need to records(); if athering back(i,2)(cotbs(i));Dotbs(amd,1)); add add aff tollraw1 h(i) forlingw1, no need to records(); if athering back(i,2)(cotbs(i));Dotbs(amd,1)); add add aff tollraw1 h(i) forlingw1, no need to records(); if athering back(i,2)(cotbs(i));Dotbs(amd,1)); add add aff tollraw1 h(i) forlingw1, no need to records(); if athering back(i,2)(cotbs(i));Dotbs(amd,1)); add add aff tollraw1 h(i) forlingw1, no need to records(); if athering back(i,2)(cotbs(i));Dotbs(amd,2);Dotbs(</pre>	<pre>if not(isempty(anyOs)) disp('fix this'(shouldn''t happen) - 1 - timeadj') disp('fix this'(shouldn''t happen) - 1 - timeadj') disp('fix this'(shouldn''t happen) - 1 - timeadj') reflor(datinfo_back(1,5)/0.40%6); if datinfo_back(1,5)/0.40%6); if datinfo_back(1,5)/0.50%6; if datinfo</pre>
<pre>iturn.eed=[brit/brik]; iturn.eed=[brit/brik]; iturn.eed=[brit/brik]; iturn.eed=[brit/brik]; atilise[Cot20[Cot2015(1)]-2:Oct20[Cot2015(1:1]-1]+1+2;:) end [crt,tcos]=sise(checkrows); for k=2:ert [tin where the switch is crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur k=1:0 / itur k=1:0 / itur k=1:0 / itur crt k=1:0 / itur crt k=1:0 / itur k=1:0 / itur</pre>	<pre>if not(isempty(arg0s)) disp('fix this(ishold''t happen) - 1 - timeadj') disp('fix this(ishold''t happen) - 1 - timeadj') disp('fix this(ishold''t happen) - 1 - timeadj') reflor(dataInto_back(1,5)/0.4096); if dataInto_back(1,5)/0.4096); if notLisempty(fietwitte(ifietwitte(ifi(fietwitte(fietwitte(ifi(fietwitte(ifi(fietwitte(fietwitte(ifi(fietwitte(fietwitte(fietwitte(ifi(fietwitte(ifi(fietwitte(fie</pre>
<pre>% use of the second secon</pre>	<pre>if notisempty(amyCan) disp('fix this' (shouldn''t happen) - 1 - timeadj') disp('fix this' (shouldn''t happen) - 1 - timeadj') disp('fix this' (shouldn''t happen) - 1 - timeadj') reflor(datain(o_back(1,5)/0,40%); if datain(o_back(1,5)/0,40%); if datain(o_back(1,5)/0,40%); if datain(o_back(1,5)/0,40%); if datain(o_back(1,5)/0,40%); ison point w/rollower if datain(o_back(1,5)/0,40%); ison point w/rollower if datain(o_back(1,5)/0,40%); ison point w/rollower ison point w/rollow</pre>
<pre>iturndesch[birt]birtBirt]; iturndesch[birt]birtBirt]; iturndesch[birt]birtBirt]; iturndesch[birt]birtBirt]; atilts(bct0s(bct0s(a))) for[bct0s(bct0s(a))] for[bct0s(bct0s(a))] for[bct0s(bct0s(a))] for[bct0s(bct0s(a))] for[bct0s(bct0s(a))] for[bct0s(bct0s(a))] for[bct0s(bct0s(a))] for[bct0s(bct0s(a))] for[bct0s(bct0s(a))] for[bct0s(bct0s(bct0s(a))]] for[bct0s(bct0s(bct0s(a))]] for[bct0s(bct</pre>	<pre>if not(isempty(amyOs)) disp('fix this' (knolds''t happen) - 1 - timead)') disp('fix this' (knolds''t happen) - 1 - timead)') disp('fix this' (knolds''t happen) - 1 - timead)') reflor(datain6_back(1,5)/0,4096); if datain6_back(1,5)/0,4096); if otimepty(resource); last(removerous)=mi-1; lif previous row was removed if otimepty(removerous); last(removerous)=mi-1; lif previous row was removed if otimepty(removerous) = last(removerous)=mi-1; lif previous row was removed if datainfo_backenta_info(tatilds(li:l:l:l:l:l)); ed if datainfo_backenta_info(tatilds(li:l:l:l:l);); ed if datainfo_backenta_info(tatilds(li:l:l:l:l);); ed if datainfo_backenta_info(tatilds(li:l:l:l);); ed if datainfo_backenta_info(tatilds(li:l:l:l);); ed if datainfo_backenta_info(tatilds(li:l:l);); ed if datainfo_backenta_info(tatilds(li:l:l);); ed if datainfo_backenta_info(tatilds(li:l:l);); ed if datainfo_backenta_info(tatinfo_back(l</pre>

timeadjuster.m, cont.

<pre>((datainfo_back(1,4)+.4096)-datainfo_back(1,5))<.02</pre>	atiOL=atiOL-1;
<pre>disp('dingalingaling'); %remove if this never occurs atiNss(i,:)</pre>	end
[datainfo_back_nxro abs((datainfo_back(1,3)+n* 4096)-datainfo_back(1,5))] % atiNss(i,4)=-100;	if not(isempty(ati0ss)) %need to reorder ati0ss so we go consecutively
<pre>else atiNss(i,4)=0;</pre>	atiOss_sorthyoriginalindex=sortrows(atiOss,6); atiOss=sortrows(atiOss_sorthyoriginalindex,3);
ati0ss=[ati0ss;atiNss(i,:)];ati0ssL=ati0ssL+1; removerows=[removerows;i];%can't remove now or will mess up rest of loop	end % atiOss(:,2)=atiOss(:,2)/1000
end end	for i=1:ati0ssL
ena d	CTfollowing=find(atir(:,3) <atiuss(1,3)); CTfollowing=find(atir(:,3)>atiOss(i,3));</atiuss(1,3));
<pre>not(lsempty(removerows)) atiNs(removerows,:)=[]; atiNs(removerows,:)=[];</pre>	if notisempty(CTprevious) & notisempty(CTfollowing) %atiOss(i,:) is in the middle o
alinasi-alinasi-imigli (imiovarowa)) d	<pre>if last(CTprevious)+1==CTfollowing(1) %ati0ss(i,:) has a unique CT</pre>
$\Lambda_{2}^{n}\Lambda_{2}^{n}\Lambda_{2}^{n}$	<pre>if atir(last(CTprevious),2)>1 atir(last(CTprevious),4)==0 %if previous ro is notantially had include more rows</pre>
r i=2:atiNsL if atiNsL	earlierstart=last(CTprevious)-10; if earlierstart<1
n=atiNss(1,4)/(-50); end	earlierstart=1; end
<pre>if atiNss(i,4)==1 atiNss(i,7)=atiNss(i-1,7)+atiNss(i,1)-atiNss(i-1,1);</pre>	<pre>r2f=last(CTprevious) -earlierstart + 2; %+1 for loc of last(CTprevious), 1 for loc of row2fix</pre>
if atiNss(i,7) <atinss(i-1,7) %missed="" &="" (can="" 0)<="" happen="" if="" many="" marked="" packets="" rollover="" td=""><td><pre>else earlierstart=last(CTprevious);</pre></td></atinss(i-1,7)>	<pre>else earlierstart=last(CTprevious);</pre>
<pre>n=floor((atiNss(i,3)-atiNss(i-1,3))/0.4096); if atiNss(i,1)-atiNss(i-1,1)<0</pre>	r2f=2; end
n=n+1; end	<pre>fixedrows=nta_fix([atir(earlierstart:last(CTprevious),:); ati0ss(i,:) 0; at (CTfollowing(1),:)],r2f); %use nta_fix</pre>
atiNss(i,7)=atiNss(i-1,7)+atiNss(i,1)-atiNss(i-1,1)+n*.4096; atiNss(i,4)=-1.1*n; %change to -lln to indicate missed rollover	<pre>atir=[atir(1:earlierstart-1,:);fixedrows;atir(CTfollowing(1)+1:atirL,:)]; </pre>
enc elseif atiNss(i,4)<0	<pre>eise %at10ss(i,:) has a repeat-CT reorderedrows=nta_fix_rr(atir(last(CTprevious):CTfollowing(1),:),at10ss(i,:)</pre>
<pre>atinss(i,/)=atinss(i-1,/)+atinss(i,1)+n*.4096-atinss(i-1,1); elseif atinss(i,4)==0 disso(fin this (chemidelle homes))</pre>	<pre>/ suse nta_fix_rr atir=[atir(1:last(CTprevious)-1,:);reorderedrows;atir(CTfollowing(1)+1:atir)); stim=[atir(1:last)];</pre>
<pre>usp(ix clis: (shouldn''t happen) = 3 = timeady'); atiMss(i,:) and</pre>	end
errer d	<pre>elseif isempty(CTprevious) %atiOss(i,:) is at the beginning of atir if CTfollowing(1)==1 %atiOss(:) has a unique CT t is the arm arm 1</pre>
$x_{x_{x_{x_{x_{x_{x_{x_{x_{x_{x_{x_{x_{x$	<pre>fixrows=[atiOss(i,:) 0; atir(1,:)]; fixrows(2,4)=-50; %set to -50 so okay i a rollover (wouldn't be -50 since it's the first row, and nta fix will look for = -50.</pre>
<pre>neck time difference between comptime and calculated time start to finish imel=atiNss(atiNssL.3)-atiNss(1.3)-atiNss(atiNssL.7);</pre>	<pre>fixedrows=nta_fix(fixrows,1); atir=[fixedrows:atir('ciatirL.')]; atir=fixedrows:atir('ciatirL.')];</pre>
	<pre>else %atiOss(i,:) has a repeat-CT reorderedrowstempenta fix rr(10 2 atir(1.3)-5 0 atir(1.5) 0 atir(1.7)-5 atir(1.5)</pre>
^%^%^%^% need to reconstruct full matrix; add atiOss back in	<pre>(1:CTfollowing(1),:)],atiOss(i,:)); % use nta fix rr with fake first row reorderedrows=reorderedrowstemp(2:CTfollowing(1)+2.:);</pre>
ir=atiNss; atirL=atiNssL; ti0L, toss]=size (ati0ss);	atir=[reorderedrows;atir(CTfollowing(1)+1:atirL,:)]; atirL=atirL+1; end
at rid of "fake" missing shoe data	else %isempty(CTfollowing) -> atiOss(i,:) is at the end of atir
oneshoemissing>0 ati0ssmati0ss(2:ati01);	<pre>if last(CTprevious)==atirL %ati0ss(:,) has a unique CT & is the new end row if atir(last(CTprevious).2)>1 atir(last(CTprevious).4)==0 %if previous ro</pre>
\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20
\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows caliestart=Lass(Cfreevious)-10;	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notiseepty(rigperfieg) tringportfind(stri(s))==1);
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rowe interstart-in(Cfreerious)-10; if earlierstart-i earlierstart-i;</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if noliemety/tidajet::2)==1); tf sempty/tidajet::2)==1); if sempty(trigapot) length(trigapot)>1 if reror('problew lich finding triggetflag ')
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows earlierstart=list(CTprevious)-10;</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if noisempty(triggerflag) triggpotFind(atif(;,2)=-1); if isempty(triggpot)) ength(triggpot)>1 erds triggerflag ') end
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potntially bad, include more rows earlierstart=last(CTprevious)-10; if earlierstart=1; end earlierstart=1; end earlierstart + 2; %+1 for loc of last(CTprevious), + 4 for loc f row2fix ele</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisempty(triggerflag) if notisempty(triggerflag) if isempty(triggerflag) length(triggerflag ') end if arror('problem with finding triggerflag ') end if atir(triggpot,5)=1 %if left shoe timetosubtractatir(triggpot, ');
<pre>\ajm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20 if noticesety(triggeflag) trigpotefind(atir(;2)=-1); if issmpt(trigpot) length(trigget)>1 error('problem with finding triggeflag ') end if atir(trigpot,5)==1 \if left shoe timetosubtractatir(triggot,7); else timetosubtract=atir(triggot,7)=int;
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisempty(riagerflag) if isempty(riagerflag) if isempty(riagerflag) end crot('problew with finding triggerflag ') end trat(ringspot,5)==1 %if left shoe timescombtractwatir(trigger,7); els imescombtractwatir(trigger,7)-intr end alif(:,8)=alif(:,7)-timetoubtract;
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisempty(trigpot1ag) trigpottind(atir(12))==-1) trigpottind(atir(12))==-1) error('problew uhf finding triggot1); end if atir(trigpot,5)==1 wif left shoe timetombtract=atir(trigpot,7)-int; end imetombtract=atir(trigpot,7)-int; end end atir(:,0)=atir(:,7)-timetombtract; omsebsforstrig=find(atir(1:trigpot-1,5)==1);
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows actilerstart(str(Gprevious)-10; if earlierstart(str(Gprevious)-10; if earlierstart(str(Gprevious)-arlierstart + 2; \+1 for loc of last(CTprevious), + < for loc of rowVik earlierstart=last(CTprevious); ind if=; ind if=; ind if=; ind ind</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if noliemety(tidgetflag) ti fisempty(tidgetflag) ti fisempty(tidgetflag) end if atir(trigget) length(tidgetflag) end if atir(trigget,5)==1 bif left shoe timetombtractatir(tidget,7); end atir(,8)=atir(,7)-timetombtract; comebeforetidgetflag(tidgetflag) atir(last(comebeforetidg),8); tidgepround(-atir(last(comebeforetidg),8),0134)-1; humber of dropped packets
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more cross antiperstartly (Typervious)-10; if earlierstartly end ad ad</pre>	<pre>C:\\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisempty(triggerflag) trigspote(tailing) ===1); if serve('problem with finding triggerflag ') end if stir(trigspot.5)==1 if laft shos timetonubtract=atir(trigspot.7); eld ali(:,0)=atir(:,7)-timetosubtract; consubtract=atir(trigspot.1,5)==1); atis(last(consubtract=tir(last(consubtract; timetonutract); else timetonutractir(last(consubtract; trigspercound-stir(last(consubtract; stimetonutractir(last(last(last(last(last(last(last(last</pre>
<pre>\ajm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bed, include more rose earlierstart=last(Cfprevious)=10; if earlierstart=last(Cfprevious)=0; earlierstart=last(earlierstart + 2; +1 for loc of last(Cfprevious), + < earlierstart=last(earlierstart=last(earlierstart + 2; +1 for loc of last(Cfprevious), + < earlierstart=last(earlierstart=last(earlierstart=last); earlierstart=last(earlierstart=last); earlierstart=last(earlierstart=last); earlierstart=last(earlierstart=last); earlierstart=last(earlierstart=last); earlierstart=last(earlierstart=last); earlierstart=last(earlierstart=last); earlierstart=last; earlierstart</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisempty(triggerflag) trigpotefind(atif(;2)=1); if arror((problem with finding triggerflag ') end if atif(trigpot,5)=1 if 1 aft shoe timetombtractwatif(trigpot,7); elsimetombtractwatif(trigpot,7); atimetombtractwatif(trigpot,7)=1; atimetombtractwatif(trigpot,7)=1; atimetombtractwatif(trigpot,7)=1; atif(atf(asedbergot,1); triggerround-atif(atf(atf(asedberflag),8)/.0134)=1; tummber of dropped packets else tif no trigger provided, then just set time in first row = 0 (if it inn't already i.e. if is/is/signer(7)=1;</pre>
<pre>\sjm\Research\Watlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if noticempty(triggerflag) trigpotefind(atif(:,2)=-1); if ismpty(trigpot) length(trigget)>1 end error('problem with finding triggerflag ') end if atir(trigpot,5)=1 %if left shoe timetonubtractwatir(trigpot,7); else end atir(:,8)=ai(:,7)-timetonubtract; onetworkforterigf(atif(1:fingpot-1,5)==1); triggepromd(-atir(last(onebeforetrig),8)/.0134)-1; tumber of dropped packets else %if at10.5; contained the real first row = 0 (if it isn't already i.e. if at10.5; contained the real first row) if at(1,7)?else(it.7)?timetonubtract; else %if at the shouldn't happen; timeadj no trig() dig('thooght this shouldn't happen; timeadj no trig()
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisespty(triggerflag) triggeotfind(si(x))==1); if isempty(triggeot) iempt)(triggeot)>1 error('problew with finding triggeotlag') end if atir(triggeot,5)==1 %if left shoe timetosubtract=ati(triggeot,7): else asimetosubtract=ati(triggeot,7): else immetosubtract=atir(triggeot,7): else inter(s)=atir(:,7)=timetosubtract; onesbeforetrig/s); triggeoround(=atir(lartigeot=1,5)==1); atir(s)=atir(:,7)=timetosubtract; inter(s)=atir(:,7)=timetosubtract; inter(s)=atir(:,7)=timetosubtract; inter(s)=atir(:,7)=timetosubtract; inter(s)=atir(:,7)=atir(s); else if so trigger provided, then just set lime in first row = 0 (if it int='t already i.e. if so trigger provided, then just set lime in first row = 0 (if it int='t already i.e. if so trigger provided, then just set lime in first row = 0 (if it int='t already i.e. if so trigger provided, then just set lime in first row = 0 (if it int='t already i.e. if so trigger provided, then just set lime in first row = 0 (if it int='t already i.e. if so trigger provided, then just set lime in first row = 0 (if it int='t already i.e. if so trigger provided, then just set lime in first row = 0 (if it int='t already i.e. if so trigger provided, then just set lime in first row = 0 (if it int='t already i.e. if so trigger provided, then just set lime in first row = 0 (if it int='t already i.e. if so trigger provided, then just set lime int s</pre>
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisespty(trigpot1ag) trigpottind(sit(1;2)=-1); if isempty(trigpot) length(trigpot)>1 error('problew with finding triggof1ag ') end if atir(trigpot,5)==1 wif left shoe timetosubtract=atir(trigpot,7)=1; else imetosubtract=atir(trigpot,7)=1; else imetosubtract=atir(trigpot,7)=1; atiggaprional(=atir(1:trigpot-1;5)==1); atiggaprional(=atir(1:trigpot-1;5)==1); atiggaprional(=atir(1:trigpot-1;5)==1); atiggaprional(=atir(1:trigpot-1;5)==1); atiggaprional(=atir(1:trigpot-1;5)==1); atir(1:s)=atir(:,7)=tir(1:trigpot-1;5)==1); else with no triggar provided, then just set time in first row = 0 (if it inn't already i.e. if a totos cost index then a tirt row) i atir(:,8)=atir(:,7)=atir(1:7); else i atir(:,1)=atir(:,7)=atir(1:7); else i atir(:,1)=atir(:,7)=atir(1:7); else i atir(:,8)=atir(:,7)=atir(1:7); else i atir(:,8)=atir(:,7)=atir(1:7); else i atir(:,8)=atir(:,7)=atir(1:7); else i atir(:,8)=atir(:,7)=a</pre>
<pre>\sip\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20 if notisempty(triggerflag) if notisempty(triggerflag) if isempty(triggerflag) if asempty(triggerflag) end if atir(triggerflag) end itimesombtractastir(triggerc, 7); end itimesombtractastir(triggerc, 7); end itin(:,8) atir(:,7); end itin(:,8) atir(:,7); end itin(:,8) atir(:,7); end itin(:,8) atir(:,7); end itin(:,8) atir(:,7); end atir(:,8) atir(:,7); end itin(:,8) atir(:,7); end atir(:,8) atir(:,7); end atir(:,7); end atir(:,7); end atir(:,7); end atir(:,7); end atir(:,7);</pre>
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisempty(triggerflag) if ispersy(triggerflag) if approxed is a second seco
<pre>\sjm\Research\Watlab\THESIS\timeadjuster.m Page 7 y12, 2004 7:03:20 AM potentially bad, include more rows</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisempty(triggerflag) trigpotefind(atif(;,2)=1); if arrc('problem with finding trigger); end if atif(trigpot,5)==1 if 16 shoe istationubtractwatif(trigpot,7)=int; end atif(;,8)=atif(;,7)=timtcoubtract; oneshforetrigf(ndirf(trigpot,7)=int; end atif(;,8)=atif(;,7)=timtcoubtract; oneshforetrigf(ndirf(trigpot,7)=int); atif(last(comeabeforetrig),8)/(0134)=1; timeher of dropped packets else tif no trigper provided, then just set time in first row = 0 (if it inn't already istif(;,8)=atif(;,7)=atif(;,7); else if (it(;,7)=atif(;,7); else if(;,8)=atif(;,7)=atif(;,7); else if(;,8)=
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisespty(triageoflag) if isempty(triageoflag) if isempty(triageoflag) if isempty(triageoflag) if isempty(triageoflag) if atir(triageoflag) if atir(the second seco
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if nntisempty(triggerflag) tignerpottind(stri(s))===1); if isempty(triggerf) immgh(trigger)); end if attr(trigger),)==1 wif left shoe timetoubbractwatir(trigger, 7); els immoubbractwatir(trigger, 7)=1; itintoubbractwatir(trigger, 7)=1; itinger, 7)=0; if attr(:,8)=attr(:,7)=attr(:,7); elseif attr(:,7)=1; attr(:,8)=attr(:,7)=attr(:,7); elseif attr(:,7)=1; attr(:,8)=attr(:,7)=1; attr(:,8)=attr(:,7)=1; attr(:,8)=attr(:,7)=1; attr(:,8)=attr(:,7)=1; attr(:,8)=attr(:,7)=1; iting</pre>
<pre>\sgim\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if mitseepy(rirger1ag) if instreepy(rirger1ag) if isempty(rirger) isempt(rirger1ag); end if atir(rirger0,)==1 if laft shoe timetombtractatir(triger0,)); eld eldetombtractatir(triger0,)); else iimtombtractatir(triger0,)); else iimtombtractatir(triger0,))=1); atir(1as(comsbefractirg), 0).013()-1; humber of dropped packets else bif no trigger provided, then yut set time in first row = 0 (if it inn't already ii.e. if atios contained the real first row) if atir(1,7)0 disp('thought this shouldn't, huppen: timedg) no trig() atir(:0) satir(:)) atir(1,7); else min(:0) satir(:)) atir(:)) trigger[]; trigger provided, then yut set time in first row = 0 (if it inn't already ii.e. if atir(:)) atir(:)) else min(:0) satir(:)) atir(:)) trigger[]; trigger provided, then yut set time in first row = 0 (if it inn't already ii.e. if atir(:)) atir(:)) else min(:0) atir(:)) atir(:)) trigger[]; trigger provided, then yut set time in first row = 0 (if it inn't already if atir(:)) atir(:)) atir(:)) trigger[]; trigger provided, then yut set time in first row = 0 (if it inn't already if atir(:)) atir(:)) trigger[]; trigger provided, then yut set time in first row = 0 (if it inn't already if atir(:)) atir(:)) trigger[]; trigger provided, then yut set time in first row = 0 (if it inn't already intris) atir(:)) timetempermonoratir(ifri); timetempermonor</pre>
<pre>Ny12, 2004 Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	<pre>C:\\sjm\Research\Matlab\THESIS\timeadjuster.m Ray May 12, 2004 7:03:20 if notisempty(trigger1ag) if notisempty(trigger1ag) if isempty(trigger1ag) if antic(trigger1ag) end if attr(trigger2,5)==1 bif loft shoe timesonbitactsatic(trigger2,1); end itimesobtractsatic(trigger2,1)==1); attr(lat(consebforetrig),0); timesobtractsatic(trigger2,1,5)==1); attr(lat(consebforetrig),0); timesobtractsatic(trigger2,1,5)==1); attr(lat(consebforetrig),0); timesobtractsatic(trigger2,1,5)==1); attr(lat(consebforetrig),0); timesobtractsatic(trigger2,1,5)==1); attr(lat(consebforetrig),0); timesobtractsatic(trigger2,1,5)==1); attr(lat(consebforetrig),0); timesobtractsatic(trigger2,1,5)==1); attr(lat(consebforetrig),0); timesobtractsatic(trigger2,1,5)==1); attr(lat(consebforetrig),0); timesobtractsatic(trigger2,1,5)==1); attr(lat(consebforetrig),0); timesobtractsatic(trigger2,1,5)==1); timesobtractsatic(tr</pre>
<pre>Alogin Research Matlab/THESIS/timeadjuster.m Page 7 y 12, 2004</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Page May 12, 2004 7:03:20 if notisempty(triggerIag) if notisempty(triggerIag) if a contempt and the second seco
<pre>\sjn\Research\Matlab\THESIS\timeadjuster.m Page 7 y12,2004</pre>	C:\sjm\Research\Matlab\THESIS\timeadjuster.m Pag May 12, 2004 7:03:20 if notisespty(trigger[ag) if isempty(frigger[ag) if isempty(frigger[ag] if isempty[if isempty[if isempty] if isempty[if isempty[if isempty] if isempty] if isempty[if isempty] if
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y12,2004 rc3320 AM potentially bad, include more rows</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timeadjuster.m May 12, 2004 T() if notisempty(trigger[ag) if notisempty(trigger[ag) if isempty(trigger[ag) if isempty(trigger[ag]) if isempty(trigger[ag]) if isempty(trigger[ag]) if isempty(trigger[ag]) if atic(trigger[ag]) if</pre>
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y 12, 2004 7:03:20 AM potentially bad, include more rows</pre>	C:\sjm\Research/Matlab/THESIS/Lineadjuster.m Page 7:03:20 if antiesepty(ringport[a) if instreepty(ringport] if isempt(trigport) instph(trigport)); end if altr(trigport,5)==1 if laft shoe timetombtractsatir(trigport,7); els end if altr(trigport,5)==1 if laft shoe timetombtractsatir(trigport,7); els end if altr(trigport,5)==1 if laft shoe timetombtractsatir(trigport,7); els end if altr(trigport,5)==1 if laft shoe timetombtractsatir(trigport,7)=1; end if altr(trigport,1)==1 if laft shoe timetombtractsatir(trigport,1)=1; end else lif no trigger provided, then just set lim in first row = 0 (if it inst laneady if it is 0; rowing the shoelds('t happen; timed) no trig() else lif no trigger provided, then just set lim in first row = 0 (if it inst laneady if it(t; 0) satir(t; 7)=tri(1); else if(t; 0) satir(t; 7)=tri(1); if(t) satir(t; 0) satir(t; 7)=tri(1); else if(t) trigger []; trigger irelevant and trigger[]; trigger and the table trigger [f,tcos]=find(umortatir(t,5)==); timeLemp=unstatif(t); 1 satir the prise trigger if provided that(t); 1 satir the prise trigger if provided that(t); 1 satir(t) trigger irelevant if cor(timepy(Bebechdow)); dep('ercor in secontracting time'; read if or(timepy(Bebechdow)); dep('ercor in secontracting time'; read if or(timept(t)); timet-timeBetom(t); 0); timet trimetemp(t); 0);
<pre>NajmlResearch/Matlab/THESIS/timeadjuster.m Page 7 y 12, 2004 r;03:20 AM rjotentially bad, include more rows</pre>	C:\sjm\Research\Matlab/TESIS\timeadjuster.m Page 7:03:20 if foriempty(triggerflag) if foriempty(triggerflag) if atir(triggerflag) if atir(triggerflag) is a cros('problem tinggerflag)' end itsetoshtrattatir(trigger,'); end itsetoshtrattatir(trig
<pre>Ny12, 204 Page 7</pre>	<pre>C:\sjm\Research\Matlab/THESIS\timeadjuster.m May 12, 2004 ff notiemety(triggerflag) ff interprotection f notiemety(triggerflag) ff interprotection fi amprotection fi amp</pre>
<pre>\sjm\Research\Matlab\THESIS\timeadjuster.m Page 7 y12,2004 7:03:20 AM potentially bad, include more rows</pre>	<pre>C:\sjm\Research\Matlab\THESIS\timeadjuster.m</pre>

dataadjuster.m

<pre>ideal_production_context_pr</pre>	<pre>[realtnew,fakejnew,newrownums,gaps]=gapfiller(time,fakedata,2*.00668); %to get record of points where gaps are filled in</pre>
	pornes where gaps are riried in
<code-block><pre>start with the start with the </pre></code-block>	<pre>dataadj(:,21)=realtnew(:,1);</pre>
	<pre>dataadj(:,22)=realtnew(:,2)*0.1; %0.1 is the flag for "generated" data during time gaps [dataadjlength,cols]=size(dataadj);</pre>
	for j=1:dataamount
	<pre>suge find outliers ss datatoreplace=findoutliers(data(:,j),time,sigthr(j),10+j,2,suspectrows);</pre>
	%%Replace outliers %%
	<pre>[dtrL,toss]=size(dtatoreplace); for i=l:dtrL</pre>
<pre>statustical statustical s</pre>	<pre>%replace data if datatoreplace(i,1)==1 datatoreplace(i,1)==datalength %if very first or v ¥</pre>
<pre>// *** *** *** *** *** *** **** **** *</pre>	ery last needs replacing, set = mean for that data data(datatoreplace(i),j)=mean(data(:,j));
<pre>listence de la constant de la c</pre>	else p=polyfit([time(datatoreplace(i,1)-1,1);time(datatoreplace(i,1)+1,1)],[da ď
	<pre>ta(datatoreplace(i,1)-1,j);data(datatoreplace(i,1)+1,j)],1);</pre>
	end
<pre>im it is for fug</pre>	<pre>[r,toss]=find(newrownums==datatoreplace(i)); %note that data was replaced</pre>
<pre>mining in the intervention of the interve</pre>	<pre>twoflag=dec2base(dataad)(r,22),3,18); twoflag()=num2str(datatoreplace(i.2));</pre>
<pre>memory memory memo</pre>	<pre>dataadj(r,22)=base2dec(twoflag,3); if datatoreplace(i,2)==1</pre>
<pre>statistical_property is a statistical_property is a statistical_p</pre>	<pre>qualinfo(1,6)=qualinfo(1,6)+1; elseif datatoreplace(i,2)==2</pre>
<pre>sective.com/s</pre>	<pre>qualinfo(1,5)=qualinfo(1,5)+1; end</pre>
<pre>Interactions for the first firs</pre>	end end
<pre>def_ distancesticity: distancestici</pre>	<pre>[toss,newdata]=gapfiller(time,data(:,j),2*.00668); %to get record of points where ga & ps are filled in </pre>
<pre>Numerical state sta</pre>	<pre>aataadj(:,j)=newdata; end</pre>
<pre>interacting interacting i</pre>	%% Graph if desired %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
<pre>manual</pre>	for i=0:1:4 figure (fig+1) clf;
<pre>p(s) sell; sell; sell; is fightEquipted for fightEquipted for fightEquiptEquiptEquipted for fightEquipt</pre>	end
<pre>set(); in</pre>	% figs=[fig fig fig fig fig fig fig fig fig+1 fig+1 fig+1 fig+1 fig+1 fig+1 fig+1 \varkappa];
<pre>tile="title</pre>	figs=[fig fig fig fig fig+1 fig+1 fig+1 fig+1 fig+2 fig+2 fig+2 fig+2 fig+3 fig+3 fig +3 fig+3 fig+4 fig+4 fig+4 fig+4];
<pre>id</pre>	titles=['hm 1';'hl 2';'mm 3';'ml 4';'heel 5';'toe 🖌 6';'bend ins 7';'bend ank 8';'gyro z 9';'accel z 10';'gyro y 11';'accel y 12'¢
<pre>Reserch/Matlab/THUSIS/datadjuster.m</pre>	<pre>// // // // // // // // // // // // /// /// ////</pre>
<pre>extraction, ext[1,20]; 13 4 15 3 4 12 3 4 12 3 4 12 3 4 12 3 4 12 3 4 12 3 4 1; xr [0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</pre>	
<pre>med(1,20); i 1 3 4 1 2 3 1 4 1 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1</pre>	
<pre>2 4 1 2 3</pre>	
<pre>i U O O O O O O O O O O O O O O O O O O</pre>	
<pre>itataanoont i</pre>	
<pre>prime (i); public to (j), na (j), ps(j)); gad=rg(; tadjdata (dataad); (,j), dataad); (,22), j, figpad); tadjdata (dataad); (,j), dataad); (,22), j, figpad); tadidata (dataad); (,j), dataad); (,22), j, figpad); talidata (dataad); (,j), dataad); (,j) = np); talidata (dataad); (,j), dataad); (,j) = nj; talidata (dataad); (,j) = nd; talidata (dataad</pre>	
<pre>indicated(ic,1).datad)(i,21).datad)(i,22).j,figped); leitlies(i,1); laitlies</pre>	
<pre>leakistatis(larval(1:2) lowar(j) highar(j)); ===================================</pre>	
<pre>i-wobj 2-trial 3-show 6-trigges 5-replatational 6-replate total ones 1-wobjects 5-rotal droppedpackets 9-hortesttretchofdroppedpack 4 setteretch 1-total sets of replacedpackets 10-freq packets 10-freq packets - length 1, 4 53 ty (ges) 1(a) = ges[eqs(r,2)]; toolumn 2 contains number of dropped packets at each g 4 2(1,9) =man(gaps(r,2)); toolumn 3 contains 1 if dropped packets were replaced 1(1,10) =man(gaps(r,2)); toolumn 3 contains 1 if dropped packets were replaced 1(1,0) =man(gaps(r,2)); toolumn 3 contains 1 if dropped packets were replaced 1(a) =man(gaps(r,2)); toolumn 3 contains 1 if dropped packets were replaced 1(a) =man(gaps(r,2)); toolumn 3 contains 1 if dropped packets were replaced 1(a) =man(gaps(r,2)); toolumn 3 contains 1 if dropped packets were replaced 1(a) =man(gaps(r,2)); toolumn 3 contains 1 if dropped packets were replaced 1(a) =man(gaps(r,2)); toolumn 3 contains 1 if dropped packets were replaced 1(a) =man(gaps(r,2)); toolumn 3 contains 1 if dropped packets were replaced 1(a) =man(gaps(r,2)); toolumn 4 contains 1 if dropped packets were replaced 1(a) =man(gaps(r,2)); toolumn 5 contains 1 if dropped packets were replaced 1(b) =man(gaps(r,2)); toolumn 5 contains 1 if dropped packets were replaced 1(c) =man(gaps(r,2)); toolumn 6 contains 1 if dropped packets were replaced 1(c) =man(gaps(r,2)); toolumn 7 contains 1 if dropped packets were replaced 1(c) =man(gaps(r,2)); toolumn 7 contains 1 if dropped packets were replaced 1(c) =man(gaps(r,2)); toolumn 8 contains 1 if dropped packets were replaced 1(c) =man(gaps(r,2)); toolumn 8 contains 1 if dropped packets were replaced 1(c) =man(gaps(r,2)); toolumn 9 contains 1 if dropped packets were replaced 1(c) =man(gaps(r,2)); toolumn 9 contains 1 if dropped packets were replaced 1(c) =man(gaps(r,2)); toolumn 9 contains 1 if dropped packets were replaced 1(c) =man(gaps(r,2)); toolumn 1 contains 1 if dropped packets were replaced 1(c) =man(gaps(r,2)); toolumn 1 contains 1 if dropped packets were replaced 1(c) =man(gap</pre>	
<pre>-mbj 2-trial 3-mbe 4-triggsp 5-replatatiotalize 6-replatatiotalizes -totalstatioproposkets #-totalreplacedpackets 13-freplackets-length1, 4' 3 4' (1,7) =spicedpackets 12-totalreplacedpackets 13-freplackets-length1, 4' 4' (1,7) =spicedpack (1,8) =sma(spgs(1,2)); (1,1) =sma(spgs(1,2</pre>	
<pre>id=cotingtestion: il=cotingtestionTreplacedpackets 12-totalreplacedpackets 13-freplackets-length1, </pre>	
<pre>interview imprivations moth.toss]=sime(sps); fo(1,7) =sim(sps(1,2)); fo(1,1) =sim(sps(1,2)); fo(1,1) =sim(sps(1,2)); fo(1,1) =sim(sps(1,2)); fo(1,1) =sim(sps(1,2)); fo(1,1) =sim(sps(1,2)); fo(1,1) =sim(spr(1,2)); fo(1,1) =sim(spr(1,2)); fo(1</pre>	
<pre>yuupas (1,1) = spaingapi (1,3) = smi(gapg(:,2)); (1,4) = smi(gapg(:,2)); (1,1) = sma(gapg(:,2)); (1,1) = sma(gapg(:,2));</pre>	
<pre>1,9) ===i(qsps(i,2)); toolum 2 contains number of dropped packets at each g </pre> 1,9) ==i(qsps(i,2)); 1,10) ==su(qsps(i,2)); 1,10) ==su(qsps(i,2)); toolum 3 contains 1 if dropped packets were replaced #sectroms(qsps)(onsetart(in(sortqsps2(:,2)=i)); 1,12) ==su(qspc(in(sortqsp2(:,2)=i)); and er(qsps(:,2)); and er(qsps(:,2)=i)); amefidic(sortqsp2(:,2)=i)); numtwordind(sortqsp3(:,2) = amefidi(sortqsp2(:,2)=i)); mpty(numthrees); qualif6(1,1)=length(numthrees); end mpty(numthrees); qualif6(1,1)=length(numthrees); end	
<pre>Ga(1,10) =max(appa(1,2)); Ga(1,10) =max(appa(1,2)); Ga(1,10) =max(appa(1,2)); Ga(1,10) =max(actapp3(1,0)=1); Ga(1,0); Jam); Jam) = Jam); Ja</pre>	
<pre>sl(1) sum(daps(;,)); toolum 3 contains 1 if dropped packets were replaced disactrow(daps,)]onestart(lostestart),2); tum just of packets wh (1,12) sum(dortgaps2(nestart(l):last(constart),2); tum just of packets wh disactrow(daps2, (r,2)=3); seesefin(dortgaps2(r,2)=3); seesefin(dortgaps2(r,2)=3); seepty(numbs/s); qualinfo(1,13)=length(numbs/s) end empty(numbs/s); qualinfo(1,15)=length(numbs/s) end ampty(numbs/s); qualinfo(1,15)=length(numbs/s) end</pre>	
<pre>12) ==uu(sortgapa3(onestart())(lat(onestart),2)); uum just of packets uh </pre> dear (upp:); jummensefind(sortgapa3(:,2)==1); numtwosfind(sortgapa2(:,2 of ind(sortgapa3(:,2)==3); cy(numones); qualinfo(1,1)=length(numtwos); and cy(numtwos); qualinfo(1,1)=length(numthrees); and cy(numthrees); qualinfo(1,1)=length(numthrees); and	
<pre>2=sortrows (gaps, 2) rummos=find(sortgaps2(:,2)=1) rumtos=find(sortgaps2(:,2 < ess=find(sortgaps2(:,2)==3); empty(numces); qualinfo(1,13)=length(numones); end empty(numthrees); qualinfo(1,15)=length(numthrees); end</pre>	
<pre>mpty(mumones); qualinf(1,13)=length(mumons); end mpty(numthrees); qualinf(1,15)=length(mumthrees); end</pre>	
<pre>pty(numthrees); qualinro(1,10)=length(numthrees); end</pre>	
<pre>12 3 4 1 2 3 4 1</pre>	

findoutliers.m



gapfiller.m

<pre>\\sjm\Research\Matlab\General\gapfiller.m Page 1 vy 12, 2004 7:12:44 AM</pre>	C:\sjm\Research\Matlab\General\gapfiller.m Page May 12, 2004 7:12:44 A
nction (xnew.vnew.rownums.gaps)=gapfiller(x.v.in.fig)	else
	<pre>xnew=x(1:gaps(1,1),1); %first block of good data</pre>
APFILLER(x,y,in,fig,Np,BF,figt)	<pre>xnew(1,2)=0; %(to start second column)</pre>
	ynew=y(1:gaps(1,1),1);
x: time, y: data, in: sample interval (default: 2/150)	rownums=rownums_orig(1:gaps(1,1),1);
	inew=gaps(1)+1; %inew: where new data will start
rig: value for the righte humber for graph of ynew, of u for ho graphs	for intert
(of if nome of the other variables are used, it can be left blank)	if gaps/i 2)// 2// 2/iii in three or fever missing packets
figt: like fig. but to plot ynew	$p_{ij}(x_i) = (x_i p_i x_j) + (x_i p_i x_j) $
	(1)[,1);
(c)2003 Stacy J. Morris ~ sjm@alum.mit.edu	
	<pre>for j=1:gaps(i,2)</pre>
	<pre>xnew(inew,1)=x(gaps(i,1))+j*in;</pre>
nargin<4	xnew(inew,2)=1; %flag to say this data is generated
fig=0;	<pre>ynew(inew, 1)=polyval(p, xnew(inew, 1));</pre>
d	rownums (inew, 1)=0;
	inew=inew+1;
······································	end
	<pre>gaps(1,3)=1; %1 to indicate this gap data was replaced</pre>
alength-length(x);	erica.
wargen (y) /	if idea.
error('SAPETLLER' x.v must be the same length')	<pre>versex: x(gaps(i,1)+1:gaps(i+1,1),1) zeros(gaps(i+1,1)-(gaps(i,1)+1)+1,1)</pre>
	1; %add in next block of good data (new values already added in)
	<pre>vnew=[vnew; v(gaps(i,1)+1:gaps(i+1,1),1)];</pre>
• • • • • • • • • • • • • • • • • • • •	rownums=[rownums; rownums orig(qaps(i,1)+1:qaps(i+1,1),1)];
* * * * * * * * * *	inew=length(xnew)+1;
i gaps	else
= [];	<pre>xnew=[xnew; x(gaps(i,1)+1:datalength,1) zeros(datalength-(gaps(i,1)+1)+1,1)];</pre>
	%add in last block of good data (new values already added in)
nge=diff(x(:,1));	<pre>ynew=[ynew; y(gaps(1,1)+1:datalength,1)];</pre>
<pre>ige(r, z)=round(xchange(r, 1)/in)=1; hereised(unberge(z, 1)/in)=1;</pre>	rownums=[rownums; rownums_orig(gaps(1,1)+1:datalength,1)];
Cosaj=tina (kenange (; /2) /0) /	and
<pre>state competent (); .tosslerize (cons);</pre>	end
, construction of the second	
t up rownums	%Graph, if desired
i=1:datalength	if fig>0
rownums_orig(i,1)=i;	figure(fig);clf;hold on;
	plot(x,y,'b.')
	axval=axis;axis([axval(1:2) 0 1]);
	plot(xnew(:,1),ynew,'color',[0.1 0.8 0.2])
	plot(xnew(:,i),ynew,'r.')
	if iconv(app)
ning off MatlaB:polyfit:PolyNotDoingo:	slabel (No. Gapt))
teckmen=menn(v);	changefont('arial',12,'x');
<pre>xxtremadiff=max(max(y)-checkmean,checkmean-min(y));</pre>	end
	end
isempty(gaps)	1
xnew=x;	1
xnew(1,2)=0;	1
ynew=y;	1
rownums=rownums_orig;	1
	1
disp('No gaps!!')	
disp('No gaps!!')	
disp('No gaps!!')	

Ι

plotadjdata.m

unction plotadjdata(data,time,adjustments,j,fig) plotadjdata(data.time.adjustments.i,fig)	
plotadidata(data.time.adjustments.j.fig)	plot(sdta2s(1,2),sdta2s(1,1),'k.','linewidth',0.5,'markersize',ms_k); %plot
	as black instead of blue dots elseif twoflagj==1 %replaced data
takes in column vector of data, column vector of adjustments (as labeled by dataadjuster.m), "j" identifying data vector's location and fig (i.e. from gcf), and plots adjustments	<pre>plot(stat2s(i,2),sdta2s(i,1),'ki', 'markerine',ms_g,'markerinecolor','g')kc lor', l1 0(,'linstiyie',','linstixi','o'); bjot as molium green dimonds plot(stat2s(i,2),sdta2s(i,1), 'ky', 'markerine',ms_r',markerinecolor','r') % olor', l0.10, 5), 'linstiyie',', ', 'linstixi','0.3); bjot as red partages</pre>
(c)2003 Stacy J Morris - sjm@alum.mit.edu	else dirp()ub_ab()
	end
<pre>igure(fig);hold on; lot(time,data,'k'); %'color',[0.1 0.6 0.1]); %plot line of data in green</pre>	end
<pre>tar(data time adjustments); demaortrew(da.3); foot by adjustments. dtal=langth(data); </pre>	
s_r=10;	
<pre>C0_toss]=find(sdta(i,3)==0); %ind the zeros - these are original, unchanged data Ou=length(fo); %eros will all be in order lot(sdta(c0(1):=0(f0L),2),adta(f0(1):=0(f0L),1),*k.','linewidth',0.5,'markersize',m %plot as black dots instead of blue</pre>	×
<pre>rl.cl]=find(ddt;,)==0.1); %find the point ones - these are data added by gapfill f how(issepty(rl)) rlb=langth(rl); nones will all be in order plot(sddtr(rl)):rl(rlcl,),sddtr(rl):rl(rlcl,),n,'ts','markersize',ms_gray,'mark color'(0.5 0.6 0.8)); %plot as gray dots instead of as (0 0.9 1), light blue dots ad</pre>	
f isempty(r1) & r0L==sdtaL %no ones or twos/changes	
<pre>sdta2s=[]; lseif isempty(r1) &r0L<sdtal %no="" -="" ones=""> everything after zeros is twos/changes edta2e=edta(r0(r01)+1)rdta1 r);</sdtal></pre>	
<pre>sutars=sutarro(rol)+r=sutar,;;; lseif r0(rol)+1==r1(1) & r1(rlL)==sdtaL %ones immediateLy follow zeros, no more dat o twos</pre>	4
<pre>sdta2s=[]; lseif r0(r0L)+l==r1(1) & r1(r1L)<sdtal %ones="" af<="" data="" follow="" immediately="" more="" td="" zeros,=""><td>4</td></sdtal></pre>	4
<pre>sdta2s=sdta(r1(r1L)+1:sdtaL,:); lse</pre>	
disp('what did I miss?') nd	
<pre>f not(isempty(sdta2s)) %twos could be for this sensor or could be for others. [sdta2sL,toss]=size(sdta2s);</pre>	
<pre>for i=1:sdta2L twoflagdec2base(sdta2s(i,3),3,18); twoflagj=str2num(twoflag(j));</pre>	
if twoflagj==0 %original data	

gencalibrations.m



clippedmean.m



linear_integrator.m



linear_integrator_endadjust.m

ay 12, 2004 8:40:01 AM	May 12, 2004 8:40:01 A
<pre>yy 12, 2004 member [In_data,madge_value]=linear_integrator_endadjust(data,time,initialvalue,endval *) [INEXES_INTEGRATOR_ENDATOR_CALL, ine, initialvalue, notvalue, toporbottom) integrates the * lata * function v. time, with the assumption that DelT is small enough that the data function * s linear Butween each belT. This is for use with data where the zero offset is not absolutely known. It integrates * he data as is, and then checks the last value against endvalue, and iterates mudging the data up/ * zesulting endvalue is leasthan is of fullscale LESS THAN the given endvalue. To return the integrated function, and the modge value (subtract the modge value to a *</pre>	<pre>NBY 17, 2004 E:2011 A E:2</pre>
ust)	tory=[nudge_value;nudgehistory(1)];
(c)2003 Stacy J. Morris - sjm@alum.mit.edu	
	end
initial integration t_data=linear_integrator(data,time,initialvalue);	if getoutofjailfree 4 isempty(smallbutpos) 4 abs(last(int_data))<10*endlimit; %e.g. w ithin 1% here as well as above in case the round which resulted in the getoutofjailfree for the second state of the
endlimit = .5% fullscale	<pre>% disp(['qet out of jail free!! - at time ' num2str(last(time))]);break;</pre>
dlimit=.001*(max(int_data)-min(int_data));	end
capehatch=50;	
<pre>ta_auj-wata; tsgooddiff=0; getoutofjailfree=0;smallbutpos=[]; usesmallbutpos=0; rsttime=1;nudge_value=0; lastdiff=0;nudgehistory=[0; 0]; tble accurace/statb20</pre>	=-1; end
<pre>%break once within limit if abs(endvalue-last(int_data))<endlimit< pre=""></endlimit<></pre>	data_adj=data+nudge_value;
<pre>ir endvalueslast(int_data) getoutorjaiirree break; else</pre>	<pre>sre-integrate int_data=linear_integrator(data_adj,time,initialvalue);</pre>
<pre>if isempty(mmallbutpos) smallbutpos(undge_value abs(endvalue-last(int_data))); elseif abs(endvalue-last(int_data))<mmallbutpos(1,2) abs(endvalue-last(int_data)));<="" pre="" smallbutpos(undge_value=""></mmallbutpos(1,2)></pre>	<pre>escapehatch=scapehatch=1; firsttime=0; if scapehatch=0;disp([' at time ' num2str(last(time)) ', escape hatch flipped']);end end</pre>
end end end	
<pre>%set nudge value (e.g. if endvalue is 0 and last(int_data) is negative, nudge value w l be positive)</pre>	
<pre>%if stable, signs should switch, and diff should get smaller if (firsttime [(oppsigns([endvalue-last(int_dats)),lastdiff) & abs(endvalue-last(in '' dats))==lass(rabs(endvalue-last(int_dats)),abs(lastdiff))))</pre>	
<pre>nudge_value=nudge_value+(endvalue=last(int_data)); lastdiff=(endvalue=last(int_data)); nudgehistory=[nudge_value;nudgehistory(1)]; else</pre>	
<pre>if getoutofjailfree & isempty(smallbutpos) & abs(last(int_data))<10*endlimit; %e. & within 1% disp(['get out of jail free! - at time ' num2str(last(time))]);break;</pre>	

simpsonsintegrator



simpsonsintegrator_endadjust.m



accel_integrator.m

C:\sjm\Research\Matlab\Analysis\accel_integrator.m Page 1 May 12, 2004 7:48:55 AM	C:\sjm\Research\Matlab\Analysis\accel_integrator.m Page : May 12, 2004 7:48:55 AM
<pre>unction [accintresult_sroom,accintresult_yroom,accintresults_schoe]=accel_integrator(M < say,MeasY,time,pitch,stack,subj,boundlocs,fig) f argint@ fig==0;</pre>	<pre>Yiisp con(boundlocsY(i,i):boundlocsY(i,j):houndlocsY(i,j)</pre>
ind	x,1))=0;
i − 8 − 8 Constants 8 − 8 − 8 − 8 − 8 − 8 − 8 − 8 − 8 − 8	Xdisp_shoe=linear_integrator(Xvel_shoe,time,0); %units in m Xdisp_room=linear_integrator(Xvel_room,time,0); %units in m
<pre>- % Load calibration info % - % - % - % - % - % - % - % - % - %</pre>	<pre>if figo0 figure(fig);clf; subplot(2,1,1) hold on;plot(time, fax Md]);title('x accels'); plot(time(boundlocsX(:,1)), Ax(boundlocsX(:,1)), 'gx');plot(time(boundlocsX(:,2)), Ax boundlocX(:,1), rat) x boundlocX(:,1), rat) x boundlocX(:,1), rat) x boundlocX(:,1), rat)</pre>
<pre>come = ()putuatizet=tendecus (utilizational and the state of t</pre>	<pre>subplot(2,12) hold on;plot(time,[Ay]);title('Y accel'); plot(time(boundlocsY(:,1)),Ay(boundlocsY(:,1)),*gx*);plot(time(boundlocsY(:,2)),A v(boundlocsY(:,2)),*x*);</pre>
<pre>f2g= 0.5*(scales(1,:)-scales(2,:))/(9.81); %scale factor to convert to m/s2</pre>	<pre>figuring+ifigure (fig)rolf; subplot(2,1,1);hold onplot(time,(Xvel_room Xvel_shoe));title('X velocities'); subplot(2,1,2);hold onplot(time,(Xvel_room));title('Y velocities'); figwfig+ifigure(fig)rolf; subplot(2,1,2);hold onplot(time,(Xdisp_room Xdisp_shoe));title('X disps'); amblot(2,1,2);hold onplot(time,(Xdisp_room));title('Y disps'); end</pre>
his=[thts+alphas(1,1) thts+alphas(1,2)]; - k - k Dynamic set up k - k - k - k - k - k - k - k - k - k	accintresults_xroom=[Ax Xval_room Xdisp_room time]; accintresults_xrhow=[Mx Xval_show Xdisp_room time]; accintresults_yroom=[Ay Yval_room Ydisp_room time];
<pre>kx=(MeasX-Gx); %dyn=meas=stat ty=(MeasY-Gy);</pre>	
$1 - 1 - 1$ Determine M_2 , M_2 $1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 $	
xx=-(Mx.*spy=My.*apx)./(cpx.*spy=cpy.*spx); %=sign y==-(Mx.*cpy=My.*cpx)./(spx.*cpy=spy.*cpx); %=sign	
woundlocsX=boundlocs; boundlocs;	
<pre>Gr i=lrime(houndlocx(i)) Xwel, hoke(bundlocx(i)) houndlocx(i,2),1) =linesr_integrator(Mr(houndlocx(i,1):ho w mdlocx(i,2)), time(houndlocx(i,1):houndlocx(i,2)),0); units in m/s Xwel room(houndlocx(i,1):houndlocx(i,2)),0); units in m/s mdlocx(i,2)), time(houndlocx(i,1):houndlocx(i,2)),0); units in m/s Twel room(houndlocx(i,1):houndlocx(i,2)),0); units in m/s mdlocx(i,2)), time(houndlocx(i,1):houndlocx(i,2)),0); units in m/s</pre>	

Τ

gyroz_linint.m

C:\sjm\Research\Matlab\THESIS\gyroz_linint.m	Page 1	C:\sjm\Research\Matlab\THESIS\gyroz_linint.m	Page 2
day 12, 2004	8:36:56 AM	May 12, 2004	8:36:56 AM
function [pli,midptbounds,gyrozcal]=gyrozlinint(gyroz_raw,time,stac	k,hsto,fig,mghgyroztim 🖌	end	
5)		end	
		% bounds	
^^	2	<pre>if isempty(bounds) bi<=3/if rig>0/rigure(rig)/cir/subplot(2,2,1 ;midptbounds=0/gyrozcal=gyroz/scalefactor;end;return;end</pre>)/piot(time,gyroz)/pii=0
load c:\sjm\matlab\transferred\gyrocalfac		for i=2:bL	
Load C:\SJM\Kesearch\Data\CallDIactors\gyrocalIac		midpt=round (U.S^ (Dounds (1,1) + Dounds (1-1,2))))	
recordedshift=gyrocal st1(3,1);		end	
scalefactor=gyrocal st1(3,2);		bounds=bounds(2:bL-1,:);	
alse		midptbounds=midptbounds(2:bL-1,:);bL=bL-2;	
recordedshift=gyrocal_st2(3,1);		8^^ Linear Integration^^^^	
scaleractor=gyrocal_st2(3,2);		for i=1:bl	
		<pre>pitch li cmzo(midpthounds(i,1):midpthounds(i,2))=lipear integ</pre>	rator (gyroz (midpthounds)
^^ Find zero-offset^^^^^^	^ <i>v</i>	i,1):midptbounds(i,2),1)/scalefactor,time(midptbounds(i,1):midptb	ounds(i,2),1),0);
=]ength(gyroz_raw):mgz=mean(gyroz_raw):nf= 001;		pitch li cmzo(L)=0:	
		gyrozcal=gyroz/scalefactor; %for return argument	
abovemean=0;stophereformean=10;			
for i=1:L %walk forward until significantly more than mean		\$^^ Plot, if desired^^^	^^
1f gyroz_raw(1)>mgz+10*nf		16 61-00	
end		figure (fig) : c) f:	
and		subplot(2,2,1); hold on;	
for i=abovemean:-1:greater(10,greater(floor(.75*abovemean),abovemea	n-30)) %walk backward 🖌	<pre>plot(time,gyroz/scalefactor,'b'); %plot(tt,yy,'k.');</pre>	
intil close to mean OR first occurence of: 30 steps back / .75 sign	more	<pre>plot(time,gyroz/scalefactor,'b.');axvalg=axis;</pre>	
11 gyroz_raw(1) <mgz+ni stophereformezni brezk</mgz+ni 		plot(time(bounds(:,1)),gyroz(bounds(:,1))/scalefactor,'r.')	
end		plot(time(midptbounds(:,1)), gyroz(midptbounds(:,1))/scalefact	or, 'v, ')
end		plot(time(midptbounds(:,2)),gyroz(midptbounds(:,2))/scalefact	or, 'y.')
to=clippedmean(gyroz_raw(1:stophereformean));		<pre>axis([time(1)1 last(time)+.1 axvalg(3:4)]);</pre>	
gyroz=gyroz_raw-zo;			
		supplot(2,2,3) food on/	
^ Dund		axis([time(1)1 last(time)+.1 axval(3:4)]);	,
oounds=gyrozlinintboundfinder(gyroz,time,hsto,nf);		stretchgraphs(2,2);	
[bL,toss]=size(bounds);[htL,toss]=size(hsto);			
bounds		subplot(1,2,2); hold on;	
bLinit=bLthoundrinit=boundry		<pre>plot(mgngyroztime(:,2),mgngyroztime(:,1), 'r', 'linewidth',2);a plot(time_pitch_li_cmro_lki_linewidth',2);</pre>	xvaim=axis/
iter=2/incrnf=[5/2]/		axis([axvalm(1:2) axval(3:4)]);	
while iter>0		end	
bounds=gyrozlinintboundfinder(gyroz,time,hsto,nf*incrnf(ite	z));	<pre>pli=[pitch_li_cmzo',time];</pre>	
[bL,toss]=size(bounds);			
if pl==ntl+1/preak/end			
end			
bounds			
%using nf is always more accurate! so replace:			
for iDinit=2:bLinit %first gets tossed			
<pre>bounds (loc, 1)=boundsinit (ibinit, 1);</pre>			
end			
for ibinit=1:bLinit-1 %last gets tossed			
<pre>[toss,loc]=min(abs(boundsinit(ibinit,2)-bounds(:,2)));</pre>			
bounds (loc, 2)=boundsinit (lbinit, 2);			

gyrozlinint_withspline.m

unction [pli,psi,midptbounds,gyrozcal,splinefitpts]=gyrozlinint_withspline(gyroz_raw,tim 🖌	iter=iter-1; end
, stack, iis	^^	enu
load c:\ oad c:\sj	sjm\matlab\transferred\yrccalfac m\Research\Data\calibfactors\gyrccalfac	<pre>for ibinit=2:bLinit %first gets tossed [toss,loc]=min(abs(boundsint(ibinit,l)-bounds(:,l)));</pre>
f stack== record	1 edshift=gyrocal stl(3,1);	end for ibinit=1:bLinit-1 %last gets tossed
scalef	actor=gyrocal_st1(3,2);	<pre>[toss,loc]=min(abs(boundsinit(ibinit,2)-bounds(:,2))); bounds(loc,2)=boundsinit(ibinit,2);</pre>
record scalef	edshitt=gyrocal_st2(3,1); actor=gyrocal_st2(3,2);	end end if isemptv(hounds) bL==2;if fig>0;figure(fig);clf;subplot(2,2,1);plot(time,gvroz);pli
^_	^ Find zero-offset^^^^^^];psi=[];midptbounds=0;gyrozcal=gyroz/scalefactor;end;return;end
=length(g gz=mean(g	yroz_raw);	<pre>for i=2:bL midpt=round(0.5*(bounds(i,1)+bounds(i-1,2))); midptbounds(i-1,2)=midpt;midptbounds(i,1)=midpt;</pre>
f=.001;		<pre>end bounds=bounds(2:bL-1,:);</pre>
or i=1:L	0;stophereformean=10; %walk forward until significantly more than mean or raw(i)bmmz+10%nf	midptbounds=midptbounds(2:bL-1,:);bL=bL-2;
ab end	ovemean=i;break	<pre>for i=1:bL [pitch_li(midptbounds(i,1):midptbounds(i,2)),nudges(i,1)]=linear_integrator_endadju</pre>
nd [aboveme	an floor(.75*abovemean) abovemean-30)	<pre>(gyroz(midptbounds(i,1):midptbounds(i,2),1)/scalefactor,time(midptbounds(i,1):midptboun (i,2),1),0,0);</pre>
ntil clos if qyr	e to mean OR first occurence of: 30 steps back / .75 signmore oz raw(1) <mgz+nf< td=""><td><pre>pitch_li(L)=0;pitch_li(midptbounds(i,2))=0; %(other midptbounds get set to 0 during fol wing integration)</pre></td></mgz+nf<>	<pre>pitch_li(L)=0;pitch_li(midptbounds(i,2))=0; %(other midptbounds get set to 0 during fol wing integration)</pre>
end	ophereformean=1; break	%^^ Set up gyrozcal,pitch_li for return argument^^
nd o=clipped	mean(gvroz raw(l:stophereformean));	gyrozcal=gyroz/scaletactor; pitch_li=pitch_li'; %^^^ Look for significant gaps, use splines as necessary^
in case d	ata doesn't start with the foot flat on the floor (most do, so above procedure \varkappa	<pre>%look up gaps usesplines=[];</pre>
f abs(zo-), use recordedshift instead, and allow a bigger noise floor recordedshift)>.05 ordedshift)=.05	tor i=1:bL %check data rate - if less than 70 Hz, then check for consecutive dropouts of 4 pts r locars - if any exist we enline for this star.
nd zu=rec	Generalize, nr031	<pre>if (midptbounds(i,2)-midptbounds(i,1))/(time(midptbounds(i,2))-time(midptbounds(i,1))/(time(midptbounds(i,2))-time(midptbounds(i,1))/(time(midptbounds(i,2)))<70</pre>
yroz=gyro	z_raw-zo;	<pre>for m=midptbounds(i,1)+1:midptbounds(i,2) if (time(m)-time(m-1))>5*(2*.00668) %dropout of 4 pts = gap of 5*.0134 iv=reminer(d,1)=1.1</pre>
ounds=gyr	^^ Find integration bounds^^^^^^^ ozlinintboundfinder(gyroz,time,hsto,nf);	<pre>usesplines(1,1)=1; end end</pre>
bL,toss]=	size (bounds); [htL,toss]=size(hsto);	end end
f bL <htl+ bLinit</htl+ 	1 =bL; bounds:nit=bounds; timersf=[5:2];	<pre>%use splines if necessary if noticepti(user)</pre>
while bo	iter>0 unds=qyrozlinintboundfinder(qyroz,time,hsto,nf*incrnf(iter));	<pre>if length(usesplines)<bl; %make="" (if="" as="" be="" bl="" latter="" length="" n="" not,="" pre="" same="" should="" t="" usesplines(bl,1)="0;end" values="" zero)<=""></bl;></pre>
[]=		
(D if	L, Cossistie (Dounds) / bl=mtLi) break red	Gilsjm\Research\Matlab\THESIS\gyrozlinint_withspline.m
:\sjm\Re ay 12, 2 tt=[ti	L,coss siie(bounds)/ bl=mtlijbreak/end ssearch/Matlab/THESIS\gyrozlinint_withspline.m Page 3 2004 8:45:22 AM mc():0.01:last(time)';rtt=length(tt)/	G:\sjm\Research\Matlab\THESIS\gyrozlinint_withspline.m Page May 12, 2004 8:45:22
:\sjm\Re ay 12, 2 tt=[ti yy=ppv	<pre>L_toss_reise(bounds); bb=mtl+l;break;end seearch/Watlab/THESIS\gyrozlinint_withspline.m Page 3 2004 8:45:22 AM me(l):0.01:last(ims)!';tL=length(tt); ine(ims.gyrcs/scalefactor); al(pp,tt);</pre>	<pre>G1\cjm\Research\Matlab\THESIS\gyrozlinint_withspline.m Page May 12, 2004 8:45:22 if fip0p t2plot=[t2plot:[t2plot:t(cplinsintpts(p,1):plinsintpts(p,2)]); end tjust plot these as black dot</pre>
(D if (:\sjm\Re lay 12, .2 tt=[ti pp=spl yy=ppv save c	<pre>L_toss_state(bounds); bb=mtL+ljbreak;end seearch/Watlab/THESIS\gyrozlinint_withspline.m Page 3 2004 8:45:22 AM me(l):0.01:last(time)!';tL=length(tt); ine(time.gyrcs/scalefactor); al(pp.tt); :\sjm/Research/Data/calibfactors\anglepp pp</pre>	C:\cjm\Research\Matlab\THESIS\gyrozlinint_withspline.m Page May 12, 2004 E:45:22 if fip:0; t2plot=(t2plot;t(splineintpts(p,1):splineintpts(p,2)));yy2 ot=(yy2plot;y(splineintpts(p,1):splineintpts(p,2)); and tjust plot these as bias end else pitch_si=(pitch_si;pitch_1i(snidptbounds(i,2)-1));
if if :\sjm\Re lay 12, 2 tt=[ti pp=spl yy=ppv save c spline spline	<pre>L,coss=sise(bounds); bk=mtili);reak;red seearch/Watlab/THESIS\gyrozlinint_withspline.m Page 3 2004 8:45:22 AM me(l):0.01:last(time)!);rti=length(tt); ine(time.gyroz/scalefactor); al(pp.tt); \\sjm(Nessarch/Uata\califactors\anglepp pp mdptbounds(;,1)*findtimepoints(time(ideptbounds(;,1)),tt); mdptbounds(;,2)*findtimepoints(time(ideptbounds(;,2),tt);</pre>	C:\cjm\Research\Matlab\THESIS\gyrozlinint_withspline.m Page May 12, 2004 8:45:22 if fip0) tt2plet=[t2plets[t2plets[t2plets[t,1]):splineintpts[p,1]):pdlineintpts[p,2])]; ot=[yy2plet;yiqplineintpts[p,1]:splineintpts[p,2])]; end tjust plot these as black dot else pitch_si=[pitch_si;pitch_ii(midptbounds(i,1):midptbounds(i,2)-i]); end end
ID if if if if if if if if if if	<pre>L,tospissie (bounds); bb=mtFiltyreaty end 2004</pre>	C:\sjm\Research\Matlab\THESIS\gyrozlinint_withspline.m Page May 12, 2004 E:Sister C:Sister C:
if if if if ay 12, 2 tt=[ti pp=spline spline spline t====================================	<pre>L_toos_ising houses; _toos_ising houses; seearch/Matlab/THESIS/gyrozlinint_withspline.m Page 3 2004 8:45:22 AM ms(1):0.01:last(tims)';ttl=length(t); in(tims.gyroz.calefactor); al(pp.tt); 'isjnkBesserAhata/calibfactors\anglepp pp mgbbonds(r,1)~find:impoints(immediatbonds(r,1)),tt); mgbbonds(r,2)~find:impoints(immediatbonds(r,2)),tt); ****-******************************</pre>	C1\cjm\Research\Matlab\THESIS\gyrozlinint_withspline.m Page May 12, 2004 E45:22 if ipp() t2plot=(t2plot=(t2plot=t2plot=t(t2plot=t(t2plot=t(t2plot=t2plot=t(t2plot=t(t2plot=t(t2plot=t(t2plot=t2p
<pre>if if if if if if if if if if if if if i</pre>	<pre>Licesjesie(bounds); bb=mtirijvreakyend 2004 Pristale VHESIS\gyrozlinint_withspline.m Page 3 2004 8:45:22 AM ms(1):0.011ast(tims)';ttl=length(t); in(tims.gyroz.calefactor); al(pp.ti); 'isjnkesschkata(salifactors\anglepp pp mgbbounds(;,2);tl; mdpbbounds(;,2);tl; infibutors(i,2); infibutors(i</pre>	<pre>C1\cjm\Research\Matlab\THESIS\gyrozlinint_withspline.m Pag May 12, 2004 E45:22 if fip90 t2plots(t2plots(t2plots(t(splingingtes(sp,)))))) ot=(yy2plotsy(splingingtes(sp,))); splingingtes(sp,2))); and ijust plot these as black doft eide eise picch_siepitch_i(nidptbounds(i,1)); midptbounds(i,2)-1)); ise ai = [time_si; time(midptbounds(i,1); midptbounds(i,2)-1)]; end pitch_siepitch_sipitch_li(midptbounds(i,1)); midptbounds(i,2)-1)]; ise ai = [time_si; time(midptbounds(s)); midptbounds(i,2)-1)]; sidptbounds(i,3)=useplines; ito save locations where splines were used (note, then midptbounds(i,3)=useplines; ito save locations where splines were used (note, then midptbounds(i,3)=useplines; ito save locations where splines were used (note, then midptbounds(i,3)=0; set third column of midptbounds to all 0s (no splines used) end end</pre>
<pre>if if if if if if if if it it it it it if if</pre>	<pre>L_toos_state [bounds]; bb=mtiripreasy and 2004</pre>	<pre>C:\sjm\Research/Matlab\THESIS\gyrorlinint_withspline.m Pag May 12, 2004 Pig etcl: cif gip0; tt2plot=[tt2plot;tt(gplineintpts(p,1):splineintpts(p,2)]; and ijuut plot these as black dot minimized the set of the set of</pre>
<pre>iB if if if if if if if if if if if</pre>	<pre>L;cosj=size(boods); bb==mtirijireaty end 2004 B:45:22 AM mc(1).0:liast(time)';ttl=length(tt); in(time_greys:calsfactor); al(pp.tt); 'isj=lkesserA(bata(califactors)anglepp pp mdptboods(:,2)=findtimepoints(time(midptboonds(:,1)),tt); mdptboods(:,2)=findtimepoints(time(midptboonds(:,1)),tt); mdptboods(:,2)=findtimepoints(time(midptboonds(:,1)),tt); mdptboods(:,2)=findtimepoints(time(midptboonds(:,1)),tt); mdptboods(:,2)=findtimepoints(time(midptboonds(:,1)),tt); mdptboods(:,2)=findtimepoints(time(midptboonds(:,1))); test init < tripy:point=(); tible maphical()>0 kooly integrate using spline if marked, otherwise, store for smalphpoints(i); integrate using spline if marked, otherwise, store for smalphpoints(i); integrate using spline if marked, otherwise, store for smalphpoints(i); integrate using spline if marked, otherwise, store for smalphoonds(i,1)=integrate); mod end end end end end end end end end en</pre>	<pre>C:\sjm\Research\Matlab\THESIS\gyrorlinint_withspline.m Pag May 12, 2004 Page else</pre>
<pre>ip if if if if if if if if if if if</pre>	<pre>L,cosj=sise(bounds); bb==mtiriyreakyend seearch/Matlab/THESIS/gyrozlinint_withspline.m Dage 3 2004 8:45:22 AM ms(1).0.0:llast(time)';ttt=length(tt); ins(time.gyrcs/caldfactor); al(pp.tb); 'iyjalkesearch(bata/caldfactor)anglepp pn mgbbounds(;,2)=find:imspoints(time(midgbbounds(i,2)),tt); mgbbounds(;,2)=find:imspoints(time(midgbbounds(i,2)),tt); mgbbounds(;,2)=find:imspoints(time(midgbbounds(i,2)),tt); mgbbounds(i,2)=find:imspoints(time(midgbbounds(i,2)),tt); mgbbounds(i,2)=find:imspoints(time(midgbbounds(i,2)),tt); midpbounds(i,2)=find:imspoints(time(midgbbounds(i,2)),tt); mgbbounds(i,2)=find:imspoints(time(midgbbounds(i,2)),tt); mgbbounds(i,2)=find:imspoints(time(midgbbounds(i,2))-1); test init < tot()pybbolks(i,2)=find(imspoints(i,2)),tt); midbbounds(i,2)=find(imspoints(i,2)),tt); if(time(midgbbounds(i,2))-1 if(time(midgbbounds(i,2))-1 if(time(midgbbounds(i,2)),tt); midbbounds(i,2)=find(imspoints(i,2)),tt); intpr</pre>	<pre>C:\sjm\Research\Matlab\THESIS\gyrozlinint_withspline.m Pag May 12, 2004 Page if fig>0; tt2plot=[tt2plot;tt(aplineintpts(p,l):splineintpts(p,2)]; ot=[y2plot;yy2plots[tt2plot=[tt2plot;tt(aplineintpts(p,2)]; and iuux plot these as black dot item sit=[tims_ii]; time indeptbounds(i,l):sndptbounds(i,2)-1]; item sit=[tims_ii]; time indeptbounds(bL,2):t]; item sit=[tims_ii]; time indeptbounds(bL,2):t]; sndptbounds(;)fursuperplinas; plot some locations where splines were used (note, then into sit=[pitch_sisplitch_li(sndptbounds(bL,2):t]); sndptbounds(;)fursuperplinas; plot some locations where splines were used (note, then into sit=[pitch_sisplitch_li(sndptbounds(bL,2):t]); sndptbounds(;)fursuperplinas; plot colum of andptbounds to all 0s (no splines used) end into fitsp0 (sit=[pitch_sisplicat]); fursuperplication of the column of andptbounds to all 0s (no splines used) if fitsp0 fursuperplications; plot(tt2plotyy2plot,'k.');splinefitpt=(y2plot tt2plot); if oftigerplic(sispline);plot(tt2plotyy2plot,'k.');splinefitpt=(y2plot tt2plot); if fitsp0</pre>
<pre>in [] [] [] [] [] [] [] [] [] [] [] [] []</pre>	<pre>L_toos_state bounds;; bb==mtiripreasy and 2004</pre>	<pre>C:\sjm\Research\Matlab\THESIS\gyrozlinint_withspline.m Pag May 12, 2004 Page 2007 if fig>0; tt2plot=(tt2plot;tt(splineintpts(p,l)):splineintpts(p,2));ynd ot=[y2plot:rytgblineintpts(p,l):splineintpts(p,2)]; and your plot these as black dot plot(n, sic[pitch, sirpitch li(sidptbounds(i,l):sidptbounds(i,2)-1)]; time_ii = [time_ii]; time_iidptbounds(b,2):b]; and on the sic[pitch, sirpitch li(sidptbounds(b,2):b]; time_ii = [time_ii]; time_iidptbounds(b,2):b]; time_ii = [time_ii]; time_iidptbounds(b,2):b]; sidptbounds(;) abuse planes; the on need to stress ploth_sic[pitch_sic]; time_iidptbounds(b,2):b]; sidptbounds(;) abuse planes; the on need to stress ploth_sic[pitch_sic]; time on need to stress splitch_sic[pitch_sic]; time on need to stress splitch_sic[pitch_sic]; time_iid=[1; timese on need to stress] sidptbounds(;) abuse here one on the stress splitch_sic[pitch_sic]; time_iid=[1; timese one of the stress splitch_sic[pitch_sic]; timese one of the stress of the stress splitch_sic[pitch_sic]; timese on</pre>
in in it is a second se	<pre>L_toos_state bounds); bb==mtiritreesymed 2004</pre>	<pre>C:\sjm\Research\Matlab\THESIS\gyrozlinit_withspline.m May 12, 2004</pre>
in in in it is a second	<pre>Lices = is is [bounds]; bb==mtiri/irreaty and 2004</pre>	<pre>C:\sjm\Research\Matlab\THESIS\gyrozlinit_withspline.m May 12, 2004</pre>
<pre>if if i</pre>	<pre>L_toos_mission definition in the station of th</pre>	<pre>C:\sjm\Research\Matlab\THESIS\gyrozlinint_withspline.m Pag May 12, 2004 Page if fig=0; tt2plot=[tt2plot;tt(splineintpts(p,l):splineintpts(p,2)]; ot=[y2plot;yy2plot;jt2plot=[tt2plot;tt(splineintpts(p,2)]; and iuu piot these as black dot give its_its_its_its_its_its_its_its_its_its_</pre>
<pre>i h gjm RR if if</pre>	<pre>L_tooj=sise(boods); bb=mtlij:realy end 2004 BitSise(yrozlinit_withspline.m Page 3 2004 BitSize2 AM mc(1).0.0:last(time)]';ttl=despth(tt); in(time.grow_coalsdactor); al(pp.tt); 'isjn(Aksearch(bata(calificators(anglepp pp mcbboods(r,2)=findimepoints(time(midptboonds(r,1)),tt); mobboods(r,2)=findimepoints(time(midptboonds(r,2)),tt); </pre>	<pre>C:\sjm\Research/Matlab\THESIS\gyrolinit_withspline.m Page May 12, 2004</pre>
if if if if if if if if if if if if if i	<pre>L_tooj=sise(boods); bb==ntirjreasy end seearch/Matlab/THESIS/gyrozlinint_withspline.m Dage 3 2004 B:45:22 AM ms(1).0.0:Last(time)'/itt=length(tt); int(time.gyres/callsfactor); al(pp.tt); 'idjalkessarch(bata/callsfactor); al(pp.tt); 'idjalkessarch(bata/callsfactor); alpp.tt); 'idjalkessarch(bata/callsfactor); alpp.tt); 'idjalkessarch(bata/callsfactor); alpp.tt); 'idjalkessarch(bata/callsfactor); alpp.tt); 'idjalkessarch(bata/callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessarch(callsfactor); 'idjalkessar</pre>	<pre>C:\sjm\Research/Matlab\THESIS\gyrczlinint_withspline.m Page May 12, 2004</pre>
<pre>i kgjm\Ray12, ' i kgjm\Ray12, ' i tt=[ti tt=[t</pre>	<pre>Licesfield bounds); hb=mtliftread with the second sec</pre>	<pre>C:\sjm\Research\Matlab\THESIS\gyrozlinit_withspline.m Page May 12, 2004 E:45:22 ff fip0(tt2plot=(tt2plot;tt(splineintpts(p,1);splineintpts(p,2)));yd ot=[y2plot;y(splineintpts(p,1):splineintpts(p,2)]; end %ut plot these as black det eise pich_sis[pich_sispitch_limidpbounds(1,1):midpbounds(1,2)-1)]; end end end eise [ttim_gii = [tim_gii _ time[midpbounds(L_2):1]]; midpbounds(up):this spline(th_1):midpbounds(1,2)-1)]; midpbounds(up):this is any too when splines were used (note, these indpbounds(up):this midpbounds(L_2):1]; midpbounds(up):this midpbounds(L_2):1]; midpbounds(up):this midpbounds(L_2):1]; midpbounds(up):this thid column of midpbounds to all 0s (no splines used) end end if fip0 figure(fig):clf; mubplot(d_2,1);bloid on; plot(time_yrorisolfactor, "p); if notimepty(usespline(ttaplot ttaplot yyplot, 'k.'):splinefitpts=[yyplot ttaplot] plot(time_yrorisolfactor, "p); plot(time_yrorisolfactor, "p); plot(time_</pre>
<pre>in f if if</pre>	<pre>L_cosperse here to be a set of the set</pre>	<pre>cityph/Research/Matlab/THESIS/gytozlinit_withspline.</pre>
<pre>if if if if if if if if if if if if if i</pre>	<pre>L_cospersion bounds); bb==mtiri/recaly and 2004</pre>	<pre>C:\cjm\Research\Matlab\THESIS\yzclinit_withspline. May 12, 2004 SetSize if figP0: tt2plot=(tt2plot;t(splineintpts(p,l)):pilmeintpts(p,l));yd or(yzplot;yylplineintpts(p,l):pilmeintpts(p,l)); add yut plot these ab lake det</pre>
<pre>if if if if if if if if if if if if if i</pre>	<pre>L_cosplaint(back) bb=mtlij/realy end 2004</pre>	<pre>citypinkesearch/Matlab/THESIS/gyrozlinit_withsplines main property topics(tipping) citypingesearch/Matlab/THESIS/gyrozlinit_withsplines main property topics(tipping); and with property topics(tipping); citypingesearch/Matlab/THESIS/gyrozlinit_withsplines main property topics(tipping); and with property topics(tipping); citypingesearch/Matlab/THESIS/gyrozlinit_withsplines main property topics(tipping); and with property topics(tipping); citypingesearch/Matlab/THESIS/gyrozlinit_withsplines main property topics(tipping); and with property topics(tipping); citypingesearch/Matlab/THESIS/gyrozlinit_withsplines(tipping); citypingesearch/Matlab/THESIS/gyrozlinit_withsplines(tipping); citypingesearch/Matlab/THESIS/gyrozlinit_(tippinge); citypingesearch/Matlab/THESIS/gyrozlinit_(tippinge); citypingesearch/Matlab/THESIS/gyrozlinit_(tippinge); citypingesearch/Matlab/THESIS/gyrozlinit_(tip); citypingesearch/Matlab/THESIS/gyrozlinit_(tip); citypingesearch/Matlab/THESIS/gyrozlinit_(tip); citypingesearch/Matlab/THESIS/gyrozlinit_(tip); citypingesearch/Matlab/THESIS/gyrozlinity/file citypingesearch/Matlab/THESIS/gyrozlinity/file citypingesearch/Matlab/THESIS/gyrozlinity/file citypingesearch/Matlab/THESIS/gyrozlinity/file citypingesearch/Matlab/THESIS/gyrozlinity/file citypingesearch/Matlab/THESIS/gyrozlinity/file citypingesearch/Matlab/THESIS/gyrozlinity/file citypingesearch/Matlab/THESIS/gyrozlinity/file citypingesearch/Matlab/THESIS/gyrozlinity/file citypingesearch</pre>
<pre>in first state stat</pre>	<pre>Licesfield bounds); hb==ntfiftready end search/Watlab/HESIS/gyrozlinit_withspline.m Page 3 2004 8:45:22 AM ms(1):0.01:last(tims)';ttL=length(tt); in(tims.gyroz/caslefactor); al(p,tt); 'isjhkesarch/bata/calificators/anglepp pp msptbounds(r,1)=find:impoints(tims(indeptbounds(r,1)),tt); msptbounds(r,1)=find:impoints(tims(indeptbounds(r,1)),tt); ===================================</pre>	<pre>cityplexeserch/Watlab/THESIS/gyzozlinit_withsplines </pre>

gyroz_postspline.m

C:\sjm\Research\Matlab\THESIS\gyroz_postspline.m Page 1 May 12, 2004 7:45:25 AM function [splinepeaksH.splinepeaksL,biggestgap]=gyroz_postspline(glinint,time,fig) [peak_resultsH,peak_resultsL,biggestgap]=pitchpeakfinder(glinint,time,l); peak_results=[peak_resultsH;peak_resultsL]; if notisempty(peak_results)
peak_results(:,2),time);
for i=lisic(peak_results(:,2),time);
tor i=lisic(peak_results(:,1))+(1):.0034*time(peak_results(:,3)+4,1);
timespline=[time(peak_results(:,1)+4,1)]*(1);
timespline=[time(peak_results(:,1)+4,1)]*(1);
sults(:,1)+4,1);
timespline(:,1)+4,1);
timespline(:,1),timespline);
sults(:,1),nextimept);
eaks((:,1),nextimept); splinepeaksH=[splinepeaks[:size(peak_resultsH,1),:) peak_resultsH(;,3)];
sl(r,3)];
sl(r,3)];
else splinepeaksH=[splinepeaksL=[]
end

hstos_fsrsum_spline.m



pitch_compare_postspline.m

May 12, 2004 7:53:00 AM	May 12, 2004 7:53:00 A
function pitch_compare_postspline(start, stop,legendon)	<pre>[shoehR, shoelR,sbggapR]=gyrozlinint_postspline(dataR(:,15),dataR(:,21),gcf);</pre>
load c:\sjm\research\Matlab\run02\run02_ssAandB load c:\sjm\research\Matlab\run02\run02_ssCandD	<pre>[r1R,r2R]=compressmgh(mghdata(:,11));adjR=findmghoutliers(mghdata(r1R:r2R,11),mgh data(r1R:r2R,31),5,0);</pre>
<pre>load c:tyjw/cesarch/Matiab/run02/run02_sndada isad c:tyjw/cesarch/Matiab/run02/run02_sndada d:tyjw/cesarch/Matiab/run02/run02_sndata_adj isad c:tyjw/cesarch/Matiab/run02/run02/run02_sndata_adj isad c:tyjw/cesarch/Matiab/run02/run02/run02_sndarch/Data isad c:tyjw/cesarch/Matiab/run02/run02_sndarch/Data isad c:tyjw/cesarch/Data isad c:tyjw/cesarch/</pre>	<pre>data(rik:22,31),5,0);</pre>

Page 2 7:57:11 AM

pitchpeakfinder.m



getmghdataorder.m

<pre>Subtion mpdata_return-sptsphatacoder(vers) Latituditatives) Latituditatives Latituditatives) Latitudita</pre>	C:\sjm\Research\Matlab\THESIS\getmghdataorder.m May 12, 2004	Page 1 8:03:06 AM	C:\sjm\Research\Matlab\THESIS\getmghdataorder May 12, 2004	.m Page 8:03:06 #
<pre>varbure to the state of th</pre>	function mghdata_return=getmghdataorder(vers)		%12 Left Shank Roll	
<pre>vortex 1, das vetures is tils order:</pre>	RGETMGHDATA(vers)		%13 Left Shank Pitch %14 Left Shank Yaw	
<pre>i distant, dui scritting in tud cont i fathant(x,y,r), sliphthal(x,y,r), filthal(x,y,r), filthal(x,y,r),</pre>			shankLrot=mghdata(:,13:15);	
<pre>i perfahaki(x,y,z,P), RiphAnk(x,y,z), real/pitch(yx),</pre>	<pre>% LeftFoot(x,v,z,roll,pitch,vaw), RightFoot(x,v,z,roll,pitch,vaw),</pre>		%57 Right Foot Roll	
<pre>% LeftAhls(x,y,z), HydrAhls(x,y,z), Time % % % % % % % % % % % % % % % % % % %</pre>	% LeftShank(x,y,z,roll,pitch,yaw), RightShank(x,y,z,roll,pitch,yaw),		%58 Right Foot Pitch	
<pre>% % (c)2003 fkey J Moris - sjellem.mit.edu % (c)2003 fkey J Moris - sjell</pre>	% LeftAnkle(x,y,z), RightAnkle(x,y,z), Time		%59 Right Foot Yaw	
<pre>% (c)2003 Stery 3 Morris - signalum.mit.eds mbdtareseaksphdof; allargewhy; [sizeargs.col]=size(sharks[; Signalum.mit.eds] % K Right Hanks X Yw allargewhy; error(You mut enter a varient number and provide dats. See help.'); end [lempth.toss]=size(mphdata); if toss=-120 error(You mut enter a varient number and provide dats. See help.'); end [lempth.toss]=size(mphdata); if toss=-120 error(You mut enter a varient number and provide dats. See help.'); end [lempth.toss]=size(mphdata); if toss=-120 error(You mut enter a varient number and provide dats. See help.'); end [lempth.toss]=size(mphdata); if toss=-120 error(You mut enter a varient number and provide dats. See help.'); end [lempth.toss]=size(mphdata); if toss=-120 error(You mut enter a varient number and provide dats. See help.'); end [lempth.toss]=size(mphdata); if toss=-120 error(You mut enter a varient number and provide dats. See help.'); end [lempth.toss]=size(mphdata); if toss=-120 error(You mut enter a varient number and provide dats. See help.'); end [lempth.toss]=size(mphdata); if toss=-120 error(You mut enter a varient number and provide dats. See help.'); end [lempth.toss]=size(mphdata); if toss=-120 error(You mut enter a varient number and provide dats. See help.'); end [lempth.toss]=size(mphdata(;,fill)] [lempth.toss]=size(mphdata(;,fil</pre>	-		footRrot=mghdata(:,58:60);	
<pre>sphitereadepubdif; allargershop allargershop allargershop allargershop atterargeQ error(Yuu mat enter a version number and provide data. See help.'); end (length.toss)=size(mpddata); if itoss=120 error(Theta is vrong ise!') end (theta=100 error(Theta is vrong ise!') end (theta=100 (theta=100) (t</pre>	% (c)2003 Stacy J Morris - sjm@alum.mit.edu		%66 Right Shank Roll	
<pre>mpunter teamspace; impunter teamspace; im</pre>	mabdata=readmabdof.		867 Right Shank Fitch	
allargenebo: [sizearge.col]=size(slizege): if sizearge.col]=size(slizege): error(Yum mat enter a version number and provide data. See help.'); end if tess=ric0 error(Tites is versg iss!') end if tess=ric0 error(Tites is versg iss!') end it lace from test is version number and provide data. See help.'); end it cols=nice(supdata): if tess=ric0 error(Tites is versg iss!') end it lace from test issue from test iss	inderen en - renemed i en r		shankRrot=mghdata(:,67:69);	
<pre>id= isingrad is isingrad is isingrad is isonally isonaly isonally isonally isonally isonally isonally isonally ison</pre>	allargs=who; [sizeargs.coll=size(allargs);		SSix DOF Joint Positions	
<pre>if sizerysCd error('Vou must enter a version number and provide data. See help.'); error (length,tess=size(expdata); if tess=size error('bat is wrong size(') error('bat is wrong size(') is for for is is an is a for is a fo</pre>	(%6 Left Ankle X	
<pre>error(You mut enter a version number and provide data. See help.'); end [length.toss]=size(mphdata); if tass=120 enterprof/toks is verong size(') enterprof.toks is verong size(') is size(') is</pre>	if sizeargs<2		%7 Left Ankle Y	
<pre>and anisignerminatis()(3)() [length.tess]=ise(mphates); if tess=120 error(bits is wrong size(') error(') tai is wrong siz</pre>	error('You must enter a version number and provide data. See help.');		%8 Left Ankle Z	
<pre>[length toss]=size(mphdsts); if toss=r10 min toss=r1</pre>	end		ankleLpos=mghdata(:,7:9);	
<pre>if test=120 strong fields is wrong site() arrow(that is wrong site()) and that is manufactors the strong field for i.t strong site() the strong sit</pre>	[length,toss]=size(mghdata);		%60 Right Ankle X	
<pre>and thasign data namestituttituttituttituttituttituttituttit</pre>	if toss=120		S61 Right Ankle I 862 Right Ankle Z	
<pre>end thasign dis manettituttituttituttituttituttituttitutti</pre>	error('Data is wrong size!')		ankleRpos=mghdata(:,61:63);	
<pre>itkating data namesittiittiittiittiittiittiittiittiittiit</pre>	end			
<pre>time():); time():); Six DO Segment Positions 0 Left Poot X 11 Left Poot X 12 Left Poot X 13 Left Poot X 14 Left Poot X 15 Left Poot X 15 Left Poot X 16 Left Shank X 10 L</pre>	**Assign data names************************************	22	%%Assign timepoints%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%	
<pre>iter = iter = iter</pre>			time=[0;0];	
<pre>size DD Segment Positions 0 Left Poot X 11 Left Poot X 12 Left Poot X 13 Left Poot X 14 Left Poot X 15 Left Poot X 16 Left Shank X 10 Lef</pre>	%********* read file 6dof_rl.txt for matlab matrix notation (i.e starts a	t 1 not 0).**** 🖌	for 1=1:length	
<pre>10 Leff Poot X Leff Poot X 12 Leff Poot X 12 Leff Poot X 13 Leff Poot X 14 Leff Poot X 15 Leff Poot X 15 Leff Bank X 16 Leff Bank X 16 Leff Bank X 16 Leff Bank Z 17 Leff Bank Z 16 Leff Bank Z 16 Right Foot X 16 Right Bank X 16 Right Coot Right A 16 Right Bank X 16 Right Coot Right A 16 Right Coot</pre>	%Six DOF Segment Positions		end	
<pre>13 Left Poot Y Loft Foot Z Left Poot Z Left Bank X Loft Eank X Loft Eank X Loft Eank Z Loft Eank</pre>	%0 Left Foot X			
Left foot	%1 Left Foot Y			
<pre>10:121 myhdata(,13:15) myhdata(,61:6) myhdata(:,67:69) myhdata(:,67:69) myhdata(:,61:6) 10 Left Shank X 10 Left Shank X 10 Left Shank Z 1</pre>	<pre>%2 Left Foot Z footIppr=mchdata(, 1,3);</pre>		mondata return=[mondata(, 1,3) mondata(, 4,6) mondata(- 55,57) mahdata(, 58,60) mahdata()
<pre>19 Laft Shank X 10 Laft Shank X 11 Laft Shank Z 15 Right Foot Y 15 Right Foot Y 15 Right Foot Y 15 Right Shank Z 15 Right Shank X 15 Right Shank X 15 Right Shank X 15 Right Shank Y 15 Right Shank Y 15 Right Shank Z 15 Laft Foot Rall 15 Laft Foot Rall</pre>	Loocapos-agamenta (.) 2. 5))		,10:12) mohdata(:,13:15) mohdata(:,64:66) mohdata(:,67	(59) mghdata(:,7:9) mghdata(:,61:6)
<pre>10 Left Bhank Y 11 Left Bhank Z shankDosemphdata(;,1012); 15 Right Foot X 15 Right Cont X 15 Right Bhank X 15</pre>	%9 Left Shank X) time];	
11 Left Bank 2 12 Left Bank 2 13 Right Foot X 15 Right Foot X 15 Right Foot X 15 Right Shark X 16 Right Shark X 17 Right Shark X 18 Right Shark X 19 Right Shark X 10 Right Shark X 10 Right Shark X 10 Right Shark X 10 Right Shark X 10 <td>%10 Left Shank Y</td> <td></td> <td></td> <td></td>	%10 Left Shank Y			
154 Right Foot X 155 Right Foot X 156 Right Foot X 157 Right Foot X 158 Right Foot X 159 Right Foot X 161 Right Ebank X 162 Right Ebank X 163 Right Ebank X 164 Right Ebank X 165 Right Ebank X 161 DOF Segment Restions 162 Light Foot Platch 163 Light Foot Platch 164 Light Foot Platch 165 Light Foot Platch 165 Light Foot Platch 165 Light Foot Platch	<pre>%11 Left Shank Z shankLpos=mghdata(:,10:12);</pre>			
135 Right Foot 7 136 Right Foot 7 footBpoarmphdata(;,5537);				
156 Right: Foot 2 fotRpoarsphdata(:,55:57);	855 Right Foot Y			
footboomsphdata(1,55:7), footboomsphdata(1,55:7), footboomsphdata(1,55:7), footboomsphdata(1,54:66), footboomsphdata(1,64:66), footboomsphdata(1,64:66), footboomsphdata(1,64:67)	\$56 Right Foot Z			
NG3 Right: Shank X V64 Right: Shank X V65 Right: Shank X Six DD7 Segment. Notations Six DD7 Segment. Notations V3 Left: Foot Nall V4 Left: Foot Pitch V5 Left: Foot Yaw ControbumpHate(1;4:6); Si	footRpos=mghdata(:,55:57);			
164 Right Shank Y 65 Right Shank Z shankRoommghdata(, 46, 66); 361a DDF Zeparat. Hostalous 18 Left: Foot Roll 18 Left: Foot Roll 19 Left: Foot Plath 15 Left: Foot Yaw Controstmghdata(, 46, 6);	%63 Right Shank X			
<pre>tk0 Right Bhank 2 kbank 2</pre>	864 Right Shank Y			
anninequosessinta (; yet very) billis DDT Equerations 1 3 Left Foot Roll 4 Left Foot Pitch 5 Left Foot Yaw 1 foothortmythat (; yet 6) ;	%65 Right Shank Z should be a start of start			
Usix DDF Segment Rotations 3 Left Poot Roll 44 Left Poot Roll 50 Left Poot Roll 60 Contextpodd(a) (+ (6))	shankrpos-myndaca(:,ex:ee);			
AS Left FOUL AULT 4 Left FOUL FILCH 15 Left FOUL Yaw fourthermodels (,4:6) /	Six DOF Segment Rotations			
15 Left Foot Yaw To Charlength (1,40)	10 Left Foot Pitch			
footLrot=mghdata(:,4:6);	%5 Left Foot Yaw			
	footLrot=mghdata(:,4:6);			

findmghoutliers.m



genleaveoutids.m

::\sjm\Research\Matlab\THESIS\genleaveoutids.m May 12, 2004	Page 1 8:14:32 AM	C:\sjm\Research\Matlab\THESIS\genleaveoutids.m May 12, 2004	Page 8:14:32 #
unction makehypCmats		r02pr04Fadj(findthis,:)=[];	
<pre>isp('disabled');return</pre>		<pre>findthis=find(r02pr04Fadj(:,1)==12); r02pr04Fadj(findthis,:)=[];</pre>	
oad c:\sjm\research\matlab\run02\r02pr04_females;		<pre>findthis=find(r02pr04Fadj(:,1)==16); r02pr04Fadj(findthis,:)=[];</pre>	
ubgait=[r02pr04F(:,1) r02pr04F(:,3)];		r02pr04Fadj(findthis,:)=[];	
eaveoutids=zeros(length(r02pr04F(:,1)),9);		findthis=find(r02pr04Fadj(:,1)==19);	
or f=1:length(femalenums);		r02pr04Fadj(findthis,:)=[];	
for g=10:10:30		remaienums=[15]1/]20]22]25]]	
rows=findtwocols(subgait,[femalenums(f) g]);		%Groups 1 and 4	
<pre>subj lengths(1,g/10)=length(rows);</pre>		findthis=find(r02pr04Fadj(:,1)==15);	
<pre>eval(['subj_' gnames(g/10,:) '=rows;']);</pre>		r02pr04Fadj(findthis,:)=[];	
end		findthis=find(r02pr04Fadj(:,1)==17);	
for r=1:3		r02pr04Fadj(findthis,:)=[];	
<pre>sL=subj_lengths(1,r);</pre>		femalenums=[11;12;16;18;19;20;22;25];	
eval(['data=subj_' gnames(r,:) ')'))		end	
if st<9		traipandtestset=r02pr04Fadi;	
<pre>dups=[data(:,1) rand(sL,1)]; %new set of random numbers dups=sortrows(dups,2);</pre>		clear trainandtestclass;	
if sL>4		if c==2	
data(sL+1:9,:)=dups(sL-(9-sL)+1:sL,:);		<pre>for t=1:length(trainandtestset(:,1)); if trainandtestset(t,</pre>	4) == 1; trainandtestcla
else %shortest are 4 long		<pre>s(t,1)=1;elseif trainandtestset(t,4)==2;trainandtestclass(t,1)=-1;</pre>	else;error('error in :
data(5:8,:)=dups;		tup of matrices in trainsymresults');end;end	
dups=[data(1:4,1) rand(sL,1)]) wanother new set or	random numbers	elself c==>	
data(9, :)=dups(4, :):		end	-crainanucescsec(c,4)
end		else	
end		<pre>for t=1:length(trainandtestset(:,1));if trainandtestset(t,4)</pre>)==1 trainandtestse
<pre>data=sortrows(data,2); eval(['subj_' gnames(r,:) '=data(:,1);']);</pre>		<pre>(t,4)==2 ; trainandtestclass(t,1)=1;elseif trainandtestset(t,4)==4; =-1;else;error('error in setup of matrices in trainsvmresults');end</pre>	trainandtestclass(t,1 d;end
end		end	
rowtrips=[subj_free_subj_dist_subj_pace];			17.21) +
for g=1/3		t/, 33:35) trainandtestset(, 45:47)1;	, 17.51) crainanducescs
<pre>leaveoutids(rowtrips(r,q),r)=1;</pre>		trainandtest=[trainandtestset(:,9) trainandtestclass trainandte	estsetdata trainandtes
end		<pre>set(:,57:65)]; %leaveoutrows are at end</pre>	
end nd		<pre>eval([clabels(c,:) 'C all=trainandtest;']);</pre>	
labels=[' nyp2')'nyp1_1')'nyp1_2')'nyp1_3')' nyp1'))		FOR 1=1:9	
r02pr04Fadi=[r02pr04F]eaveoutids]:		trainandtestsubset=sortrows(trainandtestsubset.29);	(1)))
if c==2			
%Groups 1 and 2		findteststart=find(trainandtestsubset(:,29)==1);	
findthis=find(r02pr04Fadj(:,1)==20);			
r02pr04Fadj(findthis,:)=[];		<pre>training=trainandtestsubset(1:findteststart(1)-1,:);</pre>	
findthis=find(r02pr04Fadj(:,1)==22);		testing =trainandtestsubset(findteststart(1):length(trainan	<pre>idtestsubset(:,1)),:);</pre>
ru2pru4radj(rindtnis,:)=[];			
r02pr04Fadi(findthis.:)=[];		eval((clabels(c,:) 'C lo' num2str(i) 'traterting(')))	
femalenums=[11;12;16;18;19;15;17];		eval([clabels(c,:) 'C lo' num2str(i) '=[training;testing	(171)7
elseif c==3		end	
%Groups 2 and 4		end	
C1 10 1 C1 10 00 000 110 01 000		rave citrin Research Matlah classw hum@mats hum# leaveoutids	

REFERENCES

- Lemmon D; Shiang TY; Hashmi A; Ulbrecht JS; Cavanagh PR. The effect of insoles in therapeutic footwear -- a finite element approach. J. Biomech, 1997 Jun; 30(6): 615-20.
- [2] Morris ME; Huxham F; McGinley J; Dodd K; Iansek R. The biomechanics and motor control of gait in Parkinson disease. Clin Biomech, 2001 Jul; 16(6): 459-70.
- [3] Kernozek TW; LaMott EE; Dancisak MJ. Reliability of an in-shoe pressure measurement system during treadmill walking. Foot Ankle Int, 1996 Apr; 17(4): 204-9.
- [4] Zhu HS; Maalej N; Webster JG; Tompkins WJ; Bach-Y-Rita P; Wertsch JJ. An umbilical data-acquisition system for measuring pressures between the foot and shoe. IEEE Trans Biomed Eng, 1990 Sep; 37(9): 908-11.
- [5] Zhu HS; Wertsch JJ; Harris GF; Loftsgaarden JD; Price MB. Foot pressure distribution during walking and shuffling. Arch Phys Med Rehabil, 1991 May; 72(6): 390-7.
- [6] Zhu HS; Wertsch JJ; Harris GF; Alba HM; Price MB. Sensate and insensate in-shoe plantar pressures. Arch Phys Med Rehabil, 1993 Dec; 74(12): 1362-8.
- [7] Hausdorff JM; Ladin Z; Wei JY. Footswitch system for measurement of the temporal parameters of gait. J Biomech, 1995 Mar; 28(3): 347-51.
- [8] Hausdorff JM; Zemany L; Peng C.-K.; Goldberger AL. Maturation of gait dynamics: stride-to-stride variability and its temporal organization in children. J Appl Physiol, 1999 Mar; 86(3): 1040-7.
- [9] Hausdorff JM; Rios DA; Edelberg HK. Gait variability and fall risk in communityliving older adults: a 1-year prospective study. Arch Phys Med Rehabil, 2001 Aug; 82(8): 1050-6.
- [10] Hausdorff, JM; Purdon PL; Peng C.-K; Ladin Z; Wei J.-Y.; Goldberger AL. Fractal dynamics of human gait: stability of long-range correlations in stride interval fluctuations. J. Appl. Physiol, 80: 1448–1457, 1996.
- [11] Morley RE Jr; Richter EJ; Klaesner JW, Maluf KS; Mueller MJ. In-shoe multisensory data acquisition system. IEEE Trans Biomed Eng, 2001 Jul; 48(7): 815-20.
- [12] Maluf KS; Morley RE Jr; Richter EJ; Klaesner JW; Mueller MJ. Monitoring in-shoe plantar pressures, temperature, and humidity: reliability and validity of measures from a portable device. Arch Phys Med Rehabil, 2001 Aug; 82(8): 1119-27.
- [13] Pappas IP; Popovic MR; Keller T; Dietz; Morari M. A reliable gait phase detection system. IEEE Trans Neural Syst Rehabil Eng, 2001 Jun; 9(2): 113-25.

- [14] Pappas IP; Keller T; Mangold S. A Reliable, Gyroscope based Gait Phase Detection Sensor Embedded in a Shoe Insole. Presented at the 2002 IEEE International Conference on Sensors, Orlando, FL.
- [15] Cutlip RG; Mancinelli C; Huber F; DiPasquale J. Evaluation of an instrumented walkway for measurement of the kinematic parameters of gait. Gait Posture, 2000 Oct; 12(2): 134-8.
- [16] Giacomozzi C; Macellari V. Piezo-dynamometric platform for a more complete analysis of foot-to-floor interaction. IEEE Trans Rehabil Eng, 1997 Dec; 5(4): 322-30.
- [17] See http://www.cse.ohio-state.edu/~jwdavis/publications.html
- [18] Yam CY, Nixon MS, Carter JN. Automated person recognition by walking and running via model-based approaches. Pattern recognition, 2004; 37(Part 5): 1057-1072
- [19] MIT AI Human ID Results [online]. Available at http://www.ai.mit.edu/people/llee/ HID/mitai_data_avg_spec.htm [accessed 30 April 2004].
- [20] Johnson AY, Sun J, Bobick AF. Predicting large population data cumulative match characteristic performance from small population data. In the 4th International Conference on Audio- and Video-Based Biometric Person Authentication, Guildford, UK; June 9-11, 2003.
- [21] Walk the Walk: Gait Recognition Technology Could Identify Humans at a Distance; Georgia Tech Research News [online]. Available at http://gtresearchnews.gatech.edu/newsrelease/GAIT.htm [accessed 30 April 2004].
- [22] Kidd CD, Orr R, Abowd GD, Atkeson CG, Essa IA, MacIntyre B, Mynatt E, Starner TE, Newstetter W. The Aware Home: A living laboratory for ubiquitous computing research. In Proceedings of CoBuild '99: Second International Conference on Cooperative Buildings: 191-198.
- [23] Suutala J, Pirttikangas S, Riekki J, Röning J. Reject-optional LVQ-based two-level classifier to improve reliability in footstep identification.
- [24] The Miburi performance system, Yamaha Corporation, 1996 [originally available online, at "http://www.yamaha.co.jp/news/96041001.html].
- [25] See http://www.probalance.com
- [26] Acceleron Technologies. The Leader in Smart Technologies for Health, Fitness and Beyond: Technology FAQ's [online]. Available at http://www.xlrn.com/faq.html [Accessed 22 February 2002].
- [27] See http://www.traxtar.com
- [28] See http://www.fitsense.com
- [29] See http://www.vectrasense.com/
- [30] Marriott M. The Bionic Running Shoe. The New York Times, May 6, 2004.

- [31] See http://www.adidas.com/
- [32] F-Scan System Features, Tekscan, Inc., 2004 [online]. Available from http:// www.tekscan.com/medical/specs_fscan1.html [accessed 4 April 2004].
- [33] See http://www.clevemed.com.
- [34] Choi I, Ricci C. Foot-mounted gesture detection and its application in virtual environments. 1997 IEEE International Conference on Systems, Man, and Cybernetics. Computational Cybernetics and Simulation, Oct. 1997; 5(12-15):4248-53
- [35] http://www.portablegaitlab.com/
- [36] Acceleron Technologies, LLC. System and Method for Measuring Movement of Objects. United States Patent #5,899,963. 4 May 1999.
- [37] Acceleron Technologies, LLC. System and Method for Measuring Movement of Objects. United States Patent #6,122,960. 26 Sept 2000.
- [38] Paradiso JA; Hsiao K; Benbasat AY; Teegarden Z. Design and implementation of expressive footware. IBM systems journal, 2000; 39 (3): 511-519.
- [39] Krebs DE; Edelstein JE; Fishman S. Reliability of observational kinematic gait analysis. Phys Ther, 1985 Jul; 65(7): 1027-33.
- [40] Eastlack ME; Arvidson J; Snyder-Mackler L; Danoff JV; McGarvey CL. Interrater reliability of videotaped observational gait-analysis assessments. Phys Ther, 1991 Jun; 71(6): 465-72.
- [41] McGinley JL; Goldie PA; Greenwood KM; Olney SJ. Accuracy and reliability of observational gait analysis data: judgments of push-off in gait after stroke. Phys Ther, 2003 Feb; 83(2): 146-60.
- [42] Whittle MW. Clinical gait analysis: A review. Hum Mov Sci, 1996; 15 (3): 369-388.
- [43] Sutherland DH. The evolution of clinical gait analysis part 1:kinesiological EMG. Gait Posture, 2001 Jul; 14(1): 61-70.
- [44] Diss CE. The reliability of kinetic and kinematic variables used to analyse normal running gait. Gait Posture, 2001 Oct; 14(2): 98-103.
- [45] Fontaine D, David D, Caritu Y. Sourceless Human Body Motion Capture. Available online at http://www.grenoble-soc.com/proceedings03/Pdf/30-fontaine.pdf [accessed 9 May 2004].
- [46] Antonnson EK. A three-dimensional kinematic acquisition and intersegmental dynamic analysis system for human motion. Ph.D. Thesis, MIT, 1982.
- [47] Riley PO; Mann RW; Hodge WA. Modelling of the biomechanics of posture and balance. J. Biomech, 1990; 23(5): 503-6.
- [48] Riley PO; Schenkman ML; Mann RW; Hodge WA. Mechanics of a constrained chair-rise. J Biomech, 1991; 24(1): 77-85.

- [49] Vicon Camera Technical Sheet, Vicon Motion Systems, 2003 [online]. Available at http://www.vicon.com/main/support/downloads/tech%20sheets/ VMS027S_cameras_hi-res.pdf [accessed 27 October 2003].
- [50] Comparison Matrix: Eagle Digital, Hawk Digital, and Falcon Analog Systems, Motion Analysis Corporation, 2003 [online]. Available at http://www.motionanalysis.com/pdf/systemcomp.pdf [accessed 26 October 2003].
- [51] FASTRAK data sheet, Polhemus, 2001 [online]. Available at http://www.polhemus.com/FastTrak/fastrak.pdf [accessed 3 November 2003].
- [52] Raab FH, Blood EB, Steiner TO, Jones RJ. Magnetic position and orientation tracking system. IEEE Transactions on Aerospace and Electronic Systems, September 1979; AES15(5): 709-718.
- [53] Flock of Birds data sheet, Ascencion Technology Corp., 2000 [online]. Available at http://www.ascension-tech.com/products/flockofbirds.pdf [accessed 3 November 2003].
- [54] Hausdorff JM; Ladin Z; Wei JY. Gait variability and basal ganglia disorders: strideto-stride variations of gait cycle timing in Parkinson's disease and Huntington's disease. Mov Disord, 1998 May; 13(3): 428-3.
- [55] Thaut MH; Kenyon GP; Schauer ML; McIntosh GC. The connection between rhythmicity and brain function. IEEE Eng Med Biol Mag, 1999 Mar-Apr; 18(2): 101-8.
- [56] Thaut MH; McIntosh GC; Rice RR, Miller RA, Rathbun J, Brault JM. Rhythmic auditory stimulation in gait training for Parkinson's disease patients. Movement Disorders, 1996; 11(2): 193-200.
- [57] McIntosh GC; Rice RR; Hurst CP, Thaut MH. Long-term training effects of rhythmic auditory stimulation on gait in patients with Parkinson's disease. Movement Disorders, 1998; 13(suppl 2): 212.
- [58] Pacchetti C; Mancini F; Aglieri R; Fundaro C; Martignoni E; Nappi G. Active music therapy in Parkinson's disease: an integrative method for motor and emotional rehabilitation. Psychosom Med, 2000 May-Jun; 62(3): 386-93.
- [59] Hale LA; Eales CJ. Recovery of walking function in stroke patients after minimal rehabilitation. Physiother Res Int, 1998;3(3):194-205.
- [60] Bessou P; Dupui P; Montoya R; Pages B. Simultaneous recording of longitudinal displacements of both feet during human walking. J Physiol, 1988-89; 83(2): 102-10.
- [61] Ebersbach G; Sojer M; Müller J; Heijmenberg M; Poewe W. Sociocultural differences in gait. Mov Disord, 2000 Nov; 15(6): 1145-7.
- [62] S700 Joint Angle Shape Sensor data sheet, Measureand Inc., 2004 [online]. Available at http://www.measurand.com/manuals/S700.pdf [accessed 4 April 1004].
- [63] Baxter LK. Capacitive Sensors. IEEE Press, New Jersey, 1997.

- [64] As cited in Hausdorff JM; Zemany L; Peng C.-K.; Goldberger AL. Maturation of gait dynamics: stride-to-stride variability and its temporal organization in children. J Appl Physiol, 1999 Mar; 86(3): 1040-7.
- [65] Benbasat AY, Morris SJ, Paradiso JA. A wireless modular sensor architecture and its application in on-shoe gait analysis. Proc. of the IEEE International Conference on Sensors, Oct. 21-24 2003; p 1086-1091.
- [66] Benbasat AY. Stack: A Modular Platform for High Density Wireless Sensing [online]. Available from http://www.media.mit.edu/resenv/Stack/ [accessed 22 February 2002].
- [67] Winter DA, Sidwall HG, Hobson DA. Measurement and reduction of noise in kenematics of locomotion. J. Biomech, 1974; 7:157-159.
- [68] Winter DA. Camera speeds for normal and pathological gait analyses. Med Biol Eng Comput, 1982; 20:408-412.
- [69] Winters DA. Biomechanics and motor control of human movement. John Wiley & Sons, 1990, 2nd ed.
- [70] Weinberg, Harvey. MEMS (Micro Electro-Mechanical Systems) Technology. [online]. Available at http://www.sensorland.com/HowPage023.html [accessed 12 December 2003].
- [71] ADXL202E Data Sheet, Rev A, Analog Devices, Inc., 2000 [online]. Available at http://www.analog.com/UploadedFiles/Data_Sheets/70885338ADXL202_10_b.pdf [accessed 29 May 2002].
- [72] Yazdi N, Ayazi F, Najafi K. Micromachined inertial sensors (invited paper). Proceedings of the IEEE, 1998; 86(8):1640-1659.
- [73] Murata ENC-03J Specifications [online]. Available from http://search.murata.co.jp/ Ceramy/owa/CATALOG.showcatalog?sHinnmTmp=ENC-03J&sLang=2&sNhnm=ENC-03J&sHnTyp=OLD [accessed 28 October 2002].
- [74] Krakauer D. A unique angular-rate-sensing gyro [online]. Sensors Magazine, Dec 16, 2003. Available at http://www.sensorsmag.com/articles/0903/53/main.shtml [accessed 12 December 2003].
- [75] ADXRS150 Data Sheet, Rev A, Analog Devices, Inc., 2003 [online]. Available at http://www.analog.com/UploadedFiles/Data_Sheets/778386516ADXRS150_B.pdf [accessed 15 April 2003].
- [76] Fraden J. Handbook of modern sensors: physics, designs, and applications. Springer Verlag, 1997, 2nd ed.
- [77] FSR integration guide and evaluation parts catalog, with suggested electrical interfaces, Interlink Electronics [originally available online, from http://www.interlinkelec.com].
- [78] Images Co. FLX-01 Specifications [online]. Available from http://www.imagesco.com/catalog/flex/FlexSensors.html [accessed 22 February 2004].

- [79] American Piezo Ceramics. Piezoelectric Ceramics: Principles and Applications. APC International, Ltd. [year to be added]
- [80] Piezo Film Sensors Technical Manual, Measurement Specialties [online]. Available at http://www.msiusa.com/piezo_download_listing.htm [accessed 8 April 2004].
- [81] Ohr, S. Non-contact sensor discerns passenger size for airbags. EE Times, February 6, 2003 [online]. Available at http://www.eetuk.com/tech/news/ OEG20030206S0035 [accessed 4 April 2004].
- [82] MC3394 data sheet, Rev 6.0, Motorola, Inc., Feb 2003 [online]. Available at http:// e-www.motorola.com/files/analog/doc/data_sheet/MC33794.pdf [accessed 17 December 2003].
- [83] Burke SE, Paradiso JA. High-Resolution Piezopolymer Acoustic Bearing Estimator. Presented at the Second Technical Conference on Telecommunications R&D in Massachusetts, University of Massachusetts, Lowell MA, March 12, 1996.
- [84] Coon A. Capabilities and Applications of SAW Coupled-Resonator Filters (AN 23), RF Monolithics, Inc., [online]. Available at http://www.rfm.com/corp/appdata/ An23.pdf [accessed 30 April 2004].
- [85] DR3000-1 Data Sheet, RF Monolithics, Inc., [online]. Available at http:// www.rfm.com/products/data/dr3000-1.pdf [accessed 24 July 2002].
- [86] Titterton D, Weston J. Strapdown Inertial Navigation Technology. IEEE, 1997.
- [87] Brown RG, Hwang PYC. Introduction to Random Signals and Applied Kalman Filtering with Matlab Exercises and Solutions. Wiley Text Books, 1996, 3rd ed.
- [88] Allen BD, Bishop G, Welch G. Tracking: Beyond 15 Minutes of Thought. 2001 SIG-GRAPH Course Notes, Course 11.
- [89] Kuipers JB. Quaternion and rotation sequences: a primer with applications to orbits, aerospace, and virtual reality. Princeton University Press, 1999.
- [90] Hogan N, Krebs D, et al. MIT 2.75s Couse Outline; June 19-23, 1995.
- [91] Lawrence A. Modern inertial technology: navigation, guidance, and control. Springer-Verlag, 1998.
- [92] Insole pressure data during gait, collected using the Tekscan F-Scan; received via email communication on 19 Apr 2004 with Mike Harty and Thomas Papakostas at Tekscan, Inc.
- [93] Breiman L, Friedman J, Olshen R, Stone C. Classification and Regression Trees. Pacific Grove: Wadsworth, 1984.
- [94] Steinberg D, Colla P. CART Classification and Regression Trees. San Diego, CA: Salford Systems, 1997.
- [95] Bayes, Rev. T., "An Essay Toward Solving a Problem in the Doctrine of Chances", Philos. Trans. R. Soc. London, 1763, 53:370-418.

- [96] Duda RO, Hart PE, Stork DG. Pattern Classification. Wiley-Interscience, 2000, 2nd ed.
- [97] Witten IH, Frank E. Data Mining: Practical Machine Learning Tools and Techniques with Java IMplementations, Chapter 8: WEKA Machine Learning algorithms in Java. Morgan Kaufmann Publishers, 2000.
- [98] Boser B, Guyon IM, and Vapnik VN. A Training Algorithm for Optimal Margin Classifiers, Computational Learing Theory 1992.
- [99] Support Vector Machines: http://www.support-vector.net/
- [100] See http://www.isis.ecs.soton.ac.uk/resources/svminfo/
- [101] Gunn SR. Support Vector Machines for Classification and Regression. Technical Report, Image Speech and Intelligent Systems Research Group, University of Southampton, 1998.
- [102] Cristianini N, Shawe-Taylor J. An Introduction to Support Vector Machines and Other Kernel-based Learning Methods. Cambridge University Press, 2000.
- [103] Reed RD, Marks RJ. Neural Smithing: Supervised Learning in Feedforward Artificial Neural Networks. MIT Press, 1999.
- [104] Neural Network Toolbox User's Guide. The MathWorks, Inc. [online]. Available at http://www.mathworks.com/access/helpdesk/help/pdf_doc/nnet/nnet.pdf [accessed 25 March 2004].
- [105] Perry J. Gait Analysis: normal and pathological function. SLACK Inc, 1992, 1st ed.
- [106] Theodoridis S, Koutroumbas C. Pattern Recognition. Academic Press, 2003, 2nd ed.
- [107] Paradiso JA, Morris SJ, Benbasat AY, Asmussen E. Interactive Therapy with Instrumented Footwear. Proceedings of CHI 2004, April 24-29, 2004, Vienna, Austria.
- [108] Wigram T, Nygaard Pedersen I, Ole Bonde L. A Comprehensive Guide to Music Therapy: Theory, Clinical Practice, Research and Training. Jessica Kingsley Publishers, London, England, 2003.
- [109] Rosenboom D (ed.). Biofeedback and the Arts: Results of Early Experiments. A.R.C. Publications, Vancouver, 1976.
- [110] Gromala D, et al. Meditation Chamber. SIGGRAPH 2001 Conf. Abstracts and Applications, ACM Press, 2001; p. 127.
- [111] Ellis P. The music of sound: a new approach for children with severe and profound and multiple learning difficulties. British Journal of Music Education, 1997; 14(2): 173–186.
- [112] Gerasimov V. Swings That Think [online]. Available from http:// vadim.www.media.mit.edu/stt/bat.html [accessed 4 April 2004].
- [113] Asmussen E. An Auditory Biofeedback Program for Gait Analysis. Responsive Environments Group Technical Report, MIT Media Lab, August 2003.

- [114] See http://www.cycling74.com/products/maxmsp.html
- [115] See http://www.keyspan.com/products/usb/usa19hs/
- [116] See http://www.midiman.com/
- [117] See http://www.emu.com/
- [118] See http://www.pdf.org/
- [119] See http://www.parkinson.org/
- [120] See http://www.msiusa.com/piezo/ultrasonic_transducers.htm
- [121] See http://www.gibsonteched.net/vma40a5r.html